## UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

### FORM 8-K

#### **CURRENT REPORT**

Pursuant to Section 13 OR 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): September 17, 2015

## ANTERO MIDSTREAM PARTNERS LP

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation)

001-36719 (Commission File Number) 46-4109058 (IRS Employer Identification No.)

1615 Wynkoop Street Denver, Colorado 80202

(Address of principal executive offices) (Zip Code)

Registrant's telephone number, including area code: (303) 357-7310

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

#### Item 1.01 Entry into a Material Definitive Agreement.

#### Contribution Agreement

On September 17, 2015, Antero Midstream Partners LP (the "Partnership") entered into a Contribution, Conveyance and Assumption Agreement (the "Contribution Agreement") by and among the Partnership, Antero Resources Corporation ("Antero") and Antero Treatment LLC, a wholly-owned subsidiary of the Partnership ("Antero Treatment"). Pursuant to the terms of the Contribution Agreement, the Partnership has agreed (i) to acquire from Antero all of the outstanding limited liability company interests of Antero Water LLC ("Antero Water"), a wholly-owned subsidiary of Antero that owns and operates Antero's fresh water distribution assets, and (ii) to cause Antero Treatment to acquire from Antero all of the assets, contracts, rights, permits and properties owned or leased by Antero and used primarily in connection with the construction, ownership, operation, use or maintenance of Antero's advanced wastewater treatment complex to be constructed in Doddridge County, West Virginia (collectively, (i) and (ii) are referred to herein as the "Contributed Assets"). In consideration for the acquisition of the Contributed Assets, the Partnership has agreed to pay Antero aggregate consideration of \$1.05 billion, consisting of (i) a cash distribution equal to \$552.45 million, less any indebtedness assumed by the Partnership in connection with the closing of the Transaction (as defined below) and (ii) 23,886,421 common units representing limited partner interests in the Partnership ("Common Units") issued to Antero. In addition, the Partnership has agreed to pay Antero (a) \$125 million in cash if the Partnership's delivered fresh water volumes average 161,000 barrels per day or more between January 1, 2017 and December 31, 2019 and (b) an additional \$125 million in cash if the Partnership's delivered fresh water volumes average 200,183 barrels per day or more between January 1, 2018 and December 31, 2020, all in accordance with the terms of the Contribution Agreement (the "Transaction"). Upon completion of the Partnership's concurrent private placement of Common Units (described below), an amount of cash equal to the net proceeds will be paid to Antero and the number of Common Units issuable to Antero will be reduced by an equivalent number to that issued in the private placement. The Partnership intends to fund the cash component of the consideration with a combination of borrowings under the Partnership's existing credit facility and cash proceeds from the private placement described below. Upon completion of the Transaction, the Partnership will own all of Antero's fresh water distribution and wastewater treatment assets (the "Water Assets").

The Contribution Agreement includes customary representations and warranties regarding the Water Assets and the Transaction, as

well as customary covenants and indemnity provisions. The parties have agreed to indemnify each other with regards to breaches of their respective representations, warranties and covenants set forth in the Contribution Agreement. In addition, the Partnership has agreed to indemnify Antero with respect to certain liabilities related to the business and operations of the Water Assets, subject to certain exceptions as set forth in the Contribution Agreement.

The consummation of the Transaction is subject to the satisfaction of customary closing conditions, including, among other things, the completion of the concurrent private placement of Common Units by the Partnership, the absence of legal impediments prohibiting the Transaction, the performance by the parties, in all material respects, of their respective covenants as set forth in the Contribution Agreement and the accuracy of certain of their respective representations and warranties as set forth in the Contribution Agreement. There is no assurance that all of the conditions to the consummation of the Transaction will be satisfied. The Partnership currently expects the Transaction to close on or about September 23, 2015, although the Contribution Agreement will be effective as of October 1, 2015.

In connection with the closing of the Transaction, the Partnership expects that the lenders under its revolving credit facility will increase the commitments thereunder by \$500,000,000. Also in connection with the closing of the Transaction, Antero will enter into the following commercial agreements with the Partnership, Antero Resources Midstream Management LLC, the general partner of the Partnership (the "*General Partner*"), Antero Midstream LLC, a wholly-owned subsidiary of the Partnership, Antero Treatment and/or Antero Water, as applicable:

*Water Services Agreement* — Antero Water will agree to provide certain fluid handling services to Antero within an area of dedication in defined service areas in Ohio and West Virginia, and Antero will agree to pay monthly fees to Antero Water for all fluid handling services provided by Antero Water in accordance with the terms of the Water Services Agreement. The initial term of the Water Services Agreement will be twenty years from the date thereof and from year to year thereafter. Under the agreement, Antero will pay a fixed fee of \$3.685 per barrel in West Virginia and \$3.635 per barrel in Ohio for freshwater deliveries by pipeline directly to the well site, subject to annual CPI adjustments. In addition, Antero will commit to pay a fee on a minimum volume of freshwater

2

deliveries in calendar years 2016 through 2019. Minimum volume commitments are 90,000 barrels per day in 2016, 100,000 barrels per day in 2017 and 120,000 barrels per day in 2018 and 2019.

Secondment Agreement — Antero will agree to provide seconded employees to perform certain operational services with respect to the Partnership's gathering and compression facilities and the Water Assets, and the Partnership will agree to reimburse Antero for expenditures incurred by Antero in the performance of those operational services. The initial term of the Secondment Agreement will be twenty years from November 10, 2014, and from year to year thereafter.

Amended and Restated Services Agreement — Antero, the Partnership and the General Partner will amend and restate their Services Agreement, dated November 10, 2014, to remove provisions relating to operational services in support of the Partnership's gathering and compression business (which will now be covered by the Secondment Agreement) and to provide that Antero will perform certain administrative services for the Partnership and its subsidiaries, and the Partnership will reimburse Antero for expenditures incurred by Antero in the performance of those administrative services.

The terms of the Transaction were unanimously approved on behalf of the Partnership by the Board of Directors of the General Partner, after the Conflicts Committee of the Board of Directors of the General Partner (the "*Conflicts Committee*") unanimously recommended that the Board of Directors of the General Partner approve the Transaction. The Conflicts Committee, composed of independent members of the Board of Directors of the General Partner, retained legal and financial advisors to assist it in evaluating and negotiating the Transaction. In approving the Transaction, the Conflicts Committee based its decisions in part on an opinion from its independent financial advisor that the consideration to be paid by the Partnership is fair to the Partnership and its subsidiaries and the unaffiliated common unitholders of the Partnership from a financial point of view.

The Contribution Agreement is filed as Exhibit 2.1 to this Current Report on Form 8-K, and the foregoing description of the Contribution Agreement is qualified in its entirety by reference to such exhibit. The above description of the Contribution Agreement is a summary only and is qualified in its entirety by reference to the complete text of the Contribution Agreement. The Contribution Agreement is filed herewith to provide investors with information regarding its terms. It is not intended to provide any other factual information about the parties. In particular, the assertions embodied in the representations and warranties contained in the Contribution Agreement were made as of the date of the Contribution Agreement only and are qualified by information in confidential disclosure schedules provided by the parties to each other in connection with the signing of the Contribution Agreement. These disclosure schedules contain information that modifies, qualifies and creates exceptions to the representations and warranties set forth in the Contribution Agreement. Moreover, certain representations and warranties in the Contribution Agreement may have been used for the purpose of allocating risk between the parties rather than establishing matters of fact. Accordingly, you should not rely on the representations and warranties in the Contribution Agreement as characterizations of the actual statements of fact about the parties.

#### Unit Purchase Agreement

On September 17, 2015, the Partnership entered into a Common Unit Purchase Agreement (the "*Unit Purchase Agreement*") with certain institutional investors (the "*Investors*") to sell 12,898,000 Common Units in a private placement for gross proceeds of approximately \$243 million (the "*Private Placement*"). As described above, an amount equal to the net proceeds of the Private Placement will be paid to Antero as a part of the cash consideration for the Transaction. The closing of the Private Placement is subject to the closing of the Transaction and certain other customary conditions, and the Private Placement is expected to close concurrently with the Transaction. Upon completion of the Private Placement, the total cash consideration paid by the Partnership to Antero will be \$794 million and 10,988,421 Common Units, respectively, plus the potential earn-out payments described above.

The Unit Purchase Agreement contains customary representations, warranties and covenants of the Partnership and the Investors. The Partnership, on the one hand, and each of the Investors (severally and not jointly), on the other hand, have agreed to indemnify each other and their respective affiliates, officers, directors and other representatives against certain losses resulting from any breach of their representations, warranties or covenants contained in the Unit Purchase Agreement, subject to certain limitations and survival periods.

Pursuant to the Unit Purchase Agreement, the Partnership has agreed to enter into a Registration Rights Agreement with the Investors in connection with the closing of the Private Placement, pursuant to which the

Partnership will agree to file and maintain a registration statement with respect to the resale of the Common Units on the terms and conditions set forth therein.

The foregoing description is not complete and is qualified in its entirety by reference to the full text of the Common Unit Purchase Agreement, which is filed as Exhibit 10.1 to this Current Report on Form 8-K and incorporated herein by reference.

#### Relationships

Certain individuals, including officers and directors of Antero and the General Partner, serve as officers and/or directors of more than one of Antero, the Partnership, Antero Treatment and Antero Water. Antero owns 29,940,957 Common Units (not including the Common Units anticipated to be issued to Antero in connection with the closing of the Transaction) and all of the subordinated units representing limited partner interests in the Partnership. In addition, certain officers, directors and private equity sponsors of Antero and the General Partner are members of Antero Resources Investment LLC, which owns and controls (and appoints all the directors of) the General Partner, which owns a non-economic general partner interest in the Partnership and all of the Partnership's incentive distribution rights.

#### Cautionary Statements

This Current Report on Form 8-K contains "forward-looking statements" within the meaning of federal securities laws. The forward-looking statements contained herein include statements about the future closing of the Transaction, including: (i) the entrance into the water services agreement, secondment agreement and amended and restated services agreement; (ii) the satisfaction of closing conditions, including, among other things, the completion of the concurrent private placement of Common Units by the Partnership, the absence of legal impediments prohibiting the Transaction, the performance by the parties, in all material respects, of their respective covenants as set forth in the Contribution Agreement; and (iii) the form of the consideration payable to Antero by the Partnership. Such forward-looking statements are subject to a number of risks and uncertainties, many of which are beyond the Partnership's control. All statements, other than historical facts included in this Current Report on Form 8-K, are forward-looking statements. All forward-looking statements speak only as of the date of this Current Report on Form 8-K. Although the Partnership believes that the plans, intentions and expectations will be achieved. Therefore, actual outcomes and results could materially differ from what is expressed, implied or forecasted in such statements.

The Partnership cautions you that these forward-looking statements are subject to all of the risks and uncertainties, most of which are difficult to predict and many of which are beyond the Partnership's control, incident to the gathering and compression business. These risks include, but are not limited to, commodity price volatility, inflation, environmental risks, drilling and completion and other operating risks, regulatory changes, the uncertainty inherent in projecting future rates of production, cash flow and access to capital, the timing of development expenditures, and the other risks described under the heading "Item 1A. Risk Factors" in the Partnership's Annual Report on Form 10-K for the year ended December 31, 2014 and Quarterly Reports on Form 10-Q for the quarters ended March 31, 2015 and June 30, 2015.

#### Item 3.02 Sale of Unregistered Units.

The information regarding the Private Placement set forth in Item 1.01 of this Current Report on Form 8-K is incorporated herein by reference. The foregoing transactions were undertaken in reliance upon the exemption from the registration requirements in Section 4(a) (2) of the Securities Act of 1933, as amended. The Partnership believes that exemptions other than the foregoing exemption may exist for these transactions.

#### Item 7.01 Regulation FD Disclosure.

On September 18, 2015, the Partnership issued a press release relating to the Transaction and the Private Placement, which is attached hereto as Exhibit 99.1.

The information included in Item 7.01 of this Current Report on Form 8-K, including Exhibit 99.1, is being furnished pursuant to Item 7.01 of Form 8-K and shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, or otherwise subject to liabilities of that section.

#### (d) Exhibits.

EXHIBIT	DESCRIPTION
2.1*	Contribution, Conveyance and Assumption Agreement, dated as of September 17, 2015, by and among Antero Resources Corporation, Antero Midstream Partners LP and Antero Treatment LLC.
10.1	Common Unit Purchase Agreement, dated as of September 17, 2015, by and among Antero Midstream Partners LP and the Purchasers named therein.
99.1	Press Release of Antero Midstream Partners LP issued September 18, 2015, relating to the Transaction and the Private Placement.

\* Pursuant to Item 601(b)(2) of Regulation S-K, the Partnership agrees to furnish supplementally a copy of any omitted exhibit or schedule to the U.S. Securities and Exchange Commission upon request.

5 SIGNATURES Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized. ANTERO MIDSTREAM PARTNERS LP

By: Antero Resources Midstream Management LLC, its general partner

By: /s/ GLEN C. WARREN, JR. Glen C. Warren, Jr.

President and Chief Financial Officer

Dated: September 18, 2015

## EXHIBIT INDEX

6

EXHIBIT	DESCRIPTION
2.1*	Contribution, Conveyance and Assumption Agreement, dated as of September 17, 2015, by and among Antero Resources Corporation, Antero Midstream Partners LP and Antero Treatment LLC.
10.1	Common Unit Purchase Agreement, dated as of September 17, 2015, by and among Antero Midstream Partners LP and the Purchasers named therein.
99.1	Press Release of Antero Midstream Partners LP issued September 18, 2015, relating to the Transaction and the Private Placement.

\* Pursuant to Item 601(b)(2) of Regulation S-K, the Partnership agrees to furnish supplementally a copy of any omitted exhibit or schedule to the U.S. Securities and Exchange Commission upon request.

7

#### CONTRIBUTION, CONVEYANCE AND ASSUMPTION AGREEMENT

#### by and among

#### ANTERO RESOURCES CORPORATION,

#### ANTERO MIDSTREAM PARTNERS LP

and

#### ANTERO TREATMENT LLC

#### dated as of

#### September 17, 2015

#### **TABLE OF CONTENTS**

ARTICLE I DEFINITIONS		2
ARTICLE II CONTRIBUTIONS, CONVEYANCES, A	CKNOWLEDGMENTS AND DISTRIBUTIONS	12
2.1 Contributions		12
2.2 Consideration		12
2.3 Earn-Out Payments		13
2.4 Transfer Taxes		16
ARTICLE III REPRESENTATIONS AND WARRANTI	ES OF ANTERO	16
3.1 Organization and Existence		16
3.2 Authority and Approval; Enforceability		17
3.3 No Conflict		17
3.4 Consents		18
3.5 Laws and Regulations; Litigation		18
3.6 Environmental Matters		19
3.7 Water Interests		19
3.8 Water Assets		19
3.9 Permits		20
3.10 Insurance		20
3.11 Brokerage Arrangements		21
3.12 Investment		21
3.13 Taxes		21
3.14 Financial Statements		22
3.15 Material Contracts		23
3.16 No Adverse Changes		23
3.17 Employees		23
3.18 Investment Company Act		23
3.19 No Other Representations or Warranties; Sched	ules	23
-		
	ES OF THE PARTNERSHIP ACQUIRING ENTITIES	24
4.1 Organization and Existence		24
4.2 Authority and Approval; Enforceability		24
4.3 No Conflict		24
4.4 Brokerage Arrangements		25
4.5 New Common Units		25
4.6 Investment		25
4.7 No Other Representations or Warranties; Schedu	les	26
ARTICLE V COVENANTS, ETC.		26
5.1 Conduct of the Business		26
5.2 Financial Statements; Financing Cooperation		20
5.3 Independent Investigation		27
5.4 Post-Closing Receivables and Payments		28
5.7 1 Ost-Crosing receivables and 1 ayments		20

<ul> <li>5.5 Further Assurances</li> <li>5.6 NYSE Listing</li> <li>5.7 Tax Covenants</li> <li>5.8 Consents</li> <li>5.9 Access</li> </ul>	28 28 28 31 32
<ul> <li>ARTICLE VI CONDITIONS TO CLOSING</li> <li>6.1 Conditions to Each Party's Obligation to Effect the Transactions</li> <li>6.2 Conditions to the Obligation of the Partnership Acquiring Entities</li> <li>6.3 Conditions to the Obligation of Antero</li> </ul>	33 33 33 34
ARTICLE VII CLOSING 7.1 Closing 7.2 Deliveries by Antero 7.3 Deliveries by the Partnership Acquiring Entities	35 35 35 35
ARTICLE VIII INDEMNIFICATION 8.1 Indemnification of Antero and Other Parties 8.2 Indemnification of the Partnership and Other Parties 8.3 Indemnification Procedures 8.4 Calculation and Payment of Damages 8.5 Waiver of Certain Damages 8.6 Limitations on Indemnification 8.7 Survival 8.8 Mitigation 8.9 Sole Remedy 8.10 Consideration Adjustment	36 36 37 38 38 38 39 40 40
ARTICLE IX TERMINATION 9.1 Events of Termination 9.2 Effect of Termination	40 40 41
ARTICLE X MISCELLANEOUS 10.1 Expenses 10.2 Notices 10.3 Governing Law and Venue 10.4 Public Statements 10.5 Form of Payment 10.6 Entire Agreement; Amendments and Waivers 10.7 Binding Effect and Assignment 10.8 Severability 10.9 Interpretation 10.10 Headings and Schedules 10.11 Counterparts 10.12 Determinations by the Partnership	41 41 42 42 42 43 43 43 44 44 44
· · · · · · · · · · · · · · · · · · ·	

iii

#### EXHIBITS AND SCHEDULES

Exhibit A	Form of Assignment and Bill of Sale of Treatment Assets
Exhibit B	Form of Assignment of Water Interest
Exhibit C	Form of Water Services Agreement
Exhibit D	Form of Secondment Agreement
Exhibit E	Form of Amended and Restated Services Agreement
Schedule 1	Excluded Liabilities
Schedule 1A	NOVs
Schedule 1-1	Excluded Assets
Schedule 1-1A	Water Withdrawal Points and Withdrawal Agreements
Schedule 1-2	Treatment Assets and Liabilities
Schedule 1-3	Water Assets
Schedule 3.3	No Conflict
Schedule 3.4	Consents
Schedule 3.5	Laws and Regulations; Litigation
Schedule 3.6	Environmental Matters
Schedule 3.8(b)	Liens
Schedule 3.13	Taxes
Schedule 3.16	Adverse Changes
Schedule 4.3	No Conflict

#### **CONTRIBUTION, CONVEYANCE AND ASSUMPTION AGREEMENT**

This Contribution, Conveyance and Assumption Agreement (this "<u>Agreement</u>") is made and entered into as of September 17, 2015 by and among Antero Resources Corporation, a Delaware corporation ("<u>Antero</u>"), Antero Midstream Partners LP, a Delaware limited partnership (the "<u>Partnership</u>"), and Antero Treatment LLC, a Delaware limited liability company ("<u>Antero Treatment</u>" and, together with the Partnership, the "<u>Partnership Acquiring Entities</u>"). Antero, the Partnership and Antero Treatment are sometimes referred to in this Agreement individually as a "<u>Party</u>" and collectively as the "<u>Partners</u>."

#### RECITALS:

WHEREAS, Antero is the sole member of Antero Water LLC, a Delaware limited liability company ("Antero Water");

WHEREAS, the Partnership is the sole member of Antero Treatment;

WHEREAS, pursuant to the Amended and Restated Contribution Agreement, dated November 10, 2014, by and between Antero and the Partnership (the "<u>Initial Contribution Agreement</u>"), the Partnership has the option to purchase, in its sole discretion, the Water Assets (as defined herein) at fair market value (the "<u>Water Option</u>");

WHEREAS, on July 9, 2015, the Partnership notified Antero in writing of its intent to exercise the Water Option;

WHEREAS, in connection with the Partnership's exercise of the Water Option, Antero intends to contribute (a) the Treatment Assets to Antero Treatment and (b) 100% of the outstanding limited liability company interests in Antero Water to the Partnership (the "<u>Water Interests</u>," and together with the Treatment Assets, the "<u>Contributed Assets</u>") in exchange for the consideration, and on the terms and conditions, set forth in this Agreement;

WHEREAS, the Conflicts Committee of the Board of Directors (the "<u>Conflicts Committee</u>") of Antero Resources Midstream Management LLC, a Delaware limited liability company and the general partner of the Partnership (the "<u>General Partner</u>") has (i) received an opinion of Evercore Group L.L.C., the financial advisor to the Conflicts Committee (the "<u>Partnership Financial Advisor</u>"), that the consideration to be paid by the Partnership as consideration for the Contributed Assets pursuant to this Agreement is fair to the Partnership from a financial point of view and (ii) found the transactions contemplated by the Transaction Documents to be not adverse to the Partnership and recommended that the Board of Directors of the General Partner approve the transactions contemplated by the Transaction Documents and, subsequently, the Board of Directors of the General Partner has approved the transactions contemplated by the Transaction Documents;

WHEREAS, the Special Committee of the Board of Directors of Antero (the "<u>Special Committee</u>") has (i) received an opinion of Robert W. Baird & Co. Incorporated, the financial advisor to the Special Committee (the "<u>Antero Financial Advisor</u>"), that the aggregate consideration to be received by Antero as consideration for the Contributed Assets pursuant to

1

this Agreement is fair, from a financial point of view, to Antero's shareholders (other than Affiliates of Antero) and (ii) found the transactions contemplated by the Transaction Documents to be fair to and in the best interests of Antero and its shareholders (other than Affiliates of Antero) and recommended that the Board of Directors of Antero approve the transactions contemplated by the Transaction Documents and, subsequently, the Board of Directors of Antero has approved the transactions contemplated by the Transaction Documents; and

WHEREAS, on the Effective Date (as defined herein), on the terms and conditions set forth in this Agreement, each of the events and transactions set forth in <u>Section 2.1</u> below shall be deemed to have occurred.

NOW, THEREFORE, in consideration of the mutual undertakings and agreements contained in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereto agree as follows:

#### ARTICLE I DEFINITIONS

Capitalized terms used but not otherwise defined herein shall have the respective meanings ascribed to such terms below:

"<u>AAA</u>" means the American Arbitration Association.

"AAA Rules" has the meaning set forth in Section 2.3(c)(i).

"Additional Cash Consideration" has the meaning set forth in Section 2.2.

"<u>Affiliate</u>" means, with respect to any Person, any other Person that, directly or indirectly, Controls, is Controlled by or is under common Control with, such specified Person through one or more intermediaries or otherwise; *provided, however*, that (a) with respect to Antero, the term "Affiliate" shall not include the General Partner or any member of the Partnership Group, (b) with respect to the

Partnership Group, the term "Affiliate" shall exclude Antero and its other Subsidiaries and (c) with respect to each of the Parties, the term "Affiliate" shall exclude Antero Resources Investment LLC, a Delaware limited liability company.

"Agreement" has the meaning set forth in the preamble to this Agreement.

"Antero" has the meaning set forth in the preamble to this Agreement.

"Antero Closing Certificate" has the meaning set forth in Section 6.2(c).

"Antero Financial Advisor" has the meaning set forth in the recitals to this Agreement.

"Antero Indemnitees" has the meaning set forth in Section 8.1.

"Antero Treatment" has the meaning set forth in the preamble to this Agreement.

"Antero Water" has the meaning set forth in the recitals to this Agreement.

2

"Antero Water Net Working Capital" means the Current Assets less the Current Liabilities.

"Arbitrator" has the meaning set forth in Section 2.3(c)(i).

"<u>Asset Taxes</u>" means sales, use, ad valorem, property, excise or similar Taxes based upon the operation or ownership of the Water Assets but excluding, for the avoidance of doubt, (a) Income Taxes and (b) Transfer Taxes.

"<u>Assignment and Bill of Sale of Treatment Assets</u>" means that certain Assignment and Bill of Sale of Treatment Assets in the form attached as <u>Exhibit A</u> hereto.

"Assignment of Water Interest" means that certain Assignment of Water Interest in the form attached as Exhibit B hereto.

"Barrel" means forty-two U.S. gallons, each of which is equal to 231 cubic inches.

"<u>Business</u>" means, collectively, the operations and business (a) as historically conducted by Antero and its Affiliates using the Water Assets and (b) as contemplated to be conducted following the Closing Date by the Partnership Group using the Water Assets.

"Cap" has the meaning set forth in Section 8.6(a).

"Cash Consideration" has the meaning set forth in Section 2.2.

"Claim Notice" has the meaning set forth in Section 8.3(a).

"Closing" has the meaning set forth in Section 7.1.

"Closing Date" has the meaning set forth in Section 7.1.

"Code" means the Internal Revenue Code of 1986, as amended.

"Commission" means the United States Securities and Exchange Commission.

"Conflicts Committee" has the meaning set forth in the recitals to this Agreement.

"Consent" has the meaning set forth in Section 3.4.

"<u>Contract</u>" means any contract, commitment, instrument, undertaking, lease, sublease, note, mortgage, conditional sales contract, license, sublicense, franchise agreement, indenture, settlement, Permit or other legally binding agreement (whether oral or written).

"Contributed Assets" has the meaning set forth in the recitals to this Agreement.

"<u>Control</u>" means, where used with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise, and the terms "Controlling" and "Controlled" have correlative meanings.

"Current Assets" means the sum of Antero Water's accounts receivable — affiliate plus its accounts receivable — third party.

"<u>Customary Post-Closing Consents</u>" means (a) consents, approvals, licenses, permits, orders, waivers, authorizations, regulations, declarations and filings from or to (as applicable) Governmental Authorities that are customarily obtained or made (as applicable) after closing in connection with transactions contemplated by this Agreement and (b) any consents, approvals, waivers and authorizations of third parties (other than Governmental Authorities) that cannot be unreasonably withheld by the relevant Person.

"Damages" has the meaning set forth in Section 8.1.

"Debt Financed Distribution" has the meaning set forth in Section 5.7(e)(ii).

"Deductible" has the meaning set forth in Section 8.6(a).

"<u>Delivered Volumes</u>" means MVC Credit Volumes plus, for any calendar year or calendar quarter, the aggregate number of additional Barrels of Fresh Water delivered to Persons other than Antero or any of its Affiliates through the Water Assets to the extent such volumes are not included in the calculation of MVC Credit Volumes.

"Earn-Out Report" has the meaning set forth in Section 2.3(b).

"Earn-Out Report Dispute" has the meaning set forth in Section 2.3(b).

"Earn-Out Payments" has the meaning set forth in Section 2.3(a).

"Earn-Out Term" means the First Earn-Out Term or the Second Earn-Out Term, as applicable.

"Effective Date" means October 1, 2015.

"<u>Environmental Laws</u>" means any and all applicable federal, state and local laws and regulations and other legally enforceable requirements and rules of common law, in each case, relating to the prevention of pollution or protection of human health or the environment or imposing liability or standards of conduct concerning handling or transportation of any Hazardous Materials.

"Excluded Assets" has the meaning provided such term in the Initial Contribution Agreement as well as the assets described on Schedule 1-1.

"<u>Excluded Liabilities</u>" means (a) (i) any and all Income Taxes imposed on Antero or any of its Affiliates; (ii) the Asset Taxes allocable to Antero pursuant to <u>Section 5.7(a)</u>; (iii) any Taxes imposed on or with respect to the Excluded Assets; and (iv) any and all other Taxes

4

imposed on or with respect to the Water Assets for any taxable period (or portion thereof) ending on or prior to the Effective Date; and (b) the liabilities described on <u>Schedule 1</u>.

"<u>Final Determination</u>" means (a) a decision, judgment, decree or other order by any court of competent jurisdiction, which decision, judgment, decree or other order has become final, (b) a closing agreement made under Section 7121 of the Code (or a comparable agreement under the laws of a state, local or foreign taxing jurisdiction) with the relevant Tax Authority or other administrative settlement with or final administrative decision by the relevant Tax Authority, (c) a final disposition of a claim for refund, or (d) any agreement between Antero and the Partnership which they agree will have the same effect as an item in (a), (b), or (c) for purposes of this Agreement.

"Financial Statements" has the meaning set forth in Section 3.14(a).

"Financing" has the meaning set forth in Section 2.2.

"First Earn-Out Term" has the meaning set forth in Section 2.3(a).

"Fresh Water" means either (i) raw fresh water or (ii) a mixture of raw fresh water and Treated Waste Water.

"Fresh Water Distribution Business" has the meaning set forth in Section 3.14(a).

"Fundamental Representations" has the meaning set forth in Section 8.7.

"General Partner" has the meaning set forth in the recitals to this Agreement.

"Governmental Approval" has the meaning set forth in Section 3.4.

"<u>Governmental Authority</u>" means (a) the United States of America or any state or political subdivision thereof within the United States of America and (b) any court or any governmental or administrative department, commission, board, bureau or agency of the United States of America or of any state or political subdivision thereof within the United States of America.

"Hazardous Material" means (a) any "hazardous substance" as defined in the Comprehensive Environmental Response,

Compensation and Liability Act of 1980, as amended, (b) any "hazardous waste" as defined in the Resource Conservation and Recovery Act, as amended, (c) any petroleum or petroleum product, (d) any polychlorinated biphenyl and (e) any pollutant or contaminant or hazardous, dangerous or toxic chemical, material, waste or substance regulated under or within the meaning of any applicable Environmental Law.

"<u>Hydrocarbons</u>" means oil and gas and other hydrocarbons produced in association therewith (whether in liquid or gaseous form), or any combination thereof, and any minerals produced in association therewith.

"Income Taxes" means any federal, state, local or foreign Taxes measured by or imposed on net income, gross revenue or receipts, including franchise or similar Taxes.

5

"Indemnity Claim" has the meaning set forth in Section 8.3.

"<u>Intellectual Property</u>" means (a) patents and patent applications; (b) trade secrets and confidential information, (c) copyrights, registered and unregistered; and (d) trademarks, service marks, trade names, trade dress, and domain names ("<u>Trademarks</u>").

"Initial Contribution Agreement" has the meaning set forth in the preamble to this Agreement.

"Investment Company Act" has the meaning set forth in Section 3.18.

"Law" means all laws (including common law), statutes, rules, regulations, ordinances, directives, orders or any similar provisions having the force or effect of Law of any applicable Governmental Authority.

"Liability" or "Liabilities" means any direct or indirect liability, indebtedness, Damage, deficiency, Tax, interest, penalty, amount paid in settlement, judgment, assessment, guaranty or endorsement of or by any Person, in the case of each of the foregoing, whether absolute or contingent, matured or unmatured, asserted or unasserted, accrued or unaccrued, due or to become due, liquidated or unliquidated.

"Lien" means any security interest, lien, deed of trust, mortgage, pledge, charge, claim, restriction, easement, encumbrance or other similar interest or right.

"Litigation" has the meaning set forth in Section 3.5.

"<u>Material Adverse Effect</u>" means any change, circumstance, effect or condition that, individually or in the aggregate, (a) is, or could reasonably be expected to be, materially adverse to the business, financial condition, assets, liabilities or results of operations of the Business or (b) materially adversely affects, or could reasonably be expected to materially adversely affect, Antero's ability to satisfy its obligations under the Transaction Documents, other than any changes (x) in the general state of the industries in which the Business operates or (y) in general economic conditions (including changes in commodity prices or interest rates), financial or securities markets or political conditions, *provided*, that in the case of clauses (x) and (y), the impact on the Business is not materially disproportionate to the impact on companies engaged in similar lines of business as the Business.

"<u>Material Contract</u>" means (a) any Contract (or series of related Contracts) relating to the ownership or operation of the Business or the ownership, use or operation of the Water Assets that, as of the date hereof, is reasonably expected to provide for revenues to or commitments of Antero or its Affiliates in an amount greater than \$500,000 during any calendar year or (b) any other Contract (other than any Contract granting any Permits, servitudes, easements or rights-of-way) materially affecting the ownership or operation of the Business or the ownership, use or operation of the Water Assets, the loss of which could, individually or in the aggregate, have a Material Adverse Effect.

6

"<u>MVC Credit Volumes</u>" shall have the meaning given to such term in the Water Services Agreement; provided, however, that the reference therein to "2019" shall be amended to state "2020" for purposes of this Agreement.

"New Common Units" has the meaning set forth in Section 2.2.

"Non-Hydrocarbon Systems" has the meaning set forth in the definition of Water Assets.

"Organizational Documents" means, with respect to any Person, the articles of incorporation, certificate of incorporation, certificate of limited partnership, bylaws, limited liability company agreement, operating agreement, partnership agreement, stockholders' agreement, and all other similar documents, instruments or certificates executed, adopted or filed in connection with the creation, formation or organization of such Person, including any amendments thereto.

"Partnership" has the meaning set forth in the preamble to this Agreement.

"Partnership Acquiring Entities" has the meaning set forth in the preamble to this Agreement.

"Partnership Closing Certificate" has the meaning set forth in Section 6.3(c).

"Partnership Financial Advisor" has the meaning set forth in the recitals to this Agreement.

"Partnership Fundamental Representations" has the meaning set forth in Section 8.7(b).

"Partnership Group" means, collectively, the Partnership and its Subsidiaries.

"Partnership Indemnitees" has the meaning set forth in Section 8.2.

"<u>Partnership Material Adverse Effect</u>" means any change, circumstance, effect or condition that (a) is, or could reasonably be expected to be, materially adverse to the business, financial condition, assets, liabilities or results of operations of the Partnership Group, taken as a whole or (b) materially adversely affects, or could reasonably be expected to materially adversely affect, the Partnership's ability to satisfy its obligations under the Transaction Documents, other than any changes (x) in the general state of the industries in which the Partnership operates or (y) in general economic conditions (including changes in commodity prices or interest rates), financial or securities markets or political conditions, *provided*, that in the case of clauses (x) and (y), the impact on the Partnership is not materially disproportionate to the impact on companies engaged in similar lines of business as the Partnership.

"Party" or "Parties" has the meaning set forth in the preamble to this Agreement.

"<u>Permits</u>" means permits, licenses, certificates, orders, approvals, authorizations, grants, consents, concessions, warrants, franchises and similar rights and privileges.

"Permitted Liens" has the meaning set forth in Section 3.8(b).

7

"<u>Person</u>" means an individual or a corporation, firm, limited liability company, partnership, joint venture, trust, unincorporated organization, association, government agency or political subdivision thereof or other entity.

"Required Consent" has the meaning set forth in Section 5.8(c).

"Second Earn-Out Term" has the meaning set forth in Section 2.3(a).

"Secondment Agreement" means that certain Secondment Agreement in the form attached as Exhibit D hereto.

"Securities Act" has the meaning set forth in Section 3.12.

"Services Agreement" means that certain Amended and Restated Services Agreement in the form attached as Exhibit E hereto.

"Special Committee" has the meaning set forth in the recitals to this Agreement.

"Straddle Period" means any Tax period beginning on or before and ending after the Effective Date.

"Subsidiary" means, with respect to any Person, (a) a corporation of which more than 50% of the voting power of shares entitled (without regard to the occurrence of any contingency) to vote in the election of directors or other governing body of such corporation is owned, directly or indirectly, at the date of determination, by such Person, by one or more Subsidiaries of such Person or a combination thereof, (b) a partnership (whether general or limited) in which such Person or a Subsidiary of such Person is, at the date of determination, a general or limited partner of such partnership, but only if more than 50% of the partnership interests of such partnership (considering all of the partnership interests of the partnership as a single class) is owned, directly or indirectly, at the date of determination, by such Person, by one or more Subsidiaries of such Person, or a combination thereof; or (c) any other Person (other than a corporation or a partnership) in which such Person, one or more Subsidiaries of such Person, or a combination thereof; directly or indirectly or indirectly, at the date of determination, has (i) at least a majority ownership interest or (ii) the power to elect or direct the election of a majority of the directors or other governing body of such Person; *provided, however*, that with respect to Antero, the term "Subsidiary" shall not include any member of the Partnership Group.

"<u>Tax</u>" or "<u>Taxes</u>" means any federal, state, local or foreign income tax, ad valorem tax, excise tax, sales tax, use tax, franchise tax, real or personal property tax, transfer tax, gross receipts tax or other tax, assessment, duty, fee, levy or other governmental charge, together with and including, any and all interest, fines, penalties, assessments, and additions to Tax resulting from, relating to, or incurred in connection with any of those or any contest or dispute thereof.

"Tax Authority" means any Governmental Authority having jurisdiction over the payment or reporting of any Tax.

"Tax Proceeding" has the meaning set forth in Section 5.7(d).

Authority in connection with Taxes.

"Total Accounts Payable" means the sum of Antero Water's accounts payable — capital plus its accounts payable — expense.

"<u>Total Accrued Expenses</u>" means the sum of Antero Water's (a) accrued capital plus (b) accrued operating expenses plus (c) accrued expenses — interest plus (d) accrued expenses — ad valorem tax plus (e) accrued expenses — audit/tax plus (f) accrued expenses — other.

"Total Cash Proceeds" has the meaning set forth in Section 5.71(e)(i).

"Traceable Partnership Debt" has the meaning set forth in Section 5.7(e)(i).

"Trademarks" has the meaning set forth in the definition of Intellectual Property.

"<u>Transaction Documents</u>" means this Agreement, the Initial Contribution Agreement, the Water Services Agreement, the Secondment Agreement, the Services Agreement and each of the other documents and certificates to be delivered at Closing pursuant to <u>Section 7.2</u> and <u>Section 7.3</u> hereof.

"Treated Waste Water" means treated waste water, but excluding water remaining after treatment at the Treatment Facility.

"<u>Treatment Assets</u>" means the assets, contracts, rights, permits and properties owned or leased by Antero and used primarily in connection with the construction, ownership, operation, use or maintenance of Antero's advanced wastewater treatment complex to be constructed in Doddridge County, West Virginia (the "<u>Treatment Facility</u>"), including those assets set forth on <u>Schedule 1-2</u> hereto.

"Treatment Facility" has the meaning set forth in Article I of this Agreement under the definition of "Treatment Assets."

"Transfer Taxes" has the meaning set forth in Section 2.4.

"Unscheduled Consent" has the meaning set forth in Section 5.8(b).

"Water Services Agreement" means that certain Water Services Agreement in the form attached as Exhibit C hereto.

"Water Assets" means all assets in respect of the water business of Antero, including the following:

(a) all assets and systems held by Antero or an Affiliate of Antero for the gathering or transportation of water, carbon dioxide or other non-Hydrocarbons, or the treatment, transportation, handling or disposal of waste water or other fluid waste, including the systems described on

9

Schedule 1-3 (the foregoing, collectively, the "Non-Hydrocarbon Systems");

- (b) all easements, surface use agreements, servitudes, third party permits, licenses, surface leases, sub-surface leases, rights-of-way, grazing rights, logging rights and other similar interests to the extent relating to surface operations or use or occupancy of the surface or the subsurface applicable to the Non-Hydrocarbon Systems, including the instruments and agreements described on <u>Schedule 1-3</u> (the foregoing, collectively, the "<u>Water</u> <u>Easements</u>"), to the extent reasonably determined by Antero to be assignable;
- (c) all fee and leasehold interests in real property that relate primarily to the ownership or operation of the assets described in the other clauses of this definition, including the fee and leasehold interests described on <u>Schedule 1-3</u> (the Water Easements and the interests described in this clause (c), collectively, the "<u>Water Lands</u>", and the leases under which the leasehold interests described in this clause (c) are, collectively, the "<u>Water Lands</u>"; <u>Leases</u>");
- (d) all of the Water Personal Property, including the facilities and equipment described on <u>Schedule 1-3</u> (the foregoing, collectively, the "<u>Water Equipment</u>");
- (e) all Water Related Contracts, to the extent reasonably determined by Antero to be assignable;
- (f) all Permits issued to or held by Antero or any of its Affiliates in connection with Antero's or its Affiliates' ownership or operation of the other assets described in this definition, including those Permits described on <u>Schedule 1-3</u> (the foregoing, collectively, the "<u>Water Permits</u>"), to the extent reasonably determined by Antero to be assignable;
- (g) all of Antero's and its Affiliates rights, claims and causes of action (including warranty and similar claims that may be made against a third party vendor under a master service agreement or any other Water Related Contract) to the extent, and only to the extent, that such rights, claims or causes of action (i) are associated with the Water Assets and relate to the period of time from and after the acquisition of the Water Assets by the Partnership or (ii) relate to the liabilities to be assumed by the Partnership pursuant to any purchase of the Water Assets (excluding any such rights, claims and causes of action that arise from or are related to the ownership by Antero

or its Affiliates of, or the rights of Antero or its Affiliates in respect of, the Excluded Assets) (the foregoing, collectively, the "<u>Water Claims</u>"), to the extent reasonably determined by Antero to be assignable;

(h) all prepaid expenses (other than Taxes) attributable to the Water Assets that are paid by or on behalf of Antero or its Affiliates and are attributable

to the periods of time on and after the acquisition of the Water Assets by the Partnership, including prepaid utility charges;

- (i) the Water Conveyed IP;
- (j) all of Antero's and its Affiliates' files, records and data directly and primarily relating to the items described in the preceding clauses above, including title records (including title opinions and curative documents), surveys, maps and drawings, operating data and records, maintenance records, and correspondence, including any Intellectual Property (other than Trademarks) held by Antero therein, except (i) to the extent the transfer, delivery or copying of such records may be restricted by contract with a third party or subject to a fee; (ii) all documents and instruments of Antero that may be protected by the attorney-client privilege; and (iii) all accounting and Tax files, books, records, Tax Returns and Tax work papers related to such items (the foregoing, collectively, the "<u>Water Records</u>"); and
- (k) the Treatment Assets.

Notwithstanding the foregoing, "Water Assets" shall not include any Excluded Assets, Excluded Liabilities or any Gathering and Compression Assets (as defined in the Initial Contribution Agreement).

"Water Capital Expenditures" has the meaning set forth in Section 5.7(f).

"Water Claims" has the meaning set forth in the definition of Water Assets.

"<u>Water Conveyed IP</u>" means (a) all Intellectual Property (other than Trademarks) owned by Antero or its Affiliates primarily relating to the operation of the Non-Hydrocarbon Systems and (b) the Intellectual Property described in <u>Schedule 1-3</u>, including all rights, claims and causes of action for past, present and future infringement and misappropriation of the Water Conveyed IP, including the right to seek injunctive relief and damages, and to collect and retain same.

"<u>Water Credit Facility</u>" means that certain Credit Agreement among Antero Water, Antero and certain Subsidiaries of Antero, as guarantors, certain lenders party thereto, JPMorgan Bank, N.A., as administrative agent, and the other parties thereto.

"Water Easements" has the meaning set forth in the definition of Water Assets.

"Water Equipment" has the meaning set forth in the definition of Water Assets.

"<u>Water Interests</u>" has the meaning set forth in the preamble to this Agreement.

"Water Lands" has the meaning set forth in the definition of Water Assets.

"Water Leases" has the meaning set forth in the definition of Water Assets.

"Water Option" has the meaning set forth in the preamble to this Agreement.

"Water Permits" has the meaning set forth in the definition of Water Assets.

"<u>Water Personal Property</u>" means the equipment, structures, fixtures, improvements, equipment, storage tanks, pipelines, manifolds, casing, tubing, pumps, motors, machinery, flow lines and other items of every kind and nature primarily used or held for use in connection with the Water Assets.

"Water Records" has the meaning set forth in the definition of Water Assets.

"<u>Water Related Contracts</u>" means those transportation agreements, operating agreements, balancing agreements, facilities or equipment leases, interconnection agreements, service and parts agreements and all other contracts to which Antero or an Affiliate of Antero is a party, that relate primarily to the ownership or operation of the Water Assets and that will be binding on the Partnership or any of the Water Assets after the acquisition by the Partnership, including the contracts set forth on <u>Schedule 1-3</u>, but excluding any contract that is an Excluded Asset.

#### CONTRIBUTIONS, CONVEYANCES, ACKNOWLEDGMENTS AND DISTRIBUTIONS

#### 2.1 <u>Contributions</u>.

(a) Prior to the Closing, Antero shall use reasonable best efforts, subject to its obligations in <u>Section 5.8</u>, to cause the contribution, assignment, transfer and conveyance of all Water Assets (other than the Treatment Assets) and related liabilities and obligations to Antero Water to the extent such Water Assets and related liabilities and obligations are not currently held by Antero Water. At the Closing and deemed effective as of the Effective Date, on the terms and subject to the conditions of this Agreement, Antero shall (i) contribute, assign, transfer and convey to the Partnership the Water Interests and (ii) subject to its obligations in <u>Section 5.8</u>, contribute, assign, transfer and convey to Antero Treatment the Treatment Assets and related liabilities and obligations, in exchange for the consideration set forth in <u>Section 2.2</u>, and (x) the Partnership shall accept the contribution of the Water Interests and (y) Antero Treatment shall accept the contribution of the Treatment Assets and shall assume the related liabilities and obligations. Antero acknowledges and agrees that the Excluded Liabilities will be retained by Antero and not assumed by the Partnership or Antero Treatment.

(b) <u>Net Working Capital</u>. Antero shall cause the Antero Water Net Working Capital to be \$0 as of the Effective Date.

2.2 <u>Consideration</u>. At the Closing, in consideration for the contribution of the Contributed Assets, the Partnership shall: (a) pay to Antero a cash distribution equal to \$552,450,000, less any indebtedness (including principal and accrued interest) outstanding as of the Closing under the Water Credit Facility and assumed by the Partnership in connection with the Closing (the "<u>Cash</u> <u>Consideration</u>") and (b) issue 23,886,421 common units representing limited partner interests in the Partnership (the "<u>New Common</u> <u>Units</u>") to Antero. In connection with the execution of this Agreement, the Partnership entered into definitive documentation to issue and sell common units representing limited partner interests in the Partnership in a private

12

placement (the "Financing"). The Financing is expected to be consummated simultaneously with the Closing. The Partnership agrees to pay to Antero an amount equal to the net proceeds received from the Financing (the "Additional Cash Consideration") as an increase to the Cash Consideration in exchange for a reduction in the number of New Common Units issuable to Antero at the Closing equal to the number of common units issued and sold in the Financing; provided, however, that notwithstanding the foregoing, the number of New Common Units issuable to Antero at the Closing pursuant to this Section 2.2 shall not be less than 4,777,284 New Common Units and the number of New Common Units issuable pursuant to the Financing shall not exceed 19,109,137. The Earn-Out Payments, if any, shall be paid by the Partnership in accordance with Section 2.3 and shall be deemed to be additional consideration for the Contributed Assets only if and when paid to Antero in accordance with this Agreement. The Cash Consideration (for the avoidance of doubt, including any Additional Cash Consideration) and the Earn-Out Payments, if any, shall be sourced by the Partnership from the proceeds of the Traceable Partnership Debt to the extent provided in Section 5.7(e). Notwithstanding Section 5.4 or anything else to the contrary in this Agreement, Antero agrees to retain and to pay all obligations for (i) capital expenditures incurred by Antero or its Affiliates and associated with the design and construction of the Treatment Facility that were accrued prior to July 31, 2015 and (ii) capital expenditures incurred by Antero or its Affiliates or the Partnership and its Affiliates, in each case associated with the design and construction of the Treatment Facility that were accrued after July 31, 2015 up to a maximum amount pursuant to this clause (ii) of \$12,000,000 (and Antero shall reimburse the Partnership for any capital expenditures paid by the Partnership that are within the scope of this clause (ii)). The Partnership shall reimburse Antero for any capital expenditures funded prior to the Closing by Antero or its Affiliates associated with the design, construction, ownership, operation, use or maintenance of the Treatment Facility in excess of the \$12,000,000 limit referenced in clause (ii) of the immediately preceding sentence by increasing the Cash Consideration paid at Closing by such amount. Antero shall provide the Partnership with a written description and amount of such capital expenditures no less than three days prior to the Closing.

#### 2.3 <u>Earn-Out Payments</u>.

(a) Following the Closing and as additional consideration for the contribution by Antero to the Partnership and Antero Treatment of the Contributed Assets, Antero shall be entitled to receive from the Partnership (subject to the terms and conditions of this <u>Section 2.3</u>) cash determined in accordance with this <u>Section 2.3</u> (the "<u>Earn-Out Payments</u>"). In addition to the consideration set forth in <u>Section 2.2</u> above, Antero shall be entitled to a cash payment equal to (i) \$125,000,000 if the aggregate Delivered Volumes during the period beginning on January 1, 2017 and ending on December 31, 2019 (the "<u>First Earn-Out Term</u>"), is greater than or equal to 176,295,000 barrels and (ii) an additional \$125,000,000 if the Delivered Volumes during the period beginning on January 1, 2018 and ending on December 31, 2020 (the "<u>Second Earn-Out Term</u>"), is greater than or equal to 219,200,000 barrels.

(b) Within 30 days after the end of the applicable Earn-Out Term, Antero shall in good faith prepare and deliver to the Partnership a report setting forth Antero's calculation of the Delivered Volumes during such Earn-Out Term as well as any other information that the Partnership may reasonably request in order to verify such Delivered Volumes (each such report, an "Earn-Out Report"). The Earn-Out Report and the Delivered

Volumes for the Earn-Out Term reflected thereon shall be final and binding upon Antero and the Partnership upon the approval of such Earn-Out Report by the Partnership, in a written notice that specifically states such approval and references this <u>Section 2.3(b)</u>, or the failure of the Partnership to object in a written notice to Antero that specifically references this <u>Section 2.3(b)</u> within 15 days after receipt of the Earn-Out Report by the Partnership. If the Partnership does not agree with the Earn-Out Report and the calculation of the Delivered Volumes stated thereon, and Antero and the Partnership cannot mutually agree on the calculation of Delivered Volumes for

such Earn-Out Term, then within 45 days following receipt by the Partnership of the Earn-Out Report, Antero and the Partnership shall resolve such dispute (the "<u>Earn-Out Report Dispute</u>") in accordance with <u>Section 2.3(c)</u>. If at any time the Partnership and Antero resolve the Earn-Out Report Dispute, then notwithstanding the preceding provisions of this <u>Section 2.3(b)</u> and <u>Section 2.3(c)</u>, the dispute resolution procedures in <u>Section 2.3(c)</u> promptly shall be discontinued with respect to such agreed matters and the Earn-Out Report and the Delivered Volumes shall be revised, if necessary to reflect such resolution and thereupon shall become binding upon Antero and the Partnership for purposes hereof. Each Earn-Out Payment, if any, that the Partnership becomes obligated to pay to Antero (by wire transfer of immediately available funds to an account or accounts specified by Antero) pursuant to this <u>Section 2.3</u> shall be paid to Antero within five Business Days of the determination of the Earn-Out Payment becoming final and binding upon the Parties in accordance with Section 2.3(b).

(c) If Antero and the Partnership are unable to resolve an Earn-Out Report Dispute as contemplated by <u>Section 2.3(b)</u>, then such Earn-Out Report Dispute shall be submitted to mandatory and binding arbitration pursuant to the following conditions:

(i) <u>Procedures</u>. The arbitration shall be conducted pursuant to the Federal Arbitration Act and the Commercial Arbitration Rules of the AAA, as they may be amended from time to time, except as expressly provided in this <u>Section 2.3(c)</u> (the "<u>AAA Rules</u>") with the arbitrators selected pursuant to the procedures set forth in this <u>Section 2.3(c)</u> (each, an "<u>Arbitrator</u>"). In resolving the substance of the Earn-Out Report Dispute, the Arbitrators shall apply substantive Colorado Law; provided, however, that no aspect of the Colorado Dispute Resolution Act shall be applied to the arbitration proceeding.

(ii) <u>Selection of Arbitrators</u>. The Earn-Out Report Dispute will be resolved by a panel of three Arbitrators, unless Antero and the Partnership mutually agree upon the selection of a single Arbitrator. Within 15 days after a failure to resolve any dispute regarding the calculation of the Delivered Volumes in an Earn-Out Report in accordance with <u>Section 2.3(b)</u>, each of Antero and the Partnership shall appoint one person to serve as an Arbitrator and, within 10 days after the later to occur of such appointments, the two appointed Arbitrators shall mutually agree to appoint a third Arbitrator, failing agreement on which such third Arbitrator shall be appointed by AAA in accordance with its rules. If either Antero or the Partnership does not appoint an Arbitrator within 15 days after such failure to resolve any dispute regarding the calculation of the Delivered Volumes in an Earn-Out Report in accordance with <u>Section 2.3(b)</u>, then the other party may submit an appropriate request to the AAA to initiate proceedings and AAA shall select such Arbitrator in accordance with its rules. Should any Arbitrator refuse or

14

be unable to proceed with arbitration proceedings as called for in this <u>Section 2.3(c)</u>, such Arbitrator shall be replaced by AAA in accordance with <u>Section 2.3(c)(viii)</u> below.

(iii) <u>Place of Arbitration</u>. The arbitration shall be held in Denver, Colorado.

(iv) <u>Conduct of Arbitration</u>. The Arbitrators shall allow reasonable discovery in the forms permitted by the AAA Rules, to the extent consistent with the purposes of the arbitration. Recognizing the express desire of the parties for an expeditious means of dispute resolution, the Arbitrators shall limit or allow the parties to expand the scope of discovery as may be reasonable and necessary under the circumstances. The arbitration hearing shall be commenced promptly and conducted expeditiously. If one party seeks discovery of electronic communications, such as emails, from another party, the Arbitrators shall have the authority to impose reasonable limits on the timing and extent of such discovery, and allow the producing party to recover reasonable expenses and costs associated with that discovery. Unless otherwise agreed by the parties, the arbitration hearing shall be conducted on consecutive days. The Arbitrators must give effect to legal privileges including the attorney-client privilege and work-product immunity. The Arbitrators shall act by majority vote in resolving all Earn-Out Report Disputes.

(v) <u>Arbitration Award</u>. The Arbitrators shall endeavor to render a binding, written decision within 14 days following the completion of the arbitration hearing.

(vi) <u>Binding Nature of the Arbitration Award</u>. The arbitration award shall be binding on the parties, and judgment thereon may be entered in any court of competent jurisdiction, and may not be appealed except to the extent permitted by the Federal Arbitration Act.

(vii) <u>Time of the Essence</u>. The Arbitrators are to be instructed that time is of the essence in the arbitration proceeding.

(viii) <u>Replacement of Arbitrator</u>. Should any Arbitrator refuse or be unable to proceed with arbitration proceedings as called for by this <u>Section 2.3(c)</u>, such Arbitrator shall be replaced in the same manner by which he or she was appointed (e.g., if a party appointed the departing Arbitrator, that party would appoint the departing Arbitrator's replacement, and if the two Party-appointed Arbitrators appointed the departing Arbitrator, then such party-appointed Arbitrator's replacement).

(ix) <u>Confidentiality</u>. To the fullest extent permitted by the Laws of the State of Colorado, the arbitration proceedings and award shall be maintained in confidence by the parties.

(x) <u>Expenses</u>. The Arbitrators shall have the authority to award attorneys' fees and expenses to the prevailing party.

# (xi) THE PARTIES EXPRESSLY ACKNOWLEDGE AND AGREE THAT IN ENTERING INTO THIS <u>SECTION 2.3(C)</u>, THEY ARE KNOWINGLY AND VOLUNTARILY WAIVING THEIR RIGHTS TO A JURY TRIAL.

2.4 <u>Transfer Taxes</u>. To the extent that any sales, use, transfer, purchase, filing, recordation, stamp, registration and similar Taxes are payable as a result of the transactions contemplated by this Agreement (collectively, "<u>Transfer Taxes</u>"), such Transfer Taxes shall be borne fifty percent (50%) by Antero and fifty percent (50%) by the Partnership. To the extent under applicable Law the transferee is responsible for filing Tax Returns or other documentation in respect of Transfer Taxes, the Partnership shall prepare and file all such Tax Returns or other documentation. The Parties shall provide such certificates and other information and otherwise cooperate in good faith to minimize, to the extent permitted under applicable Law, any Transfer Taxes. The Party that is not responsible under applicable Law for paying the Transfer Taxes shall pay its share of the Transfer Taxes to the responsible Party prior to the due date of such Transfer Taxes.

#### ARTICLE III REPRESENTATIONS AND WARRANTIES OF ANTERO

Antero hereby represents and warrants to the Partnership Acquiring Entities that, as of the date hereof:

#### 3.1 <u>Organization and Existence</u>.

(a) Antero has been duly organized and is validly existing and in good standing under the laws of the State of Delaware, with full corporate power and authority to own, lease and operate the properties and assets it now owns, leases and operates and to carry on its business as and where such properties and assets are now owned or held and such business is now conducted. Antero is duly qualified to transact business and is in good standing as a foreign entity in each other jurisdiction in which such qualification is required for the conduct of its business, except where the failure to so qualify or to be in good standing does not have a Material Adverse Effect.

(b) Antero Water has been duly organized and is validly existing and in good standing under the laws of the jurisdiction of its formation, with full limited liability company power and authority to own, lease, use and operate the properties and assets it now owns, leases, uses and operates and to carry on its business as and where such properties and assets are now owned or held and such business is now conducted. Antero Water is duly qualified to transact business and is in good standing as a foreign entity in each other jurisdiction in which such qualification is required for the conduct of its business, except where the failure to so qualify or to be in good standing does not have a Material Adverse Effect. Antero has delivered to the Partnership correct and complete copies of Antero Water's Organizational Documents, as

16

amended to date. There is no pending, or to the knowledge of Antero, threatened, action for the dissolution, liquidation or insolvency of Antero Water.

3.2 <u>Authority and Approval; Enforceability</u>. Antero has the corporate power and authority to execute and deliver this Agreement and each of Antero and its Subsidiaries has the corporate or limited liability company power and authority to execute and deliver any other Transaction Document to which it is or will be a party, to consummate the transactions contemplated hereby and thereby and to perform all the terms and conditions hereof and thereof to be performed by it. The execution and delivery by Antero of this Agreement and the execution and delivery by each of Antero and its Subsidiaries of any other Transaction Document to which it is or will be a party, the performance by Antero or its applicable Subsidiary of all the terms and conditions hereof to be performed by it and the consummation of the transactions contemplated hereby and thereby have been duly authorized and approved by all requisite corporate or limited liability company action of Antero and any applicable Subsidiary of Antero. Each of this Agreement and any other Transaction Document to which Antero or any Subsidiary of Antero is or will be a party constitutes or will constitute, upon execution and delivery by Antero or such applicable Subsidiary of Antero is or will be a party constitutes or such Subsidiary of Antero, enforceable against Antero or such Subsidiary of Antero in accordance with its terms, except as such enforcement may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting enforcement of creditors' rights generally and by general principles of equity (whether applied in a proceeding at law or in equity).

3.3 <u>No Conflict</u>. Other than as set forth on <u>Schedule 3.3</u> or <u>Schedule 3.4</u>, this Agreement, the other Transaction Documents to which Antero or any Subsidiary of Antero is or will be a party, the execution and delivery hereof and thereof by Antero or any Subsidiary of Antero do not and will not, and the fulfillment and compliance with the terms and conditions hereof and thereof and the consummation of the transactions contemplated hereby and thereby will not:

(a) conflict with any of the provisions of the Organizational Documents of Antero or any of its Subsidiaries;

(b) conflict with any provision of any law or administrative regulation or any judicial, administrative or arbitration order, award, judgment, writ, injunction or decree applicable to Antero or any of its Subsidiaries;

(c) conflict with, result in a breach of, constitute a default under (whether with notice or the lapse of time or both) or accelerate or permit the acceleration of the performance required by, or require any consent, authorization or approval under, or give others the right to terminate, in each case, any indenture, mortgage, Lien or Contract to which Antero or any of its Subsidiaries is a party or by which any of them is bound or to which any of the Water Interests or any of the Water Assets are subject or that could prevent or materially delay the consummation of the transactions contemplated by the Transaction Documents;

(d) result in the creation of, or afford any Person the right to obtain, any material Lien on the capital stock or other equity interests, property or assets of Antero or any of its Subsidiaries; or

(e) result in the revocation, cancellation, suspension or material modification, individually or in the aggregate, of any Governmental Approval possessed by Antero or any of its Subsidiaries that is necessary or desirable for the ownership, lease or operation of the Water Assets or the Business as conducted as of the date of this Agreement, including any Governmental Approvals under any applicable Environmental Law;

except, in the case of <u>clauses (b)</u>, (c), (d) and (e), as would not have, individually or in the aggregate, a Material Adverse Effect and except, in each case, for such matters as will have been cured at or prior to the Closing.

3.4 <u>Consents</u>. Except as set forth in <u>Schedule 3.4</u>, and except for notice to, or consent of, Governmental Authorities related to the transfer of environmental Permits, no consent, approval, license, permit, order, waiver, or authorization of, or registration, declaration, or filing with any Governmental Authority (each a "<u>Governmental Approval</u>") or other Person is required to be obtained or made by or with respect to Antero or any of its Subsidiaries in connection with:

(a) the execution, delivery, and performance of this Agreement or the other Transaction Documents, or the consummation of the transactions contemplated hereby and thereby; or

(b) the enforcement against Antero or any of its Subsidiaries of its obligations under this Agreement or the other Transaction Documents;

except, in each case, as would not have, individually or in the aggregate, a Material Adverse Effect (each of the foregoing, a "Consent").

3.5 Laws and Regulations: Litigation. Except as set forth in <u>Schedule 3.5</u>, there are no pending or, to Antero's knowledge, threatened claims, fines, actions, suits, demands, investigations or proceedings or any arbitration or binding dispute resolution proceeding (collectively, "<u>Litigation</u>") against Antero Water, or against or affecting the Business or the Water Assets or the ownership and operation of the Business or the Water Assets (other than Litigation under any Environmental Law, which is the subject of <u>Section 3.6</u>) that (i) would individually, or in the aggregate, have a Material Adverse Effect or (ii) seek any material injunctive relief with respect to the Business or the Water Assets. Except as would not, individually or in the aggregate, have a Material Adverse Effect, Antero Water is not in violation of or in default under any law or regulation or under any order (other than Environmental Laws, which are the subject of <u>Section 3.6</u>) of any Governmental Authority applicable to it. Except as would not, individually or in the aggregate, have a Material Adverse Effect, no Litigation is pending or, to Antero's knowledge, threatened to which Antero or any of its Subsidiaries is or may become a party that questions or involves the validity or enforceability of any of its respective obligations under this Agreement or seeks to prevent or delay, or seeks substantial damages in connection with, the consummation of the transactions contemplated hereby.

18

3.6 <u>Environmental Matters</u>. Except as set forth in <u>Schedule 3.6</u>, with respect to the Business or Antero Water, except as would not, individually or in the aggregate, have a Material Adverse Effect, Antero and its Subsidiaries (i) are in compliance with Environmental Laws, (ii) are not the subject of any outstanding administrative or judicial order of judgment, agreement or arbitration award from any Governmental Authority under any Environmental Law relating to the Water Assets or Antero Water and requiring remediation or the payment of a fine or penalty, (iii) have received all Permits required of them under applicable Environmental Laws necessary to conduct the Business as presently conducted or in light of the current stage of development or construction of the Water Assets, (iv) are in compliance with all terms and conditions of any such Permits, (v) are not subject to any pending Litigation involving any Environmental Law with respect to which Antero or any of its Subsidiaries has been contacted in writing, and (vi) do not have any liability in connection with the release into the environment of any Hazardous Material.

#### 3.7 <u>Water Interests</u>.

(a) The Water Interests (i) constitute 100% of the limited liability company interests in Antero Water and (ii) were duly authorized and validly issued and are fully paid and nonassessable (except as such nonassessability may be affected by Sections 18-607 and 18-804 of the Delaware Limited Liability Company Act). The Water Interests are not subject to and were not issued in violation of any purchase option, call option, right of first refusal, preemptive right, subscription right or any similar right under any provision of local or state law applicable to such interests, Antero Water's Organizational Documents, or any Contract to which Antero or any of its Subsidiaries is a party or to which it or any of its properties or assets is otherwise bound.

(b) As of immediately prior to the Closing, Antero has good and valid record and beneficial title to the Water Interests, free and clear of any and all Liens, and, except as provided or created by the Organizational Documents of Antero Water, the Securities Act or applicable securities laws, the Water Interests are free and clear of any restrictions on transfer or claims. There are no options, warrants, purchase rights, Contract rights or other securities exercisable or exchangeable for any equity interests of Antero Water, any other commitments or Contracts providing for the issuance of additional equity interests of Antero Water, or for the repurchase or redemption of the Water Interests, or any Contracts of any kind which may obligate Antero Water to issue, purchase, register for sale, redeem or otherwise acquire any of its equity interests. Immediately after the Closing, the Partnership will have good and valid record and beneficial title to such Water Interests, free and clear of any Liens (other than Permitted Liens) that have been created by, through or under Antero.

#### 3.8 <u>Water Assets</u>.

(a) At the Closing, the Water Assets, when considered together with the services provided by Antero and its Affiliates pursuant to the Services Agreement and the Secondment Agreement, will be sufficient to conduct the Business in a manner consistent in all material respects with the way the Business is conducted as of the date of this Agreement.

19

(b) Except as set forth in <u>Schedule 3.8(b)</u>, Antero and Antero Water, as applicable, have valid and indefeasible title in fee to all real property and interests in real property constituting part of the Water Assets and purported to be owned in fee, and good and valid title to the leasehold estates in all other real property and interests in real property (including rights of way) constituting part of the Water Assets, in each case except as would not have a Material Adverse Effect. Antero and Antero Water, as applicable, own all such real property and interests in real property free and clear of any Liens except (i) those set forth in <u>Schedule 3.8(b)</u>, (ii) mechanics', carriers', workmen's, repairmen's or other similar Liens arising or incurred in the ordinary course of business that are not yet delinquent or can be paid without penalty or are being contested in good faith and by appropriate proceedings in respect thereof and for which an appropriate reserve has been established in accordance with U.S. generally accepted accounting principles, (iii) Liens for current period Taxes that are not yet due and payable or are being contested in good faith and by appropriate proceedings in respect thereof and for which an appropriate reserve has been established in accordance with U.S. generally accepted accounting principles, (iv) Liens securing debt of Antero or any of its Subsidiaries that will be released prior to or as of the Closing and (v) other imperfections of title or encumbrances that, individually or in the aggregate, could not reasonably be expected to materially interfere with the ordinary conduct of the Business (the Liens described in clauses (i), (ii), (iii), (iv) and (v) above, being referred to collectively as "<u>Permitted Liens</u>").

(c) All tangible personal property included in the Water Assets is owned by Antero or Antero Water, as applicable, free and clear of all Liens except Permitted Liens. All tangible personal property included in the Water Assets is, in the aggregate, in good operating condition and repair (normal wear and tear excepted) and has been maintained in accordance with applicable laws and regulations, as well as generally accepted industry practice, and is sufficient for the purposes for which it is currently being used or held for use.

(d) <u>Schedule 1-2</u> sets forth, in all material respects, all rights, Contracts, Permits and related instruments associated with the Treatment Assets and all Liabilities associated therewith.

3.9 <u>Permits</u>. Antero or Antero Water, as applicable, holds or has a valid right to use all Permits (other than environmental Permits, which are the subject of <u>Section 3.6</u>) that are necessary for the conduct of the Business and the ownership and operation of the Water Assets in a manner consistent in all material respects with the way the Business is conducted as of the date of this Agreement, each in compliance with applicable laws and regulations of applicable Governmental Authorities, except for those Permits the failure of which to have would not, individually or in the aggregate, have a Material Adverse Effect. Antero or its Affiliates have complied in all material respects with all terms and conditions of such Permits.

3.10 <u>Insurance</u>. Antero or its Affiliates maintain policies of fire and casualty, liability and other forms of property and liability insurance related to the Water Assets and the Business in such amounts, with such deductibles, and against such risks and losses as are, in their judgment, reasonable for the Business and the Water Assets. All such policies are in full force and effect, all premiums due and payable thereon have been paid, and no notice of cancellation or termination has been received with respect to any such policy that has not been replaced on substantially similar terms prior to the date of such cancellation.

20

3.11 <u>Brokerage Arrangements</u>. Neither Antero nor any of its Affiliates has entered, directly or indirectly, into any Contract with any Person that would obligate the General Partner or any member of the Partnership Group to pay any commission, brokerage or "finder's fee" or other fee in connection with this Agreement, the other Transaction Documents or the transactions contemplated hereby or thereby.

3.12 Investment. Antero is an "accredited investor" as such term is defined in Rule 501 promulgated under the Securities Act, as amended (the "Securities Act"). Antero is not acquiring the New Common Units with a view to or for sale in connection with any distribution thereof or any other security related thereto within the meaning of the Securities Act. Antero is familiar with investments of the nature of the New Common Units, understands that this investment involves substantial risks, has adequately investigated the Partnership and the New Common Units, and has substantial knowledge and experience in financial and business matters such that it is capable of evaluating, and has evaluated, the merits and risks inherent in purchasing the New Common Units, and is able to bear the economic risks of such investment. Antero has had the opportunity to visit with the Partnership and meet with the officers of the General Partner and other representatives to discuss the business, assets, liabilities, financial condition, and operations of the Partnership and the New Common Units, and other information that Antero deems necessary or advisable to evaluate the Partnership and the New Common Units, and has made its own independent examination, investigation, analysis and evaluation of the Partnership and the

New Common Units, including its own estimate of the value of the New Common Units. Antero has undertaken such due diligence (including a review of the properties, liabilities, books, records and contracts of the Partnership) as Antero deems adequate. Antero acknowledges that the New Common Units have not been registered under applicable federal and state securities laws and that the New Common Units may not be sold, transferred, offered for sale, pledged, hypothecated or otherwise disposed of unless such transfer, sale, assignment, pledge, hypothecation or other disposition is registered under applicable federal and state securities laws or pursuant to an exemption from registration under any federal or state securities laws.

3.13 <u>Taxes</u>. Except as set forth on <u>Schedule 3.13</u> or as would not, individually or in the aggregate, have a Material Adverse Effect:

(a) all Tax Returns that are required to be filed by or with respect to Antero Water or the Water Assets on or prior to the Closing Date (taking into account any valid extension of time within which to file) have been or will be timely filed on or prior to the Closing Date and all such Tax Returns are or will be true, correct and complete in all material respects;

(b) all Taxes due and payable by or with respect to Antero Water or the Water Assets (whether or not shown on any Tax Return) have been fully paid and all deficiencies asserted or assessments made with respect to such Tax Returns have been paid in full or properly accrued for by Antero;

(c) no examination, audit, claim, assessment, levy, or administrative or judicial proceeding regarding any of the Tax Returns described in <u>Section 3.13(a)</u> or any Taxes

21

of or with respect to Antero Water or the Water Assets are currently pending or have been proposed in writing or have been threatened in writing;

(d) no waivers or extensions of statutes of limitations have been given or requested in writing with respect to any amount of Taxes of or with respect to Antero Water or the Water Assets or any Tax Returns of or with respect to Antero Water or the Water Assets;

(e) there are no Liens (other than Permitted Liens) on the Water Interests or the Water Assets that arose in connection with any failure (or alleged failure) to pay any Tax;

(f) Antero Water is not a party to or bound by any Tax allocation or sharing agreements;

(g) the Water Assets can be operated and the Business (as set forth in clause (b) of the definition thereof) can be conducted in accordance with the factual information presented in the supplemental private letter ruling request the Partnership submitted to the Internal Revenue Service on April 3, 2015 and the private letter ruling that the Internal Revenue Service issued to the Partnership on May 26, 2015. Antero expects that at least 90% of the combined gross income of the Partnership, Antero Water, Antero Treatment and the operations conducted with the Water Assets in 2015 after the Closing Date will be "qualifying income" within the meaning of Section 7704(d) of the Code; and

(h) Antero Water will, at Closing, be classified as an entity disregarded as separate from its owner for U.S. federal income Tax purposes in accordance with Treasury Regulation Section 301.7701-3.

3.14 <u>Financial Statements</u>.

(a) Antero has delivered to the Partnership balance sheets as of June 30, 2015 and December 31, 2014 and 2013 and statements of operations and comprehensive income, statements of equity and statements of cash flows for the six months ended June 30, 2015 and the years ended December 31, 2014 and 2013, in each case with respect to the fresh water distribution business (the "<u>Fresh Water Distribution Business</u>") that, to the Parties' knowledge, comprises the only "business" (as defined under Regulation S-X 11-01(d)) within the Business and the Water Assets (the "<u>Financial Statements</u>").

(b) The Financial Statements have been prepared from the books and records of Antero and its Subsidiaries in accordance with U.S. generally accepted accounting principles applied on a consistent basis throughout the periods covered thereby (except that the financial statements as of and for the six months ended June 30, 2015 do not include normal recurring year-end adjustments and footnotes required by U.S. generally accepted accounting principles for complete financial statements) and, to the knowledge of Antero, present fairly in all material respects the consolidated financial position, results of operations and cash flows of or attributable to the Fresh Water Distribution Business as of and for the periods stated in such Financial Statements.

(c) There are no liabilities or obligations related to the Fresh Water Distribution Business (whether accrued, absolute, contingent or otherwise) and there are no facts or circumstances that would result in any such liabilities or obligations, other than (i) liabilities or obligations reflected or reserved against in the Financial Statements, (ii) liabilities or obligations incurred in the ordinary course of business consistent with past practices since June 30, 2015, (iii) liabilities not required to be presented by U. S. general accepted accounting principles in unaudited financial statements, (v) liabilities or obligations under this Agreement and (vi) other

liabilities or obligations which, in the aggregate, would not have a Material Adverse Effect.

3.15 <u>Material Contracts</u>. Antero has made available to the Partnership a correct and complete copy of each Material Contract (including any amendments thereto). Each Material Contract is in full force and effect, and none of Antero, Antero Water, or, to the knowledge of Antero, any other party, is in breach or default thereunder and no event has occurred that upon receipt of notice or lapse of time or both would constitute any breach or default thereunder, except, in each case, for such breaches or defaults as would not, individually or in the aggregate, have a Material Adverse Effect. None of Antero or Antero Water has given or received from any third party any notice of any action or intent to terminate or amend in any material respect any Material Contract.

3.16 <u>No Adverse Changes</u>. Except as set forth in <u>Schedule 3.16</u> or as described in the Financial Statements, from June 30, 2015:

(a) there has not been a Material Adverse Effect; and

(b) the Business and the Water Assets have been operated and maintained in the ordinary course of business consistent with past practices or as otherwise required pursuant to the terms of this Agreement.

3.17 <u>Employees</u>. Antero Water has no employees.

3.18 <u>Investment Company Act</u>. Antero Water is not an "investment company" or a company "controlled" by an "investment company" within the meaning of the Investment Company Act of 1940, as amended, and the rules and regulations of the Securities and Exchange Commission thereunder (collectively, the "<u>Investment Company Act</u>") or a "business development company" (as defined in Section 2(a)(48) of the Investment Company Act).

3.19 <u>No Other Representations or Warranties; Schedules</u>. Except as set forth in this <u>Article III</u> and the Assignment and Bill of Sale of Treatment Assets, and except for any special warranty of title provided in any assignment or conveyance of the Water Assets from Antero (or any of its Affiliates) to Antero Water prior to the date hereof, neither Antero nor Antero Water makes any other express or implied representation or warranty with respect to the Contributed Assets, the Water Assets or the transactions contemplated by this Agreement, and disclaims any other representations or warranties. The disclosure of any matter or item in any schedule to this Agreement shall not be deemed to constitute an acknowledgment that any such matter is required to be disclosed.

23

#### ARTICLE IV REPRESENTATIONS AND WARRANTIES OF THE PARTNERSHIP ACQUIRING ENTITIES

The Partnership Acquiring Entities hereby represent and warrant to Antero that, as of the date hereof:

4.1 <u>Organization and Existence</u>. Each of the Partnership Acquiring Entities is an entity duly organized, validly existing and in good standing under the laws of the State of Delaware, and has all limited partnership or limited liability company power and authority, as applicable, to own the Water Interests and the Water Assets. Each of the Partnership Acquiring Entities is duly qualified to transact business as a limited partnership or limited liability company, as applicable, and is in good standing as a foreign entity in each other jurisdiction in which such qualification is required for the conduct of its business, except where the failure to so qualify or to be in good standing does not have a Partnership Material Adverse Effect. Antero Treatment will, at Closing, be classified as an entity disregarded as separate from the Partnership for U.S. federal income Tax purposes in accordance with Treasury Regulation Section 301.7701-3.

4.2 <u>Authority and Approval; Enforceability</u>. Each of the Partnership Acquiring Entities has the requisite power and authority to execute and deliver this Agreement and any other Transaction Document to which it is or will be a party, to consummate the transactions contemplated hereby and thereby and to perform all the terms and conditions hereof and thereof to be performed by it. The execution and delivery by each of the Partnership Acquiring Entities of this Agreement and any other Transaction Document to which it is or will be a party, the performance by it of all the terms and conditions hereof and thereof to be performed by it and the consummation of the transactions contemplated hereby and thereby have been duly authorized and approved by all requisite action of the Partnership Acquiring Entities is or will be a party constitutes or will constitute, upon execution and delivery by each of the Partnership Acquiring Entities is or will be a party constitutes or will constitute, upon execution and delivery by each of the Partnership Acquiring Entities, as applicable, the valid and binding obligation of each of the Partnership Acquiring Entities, as applicable, enforceable against each of the Partnership Acquiring Entities in accordance with its terms, except as such enforcement may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting enforcement of creditors' rights generally and by general principles of equity (whether applied in a proceeding at law or in equity).

4.3 <u>No Conflict</u>. Other than as set forth on <u>Schedule 4.3</u>, this Agreement, the other Transaction Documents to which the Partnership Acquiring Entities or any other member of the Partnership Group is or will be a party and the execution and delivery hereof and thereof by the Partnership Acquiring Entities or any other member of the Partnership Group do not and will not, and the fulfillment and compliance with the terms and conditions hereof and thereof and the consummation of the transactions contemplated hereby and thereby will not:

(a) conflict with any of the provisions of the Organizational Documents of the Partnership Acquiring Entities or any other member of the Partnership Group;

(b) conflict with any provision of any law or administrative regulation or any judicial, administrative or arbitration order, award, judgment, writ, injunction or decree applicable to the Partnership, the General Partner or any other member of the Partnership Group;

(c) conflict with, result in a breach of, constitute a default under (whether with notice or the lapse of time or both) or accelerate or permit the acceleration of the performance required by, or require any consent, authorization or approval under, or give others the right to terminate, in each case, any indenture, mortgage, Lien or Contract to which the Partnership Acquiring Entities or any other member of the Partnership Group is a party or by which any of them is bound;

(d) result in the creation of, or afford any Person the right to obtain, any material Lien on the capital stock or other equity interests, property or assets of the Partnership Acquiring Entities or any other member of the Partnership Group; or

(e) result in the revocation, cancellation, suspension or material modification, individually or in the aggregate, of any Governmental Approval (as defined below) possessed by the Partnership Acquiring Entities or any other member of the Partnership Group that is necessary or desirable for the ownership, lease or operation of its properties and other assets in the conduct of its business as now conducted, including any Governmental Approvals under any applicable Environmental Law;

except, in the case of <u>clauses (b)</u>, (c), (d) and (e), as would not have, individually or in the aggregate, a Partnership Material Adverse Effect and except, in each case, for such matters as will have been cured at or prior to the Closing.

4.4 <u>Brokerage Arrangements</u>. None of the Partnership Acquiring Entities has entered, directly or indirectly, into any Contract with any Person, other than the Partnership Financial Advisor (whose fees shall be paid by the Partnership), that would obligate Antero or any of its Affiliates to pay any commission, brokerage or "finder's fee" or other fee in connection with this Agreement or the transactions contemplated hereby.

4.5 <u>New Common Units</u>. The New Common Units being issued at Closing, when issued in consideration for the contribution by Antero of the Contributed Assets as provided by this Agreement, will be duly authorized, validly issued, fully paid (to the extent required by the Partnership's Organizational Documents) and nonassessable (except as such nonassessability may be affected by the Delaware Revised Uniform Limited Partnership Act) and free of any preemptive or similar rights (other than those set forth in the Partnership's limited partnership agreement).

4.6 <u>Investment</u>. The Partnership is an "accredited investor" as such term is defined in Rule 501 promulgated under the Securities Act. The Partnership Acquiring Entities are not acquiring the Contributed Assets with a view to or for sale in connection with any distribution thereof or any other security related thereto within the meaning of the Securities Act. The Partnership Acquiring Entities are familiar with investments of the nature of the Contributed Assets, understand that this investment involves substantial risks, have adequately investigated the Contributed Assets and the Business, and have substantial knowledge and experience in

25

financial and business matters such that they are capable of evaluating, and have evaluated, the merits and risks inherent in acquiring the Contributed Assets, and are able to bear the economic risks of such investment. The Partnership Acquiring Entities have had the opportunity to visit with Antero and meet with the officers of Antero and other representatives to discuss the business, assets, liabilities, financial condition, and operations of Antero Water, have received all materials, documents and other information that the Partnership Acquiring Entities deem necessary or advisable to evaluate the Business or the Contributed Assets, and have made their own independent examinations, investigations, analyses and evaluations of the Business and the Contributed Assets, including their own estimate of the value of the Contributed Assets. The Partnership Acquiring Entities have undertaken such due diligence (including a review of the properties, liabilities, books, records and contracts of Antero Water) as they deem adequate. The Partnership Acquiring Entities acknowledge that the Water Interests have not been registered under applicable federal and state securities laws and that the Water Interests may not be sold, transferred, offered for sale, pledged, hypothecated or otherwise disposed of unless such transfer, sale, assignment, pledge, hypothecation or other disposition is registered under applicable federal and state securities laws or pursuant to an exemption from registration under any federal or state securities laws.

4.7 <u>No Other Representations or Warranties: Schedules</u>. Except as set forth in this <u>Article IV</u>, none of the Partnership Acquiring Entities makes any other express or implied representation or warranty with respect to the New Common Units or the transactions contemplated by this Agreement, and disclaims any other representations or warranties. The disclosure of any matter or item in any schedule to this Agreement shall not be deemed to constitute an acknowledgment that any such matter is required to be disclosed.

#### ARTICLE V COVENANTS, ETC.

5.1 <u>Conduct of the Business</u>. Antero covenants and agrees that from and after the execution of this Agreement and until the Closing, except (i) as expressly contemplated by this Agreement, (ii) as required by applicable Law or (iii) with the prior written consent of the Partnership (which consent shall not be unreasonably withheld, conditioned or delayed):

(a) Antero will not, and will not permit any of its Subsidiaries to, sell, transfer, assign, lease, convey or otherwise dispose of any of the Water Interests or Water Assets except in the ordinary course of operating the Business;

(b) Antero will, and will cause Antero Water to, use commercially reasonable efforts to maintain the applicable

Water Assets in such working order and condition as is consistent with past practice;

(c) Antero will, and will cause Antero Water to, use commercially reasonable efforts to conduct the Business in the ordinary course consistent with past practices;

(d) Antero will not, and will not permit Antero Water to, permit any Lien to be imposed on the Water Interests or the Water Assets, other than Permitted Liens;

(e) Antero will not, and will not permit Antero Water to, adopt any material change in Tax methods, principles or elections;

(f) Antero will not, and will not permit Antero Water to, acquire, commence or conduct any new activity that would cause the representation in <u>Section 3.13(g)</u> to be untrue at Closing

(g) Commencing on the Effective Date, Antero will not, and will not permit Antero Water to, pay any indebtedness outstanding under the Water Credit Facility; and

(h) Commencing on the Effective Date, (i) Antero will not permit Antero Water to make any distribution with respect to its equity interests and (ii) Antero will not, and will not permit Antero Water to, make any distribution with respect to any daily cash sweep from Antero Water accounts pursuant to the Antero cash management program.

#### 5.2 <u>Financial Statements; Financing Cooperation</u>.

(a) Antero shall permit the Partnership and its representatives to contact Antero's accountants, auditors and employees, and shall cause such accountants, auditors and employees to discuss, cooperate and provide information reasonably requested by the Partnership or its representatives, in order for the Partnership to prepare audited and unaudited historical financial statements for the Water Assets and the Business and pro forma financial statements of the Partnership, in each case that meet the requirements of Regulation S-X promulgated under the Securities Act and within the timeframe specified for the Partnership to file such financial statements on Form 8-K under the Securities Exchange Act of 1934, as amended, and the rules and regulations of the Commission promulgated thereunder. Antero shall cause its accountants, auditors and employees to cooperate with the Partnership with regards to responding to any comments from the Commission on the financial statements of the Water Assets and the Business. The Partnership shall be responsible for and shall pay for or reimburse Antero for all costs incurred by Antero in connection with the external audit of any such financial statements (including reasonable accountants' fees) incurred after the date hereof. The obligations of Antero under this <u>Section 5.2</u> shall survive for three (3) years after the Closing.

(b) From and after the date of this Agreement, Antero shall, and shall cause each of its Subsidiaries to, use commercially reasonable efforts to cause its and their representatives (including its and their auditors) to provide all customary cooperation as reasonably requested by the Partnership to assist the Partnership in the arrangement of any bank debt financing or any capital markets debt or equity financing as reasonably requested by the Partnership.

5.3 <u>Independent Investigation</u>. The Partnership acknowledges that in making the decision to enter into this Agreement and to consummate the transactions contemplated hereby, it has relied solely on its own independent investigation of the Business, the Water Interests and the Water Assets and upon the express written representations, warranties and covenants in this Agreement. Without diminishing the scope of the express written representations, warranties and covenants of the Parties and without affecting or impairing its right to rely thereon, THE PARTNERSHIP ACKNOWLEDGES THAT NEITHER ANTERO NOR ANY OF ITS

27

AFFILIATES OR REPRESENTATIVES HAS MADE ANY REPRESENTATION OR WARRANTY OTHER THAN THOSE CONTAINED IN <u>ARTICLE III</u>, AND ANTERO HEREBY EXPRESSLY DISCLAIMS AND NEGATES ANY OTHER REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, RELATING TO THE WATER ASSETS OR THE BUSINESS (INCLUDING ANY IMPLIED OR EXPRESS WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE OR CONFORMITY TO MODELS OR SAMPLES OF MATERIALS).

#### 5.4 <u>Post-Closing Receivables and Payments</u>.

(a) Should Antero or any of its Subsidiaries receive any payments attributable to accounts receivable of Antero Water or the Business that relate to the operations of the Business after the Effective Date, then Antero shall or shall cause its applicable Subsidiary to, within thirty (30) days of receipt of such payments, forward such payments to Antero Water. Should the Partnership or any of its Subsidiaries receive any payments attributable to accounts receivable of Antero Water or the Business that relate to the operations of the Business on or prior to the Effective Date, then the Partnership shall or shall cause its applicable Subsidiary to, within thirty (30) days of receipt of such payments to Antero.

(b) If any demand is made on Antero or any of its Subsidiaries to pay any invoice, account payable or other amount incurred in connection with the operation of the Business after the Effective Date, then the Partnership shall be responsible, or promptly reimburse Antero or such Subsidiaries, for the same. If any demand is made on the Partnership or any of its Subsidiaries to pay

any invoice, account payable or other amount incurred in connection with the operation of the Business on or prior to the Effective Date, then Antero shall be responsible, or promptly reimburse the Partnership or such Subsidiaries, for the same.

5.5 <u>Further Assurances</u>. On and after the Closing Date, the Parties shall cooperate and use their respective commercially reasonable efforts to take or cause to be taken all appropriate actions and do, or cause to be done, all things necessary or appropriate to make effective the transactions contemplated by this Agreement and the other Transaction Documents, including the execution of any additional assignment or similar documents or instruments of transfer of any kind, the obtaining of consents which may be reasonably necessary or appropriate to carry out any of the provisions hereof and the taking of all such other actions as such Party may reasonably be requested to take by the other Party from to time to time, consistent with the terms of this Agreement or the other Transaction Documents, in order to effectuate the provisions and purposes of this Agreement and the other Transaction Documents and the transactions contemplated hereby.

5.6 <u>NYSE Listing</u>. Prior to the Closing, the Partnership will use its reasonable best efforts to obtain approval for listing, subject to notice of issuance, of the New Common Units on the New York Stock Exchange.

#### 5.7 <u>Tax Covenants</u>.

(a) The Parties agree that Antero shall bear the liability for any Asset Taxes attributable to (A) any taxable period ending on or prior to the Effective Date and (B) the portion

28

of any Straddle Period ending on and including the Effective Date. The Parties further agree that the Partnership shall bear the liability for any Asset Taxes attributable to (A) any taxable period beginning after the Effective Date and (B) the portion of any Straddle Period beginning after the Effective Date.

(b) <u>Proration of Straddle Period Taxes</u>. In the case of Taxes that are payable with respect to any Straddle Period, the portion of any such Taxes that is attributable to the portion of such Straddle Period ending on the Effective Date will be:

- (i) in the case of property or *ad valorem* or franchise Taxes or any other Taxes that are imposed on a periodic basis and which are measured by, or based solely upon capital, debt or a combination of capital and debt, deemed to be the amount of such Taxes for the entire Straddle Period (or, in the case of such Taxes determined on an arrears basis, the amount of such Taxes for the immediately preceding period), multiplied by a fraction the numerator of which is the number of calendar days in the portion of the period ending on and including the Effective Date and the denominator of which is the number of calendar days in the entire period; and
- (ii) in the case of all other Taxes, deemed equal to the amount which would be payable if the relevant Straddle Period ended on and included the Effective Date; *provided* that exemptions, allowances, or deductions that are calculated on an annual basis (including depreciation and amortization deductions) will be allocated between the portion of the Straddle Period ending on and including the Effective Date and the portion of the Straddle Period beginning after the Effective Date in proportion to the number of days in each period.

(c) With respect to any Tax Return attributable to a Straddle Period that is required to be filed after the Closing Date by Antero Water or with respect to the Water Assets, the Partnership shall cause such Tax Return to be prepared, cause to be included in such Tax Return all items of income, gain, loss, deduction and credit required to be included therein, furnish a copy of such Tax Return to Antero, cause such Tax Return to be filed timely with the appropriate Tax Authority, and the Partnership shall be responsible for the timely payment of all Taxes due with respect to the period covered by such Tax Return (but shall have a right to recover from Antero the amount of Taxes attributable to the portion of the Straddle Period ending on and including the Effective Date pursuant to <u>Section 5.7(a)</u>).

(d) The Parties shall cooperate fully, and cause their Affiliates to cooperate fully, as and to the extent reasonably requested by the other Party, (i) to accomplish the apportionment of Tax liability described in this <u>Section 5.7</u>, (ii) to respond to requests for the provision of any information or documentation within the knowledge or possession of such Party as reasonably necessary to facilitate compliance with financial reporting obligations arising under FASB Statement No. 109 (including compliance with Financial Accounting Standards Board Interpretation No. 48), and (iii) in connection with any audit, litigation or other proceeding (each a "<u>Tax</u> <u>Proceeding</u>") with respect to Taxes. Such cooperation shall include access to, the retention and (upon the other Party's request) the provision of records and information which are

reasonably relevant to any Tax Return or Tax Proceeding, and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. The Partnership and Antero will use their respective commercially reasonable efforts to retain all books and records with respect to Tax matters pertinent to the Water Assets relating to any taxable period beginning before the Effective Date until the later of six years after the Effective Date or the expiration of the applicable statute of limitations of the respective taxable periods (including any extensions thereof), and to abide by all record retention agreements entered into with any Tax Authority. Each of the Partnership and Antero agrees, upon request, to use their respective commercially reasonable efforts to obtain any certificate or other document from any Tax Authority or any other Person as may be necessary to mitigate, reduce or eliminate, to the extent permitted by applicable Law, any Tax that could be imposed with respect to the transactions

- (e) <u>Debt Financed Distribution.</u>
  - (i) To the extent (A) the sum of the Cash Consideration (for the avoidance of doubt, including any Additional Cash Consideration) and the Earn-Out Payments (such sum, the "<u>Total Cash Proceeds</u>") exceeds (B) the amount, if any, of the Additional Cash Consideration minus any indebtedness assumed by the Partnership as of the Closing under the Water Credit Facility, the Partnership shall borrow an amount equal to such excess to finance the payments of such excess amount in a manner such that the proceeds of such borrowing are allocable to the payment of the Total Cash Proceeds pursuant to Treasury Regulation Section 1.707-5(b)(1) and Temporary Treasury Regulation Section 1.163-8T (such borrowing, and any "refinancing" of such borrowing treated as the liability it refinances pursuant to Treasury Regulation Section 1.707-5(c), the "<u>Traceable Partnership Debt</u>"). For the avoidance of doubt, the Total Cash Proceeds and the excess amount shall be calculated at the Closing using a value of zero for the Earn-Out Payments and shall be recalculated at the time of the payment of any Earn-Out Payment to take into account such Earn-Out Payment and any prior Earn-Out Payment.
  - (ii) Antero and the Partnership intend that any payment of Total Cash Proceeds to Antero that is attributable to the Traceable Partnership Debt shall qualify as a "debt-financed transfer" described under Treasury Regulation Sections 1.707-3 and 1.707-5(b) (a "Debt Financed Distribution").
  - (iii) For a period of four years following the Closing Date, to the extent the total indebtedness of the Partnership exceeds the Traceable Partnership Debt (as calculated immediately after the then most-recent payment of any Total Cash Proceeds), the Partnership shall take reasonable steps to ensure (and shall cause its respective Affiliates to ensure) that the Traceable Partnership Debt will not be less than the entire outstanding principal balance of the Traceable Partnership Debt outstanding immediately after such payment.

30

(iv) Antero and the Partnership shall act (and shall cause their respective Affiliates to act) at all times in a manner consistent with the foregoing provisions of this <u>Section 5.7(e)</u>, except with the prior written consent of Antero or as otherwise required by applicable law following a Final Determination.

(f) <u>Reimbursement of Capital Expenditures</u>. Antero and the Partnership agree that all or a portion of any Total Cash Proceeds received by Antero pursuant to this Agreement is being paid to Antero as reimbursement for the capital expenditures (i) incurred by Antero (or deemed to be incurred by Antero for U.S. federal income tax purposes if incurred by Antero Water) with respect to the Water Assets before the Closing or otherwise in accordance with <u>Section 2.2</u> of this Agreement (the "<u>Water Capital Expenditures</u>") or (ii) incurred by Antero (or deemed to be incurred by Antero for U.S. federal income tax purposes if incurred by NewCo (as defined in the Initial Contribution Agreement)) with respect to the Gathering and Compression Assets (as defined in the Initial Contribution Agreement) before the Closing Date to Antero pursuant to Section 2.2 of the Initial Contribution Agreement exceeds any unreimbursed capital expenditures with respect to the Gathering and Compression Assets (as defined in the Initial Contribution subsequent to the Closing Date to Antero pursuant to Section 2.2 of the Initial Contribution Agreement), any such excess distribution will be treated as a reimbursement of the then unreimbursed Water Capital Expenditures.

(g) The Parties intend that for U.S. federal income tax purposes, (i) the contribution of the Contributed Assets shall be treated as a contribution by Antero to the Partnership pursuant to Section 721(a) of the Code, subject to Section 707 of the Code, and (ii) the payment of the Total Cash Proceeds shall qualify as a Debt Financed Distribution to the extent set forth in Section 5.7(e)(ii). Any payment of Total Cash Proceeds in excess of the amount treated as a Debt Financed Distribution shall be treated as a reimbursement of Antero's preformation expenditures within the meaning of Treasury Regulation Section 1.707-4(d) to the greatest extent applicable and in accordance with Section 5.7(g), except with the prior written consent of Antero or as otherwise required by applicable law following a Final Determination, including disclosing the distribution of the Total Cash Proceeds in accordance with the requirements of Treasury Regulations Section 1.707-3(c)(2).

5.8 <u>Consents</u>.

(a) Antero shall use commercially reasonable efforts to obtain prior to the Closing any Consents necessary to cause (i) the transfer and assignment of the Water Assets (other than the Treatment Assets) and associated liabilities to Antero Water prior to the Closing, (ii) the contribution, assignment, transfer and conveyance of the Water Interests to the Partnership at Closing and (iii) the contribution, assignment, transfer and conveyance of the Treatment Assets to Antero Treatment at Closing.

(b) If, following the date hereof but prior to the Closing Date, either Party discovers any Consent that is not set forth on <u>Schedule 3.4</u> (such discovered consent, an

"<u>Unscheduled Consent</u>"), such Party shall notify the other Party thereof and Antero shall promptly send to the holder of such Unscheduled Consent a notice in compliance with the contractual provisions applicable to such Unscheduled Consent seeking such holder's consent to the transactions contemplated by the Transaction Documents.

(c) If and to the extent that the valid and complete contribution, assignment, transfer or conveyance of any Water Asset (including any Contract or Permit) (i) to Antero Water, (ii) to Antero Treatment or (iii) indirectly as part of the contribution of the Water Interests to the Partnership would (x) be a violation of applicable Law, or (y) give rise to any requirement to obtain a consent of a third Person, as applicable, that has not been obtained or made by the Closing and the failure to obtain such consent would cause the contribution, assignment, transfer or conveyance thereof to be void or voidable or would result in the termination of the underlying agreement or instrument giving rise to such consent obligation (any consent satisfying either clause (x) or (y) above, a "<u>Required Consent</u>"), then, unless the Parties shall otherwise mutually determine, the transfer or assignment of the Water Asset contemplated by clause (i), (ii) or (iii), as the case may be, shall be automatically deemed deferred and any such purported transfer, assignment or assumption shall be null and void until such time as all legal impediments are removed or such Required Consents have been obtained or made. Notwithstanding the foregoing, any such Water Assets shall continue to constitute Water Assets for all other purposes of this Agreement.

(d) If any transfer or assignment of any Water Asset intended to be transferred or assigned from Antero to Antero Water or Antero Treatment, as the case may be, is not consummated on or prior to the Closing, whether as a result of the provisions of <u>Section 5.8(c)</u> or for any other reason, then, insofar as reasonably possible, Antero or its applicable Subsidiary retaining such Water Asset shall thereafter hold such Water Asset for the use, benefit and/or burden of the Partnership (at the expense and for the account of the Partnership) until such time as such transfer or assignment can be completed; <u>provided</u>, <u>however</u>, that Antero or its applicable Subsidiary shall hold such Water Asset for the use, benefit and/or burden of the Partnership at its own expense if the reason the transfer or assignment is not consummated is the failure to obtain any Required Consent. In addition, Antero or its applicable Subsidiary shall, insofar as reasonably possible and to the extent permitted by applicable Law, treat such Water Asset in the ordinary course of business in accordance with past practice and take such other actions as may be reasonably requested by the Partnership in order to place the Partnership in a substantially similar position as if such Water Asset had been transferred or assigned as contemplated hereby and so that all the benefits and burdens relating to such Water Asset, as the case may be, including use, risk of loss, potential for gain, and dominion, control and command over such Water Asset, as the case may be, and all costs and expenses related thereto, shall inure from and after the Closing to the Partnership.

(e) From and after the date hereof, each Party shall, and Antero shall (up to Closing) and the Partnership shall (after Closing), cause Antero Water and Antero Treatment to use its commercially reasonable endeavors to cooperate with the other Party in seeking to obtain each Customary Post-Closing Consent to the extent not previously obtained.

5.9 <u>Access</u>. Prior to the Closing but after the date of execution of this Agreement, Antero shall permit, and shall cause its Affiliates to permit, the Partnership and its respective

32

authorized representatives to (i) have reasonable access, during regular business hours upon reasonable prior notice, to the Water Assets and records relating thereto as the Partnership and its Representatives may reasonably request, (ii) inspect and make copies of the Water Asset records and all other documents and information relating predominately to the Business from time to time as reasonably requested by the Partnership and its representatives and (iii) without limiting the foregoing, meet with designated employees of Antero as reasonably requested by the Partnership; provided, however, that, in each case, any such access shall be conducted in such a manner as not to interfere unreasonably with the operation of the businesses of Antero or its Affiliates and shall not require Antero to waive any applicable privilege (including attorney-client privilege) nor to violate any contractual obligation.

#### ARTICLE VI CONDITIONS TO CLOSING

6.1 <u>Conditions to Each Party's Obligation to Effect the Transactions</u>. The respective obligation of each Party to proceed with the Closing is subject to the satisfaction or waiver by each of the Parties (subject to applicable Laws) on or prior to the Closing Date of all of the following conditions:

(a) all necessary filings with and consents (other than Customary Post-Closing Consents) of any Governmental Authority required for the consummation of the transactions contemplated by this Agreement and the other Transaction Documents shall have been made and obtained; *provided, however*, that, prior to invoking this condition, the invoking party shall have used commercially reasonable efforts to make or obtain such filings and consents;

(b) no Party shall be subject to any decree, final non-appealable order or injunction of a court of competent jurisdiction that prohibits the consummation of the transactions contemplated by this Agreement and the other Transaction Documents and no Law enacted, entered, or issued by any Governmental Authority, or other legal restraint or prohibition preventing the consummation of the transactions contemplated by this Agreement and the other Transaction Documents, shall be in effect; and

(c) the New Common Units shall have been approved for listing on The New York Stock Exchange, subject to official notice of issuance.

6.2 <u>Conditions to the Obligation of the Partnership Acquiring Entities</u>. The obligation of the Partnership Acquiring Entities to proceed with the Closing is subject to the satisfaction or waiver by the Partnership on or prior to the Closing Date of the following conditions:

(a) Antero shall have performed, in all material respects, the covenants and agreements contained in this Agreement required to be performed by it on or prior to the Closing Date;

(b) (i) the Fundamental Representations shall be true and correct (without regard to qualifications as to materiality or Material Adverse Effect contained therein) in all material respects as of the Closing Date (except to the extent such representations and warranties expressly relate to an earlier date, in which case as of such earlier date), and (ii) the other

33

representations and warranties of Antero made in this Agreement shall be true and correct (without regard to qualifications as to materiality or Material Adverse Effect contained therein) as of the Closing Date (except to the extent such representations and warranties expressly relate to an earlier date, in which case as of such earlier date), except in the case of <u>clause (ii)</u> where the failure of the representations and warranties to be true and correct, individually or in the aggregate, has not had a Material Adverse Effect;

(c) Antero shall have delivered to the Partnership a certificate dated the Closing Date and signed by an authorized officer of Antero confirming the foregoing matters set forth in <u>clauses (a)</u> and <u>(b)</u> of this <u>Section 6.2</u> (the "<u>Antero Closing Certificate</u>");

(d) Antero shall have delivered or caused the delivery of the Closing deliverables set forth in <u>Section 7.2</u>;

(e) between the date hereof and the Closing Date, there shall not have been a Material Adverse Effect; and

(f) the Financing shall have closed concurrently with the Closing.

6.3 <u>Conditions to the Obligation of Antero</u>. The obligation of Antero to proceed with the Closing is subject to the satisfaction or waiver by Antero on or prior to the Closing Date of the following conditions:

(a) the Partnership Acquiring Entities shall have performed, in all material respects, the covenants and agreements contained in this Agreement required to be performed by them on or prior to the Closing Date;

(b) the representations and warranties of the Partnership Acquiring Entities made in this Agreement shall be true and correct (without regard to qualifications as to materiality or Partnership Material Adverse Effect contained therein) as of the Closing Date (except to the extent such representations and warranties expressly relate to an earlier date, in which case as of such earlier date), except where the failure of the representations and warranties to be true and correct, individually or in the aggregate, has not had a Partnership Material Adverse Effect;

(c) the Partnership shall have delivered to Antero a certificate dated the Closing Date and signed by an authorized officer of the General Partner confirming the foregoing matters set forth in <u>clauses (b)</u> and <u>(c)</u> of this <u>Section 6.3</u> (the "<u>Partnership Closing</u> <u>Certificate</u>");

(d) the Partnership shall have delivered or caused the delivery of the Closing deliverables set forth in <u>Section 7.3;</u>

(e) between the date hereof and the Closing Date, there shall not have been a Partnership Material Adverse

Effect; and

(f) the Financing shall have closed concurrently with the Closing.

34

#### ARTICLE VII CLOSING

7.1 <u>Closing</u>. Subject to the terms and conditions of this Agreement and unless otherwise agreed in writing by Antero and the Partnership, the closing (the "<u>Closing</u>") of the transactions contemplated by this Agreement will be held at the offices of Vinson & Elkins L.L.P., 1001 Fannin Street, Suite 2500, Houston, Texas at 9:00 a.m., Houston, Texas time on the date that is three business days immediately following the date of fulfillment or waiver (in accordance with the provisions hereof) of the last to be fulfilled or waived of the conditions set forth in <u>Sections 6.1, 6.2</u> and <u>6.3</u> (other than those conditions that by their nature are to be fulfilled at the Closing, but subject to the fulfillment or waiver of such conditions). The date on which the Closing occurs is referred to as the "<u>Closing Date</u>."

7.2 <u>Deliveries by Antero</u>. At the Closing, Antero will deliver (or cause to be delivered) to the Partnership the following:

- (a) a counterpart to the Assignment of Water Interest, duly executed by Antero;
- (b) a counterpart to the Assignment and Bill of Sale of Treatment Assets, duly executed by Antero;
- (c) a counterpart to the Water Services Agreement, duly executed by Antero;
- (d) a counterparty to the Secondment Agreement, duly executed by Antero;
- (e) a counterpart to the Services Agreement, duly executed by Antero;

(f) the Antero Closing Certificate, duly executed by an officer of Antero;

(g) a certification of non-foreign status executed by Antero in the form prescribed in Treasury Regulations Section 1.1445-2(b)(2); and

(h) such other documents, certificates and other instruments as may be reasonably requested by the Partnership prior to the Closing Date to carry out the intent and purposes of this Agreement.

7.3 <u>Deliveries by the Partnership Acquiring Entities</u>. At the Closing, the Partnership Acquiring Entities will deliver (or cause to be delivered) to Antero the following:

(a) (x) the Cash Consideration, by wire transfer of immediately available funds to an account specified by Antero, and (y) the New Common Units, by issuance of such New Common Units (in book-entry form) to Antero, by instruction to the Partnership's transfer agent or otherwise, and evidence of such issuance that is reasonably satisfactory to Antero, each as set forth in Section 2.2;

(b) a counterpart to the Assignment of Water Interest, duly executed by the Partnership;

35

- (c) a counterpart to the Assignment and Bill of Sale of Treatment Assets, duly executed by Antero Treatment;
- (d) a counterpart to the Water Services Agreement, duly executed by Antero Water;
- (e) a counterparty to the Secondment Agreement, duly executed by the Partnership;
- (f) a counterpart to the Services Agreement, duly executed by the Partnership and the General Partner;
- (g) the Partnership Closing Certificate, duly executed by an officer of the General Partner; and

(h) such other documents, certificates and other instruments as may be reasonably requested by Antero prior to the Closing Date to carry out the intent and purposes of this Agreement.

#### ARTICLE VIII INDEMNIFICATION

8.1 Indemnification of Antero and Other Parties. From and after the Closing Date, subject to the other provisions of this <u>Article VIII</u>, the Partnership shall indemnify and hold Antero and its Affiliates, directors, officers, employees, agents and representatives (together with Antero, the "<u>Antero Indemnitees</u>") harmless from and against any and all damages, losses, deficiencies, costs, expenses, obligations, fines, expenditures, claims and liabilities, including court costs and reasonable attorneys', accountants' or other experts' fees and reasonable expenses of investigation, defending and prosecuting Litigation (collectively, the "<u>Damages</u>"), suffered by the Antero Indemnitees as a result of, caused by, arising out of, or in any way relating to (a) any breach of a representation or warranty of the Partnership Acquiring Entities in this Agreement, (b) any breach of any agreement or covenant under this Agreement on the part of the Partnership Acquiring Entities, (c) any demand, assertion, claim, action or proceeding, judicial or otherwise, by any third party against Antero or any of its Affiliates that pertains to the business or operations of Antero Water or the ownership of the Contributed Assets, except in the case of this clause (c), to the extent covered by <u>Section 8.2(c)</u> or arising out of a breach by Antero of any of the representations, warranties or covenants of Antero set forth in this Agreement, or (d) Environmental Law to the extent resulting from and limited to the portion attributable to the operation or ownership of the Business or the Water Assets by the Partnership or any of its Subsidiaries and occurring on or after the Closing Date.

8.2 <u>Indemnification of the Partnership and Other Parties</u>. From and after the Closing Date, subject to the other provisions of this <u>Article VIII</u>, Antero shall indemnify and hold the General Partner, the members of the Partnership Group and their respective directors, officers, employees, agents and representatives (together with the Partnership, the "<u>Partnership Indemnitees</u>") harmless from and against any and all Damages suffered by the Partnership Indemnitees as a result of, caused by, arising out of, or in any way relating to (a) any breach of a representation or warranty of Antero in this Agreement except, in the case of this <u>clause (a)</u>, to

36

the extent covered by <u>Section 8.1(d)</u>, (b) any breach of any agreement or covenant in this Agreement on the part of Antero, (c) Environmental Law to the extent resulting from and limited to the portion attributable to Antero's operation or ownership of the Business or the Water Assets and occurring before the Closing Date and (d) the Excluded Assets and Excluded Liabilities.

#### 8.3 <u>Indemnification Procedures</u>.

(a) Each indemnified party agrees that promptly after it becomes aware of facts giving rise to a claim by it for indemnification pursuant to this <u>Article VIII</u> with respect to any matter as to which it claims to be entitled to indemnity under the provisions of this Agreement, such indemnified party must assert its claim for indemnification under this <u>Article VIII</u> (each, an "<u>Indemnity</u> <u>Claim</u>") by providing a written notice (a "<u>Claim Notice</u>") to the indemnifying party allegedly required to provide indemnification

protection under this <u>Article VIII</u> specifying, in reasonable detail, the nature and basis for such Indemnity Claim (*e.g.*, the underlying representation, warranty, covenant or agreement alleged to have been breached). Such notice shall include a demand for indemnification under this Agreement. Notwithstanding the foregoing, an indemnified party's failure to send or delay in sending a third party Claim Notice will not relieve the indemnifying party from liability hereunder with respect to such Indemnity Claim except to the extent the indemnifying party is prejudiced by such failure or delay and except as is otherwise provided herein. Except as specifically provided herein, each indemnified party's rights and remedies set forth in this Agreement will survive the Closing.

(b) In the event of the assertion of any third-party Indemnity Claim for which, by the terms hereof, an indemnified party seeks indemnification from an indemnifying party, the indemnifying party will have the right, at such indemnifying party's expense, to assume the defense of same, including the appointment and selection of counsel on behalf of the indemnified party so long as such counsel is reasonably acceptable to the indemnified party. If the indemnifying party elects to assume the defense of any such third-party Indemnification Claim, it shall within 20 business days of its receipt of the Claim Notice notify the indemnified party in writing of its intent to do so. Any such contest may be conducted in the name and on behalf of the indemnifying party or the indemnified party as may be appropriate. The indemnifying party will have the right to settle or compromise or take any corrective or remediation action with respect to any such Indemnification Claim by all appropriate proceedings, which proceedings will be prosecuted by the indemnifying party to a final conclusion or settled at the discretion of the indemnifying party. The indemnified party will be entitled, at its own cost, to participate with the indemnifying party in the defense of any such Indemnifying party Indemnity Claim but fails to reasonably prosecute such Indemnify claim, or if the indemnifying party does not assume the defense of any such Indemnity Claim, the indemnified party may assume control of such defense and in the event it is determined pursuant to the procedures set forth in this <u>Article VIII</u>, the indemnifying party will bear the reasonable costs and expenses).

(c) If requested by the indemnifying party, the indemnified party agrees to cooperate with the indemnifying party and its counsel in contesting any third-party Indemnity Claim that the indemnifying party elects to contest or, if appropriate, in making any counterclaim

37

against the person asserting the third-party Indemnity Claim, or any cross-complaint against any person, and the indemnifying party will reimburse the indemnified party for reasonable expenses incurred by it in so cooperating. At no cost or expense to the indemnified party, the indemnifying party shall reasonably cooperate with the indemnified party and its counsel in contesting any third-party Indemnity Claim.

(d) Notwithstanding anything to the contrary in this Agreement, the indemnifying party will not be permitted to settle, compromise, take any corrective or remedial action or enter into an agreed judgment or consent decree, in each case, that subjects the indemnified party to any injunctive or other non-monetary relief or any criminal liability, requires an admission of guilt or wrongdoing on the part of the indemnified party or imposes any continuing obligation on or requires any payment from the indemnified party without the indemnified party's prior written consent.

#### 8.4 <u>Calculation and Payment of Damages</u>.

(a) In calculating amounts payable to an indemnified party for a claim for indemnification hereunder, the amount of any indemnified Damages shall be determined without duplication of any other Damages for which an indemnification claim has been made or could be made under any other representation, warranty, covenant or agreement and shall be computed net of (i) payments actually recovered under any insurance policy with respect to such Damages or (ii) any prior or subsequent actual recovery from any Person other than the applicable indemnifying party with respect to such Damages.

(b) The indemnification required hereunder shall be made by periodic payments of the amount thereof during the course of the investigation or defense, within 10 days as and when reasonably specific bills are received or loss, liability, claim, damage or expense is incurred and reasonable evidence thereof is delivered.

8.5 <u>Waiver of Certain Damages</u>. Notwithstanding any other provision of this Agreement, in no event shall any Party be liable pursuant to this <u>Article VIII</u> for punitive, special, indirect, consequential, remote, speculative or lost profits damages of any kind or nature, including any damages based on diminution in value or any other damages based on a multiple of earnings or other multiple, regardless of the form of action through which such damages are sought, except for any such damages recovered by any third party against an indemnified party in respect of which such indemnified party would otherwise be entitled to indemnification pursuant to the terms hereof.

#### 8.6 <u>Limitations on Indemnification</u>.

(a) To the extent the Partnership Indemnitees are entitled to indemnification for Damages pursuant to <u>Section 8.2(a)</u> or <u>8.2(c)</u>, Antero shall not be liable for those Damages unless the aggregate amount of Damages exceeds \$10,524,500(the "<u>Deductible</u>"), and then only to the extent of any such excess; *provided*, *however*, that Antero shall not be liable for Damages pursuant to <u>Section 8.2(a)</u> or <u>8.2(c)</u> that exceed, in the aggregate, \$105,245,000 (the "<u>Cap</u>") less the Deductible.

(b) Notwithstanding <u>clause (a)</u> above, (i) to the extent the Partnership Indemnitees are entitled to indemnification for Damages for claims arising from fraud or Damages for Taxes arising from a breach of the representations or warranties in <u>Section 3.13</u>, Antero shall be fully liable for such Damages without regard to the Deductible or the Cap and (ii) to the extent the Partnership Indemnitees are entitled to indemnification for Damages for claims arising from a breach of a Fundamental Representation, Antero shall be fully liable for such Damages without regard to the Deductible or the Cap.

(c) To the extent the Antero Indemnitees are entitled to indemnification for Damages pursuant to Section 8.1(a) or (d), the Partnership shall not be liable for those Damages unless the aggregate amount of Damages exceeds the Deductible; *provided*, *however*, that the Partnership shall not be liable for Damages pursuant to Section 8.1(a) or (d) that exceed, in the aggregate, the Cap less the Deductible.

(d) Notwithstanding <u>clause (c)</u> above, to the extent the Antero Indemnitees are entitled to indemnification for Damages for claims arising from a breach of a Partnership Fundamental Representation, the Partnership shall be fully liable for such Damages without regard to the Deductible or the Cap.

8.7 <u>Survival</u>.

(a) The liability of Antero for (i) the breach of any of the representations and warranties of Antero set forth in <u>Sections 3.1, 3.2, 3.7, 3.11</u>, and <u>3.12</u> (the "<u>Fundamental Representations</u>") and (ii) Damages under <u>Section 8.2(d)</u> shall be indefinite. The liability of Antero for the breach of any of the representations and warranties of Antero set forth in <u>Article III</u> other than the Fundamental Representations and the matters set forth in <u>Sections 3.6</u> and <u>3.13</u> shall be limited to claims for which the Partnership delivers written notice to Antero on or before the date that is 18 months after the Closing Date. The liability of Antero for Damages for claims related to or arising from Taxes resulting from a breach of the covenants relating to Taxes set forth in this Agreement or a breach of the representations or warranties set forth in <u>Section 3.13</u> shall be limited to claims for which the Partnership delivers written notice to Antero on or before the date that is ninety (90) days after the expiration of the applicable statute of limitations for assessment of the applicable Tax. The liability of Antero for Damages under <u>Section 8.2(c)</u> or a breach of the representations and warranties set forth in <u>Section 3.6</u> shall be limited to claims for which the Partnership delivers written notice to Antero on or before the date that is ninety (90) days after the expiration of the applicable statute of limitations for assessment of the applicable Tax. The liability of Antero for Damages under <u>Section 8.2(c)</u> or a breach of the representations and warranties set forth in <u>Section 3.6</u> shall be limited to claims for which the Partnership delivers written notice to Antero and the applicable tax.

(b) The liability of the Partnership for the breach of any of the representations and warranties of the Partnership Acquiring Entities set forth in Sections 4.1, 4.2, 4.4, 4.5 and 4.6 (the "Partnership Fundamental Representations") shall be indefinite. The liability of the Partnership for Damages arising under Section 8.1(d) or a breach of the representations and warranties of the Partnership Acquiring Entities set forth in <u>Article IV</u> other than the Partnership Fundamental Representations shall limited to claims for which Antero delivers written notice to the Partnership on or before the date that is 24 months after the Closing Date.

(c) Notwithstanding anything to the contrary in this <u>Section 8.7</u>, if an indemnified party delivers written notice in reasonable detail to an indemnifying party of a claim

39

for indemnification on or prior to the applicable expiration date for such claim, such claim shall survive until finally resolved.

8.8 <u>Mitigation</u>. The Parties agree that an indemnified party's right to recourse under this <u>Article VIII</u> for any Damages shall be limited to the extent that such indemnified party would not have suffered such Damages had such indemnified party exercised commercially reasonable efforts to mitigate such Damages following the actual discovery by such indemnified party of the fact, event or circumstance giving rise to such Damages.

8.9 <u>Sole Remedy</u>. After the Closing, no Party shall have liability under this Agreement or the transactions contemplated hereby except as is provided in this <u>Article VIII</u> (other than claims or causes of action arising from fraud, and other than claims for specific performance or claims arising under any Transaction Documents (other than this Agreement) (which claims shall be subject to the liability provisions of such Transaction Documents)).

8.10 <u>Consideration Adjustment</u>. The Parties agree to treat any payments made pursuant to this <u>Article VIII</u> as an adjustment to the Cash Consideration for all Tax purposes, except as otherwise required by applicable Law following a Final Determination.

#### ARTICLE IX TERMINATION

9.1 <u>Events of Termination</u>. This Agreement may be terminated at any time prior to the Closing:

(a) by mutual written consent of Antero and the Partnership;

(b) by either Antero or the Partnership in writing after October 30, 2015, if the Closing has not occurred by that date, provided that as of such date neither the terminating Party nor any of its Affiliates are in material breach of its representations, warranties or covenants under this Agreement;

(c) by either Antero or the Partnership in writing without prejudice to other rights and remedies the terminating Party or its Affiliates may have (provided the terminating Party and its Affiliates are not otherwise in material default or breach of this Agreement, or have not failed or refused to close without justification hereunder), if the other Party or its Affiliates shall have
 (i) materially failed to perform its covenants or agreements contained herein required to be performed by such Party or its Affiliates on or prior to the Closing Date or (ii) materially breached any of its representations or warranties contained herein; *provided, however*, that in

the case of <u>clauses (i)</u> or <u>(ii)</u>, the breaching Party shall have a period of 30 days following written notice from the non-defaulting Party to cure any breach of this Agreement if the breach is curable; or

(d) by either Antero or the Partnership in writing, if there shall be any order, writ, injunction or decree of any Governmental Authority binding on the Parties that prohibits or restrains any Party from consummating the transactions contemplated hereby; *provided, however*, that the applicable Party shall have used its reasonable best efforts to have any such

40

order, writ, injunction or decree removed but it shall not have been removed within 30 days after entry by the Governmental Authority.

9.2 <u>Effect of Termination</u>. In the event of the termination of this Agreement by a Party as provided in <u>Section 9.1</u>, this Agreement shall thereafter become void except for this <u>Section 9.2</u> and <u>Section 10.2</u>. Nothing in this <u>Section 9.2</u> shall be deemed to release any Party from any liability for any willful and intentional breach by such Party of the terms and provisions of this Agreement or to impair any rights of any Party under this Agreement. If this Agreement is terminated by either Party pursuant to <u>Section 9.1(c)</u>, then the other Party shall reimburse such terminating Party for such terminating Party's and its Affiliates out-of-pocket expenses incurred in connection with the negotiation, execution and performance of this Agreement (including, where applicable, (i) legal fees and fees paid to Antero Financial Advisor, in either case incurred by the Partnership or the Conflicts Committee and (ii) legal fees and fees paid to the Partnership Financial Advisor, in either case incurred by Antero or the Special Committee).

#### ARTICLE X MISCELLANEOUS

10.1 <u>Expenses</u>. Unless otherwise specifically provided in this Agreement, each Party shall pay its own expenses incident to this Agreement or the other Transaction Documents and all action taken in preparation for effecting the provisions of this Agreement and the other Transaction Documents.

10.2 <u>Notices</u>. Unless otherwise specifically provided in this Agreement, any notice, request, instruction, correspondence or other document to be given under or in relation to this Agreement shall be made in writing and shall be deemed to have been properly given if: (i) personally delivered (with written confirmation of receipt); or (ii) delivered by a recognized overnight delivery service (delivery fees prepaid), in either case to the appropriate address set forth below:

If to Antero or any Subsidiary of Antero, addressed to:

Antero Resources Corporation (or applicable Subsidiary) 1615 Wynkoop Street Denver, Colorado 80202 Attn: Chief Financial Officer Facsimile: (303) 357-7315

If to the Partnership Acquiring Entities, addressed to:

Antero Midstream Partners LP and/or Antero Treatment LLC c/o Antero Resources Midstream Management LLC 1615 Wynkoop Street Denver, Colorado 80202 Attn: Chief Financial Officer Facsimile: (303) 357-7315

41

Any Party may change any address to which notice is to be given to it by giving notice as provided above of such change of address.

#### 10.3 Governing Law and Venue.

(a) This Agreement shall be governed by, construed, and enforced in accordance with the laws of the State of Colorado without regard to choice of law principles.

(b) The Parties agree that the appropriate, exclusive and convenient forum for any disputes between the Parties arising out of this Agreement or the transactions contemplated hereby shall be in any state or federal court in City and County of Denver, Colorado, and each of the Parties irrevocably submits to the jurisdiction of such courts solely in respect of any proceeding arising out of or related to this Agreement. The Parties further agree that the Parties shall not bring suit with respect to any disputes arising out of this Agreement or the transactions contemplated hereby in any court or jurisdiction other than the above specified courts.

10.4 <u>Public Statements</u>. The Parties agree that prior to making any public announcement or statement with respect to this Agreement or the transaction represented herein, the Party desiring to make such public announcement or statement shall provide the other Party with a copy of the proposed announcement or statement prior to the intended release date of such announcement. The other Party shall thereafter consult with the Party desiring to make the release, and the Parties shall exercise their reasonable best efforts to

(a) agree upon the text of a joint public announcement or statement to be made by both such Parties or (b) in the case of a statement to be made solely by one Party, obtain approval of the other Party to the text of a public announcement or statement. Nothing contained in this Section 10.4 shall be construed to require either Party to obtain approval of the other Party to disclose information with respect to this Agreement or the transaction represented herein to any Governmental Authority to the extent such disclosure is required by Applicable Law or necessary to comply with disclosure requirements of the Securities and Exchange Commission, New York Stock Exchange, or any other regulated stock exchange.

10.5 Form of Payment. All payments hereunder shall be made in United States dollars and, unless the Parties making and receiving such payments shall agree otherwise or the provisions hereof provide otherwise, shall be made by wire or interbank transfer of immediately available funds on the date such payment is due to such account as the Party receiving payment may designate at least three business days prior to the proposed date of payment.

10.6 <u>Entire Agreement; Amendments and Waivers</u>. This Agreement and the documents and instruments and other agreements specifically referred to herein or delivered pursuant hereto, including the exhibits and schedules hereto, (a) constitute the entire agreement among the Parties with respect to the subject matter hereof and supersede all prior agreements and understandings, both written and oral, among the Parties with respect to the subject matter hereof and (b) are not intended to confer upon any other Person any rights or remedies hereunder except as <u>Article VIII</u> or <u>Article X</u> contemplates or except as otherwise expressly provided herein or therein. Each Party agrees that (i) no other Party (including its agents and representatives) has made any representation, warranty, covenant or agreement to or with such Party relating to this Agreement or the transactions contemplated hereby, other than those

42

expressly set forth in the documents and instruments and other agreements specifically referred to herein or delivered pursuant hereto, including the exhibits and schedules hereto, and (ii) such Party has not relied upon any representation, warranty, covenant or agreement relating to this Agreement or the transactions contemplated hereby other than those referred to in <u>clause (i)</u> above. No amendment, supplement, modification or waiver of this Agreement shall be binding unless executed in writing by the Parties. No waiver of any of the provisions of this Agreement shall be deemed or shall constitute a waiver of any other provision hereof (regardless of whether similar), nor shall any such waiver constitute a continuing waiver unless otherwise expressly provided.

10.7 <u>Binding Effect and Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective permitted successors and assigns, but neither this Agreement nor any of the rights, benefits or obligations hereunder shall be assigned, by operation of law or otherwise, by any Party without the prior written consent of the other Parties.

10.8 <u>Severability</u>. If any provision of this Agreement is determined to be void or unenforceable, in whole or in part, then (a) such provision shall be deemed inoperative to the extent it is deemed void or unenforceable, (b) the Parties agree to enter into such amendments to this Agreement in order to give effect, to the greatest extent legally possible, to the provision that is determined to be void or unenforceable and (c) the other provisions of this Agreement in all other respects shall remain in full force and effect and binding and enforceable to the maximum extent permitted by Applicable Law; provided, however, that in the event that a material term under this Agreement is so modified, the Parties will, timely and in good faith, negotiate to revise and amend this Agreement in a manner which preserves, as closely as possible, each Party's business and economic objectives as expressed by the Agreement prior to such modification.

#### 10.9 <u>Interpretation</u>.

(a) The Parties agree that they have been represented by counsel during the negotiation and execution of this Agreement and therefore waive the application of any law, regulation, holding or rule of construction providing that ambiguities in an agreement or other document will be construed against the Party drafting such agreement or document.

(b) The words "this Agreement," "herein," "hereby," "hereunder" and "hereof," and words of similar import, refer to this Agreement as a whole and not to any particular subdivision unless expressly so limited. The words "this Article," "this Section" and "this clause," and words of similar import, refer only to the Article, Section or clause hereof in which such words occur. The word "or" is exclusive, and the word "including" (in its various forms) means including without limitation.

(c) Pronouns in masculine, feminine or neuter genders shall be construed to state and include any other gender, and words, terms and titles (including terms defined herein) in the singular form shall be construed to include the plural and vice versa, unless the context otherwise requires.

43

(d) References herein to any Person shall include such Person's successors and assigns; provided, however, that nothing contained in this clause (d) is intended to authorize any assignment or transfer not otherwise permitted by this Agreement.

(e) References herein to any Law shall be deemed to refer to such Law as amended, reenacted, supplemented or superseded in whole or in part and in effect from time to time and also to all rules and regulations promulgated thereunder

(f) References herein to any Contract mean such Contract as amended, supplemented or modified (including any waiver thereto) in accordance with the terms thereof, except that with respect to any Contract listed on any schedule hereto, all such amendments, supplements or modifications must also be listed on such schedule.

(g) Each representation, warranty, covenant and agreement contained in this Agreement will have independent significance, and the fact that any conduct or state of facts may be within the scope of two or more provisions in this Agreement, whether relating to the same or different subject matters and regardless of the relative levels of specificity, shall not be considered in construing or interpreting this Agreement.

(h) Unless otherwise expressly provided herein to the contrary, accounting terms shall have the meaning given by U.S. generally accepted accounting principles.

10.10 <u>Headings and Schedules</u>. The headings of the several Articles and Sections herein are inserted for convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement. The schedules referred to herein are attached hereto and incorporated herein by this reference, and the matters disclosed in those schedules shall be deemed to qualify the representation or warranty to which they expressly relate and any other representation or warranty, but only to the extent that it is reasonably apparent on its face that such disclosure is applicable to such other representation or warranty. The Parties acknowledge and agree that (a) the schedules may include certain items and information solely for informational purposes for the convenience of the Parties and (b) the disclosure by any Party of any matter in any schedule shall not be deemed to constitute an acknowledgment by such Party that the matter is required to be disclosed by the terms of this Agreement or that the matter is material.

10.11 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, including electronic, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. In the event that any signature is delivered by facsimile transmission or by e-mail delivery of a ".pdf" format data file, such signature shall create a valid and binding obligation of the Party executing (or on whose behalf such signature is executed) with the same force and effect as if such facsimile or ".pdf" signature page were an original thereof.

10.12 <u>Determinations by the Partnership</u>. With respect to any notice, consent, approval or waiver that is required to be or may be taken or given by the Partnership (a) pursuant to the terms of this Agreement on or prior to the Closing Date or (b) pursuant to <u>Article VIII</u> after the

44

Closing Date, such notice, consent, approval or waiver shall be taken or given by the Conflicts Committee on behalf of the Partnership.

[Signature page follows]

45

IN WITNESS WHEREOF, the Parties have duly executed this Agreement as of the date first written above.

#### ANTERO RESOURCES CORPORATION

By:	/s/ Alvyn A. Schopp
Name:	Alvyn A. Schopp
Title:	Chief Administrative Officer, Regional Vice President
	and Treasurer

#### ANTERO MIDSTREAM PARTNERS LP

By: Antero Resources Midstream Management LLC, its general partner

By: /s/ Alvyn A. Schopp

Name: Alvyn A. Schopp

Title: Chief Administrative Officer, Regional Vice President and Treasurer

#### ANTERO TREATMENT LLC

By: /s/ Alvyn A. Schopp

Name: Alvyn A. Schopp Title: Chief Administrative Officer, Regional Vice President and Treasurer

#### <u>Exhibit A</u>

#### Form of Assignment and Bill of Sale of Treatment Assets

#### ASSIGNMENT AND BILL OF SALE

This Assignment and Bill of Sale (this "Assignment") is made from Antero Resources Corporation, a Delaware corporation ("Assignor") to Antero Treatment LLC, a Delaware limited liability company, whose address is 1615 Wynkoop Street, Denver, Colorado 80202 ("Assignee"), and dated as of September [], 2015, but is effective as of 00:01 a.m. (Mountain Time) on October 1, 2015 (the "Effective Time"). Assignor and Assignee are sometimes referred to in this Assignment individually as a "Party" and collectively as the "Parties." Capitalized terms used herein but not otherwise defined shall have the meanings given such terms in that certain Contribution, Conveyance and Assumption Agreement, dated as of September 17, 2015, by and among the Assignor, Antero Midstream Partners LP, a Delaware limited partnership (the "Partnership"), and Assignee (as amended, supplemented or restated from time to time, the "Contribution Agreement").

#### RECITALS

WHEREAS, Assignor desires to assign to Assignee and Assignee desires to receive, assume and bear all of Assignor's right, title and interest in and to the Treatment Assets (as defined below).

NOW, THEREFORE, for and in consideration of the mutual promises contained herein, the benefits to be derived by each party hereunder, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Assignor and Assignee agree as follows:

#### ARTICLE 1 ASSIGNMENT OF ASSETS

Section 1.1 <u>Assignment of Treatment Assets</u>. Assignor hereby GRANTS, BARGAINS, ASSIGNS, CONVEYS AND TRANSFERS unto Assignee, with a warranty of title by, through and under Assignor but not otherwise, all of Assignor's right, title and interest in and to the Treatment Assets. For purposes hereof, "*Treatment Assets*" means those assets, contracts, rights, permits and properties owned or leased by Assignor and used primarily in connection with the construction, ownership, operation, use or maintenance of Assignor's advanced wastewater treatment complex to be constructed in Doddridge County, West Virginia, including those assets set forth on <u>Exhibit A</u> and Schedule 1-2 of the Contribution Agreement, in each case, less and except the Excluded Assets.

**TO HAVE AND TO HOLD** the Treatment Assets, together with all and singular the rights, privileges, contracts and appurtenances, in any way appertaining or belonging thereto, unto Assignee, its successors and assigns, forever, subject, to the terms and conditions of this Assignment.

Section 1.2 Excluded Assets. The Treatment Assets shall not include, and Assignor hereby excepts and reserves, all of the Excluded Assets.

*"Excluded Assets"*, as used in this Assignment, means all right, title and interest of Assignor and its Affiliates in and to any assets, properties, agreements and interests not included in the definition of Treatment Assets, as applicable, including:

- (a) the Upstream Assets;
- (b) all systems primarily used or intended for the gathering or transportation of Hydrocarbons;
- (c) all of Assignor's (and its Affiliates') corporate minute books, financial records and other business records to the extent such books and records are related to Assignor's (or any of its Affiliates') business generally or are otherwise not directly related to the Treatment Assets;
- (d) All claims for refunds, credits, loss carryforwards and similar Tax assets with respect to (i) Asset Taxes allocated to Assignor pursuant to Section 5.7(a) of the Contribution Agreement, (ii) Income Taxes of Assignor or any of its Affiliates or (iii) any Taxes attributable to any of the assets or properties described in the definition of Excluded Assets.
- (e) all personal computers and associated peripherals and all radio and telephone equipment (and licenses related thereto);
- (f) all of Assignor's (and its Affiliates') computer software, patents, trade secrets, copyrights, names, trademarks,

logos and other intellectual property (except the Treatment Conveyed IP). and all rights in and to that certain License Agreement, dated November 10, 2014, between Assignor and the Partnership;

- (g) all documents and instruments of Assignor (or any of its Affiliates) that may be protected by an attorney-client privilege (other than title opinions and reports on status of title, in each case, with respect to title to any of the Treatment Assets);
- (h) all offices of Assignor and all personal property located therein;
- (i) all of the bonds, letters of credit, guarantees, deposits and other pre-payments posted by Assignor or any of its Affiliates with any Governmental Authorities or any other third parties;
- (j) all trade credits, receivables and all other proceeds, income or revenues attributable to the Treatment Assets with respect to any period of time prior to the Effective Time, or attributable to any of the assets and properties described in this definition of Excluded Assets with respect to any period of time;
- (k) all accounts (including bank accounts) and all cash on hand;
  - 2
- (1) any policy or agreement of insurance or indemnity agreement and any proceeds or awards therefrom;
- (m) all assets of Antero Resources Midstream Management LLC and all assets of any Person that directly or indirectly holds any interest in Antero Resources Midstream Management LLC;
- (n) all assets described on Exhibit B; and
- (o) any Gathering and Compression Assets (as defined in the Initial Contribution Agreement).

#### ARTICLE 2 DEFINED TERMS

#### Section 2.1 <u>Defined Terms</u>.

*"Excluded Liabilities"* means (a) (i) any and all Income Taxes imposed on Assignor or any of its Affiliates; the Asset Taxes allocable to Assignor pursuant to Section 5.7(a) of the Contribution Agreement; (ii) any Taxes imposed on or with respect to the Excluded Assets; and (iii) any and all other Taxes imposed on or with respect to the Treatment Assets for any taxable period (or portion thereof) ending prior to the Effective Time; and (b) the liabilities described on Schedule 1 to the Contribution Agreement.

"including" means including, without limitation.

*"Intellectual Property"* means (a) patents and patent applications; (b) trade secrets and confidential information, (c) copyrights, registered and unregistered; and (d) trademarks, service marks, trade names, trade dress, and domain names (*"Trademarks"*).

*"Liabilities"* means any direct or indirect liability, indebtedness, Damage, deficiency, Tax, interest, penalty, amount paid in settlement, judgment, assessment, guaranty or endorsement of or by any Person, in the case of each of the foregoing, whether absolute or contingent, matured or unmatured, asserted or unasserted, accrued or unaccrued, due or to become due, liquidated or unliquidated.

*"Treatment Conveyed IP"* means (a) all Intellectual Property (other than Trademarks) owned by Assignor primarily relating to the Treatment Assets and (b) the Intellectual Property described under the heading "Treatment Conveyed IP" in Schedule 1-2 of the Contribution Agreement, including all rights, claims and causes of action for past, present and future infringement and misappropriation of the Treatment Conveyed IP, including the right to seek injunctive relief and damages, and to collect and retain same.

*"Upstream Assets"* means all assets held by Assignor or an Affiliate of Assignor at any time in respect of the exploration, development and/or production of Hydrocarbons business of Assignor and its Affiliates, including all interests of Assignor or its Affiliates:

3

- (a) in and to any oil and gas wells and leases, including all mineral interests, royalty interests, overriding royalty interests, production payments, other payments out of or measured by the value of oil and gas production, net profits interests, carried interests, farmout or farmin rights, options, subleases, and all other rights and interests that Assignor or its Affiliates have in and to any oil and gas leases, any lands pooled or unitized therewith and any oil and gas wells;
- (b) in and to (i) any Hydrocarbons, carbon dioxide or water or other non-Hydrocarbons in and under, or which may be produced and saved from or attributable to, the leases or lands referred to in *clause (a)* above, or any interests pooled or unitized therewith; (ii) all agreements, contracts, and/or permits between Assignor and any third party

to take or draw fresh water from any source; and (iii) any fresh water the subject of clause (b)(ii) above;

- (c) in and to all Hydrocarbons in storage or existing in stock tanks, pipelines and/or plants (including inventory) on the gathering, compression, and/or transportation systems of Assignor and all carbon dioxide, water and other non-Hydrocarbons in storage or existing in stock tanks, pipelines and/or plants (including inventory);
- (d) with respect to the use and occupancy of the surface of and the subsurface depths under the lands and leases referred to in *clause (a)*, and rights of ingress and egress and similar rights and interests pertaining to, situated on or used in connection with such lands and leases;
- (e) in and to any oil, gas or mineral unitization, pooling, operating and communitization agreements, joint venture agreements, farmin and farmout agreements, exploration agreements, exchange agreements, declarations, orders, rules, regulations or other official acts of any Governmental Authority and the units created thereby, including all units voluntarily formed or formed under orders, regulations, rules or other official acts of any Governmental Authority having jurisdiction;
- (f) in and to all surface and subsurface personal property, equipment, machinery, fixtures, movable and immovable property and improvements on or appurtenant to the leases, lands or wells described in *clause (a)*, or used or obtained in connection with the exploration, development or operation of such leases, lands or wells, including any trucks and cars, drilling/workover rigs and rolling stock and all equipment, pipe and inventory that is not currently being used or currently designated for use in connection with the ownership or operation of the Treatment Assets (whether located on or off the Treatment Assets); and
- (g) in and to (i) that certain Gathering Agreement between Assignor and Antero Midstream LLC entered into effective as of December 1, 2013, and that certain Right of First Offer Agreement between Assignor and Antero

Midstream LLC entered into effective as of the December 1, 2013, and (ii) any Transaction Document to which Assignor or an Affiliate of Assignor is or will be a party.

#### ARTICLE 3 DISCLAIMER OF WARRANTIES

Section 3.1 EXCEPT AS AND TO THE EXTENT EXPRESSLY SET FORTH IN THIS ASSIGNMENT OR IN THE CONTRIBUTION AGREEMENT, (I) ASSIGNOR MAKES NO REPRESENTATIONS OR WARRANTIES, EXPRESS, STATUTORY OR IMPLIED AND (II) ASSIGNOR EXPRESSLY DISCLAIMS ALL LIABILITY AND RESPONSIBILITY FOR ANY REPRESENTATION, WARRANTY, STATEMENT OR INFORMATION MADE OR COMMUNICATED (ORALLY OR IN WRITING) TO ASSIGNEE OR ITS EMPLOYEES, AGENTS, CONSULTANTS OR REPRESENTATIVES (INCLUDING ANY OPINION, INFORMATION, PROJECTION OR ADVICE THAT MAY HAVE BEEN PROVIDED TO AN ASSIGNEE BY ANY OFFICER, DIRECTOR, EMPLOYEE, AGENT, CONSULTANT, REPRESENTATIVE OR ADVISOR OF ASSIGNOR).

EXCEPT AS AND TO THE EXTENT EXPRESSLY SET FORTH IN THIS ASSIGNMENT OR IN Section 3.2 THE CONTRIBUTION AGREEMENT, ASSIGNOR EXPRESSLY DISCLAIMS ANY REPRESENTATION OR WARRANTY, EXPRESS, STATUTORY OR IMPLIED, AS TO (I) TITLE TO ANY OF THE TREATMENT ASSETS, (II) ANY ESTIMATES OF THE VALUE OF THE TREATMENT ASSETS OR FUTURE REVENUES GENERATED BY THE TREATMENT ASSETS, (III) THE CONDITION, QUALITY, SUITABILITY OR MARKETABILITY OF THE TREATMENT ASSETS, (IV) THE CONTENT. CHARACTER OR NATURE OF ANY INFORMATION MEMORANDUM, REPORTS, BROCHURES, CHARTS OR STATEMENTS PREPARED BY ASSIGNOR OR THIRD PARTIES WITH RESPECT TO THE TREATMENT ASSETS, AND (V) ANY OTHER MATERIALS OR INFORMATION THAT MAY HAVE BEEN MADE AVAILABLE TO ASSIGNEE OR ITS EMPLOYEES, AGENTS, CONSULTANTS, REPRESENTATIVES OR ADVISORS IN CONNECTION WITH THIS ASSIGNMENT OR ANY DISCUSSION OR PRESENTATION RELATING THERETO. EXCEPT AS AND TO THE EXTENT EXPRESSLY REPRESENTED OTHERWISE IN THIS ASSIGNMENT OR IN THE CONTRIBUTION AGREEMENT, ASSIGNOR FURTHER DISCLAIMS ANY REPRESENTATION OR WARRANTY, EXPRESS, STATUTORY OR IMPLIED, OF MERCHANTABILITY, FREEDOM FROM LATENT VICES OR DEFECTS, FITNESS FOR A PARTICULAR PURPOSE OR CONFORMITY TO MODELS OR SAMPLES OF MATERIALS OF THE TREATMENT ASSETS, RIGHTS OF A PURCHASER UNDER APPROPRIATE STATUTES TO CLAIM DIMINUTION OF CONSIDERATION, IT BEING ACKNOWLEDGED BY THE PARTIES THAT, EXCEPT AS AND TO THE EXTENT OTHERWISE PROVIDED IN THIS ASSIGNMENT OR IN THE CONTRIBUTION AGREEMENT, ASSIGNEE SHALL BE DEEMED TO HAVE OBTAINED THE TREATMENT ASSETS IN THEIR PRESENT STATUS, CONDITION AND STATE OF REPAIR, "AS IS" AND "WHERE IS" WITH ALL FAULTS OR DEFECTS (KNOWN OR UNKNOWN, LATENT, DISCOVERABLE OR UNDISCOVERABLE) AND THAT ASSIGNEE HAS MADE OR CAUSED TO BE

Section 3.3 <u>Environmental Matters</u>. EXCEPT AS AND TO THE EXTENT OTHERWISE PROVIDED IN THE CONTRIBUTION AGREEMENT, ASSIGNOR HAS NOT MADE ANY REPRESENTATION OR WARRANTY TO ASSIGNEE REGARDING ANY MATTER OR CIRCUMSTANCE RELATING TO ENVIRONMENTAL LAWS, ENVIRONMENTAL PERMITS, THE RELEASE OR THREATENED RELEASE OF HAZARDOUS MATERIALS INTO THE ENVIRONMENT, EXPOSURE TO HAZARDOUS MATERIALS, OR THE PROTECTION OF HUMAN HEALTH, SAFETY, NATURAL RESOURCES OR THE ENVIRONMENT, OR ANY OTHER ENVIRONMENTAL CONDITION OF THE TREATMENT ASSETS, AND NOTHING IN THIS ASSIGNMENT OR OTHERWISE SHALL BE CONSTRUED AS SUCH A REPRESENTATION OR WARRANTY.

#### Section 3.4 THE PARTIES AGREE THAT, TO THE EXTENT REQUIRED BY APPLICABLE LAW TO BE EFFECTIVE, THE DISCLAIMERS OF CERTAIN REPRESENTATIONS AND WARRANTIES CONTAINED IN THIS *ARTICLE 3* ARE "CONSPICUOUS" DISCLAIMERS FOR THE PURPOSE OF SUCH APPLICABLE LAW.

#### ARTICLE 4 MISCELLANEOUS

Section 4.1 <u>Assignment Subject to Contribution Agreement</u>. This Assignment is made in accordance with and is subject to the terms, covenants and conditions contained in the Contribution Agreement, a copy of which can be obtained from Assignee at the above referenced address. The terms and conditions of the Contribution Agreement are incorporated herein by reference, and in the event of a conflict between the provisions of the Contribution Agreement and this Assignment, the provisions of the Contribution Agreement shall control. The execution and delivery of this Assignment by Assigner, and the execution and acceptance of this Assignment by Assigner or Assignee under the Contribution Agreement.

Section 4.2 <u>Assumed Liabilities</u>. Assignee hereby assumes and agrees to fulfill, perform, pay and discharge (or cause to be fulfilled, performed, paid and discharged) all Liabilities, known or unknown, with respect Assignee's interests in the Treatment Assets to the extent assigned to Assignee herein, subject to, and as limited by, the terms of the Contribution Agreement and as described in Schedule 1-2 thereof; *provided that* Assignee does not assume any Liabilities of Assignor resulting from, relating to or arising out of the Excluded Liabilities or, for the avoidance of doubt, relating to or arising out of the Excluded Assets.

Section 4.3 <u>Separate Assignments</u>. Where separate assignments or deeds of the Treatment Assets have been, or will be, executed for filing with and approval by applicable Governmental Authorities, any such separate assignments or deeds (a) shall evidence the Assignment and assignment of the Treatment Assets herein made, and shall not constitute any additional Assignment or assignment of the Treatment Assets, (b) are not intended to modify, and shall not modify, any of the terms, covenants and conditions, or limitations on warranties, set forth

in this Assignment and are not intended to create, and shall not create, any representations, warranties or additional covenants of or by Assigner to Assignee, and (c) shall be deemed to contain all of the terms and provisions of this Assignment, as fully and to all intents and purposes as though the same were set forth at length in such separate assignments.

Section 4.4 <u>Governing Law</u>. EXCEPT TO THE EXTENT THAT THE LAWS OF THE STATE OF WEST VIRGINIA, PENNSYLVANIA OR OHIO ARE MANDATORILY APPLICABLE TO THIS ASSIGNMENT IN CONNECTION WITH THE CONVEYANCES OF PROPERTY INTERESTS INVOLVING REAL PROPERTY LOCATED IN THE STATE OF WEST VIRGINIA, PENNSYLVANIA OR OHIO, THIS ASSIGNMENT AND ANY CLAIM, CONTROVERSY OR DISPUTE ARISING UNDER OR RELATED TO THIS ASSIGNMENT OR THE RIGHTS, DUTIES AND RELATIONSHIP OF THE PARTIES, SHALL BE GOVERNED BY AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH THE LAWS OF THE STATE OF COLORADO, EXCLUDING ANY CONFLICTS OF LAW, RULE OR PRINCIPLE THAT MIGHT REFER CONSTRUCTION OF PROVISIONS TO THE LAWS OF ANOTHER JURISDICTION.

Section 4.5 <u>Successors and Assigns</u>. This Assignment shall be binding upon, inure to the benefit of and be enforceable by the Parties and their respective successors and assigns.

Section 4.6 <u>Titles and Captions</u>. All Article or Section titles or captions in this Assignment are for convenience only, shall not be deemed part of this Assignment and in no way define, limit, extend or describe the scope or intent of any provisions hereof. Except to the extent otherwise stated in this Assignment, references to "Articles" and "Sections" are to Articles and Sections of this Assignment, and references to "Exhibits" are to the Exhibits attached to this Assignment, which are made a part hereof and incorporated herein for all purposes.

**Section 4.7 Counterparts**. This Assignment may be executed in any number of counterparts, and each such counterpart hereof shall be deemed to be an original instrument, but all of such counterparts shall constitute for all purposes one agreement.

Section 4.8 <u>Further Assurances</u>. Assignor, for itself and its successors and assigns, hereby covenants and agrees that, if this Assignment shall at any time be insufficient in whole or in part to transfer the Treatment Assets to Assignee, at the reasonable request of Assignee, Assignor or its successors or assigns shall take such further actions as are requested and execute any additional documents, instruments or conveyances of any kind which may be reasonably necessary to further effect the transactions contemplated by this Assignment; provided, however, that no such action, document, instrument or conveyance shall increase a party's liability beyond that contemplated by this Assignment.
EXECUTED as of the date of the Parties' acknowledgments below, but effective at the Effective Time.

### ASSIGNOR:

### ANTERO RESOURCES CORPORATION

By: Name: Alvyn A. Schopp Title: Chief Administrative Officer and Regional Vice President

### ASSIGNEE:

#### ANTERO TREATMENT LLC

By: Name: Alvyn A. Schopp Title: Chief Administrative Officer and Regional Vice President

### <u>Exhibit B</u>

#### Form of Assignment of Water Interest

# FORM OF ASSIGNMENT OF CONTRIBUTED INTERESTS

This Assignment (this "Assignment") is made and entered into this [] day of September, 2015, by and between Antero Resources Corporation, a Delaware corporation, (the "Assignor"), and Antero Midstream Partners LP, a Delaware limited partnership (the "Assignee"), and is effective as of October 1, 2015 (the "Effective Date"). Assignor and Assignee are each referred to herein individually as a "Party" and collectively as the "Parties."

Capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed to such terms in that certain Contribution, Conveyance and Assumption Agreement (the "*Contribution Agreement*"), dated as of September 17, 2015, by and among the Assigner, the Assignee and Antero Treatment LLC, a Delaware limited liability company and wholly-owned subsidiary of the Assignee.

# WITNESSETH:

WHEREAS, the Assignor owns all of the outstanding limited liability company interests (the "*Contributed Interests*") in, and is the sole member of, Antero Water LLC, a Delaware limited liability company ("*Antero Water*");

WHEREAS, pursuant to the Contribution Agreement, Assignor has agreed to contribute, assign, transfer and convey the Contributed Interests, which shall include, without limitation, any and all income, distributions, value, rights, benefits and privileges associated therewith or deriving therefrom (collectively, the "*Conveyed Interests*") to the Assignee, and the Assignor desires to effect such contribution, assignment, transfer and conveyance;

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties do hereby agree as follows:

1. <u>Assignment</u>. Subject to and in accordance with the terms of the Contribution Agreement, and effective as of the Effective Date, the Assignor hereby CONTRIBUTES, ASSIGNS, TRANSFERS AND CONVEYS to the Assignee the Conveyed Interests, free and clear of all Liens (other than restrictions under applicable federal and state securities Laws or under the Organizational Documents of Antero Water), and the Assignee hereby accepts the same.

2. <u>Substitution as Member</u>. From and after the Effective Date, Assignee shall be substituted for Assignor as a member of Antero Water and shall become the sole member of Antero Water. From and after the Effective Date, Assignor shall cease to be a member of Antero Water and to have or exercise any right or power as a member of Antero Water or with respect to the Conveyed Interests. The

Parties agree that the assignment of the Conveyed Interests, the admission of Assignee as a substitute member of Antero Water and the cessation of Assignor as a member of Antero Water shall not dissolve Antero Water and the business of Antero Water shall continue, notwithstanding any provision of the Organizational Documents of Antero Water to the contrary.

#### 3. Disclaimer of Warranties.

#### THE ASSIGNOR IS CONVEYING THE CONVEYED INTERESTS WITHOUT REPRESENTATION OR (a) WARRANTY, EXCEPT AS PROVIDED IN THE CONTRIBUTION AGREEMENT.

(b) The Assignor and the Assignee agree that the disclaimers contained in Section 3(a) herein, are "conspicuous" disclaimers. Any covenant implied by statute or law by use of the words "grant," "contribute," "transfer," "assign" or "convey" or any of them are hereby expressly disclaimed, waived, and negated.

#### 4. General Provisions.

Binding Effect. This Assignment will be binding upon, inure to the benefit of and be enforceable by the Parties and (a) their respective successors and assigns.

Governing Law. This Assignment shall be governed by, construed and enforced in accordance with the laws of the (b)State of Colorado without regard to choice of law principles.

Consent to Jurisdiction. The Parties agree that the appropriate, exclusive and convenient forum for any disputes (c) between the Parties arising out of this Assignment shall be in any state or federal court in City and County of Denver, Colorado, and each of the Parties irrevocably submits to the jurisdiction of such courts solely in respect of any proceeding arising out of or related to this Assignment. The Parties further agree that the Parties shall not bring suit with respect to any disputes arising out of this Assignment in any court or jurisdiction other than the above specified courts.

Amendment and Modification. This Assignment may be amended, modified or supplemented only by written (d)agreement of the Parties.

Waiver of Compliance. Any failure of any Party to comply with any obligation, covenant, agreement or condition in (e) this Assignment may be waived by the Party entitled to the benefits thereof only by a written instrument signed by the Party granting such waiver, but such waiver or failure to insist upon strict compliance with such obligation, covenant, agreement or condition shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure, nor shall any such waiver constitute a continuing waiver unless otherwise expressly provided.

No Third Party Rights. This Assignment shall be binding upon and inure solely to the benefit of the Parties and their (f) respective successors and permitted assigns, and none of the provisions of this Assignment shall be for the benefit of or enforceable by any third party, including any creditor of any Party or any of their Affiliates. No such third party shall obtain any right under any provision of this Assignment or shall by

2

reasons of any such provision make any claim in respect of any liability (or otherwise) against any other Party.

Entire Agreement. This Assignment, together with the Contribution Agreement and the other Transaction Documents, (g) constitute the entire agreement and understanding of the Parties with respect to the subject matter hereof and supersede all prior agreements and understandings, both oral and written, between the Parties with respect to such subject matter. In the event of a conflict or inconsistencies between the terms and conditions of this Assignment and the Contribution Agreement, the terms and conditions of the Contribution Agreement shall control.

Counterparts. This Assignment may be executed in one or more counterparts, including electronic, each of which shall (h)be deemed an original, but all of which together shall constitute one and the same instrument. In the event that any signature is delivered by facsimile transmission or by e-mail delivery of a ".pdf" format data file, such signature shall create a valid and binding obligation of the Party executing (or on whose behalf such signature is executed) with the same force and effect as if such facsimile or ".pdf" signature page were an original thereof.

[Signature page follows]

3

IN WITNESS WHEREOF, this Assignment has been duly executed by each of the Parties as of the date and year first above

written.

**ASSIGNOR:** 

# ANTERO RESOURCES CORPORATION

By: Name: Title:

**ASSIGNEE:** 

# ANTERO MIDSTREAM PARTNERS LP

By: Antero Resources Midstream Management LLC, its general partner

By: Name: Title:

SIGNATURE PAGE TO ASSIGNMENT

# Exhibit C

### Form of Water Services Agreement

### WATER SERVICES AGREEMENT

### **BY AND BETWEEN**

### ANTERO RESOURCES CORPORATION

### AND

# ANTERO WATER LLC

### DATED AS OF

], 2015

[

# TABLE OF CONTENTS

ARTICLE 1	DEFINITIONS	1
ARTICLE 2	PRODUCER COMMITMENTS	10
Section 2.1	Producer Commitments	10
Section 2.2	Conflicting Commitments	10
Section 2.3	Covenant Running with the Land	10
Section 2.4	Additional Oil and Gas Interests or Water Facilities	11
ARTICLE 3	SERVICES	13
Section 3.1	Antero Water Service Commitment	13
Section 3.2	Priority of Fluid Handling Services	15
ARTICLE 4	FRESH WATER SYSTEM EXPANSION AND CONNECTION OF FRESH WATER DELIVERY POINTS	15
Section 4.1	Development Plan; Fresh Water Facilities Plan; Exchange and Review of Information	15
Section 4.2	Expansion of Fresh Water System; Connection of Well Pads	16
Section 4.3	Installation and Operation of High-Rate Transfer Facilities	19
Section 4.4	Take Points	20
Section 4.5	Retention Facilities	20

Section 4.6	Pumping Facilities	21
Section 4.7	Cooperation	21
ARTICLE 5 Section 5.1 Section 5.2 Section 5.3 Section 5.4 Section 5.5	CERTAIN PROVISIONS REGARDING WASTE WATER SERVICES Access to Produced Water Receipt Points and Other Waste Water Sites Dispatch Procedures Treatment Facility: Designated Receiving Facilities Transportation Services Equipment Contract Carrier Status	21 21 22 23 23
ARTICLE 6	TERM	23
Section 6.1	<u>Term</u>	23
ARTICLE 7	FEES AND CONSIDERATION	23
Section 7.1	<u>Fees</u>	23
Section 7.2	<u>Minimum Daily Fresh Water Volume Commitment</u>	29
ARTICLE 8	CERTAIN RIGHTS AND OBLIGATIONS OF PARTIES	30
Section 8.1	<u>Operational Control of Antero Water's Facilities</u>	30
Section 8.2	<u>Maintenance</u>	30
Section 8.3	<u>Right of Way and Access</u>	30
Section 8.4	<u>Third Party Services; Capacity Allocations on the Fresh Water System</u>	31
Section 8.5	<u>Governmental Approvals; Compliance with Applicable Law</u>	32

# i

ARTICLE 9 Section 9.1 Section 9.2	FRESH WATER DELIVERY RATES Fresh Water Delivery Rates Producer Facilities	33 33 33
	<u>rioducer racinites</u>	55
ARTICLE 10	FRESH WATER NOMINATION	33
Section 10.1	Maximum Take Point Volumes	33
Section 10.2	Fresh Water Delivery Nominations	33
Section 10.3	Changes in Fresh Water Delivery Rates	33
ARTICLE 11	FRESH WATER AVAILABILITY AND QUALITY; WASTE WATER QUALITY	33
Section 11.1	Fresh Water Availability	33
Section 11.2	Take Point Fresh Water Standards	34
Section 11.3	Non-Conforming Take Point Fresh Water	34
Section 11.4	Delivery Point Fresh Water Quality Standards	34
Section 11.5	Retention Facility Contamination	34
Section 11.6	Non-Conforming Waste Water	35
ARTICLE 12	MEASUREMENT EQUIPMENT AND PROCEDURES	35
Section 12.1	Measurement Equipment	35
Section 12.2	Notice of Measurement Facilities Inspection and Calibration	36
Section 12.3	Measurement Accuracy Verification	36
Section 12.4	Special Tests	36
Section 12.5	Metered Flow Rates in Error	37
Section 12.6	Waste Water Measurement	37
Section 12.7	Units of Measurement	37
Section 12.8	Record Retention	37
ARTICLE 13	NOTICES	38
Section 13.1	Notices	38
ARTICLE 14	PAYMENTS	39
Section 14.1	Invoices	39
Section 14.2	Right to Suspend on Failure to Pay	40
Section 14.3	Audit Rights	40
Section 14.4	Payment Disputes	40
Section 14.5	Interest on Late Payments	40
Section 14.6	Credit Assurance	40
ARTICLE 15	FORCE MAJEURE	41
Section 15.1	Suspension of Obligations	41
Section 15.2	Definition of Force Majeure	41
Section 15.3	Settlement of Strikes and Lockouts	42
Section 15.4	Payments for Services Performed	42

ARTICLE 16 Section 16.1 Section 16.2	INDEMNIFICATION <u>Antero Water</u> Producer	42 42 43
ARTICLE 17 Section 17.1	CUSTODY AND TITLE	43 43
Section 17.1 Section 17.2	Custody of Fresh Water Custody of Waste Water	43
Section 17.2 Section 17.3	Title to Waste Water	44 44
Section 17.5	The to wase water	
ARTICLE 18	PAYMENTS FOR FRESH WATER; TAXES	44
Section 18.1	Payments for Fresh Water; Taxes	44
ARTICLE 19	MISCELLANEOUS	45
Section 19.1	<u>Rights</u>	45
Section 19.2	Applicable Laws	45
Section 19.3	Governing Law; Jurisdiction	45
Section 19.4	Successors and Assigns	45
Section 19.5	<u>Severability</u>	46
Section 19.6	Confidentiality	47
Section 19.7	Entire Agreement, Amendments and Waiver	48
Section 19.8	Limitation of Liability	48
Section 19.9	Headings Disktoord Demodiate	49
Section 19.10 Section 19.11	<u>Rights and Remedies</u> No Partnership	49 49
Section 19.11 Section 19.12	<u>Rules of Construction</u>	49 49
Section 19.12	No Third Party Beneficiaries	49 50
Section 19.13	Further Assurances	50 50
Section 19.15	Counterpart Execution	50
Section 19.16	Memorandum of Agreement	50
Section 17.10	Memorandum of Agreement	50
Exhibit A Init	ial Service Area	
Exhibit B Init	ial Fresh Water Facilities	
Exhibit C Take Points		
Exhibit D Initial Development Plan		
Exhibit E Initial Fresh Water Facilities Plan		
Exhibit F Form of Connection Notice		
Exhibit G Deemed Connection Notices		
Exhibit H Cos	st of Service Fee	

iii

Exhibit I

Exhibit J

Assumed Subcontracts

Memorandum of Agreement

### WATER SERVICES AGREEMENT

This Water Services Agreement (this "*Agreement*"), dated as of [ ], 2015 (the "*Effective Date*"), is by and between **ANTERO RESOURCES CORPORATION**, a Delaware corporation ("*Producer*"), and **ANTERO WATER LLC**, a Delaware limited liability company ("*Antero Water*"). Producer and Antero Water may be referred to herein individually as a "*Party*" or collectively as the "*Parties*."

#### RECITALS

A. Producer owns Oil and Gas Interests and intends to drill and complete Wells for the production of Hydrocarbons in the Initial Service Area and may from time to time own Oil and Gas Interests and may drill and complete Wells for the production of Hydrocarbons in other areas.

B. Producer requires supplies of Fresh Water in its areas of operation for hydraulic fracturing operations and other purposes and has the right to take Fresh Water from various rivers and other Fresh Water sources to use for such purposes in its operations in the Initial Service Area and may from time to time have rights to take Fresh Water from other sources for such operations and operations in other areas.

C. Antero Water has acquired from Producer certain Fresh Water Facilities, which Producer has been using to take Fresh Water from its Fresh Water sources and to make available such Fresh Water to Wells in its areas of operation in the Initial Service Area, including certain related assets. Antero Water anticipates the expansion of the Fresh Water Facilities to make available Fresh Water to

additional locations in the Initial Service Area and other areas.

D. Antero Water has also acquired from Producer certain contracts and other assets that Producer has been using (or expects that Antero Water will use in the future) in connection with the collection of Produced Water produced from Wells in the Initial Service Area and Other Waste Water generated or collected at the Well Pads or from Producer's property adjacent to any Well Pad in the Initial Service Area and the transportation, treatment, disposal, recycling, and/or sale of such Produced Water and Other Waste Water. Antero Water may in the future enter into additional contracts and/or acquire additional assets for the purpose of carrying out such activities in the Initial Service Area and other areas.

E. Producer desires to contract with Antero Water for Antero Water to provide the Fluid Handling Services in the Service Area, and Antero Water desires to provide the Fluid Handling Services to Producer, in each case in accordance with the terms and conditions of this Agreement.

NOW THEREFORE, in consideration of the premises and mutual covenants set forth in this Agreement, the Parties agree as follows:

# ARTICLE 1 DEFINITIONS

Capitalized terms used, but not otherwise defined, in this Agreement shall have the respective meanings given to such terms set forth below:

1

### Adequate Assurance of Performance. As defined in Section 14.6(a).

*Affiliate*. Any Person that, directly or indirectly through one or more intermediaries, controls or is controlled by or is under common control with another Person. *Affiliated* shall have the correlative meaning. The term "control" (including its derivatives and similar terms) shall mean possessing the power to direct or cause the direction of the management and policies of a Person, whether through ownership, by contract, or otherwise. Notwithstanding the foregoing, for purposes of this Agreement, none of Antero Midstream Partners LP, Antero Resources Midstream Management LLC, or any of their respective direct or indirect subsidiaries (including Antero Water) shall be an Affiliate of Producer, and neither Producer nor any of its direct or indirect subsidiaries (other than Antero Midstream Partners LP and its direct and indirect subsidiaries) shall be an Affiliate of Antero Water.

Agreement. As defined in the preamble hereof.

Antero Treatment. Antero Treatment LLC.

Antero Water. As defined in the preamble of this Agreement.

Antero Water Group. As defined in Section 16.2(b).

*Applicable Law.* Any applicable law, statute, regulation, rule, code, administrative order or enforcement action (whether national, local, municipal, territorial, provincial, or federal and including common law) of any Governmental Authority, including any Environmental Law, to the extent they apply to the Fluid Handling Services or the Parties.

Assumed Subcontracts. As defined in Section 7.1(b).

AST. An above-ground storage tank.

*Attributable Produced Water*. Produced Water produced from a Well operated by Producer located on the Service Area Properties and attributable to either (i) Producer's interest in such Well or (ii) the interest of non-operating parties in such Well, to the extent that Producer (as operator) has the right to dispose of, and is responsible for the disposition of, such Produced Water.

Barrel. Forty-two Gallons.

Baseline Components. As defined in Section 7.1(e).

Business Day. Any calendar Day that commercial banks in New York City are open for business.

Chemicals Baseline. As defined in Section 7.1(d).

Completion Deadline. As defined in Section 4.2(b).

Connection Notice. As defined in Section 4.2(b).

Confidential Information. As defined in Section 19.6(a).

**Conflicting Commitment.** Any agreement, commitment or arrangement that would require Producer to use Fresh Water delivered by any Person other than Antero Water to a Well in Producer's hydraulic fracturing operations on such Well or to deliver Attributable Produced Water or Other Waste Water to any Person other than Antero Water for gathering, collection, transportation, processing, treatment, recyclying, re-sale, or other disposal or disposition.

*Contract Year.* Each of (i) the period from the Effective Date to the last Day of the Month in which the first anniversary of the Effective Date occurs and (ii) each period of twelve (12) Months thereafter.

Contribution Agreement. As defined in Section 15.2.

Cost of Service Fee. As defined in Section 7.1(j).

CPI. As defined in Section 7.1(f).

CS Facility. As defined in Section 7.1(j).

*Day.* A period commencing at 10:00 a.m., Eastern Standard Time, on a calendar day and ending at 10:00 a.m., Eastern Standard Time, on the next succeeding calendar day. *Daily* shall have the correlative meaning.

Delivery Point Fee. As defined in Section 7.1(a)(i).

Designated Receiving Facility. As defined in Section 5.3.

Development Plan. As defined in Section 4.1(a).

Effective Date. As defined in the preamble of this Agreement.

*Electricity Baseline*. As defined in <u>Section 7.1(d)</u>.

*Environmental Laws*. All Applicable Laws pertaining to the presence or release of environmental contaminants (including any Hazardous Materials), or relating to natural resources (including any protected species) or the environment (including the air, water, surface or subsurface of the ground) as same are in effect at any time and including the Comprehensive Environmental Response, Compensation and Liability Act ("CERCLA"), as amended by Superfund Amendments and Reauthorization Act ("SARA"), 42 U.S.C. §§ 9601 et seq.; Resource Conservation and Recovery Act ("RCRA"), as amended by the Solid Waste Disposal Act ("SWDA"), 42 U.S.C. §§ 6901 et seq.; Federal Water Pollution Control Act ("FWPCA"), as amended by the Clean Water Act ("CWA"), 33 U.S.C. §§ 1251 et seq.; Safe Drinking Water Act, 42 U.S.C. §§ 300f et seq.; Clean Air Act ("CAA"), 42 U.S.C., §§ 7401 et seq.; and Toxic Substances Control Act ("TSCA"), 15 U.S.C., §§ 2601 et seq., as each are amended from time to time, and any similar state or local enactments by Governmental Authorities.

*Fair Market Value*. With respect to any asset, the price that would be paid by a willing buyer of such asset to a willing seller, as determined by an independent nationally known investment banking firm selected by Antero Water and reasonably acceptable to Producer.

3

*Firm Service*. Fluid Handling Services that are accorded the highest priority on the Water Facilities with respect to capacity allocations, interruptions, or curtailments, specifically including the Fresh Water Services provided to Producer hereunder. Firm Services will be the last curtailed on the relevant part of the Water Facilities in the event of an interruption or curtailment, and all Firm Services will be treated equally in the event an allocation is necessary.

Fluid Handling Services. As defined in Section 3.1.

Force Majeure. As defined in Section 15.2.

Fresh Water. Either (i) raw fresh water or (ii) a mixture of raw fresh water and Treated Waste Water.

*Fresh Water Delivery Point*. With respect to each Well Pad, the water inlet flange of the working tanks or the hydration unit located at or in the vicinity of such Well Pad being utilized by Producer and its other contractors in hydraulic fracturing operations on a Well on such Well Pad.

*Fresh Water Delivery Rate.* With respect to each Well Pad, the rate at which Fresh Water is to be delivered by Antero Water from the Fresh Water System to the Fresh Water Delivery Point at such Well Pad, which rate is specified by Producer in the Connection Notice for such Well Pad and is subject to modification pursuant to <u>Section 10.3</u>.

*Fresh Water Facilities.* Collectively, the Fresh Water System and the High-Rate Transfer Facilities, including the Fresh Water facilities described in <u>Exhibit B</u> being acquired by Antero Water from Producer as of the date hereof (including any Fresh Water assets included in the Water Assets, as such term is defined in the Contribution Agreement), including any additional System Segments constructed after the date hereof, as such Fresh Water facilities are expanded after the date hereof. The Fresh Water Facilities do not

include any Fresh Water Delivery Point or facilities at the Well Pads downstream of the Fresh Water Delivery Points, all of which facilities shall be owned and operated by Producer or its subcontractors.

# Fresh Water Facilities Plan. As defined in Section 4.1(b).

*Fresh Water Measurement Point*. The inlet to Antero Water's Measurement Facilities located at the inlet to the High-Rate Transfer Facilities located at or in the vicinity of each Well Pad.

Fresh Water Quality Standards. As defined in Section 11.2.

Fresh Water Services. Those Fluid Handling Services that are described in Section 3.1(a) through Section 3.1(e).

*Fresh Water System*. Fresh Water facilities of Antero Water and its subcontractors upstream of the interconnection with the High-Rate Transfer Facilities, including, in each case, to the extent now in existence or constructed or installed in the future, all Fresh Water pipelines, System Retention Facilities, System Pumping Stations, Take Point Facilities, Measurement Facilities, rights of way (whether for underground or surface use), fee parcels, surface rights, and permits, and all appurtenant facilities. The Fresh Water System is used to take Fresh Water from

4

the Take Points to the Well Pads, where such Fresh Water is transferred through the High Rate Transfer Facilities to the Fresh Water Delivery Points.

Fuel Gas. As defined in Section 7.1(d).

Gallon. One U.S. gallon, which is equal to 231 cubic inches.

*Gas.* Any mixture of gaseous hydrocarbons, consisting essentially of methane and heavier hydrocarbons and inert and noncombustible gases, that is extracted from beneath the surface of the earth.

Gas Baseline. As defined in Section 7.1(d).

*Governmental Approval*. Any permit, license, consent, clearance, certificate, approval, authorization or similar document or authority which any Applicable Law or Governmental Authority requires either Party to hold or obtain in order for the Fluid Handling Services to be performed, including any that are required to take Fresh Water from the Take Points.

*Governmental Authority*. Any federal, state, local, municipal, tribal or other government; any governmental, regulatory or administrative agency, commission, body or other authority exercising or entitled to exercise any administrative, executive, judicial, legislative, regulatory or taxing authority or power; and any court or governmental tribunal, including any tribal authority having or asserting jurisdiction.

*Hazardous Materials*. Collectively, (i) materials defined as "hazardous substances" in CERCLA, or any successor statute, unless such term has been given broader meaning by laws, regulations, rules, orders, ordinances, requirements or determinations of any Governmental Authority having jurisdiction with respect to the Fluid Handling Services or the Parties (including Governmental Authorities establishing common law liability), in which case such broader meaning shall apply; (ii) materials defined as "hazardous wastes" in RCRA, or any successor statute, unless such term has been given broader meaning by laws, regulations, rules, orders, ordinances, requirements or determinations of any Governmental Authority having jurisdiction with respect to the Fluid Handling Services or the Parties (including Governmental Authorities establishing common law liability), in which case such broader meaning by laws, regulations, rules, orders, ordinances, requirements or determinations of any Governmental Authority having jurisdiction with respect to the Fluid Handling Services or the Parties (including Governmental Authorities establishing common law liability), in which case such broader meaning shall apply; (c) any Hydrocarbons, Produced Water, petroleum or petroleum produce; (d) any polychlorinated biphenyl; and (e) any pollutant or contaminant or hazardous, dangerous or toxic chemical, material, waste or substance, including naturally occurring radioactive material, regulated under or within the meaning of any applicable Environmental Law.

*High-Rate Transfer Facilities*. Facilities of Antero Water and its subcontractors located at or in the vicinity of a Well Pad used to perform High-Rate Transfer Services at such Well Pad, including, as applicable, ASTs, treated water offload tanks, high-rate transfer pumps, associated hoses and lines, and all related equipment and facilities used to transfer Fresh Water from the Fresh Water System to the Fresh Water Delivery Points at the Fresh Water Delivery Rate.

High-Rate Transfer Services. As defined in Section 3.1(e).

Hydrocarbons. Gas and/or Liquid Hydrocarbons.

5

Initial Development Plan. The Development Plan attached hereto as Exhibit D.

Initial Service Area. The area described in Exhibit A hereto.

Interruptible Service. Fluid Handling Services that are accorded the lowest priority on the Water Facilities with respect to

capacity allocations, interruptions, or curtailments. Interruptible Service will be the first curtailed on the relevant part of the Water Facilities in the event of an interruption or curtailment.

Landfill Credit. As defined in Section 7.1(c).

*Liquid Hydrocarbons*. Oil, condensate, natural gasoline and all the liquid hydrocarbon production from wells, or a blend of such.

Maintenance. As defined in Section 8.2.

*Measurement Facilities*. Any facility or equipment used to measure the volume of Fresh Water, which may include meter tubes, isolation valves, tank strappings, recording devices, communication equipment, buildings and barriers.

*Minimum Annual Fresh Water Volume Commitment*. An aggregate volume of Fresh Water in the applicable calendar year equal to the number of Days in such calendar year multiplied by the Minimum Daily Fresh Water Volume Commitment with respect to such calendar year.

*Minimum Daily Fresh Water Volume Commitment*. With respect to each of calendar years 2016 through 2019, inclusive, the volume of Fresh Water, in Barrels per Day, set forth beside such calendar year in the following table:

2016	90,000 Barrels per Day
2017	100,000 Barrels per Day
2018	120,000 Barrels per Day
2019	120,000 Barrels per Day

*Minimum Quarterly Fresh Water Volume Commitment*. An aggregate volume of Fresh Water in the applicable calendar quarter equal to the number of Days in such calendar quarter multiplied by the Minimum Daily Fresh Water Volume Commitment with respect to the calendar year in which such calendar quarter falls.

*Month.* A period commencing at 10:00 a.m., Eastern Standard Time, on the first Day of a calendar month and extending until 10:00 a.m., Eastern Standard Time, on the first Day of the next succeeding calendar month. *Monthly* shall have the correlative meaning.

*MVC Credit Volumes*. With respect to each calendar year from 2016 through 2019, inclusive, or any calendar quarter in any such calendar year, as applicable, the sum of (i) the aggregate number of Barrels of Fresh Water delivered to the Fresh Water Delivery Points under

		-	
1	F		
ļ			,

this Agreement during such calendar year or calendar quarter, as applicable, plus (ii) if Antero Water fails to deliver Fresh Water to the Fresh Water Delivery Points when required to do so under this Agreement, including if such failure is excused by an event of Force Majeure (but, for the avoidance of doubt, not including Producer's inability to receive or utilize Fresh Water as a result of an event of Force Majeure), the aggregate number of additional Barrels of Fresh Water that would have been delivered to the Fresh Water Delivery Points during such calendar year or calendar quarter, as applicable, if Antero Water had delivered such Fresh Water when required to do so.

*Off-Site Tanks*. Tanks located downstream from a Well Pad where Produced Water that is entrained in Hydrocarbons and becomes separated from such Hydrocarbons downstream of a Well Pad is held pending collection, gathering, and/or disposal.

*Oil and Gas Interests*. Oil and gas leasehold interests and oil and gas mineral fee interests, including working interests, overriding royalty interests, net profits interests, carried interests, and similar rights and interests.

*Other Waste Water*. Waste water other than Produced Water that is generated from Producer's operations at Well Pads or that collects at the Well Pads or on Producer's property adjacent to the Well Pads, and including in each case all materials (including Hydrocarbons) contained in such water.

Parties. As defined in the preamble of this Agreement.

Party. As defined in the preamble of this Agreement.

*Person*. An individual, a corporation, a partnership, a limited partnership, a limited liability company, an association, a joint venture, a trust, an unincorporated organization, or any other entity or organization, including a Governmental Authority.

Planned Well. As defined in Section 4.1(a).

Planned Well Pad. As defined in Section 4.1(a).

**Previous Year Credit.** An amount with respect to each of calendar years 2017 through 2019, inclusive, equal to the excess, if any, of (1) the aggregate of the MVC Credit Volumes for the previous calendar year multiplied by the applicable Delivery Point Fee in effect on the last Day of such previous calendar year over (2) the product of the Projected Fresh Water Volumes for such previous calendar year multiplied by Weighted Delivery Point Fee in effect on the last Day of such previous calendar year. If there is no such

excess, the Previous Year Credit shall be zero.

*Produced Water*. Water produced from Wells alongside Hydrocarbons, including (i) water separated from Hydrocarbons at the wellhead upstream of a Produced Water Receipt Point located at a Well Pad through conventional mechanical separation equipment and held in tanks owned by Producer at the Well Pad, including flowback water, drilling fluids, and other fluid wastes produced from such Wells, and (ii) water entrained in Hydrocarbons that separates from such Hydrocarbons downstream of a Well Pad that a third party is not responsible for collecting,

gathering, and/or disposing of and that is held in Off-Site Tanks, and including in each case all materials (including Hydrocarbons) contained in such water.

**Produced Water Receipt Point**. The outlet flange of the Producer's Produced Water tankage located at or nearby or assigned to a Well, downstream of the Producer's separation equipment, or the outlet flange of an Off-Site Tank.

Producer. As defined in the preamble of this Agreement.

Producer Group. As defined in Section 16.1(b).

*Projected Fresh Water Volumes.* An aggregate volume of Fresh Water in the applicable calendar year equal to the number of Days in such calendar year multiplied by the number of Barrels per Day set forth beside such calendar year in the following table:

2016	126,000 Barrels per Day
2017	142,000 Barrels per Day
2018	203,000 Barrels per Day

Quarterly MVC Shortfall Fee. As defined in Section 7.2(a).

Reimbursable Landfill Costs. As defined in Section 7.1(c).

Reimbursable Waste Water Services Costs. As defined in Section 7.1(a)(vii).

Required Pumping Station. As defined in Section 4.6.

Required Retention Facility. As defined in Section 4.5.

*Retention Facility*. Each retention area, impoundment, or other similar facility used to temporarily store Fresh Water upstream of the High Rate Transfer Facilities.

Service Area. The Initial Service Area and any other area that becomes part of the Service Area pursuant to Section 2.4.

*Service Area Properties*. All Oil and Gas Interests now owned or hereafter acquired by Producer and located wholly or partly within the Service Area or pooled, unitized or communitized with Oil and Gas Interests located wholly or partly within the Service Area; provided that Service Area Properties shall not include any Oil and Gas Interests that are unitized or pooled with the properties of third parties that are not Service Area Properties if Producer is not the operator of such unit.

Specified Fee. As defined in Section 7.1(h).

System Pumping Station. As defined in Section 4.6.

8

### System Retention Facility. As defined in Section 4.5.

*System Segment*. A physically separate segment of the Fresh Water System that connects one or more Take Points to one or more System Retention Facilities, together with any underground Fresh Water lines downstream of such System Retention Facilities and any rights of way downstream of such System Retention Facilities for surface Fresh Water lines, including all Fresh Water pipelines, System Retention Facilities, System Pumping Stations, Take Point Facilities, Measurement Facilities, rights of way, fee parcels, surface rights, and permits, and all appurtenant facilities.

System Take Point. Each take point on any System Segment, including the Take Points.

*Take Point*. Those points specified in <u>Exhibit C</u> hereto, together with such additional points as are specified by Producer in accordance with <u>Section 4.4</u> from which, in accordance with agreements with the holders of water rights and/or Applicable Laws and required Governmental Approvals, Producer has procured the right for Antero Water to take Fresh Water which Antero Water will then make available to Producer for use in accordance with his Agreement.

*Take Point Facilities*. All facilities located at any Take Point that are necessary for Antero Water to take Fresh Water from the Fresh Water source at such Take Point, together with any ozonation facilities installed at such Take Point.

Target Commencement Date. As defined in Section 4.2(b).

Transportation Services. As defined in Section 3.1(f).

Treated Waste Water. Treated waste water, but excluding water remaining after treatment at the Treatment Facility.

*Treatment Facility*. The advanced 60,000 Barrel per Day wastewater treatment facility to be constructed in Doddridge County, West Virginia, and owned by Antero Treatment.

Treatment Facility Fee. As defined in Section 7.1(a)(v).

Treatment Facility Fee Supplement. As defined in Section 7.1(d).

Trucked Fresh Water Fee. As defined in Section 7.1(a)(ii).

Truck Fuel Baseline. As defined in Section 7.1(e).

USDOT. The United States Department of Transportation.

Waste Water. Produced Water and Other Waste Water.

Waste Water Quality Standards. As defined in Section 11.6.

Waste Water Services. Those Fluid Handling Services described in Section 3.1(f) and Section 3.1(g).

9

Waste Water Trucking Fee. As defined in Section 7.1(a)(vi).

Waste Water Trucking Fee Supplement. As defined in Section 7.1(e).

*Water Facilities*. The Fresh Water Facilities and any facilities owned by Antero Water through which Waste Water is gathered, collected, transported, processed, treated, recycled, or disposed of from any Oil and Gas Interests.

*Weighted Delivery Point Fee*. With respect to any Day, the sum of (i) the product of the Delivery Point Fee applicable to Fresh Water Delivery Points in West Virginia on such Day multiplied by 0.64 plus (ii) the product of the Delivery Point Fee applicable to Fresh Water Delivery Points outside West Virginia on such Day multiplied by 0.36.

*Well*. A well for the production of Hydrocarbons in which Producer owns an interest that is located on the Service Area Properties or for which Fluid Handling Services are otherwise required to be performed in accordance with this Agreement.

Well Pad. The surface installation on which one or more Wells are located.

### ARTICLE 2 PRODUCER COMMITMENTS

Section 2.1 <u>Producer Commitments.</u> Subject to the terms and conditions of this Agreement, including <u>Section 2.2</u> and <u>Section 2.3</u>, during the term of this Agreement Producer covenants and commits (a) to exclusively use Fresh Water made available by Antero Water to Producer under this Agreement for use in its hydraulic fracturing operations for all Wells operated by Producer and (b) to exclusively utilize Antero Water for the performance of the Waste Water Services for all Attributable Produced Water, as and when produced, and all Other Waste Water.

Section 2.2 <u>Conflicting Commitments</u>. Producer shall have the right to comply with any Conflicting Commitment entered into by a predecessor-in-interest to Producer that is not an Affiliate of Producer that is applicable as of the date of acquisition thereof to any Service Area Property acquired after the Effective Date (but not any entered into in connection with such acquisition); provided, however, that Producer shall have the right to comply with each Conflicting Commitment only until the first Day of the Month following the termination of such Conflicting Commitment and shall not take any voluntary action (including the exercise of any right to extend) to extend the term of such Conflicting Commitment beyond the minimum term provided for in the document evidencing such Conflicting Commitment applies to one or more Wells, but not all Wells, on a Well Pad, Producer shall have the right to comply with such Conflicting Commitment (to the extent set forth in this Section 2.2) with respect to all Wells on such Well Pad.

Section 2.3 <u>Covenant Running with the Land</u>. The covenants and commitments made by Producer under this <u>Article 2</u> are covenants running with the land. For the avoidance of doubt and in addition to that which is provided in <u>Section 19.4</u>, in the event Producer sells, transfers, conveys, assigns, grants, or otherwise disposes of any or all of its interest in any of its Service Area Properties, then any such sale, transfer, conveyance, assignment, grant, or other disposition shall

be expressly subject to this Agreement and any instrument of conveyance shall so state. Notwithstanding the foregoing, Producer shall be permitted to sell, transfer, convey, assign, grant, or otherwise dispose of Service Area Properties free of the covenant and commitment made under this <u>Article 2</u> in a sale or other disposition in which a number of net acres of Service Area Properties that, when added to the total of net acres of Service Area Properties there of the commitment made by Producer under this <u>Article 2</u>, does not exceed the aggregate number of net acres of Service Area Properties acquired by Producer after the Effective Date, including in a transaction in which Service Area Properties are exchanged for other properties located in the Service Area that would be subject to commitment made by Producer under this <u>Article 2</u>. At the request of Antero Water, the Parties shall execute and record an amendment to the memorandum of this Agreement previously entered into, as provided in <u>Section 19.16</u>, to reflect additions to the Service Area Properties.

### Section 2.4 Additional Oil and Gas Interests or Water Facilities.

If at any time Producer acquires any existing water facilities through which Fresh Water is made available to, (a) or through which Waste Water is gathered, collected, transported, processed, treated, recycled, or disposed of from, any Oil and Gas Interests, it shall, by notice delivered to Antero Water on or before the 10th day after such acquisition, which notice shall include a reasonable description of such water facilities and such Oil and Gas Interests (including an update to the Development Plan reflecting such Oil and Gas Interests that Producer is acquiring or has acquired that Producer will operate) and the price paid by Producer for such water facilities, including any liabilities assumed by Producer, offer to sell to Antero Water such water facilities, including all Fresh Water or Produced Water pipelines, Retention Facilities, pumping stations, take point facilities, delivery points, receipt points, measurement facilities, trucking facilities, disposal wells, processing, recycling, or treatment facilities, rights of way or surface rights (whether for underground or surface use, but excluding, in each case, any rights of way or surface rights to the extent such rights are granted under any Oil and Gas Interest), fee parcels, and permits, and all appurtenant facilities, in each case, to the extent the foregoing items are used in connection with such water facilities, as well as any third party contracts for services utilizing such water facilities, at the same price at which such water facilities and related assets were acquired by Producer, including the assumption of any liabilities with respect thereto assumed by Producer. Antero Water shall have the right, to be exercised by notice delivered to Producer on or before the 60th Day after delivery of Producer's notice of its acquisition of such water facilities, to acquire such water facilities and related assets at such price (including the assumption of such liabilities). If Antero Water does not deliver such notice to Producer on or before such 60th Day, Antero Water shall be deemed to have waived its right to acquire such water facilities and related assets (except in the case of a third party services offer as provided below), and (i) Producer shall have the right to own and operate such facilities to make available Fresh Water to, and/or to gather, collect, transport, process, treat, recycle, and/or dispose of Produced Water produced from and Other Waste Water associated with, the Oil and Gas Interests described in such notice and all other Oil and Gas Interests within the area within two miles of any such Oil and Gas Interest for which Antero Water is not, as of the date of Producer's initial notice of the acquisition, providing Fluid Handling Services and/or (ii) Producer shall have the right to solicit proposals from a third party service provider to acquire and operate such facilities to make available Fresh Water to, and/or to gather, collect, transport, and/or dispose of Produced Water produced from and Other Waste Water associated with, the Oil and Gas Interests described in such

11

notice owned or acquired by Producer and all Oil and Gas Interests within the area within two miles of any such Oil and Gas Interest owned or acquired by Producer for which Antero Water is not, as of the date of Producer's initial notice of the acquisition, providing Fluid Handling Services. If Producer obtains any such third party proposal, it shall, by notice to Antero Water, provide Antero Water with all the terms and conditions thereof (and, to the extent any such terms and conditions are confidential, shall use commercially reasonable efforts to obtain a waiver of any such applicable confidentiality restriction), and Antero Water shall have the right to elect, by notice delivered to Producer on or before the 60th Day after delivery of Producer's notice containing the terms and conditions of such proposal, to acquire such water facilities and related assets and provide the services offered by the applicable third party on the same terms and conditions as those offered by the third party service provider. If Antero Water does not so elect on or before such 60th Day, Producer shall have the right to contract with such third party service provider to acquire such water facilities and to provide such services, on terms no more favorable to such third party service provider than the terms and conditions of its proposal as provided to Antero Water, to the Oil and Gas Interests described in such notice in which Producer owns or acquires an interest and all other Oil and Gas Interests within the area within two miles of any such Oil and Gas Interest in which Producer owns or acquires an interest for which Antero Water is not, as of the date of Producer's initial notice of the acquisition, providing Fluid Handling Services. If Antero Water elects to acquire such water facilities, the closing of Antero Water's purchase of such water facilities and related assets from Producer shall take place as soon as reasonably practicable following Antero Water's exercise of its right to acquire such water facilities. From and after the closing of such purchase by Antero Water, all Oil and Gas Interests in which Producer owns an interest that are operated by Producer the Fresh Water for which is being made available, and the Produced Water from which is being gathered, collected, transported, processed, treated, recycled and disposed of utilizing such water facilities, as well as Oil and Gas Interests in which Producer owns an interest that are operated by Producer within the area within two miles of any such Oil and Gas Interest, unless such Oil and Gas Interests are already Service Area Properties, shall become Service Area Properties, and such area, to the extent not in or part of the then-existing Service Area, shall become part of the Service Area, and the proposed Development Plan update included in Producer's notice shall become part of the Development Plan. In any transaction in which Producer so acquires water facilities, Producer shall use reasonable efforts to cause the transaction documents for such acquisition to state a separate purchase price (and separately state any assumed liabilities) for such water facilities and related assets. If notwithstanding such reasonable efforts the transaction documents for such acquisition do not state a separate purchase price, the purchase price to be paid by Antero Water to Producer for such water facilities and related assets shall be equal to the Fair Market Value of such water facilities and related assets, and Antero Water shall assume all liabilities in respect of such water facilities and related assets to the extent arising from the ownership and operation of such water facilities and related assets and/or any occurrence with respect thereto from and after the closing of the purchase of such water facilities and related assets by Antero Water.

(b) If at any time Producer desires to have Fresh Water made available to, and/or Produced Water and Other Waste Water gathered, collected, transported, processed, treated, recycled, and/or disposed of from, any Oil and Gas Interests in which Producer owns an interest that are operated by Producer outside the then-existing Service Area, Producer shall, by notice to Antero Water specifying (i) the services it desires, including in the case of Fresh Water the take points and delivery points it desires to have connected, and (ii) the Oil and Gas Interests owned by

Producer to which Fresh Water is to be made available and from which Produced Water and Other Waste Water are to be gathered, collected, transported, processed, treated, recycled, and/or disposed of (including an update to the Development Plan reflecting such Oil and Gas Interests and the Wells planned to be drilled on such Oil and Gas Interests during the period of at least 18 Months after such notice), offer to Antero Water the opportunity to provide such Fresh Water and to gather, collect, transport, process, treat, recycle, and/or dispose of such Produced Water and Other Waste Water. Antero Water shall have the right, to be exercised by notice to Producer on or before the 60th Day after delivery of Producer's notice, to elect to perform such services as Fluid Handling Services under this Agreement. If Antero Water exercises such right as provided above, from and after the date of Antero Water's notice of exercise, the area within two miles of the Oil and Gas Interests described in such notice, to the extent not already Service Area Properties, shall become part of the Service Area Properties, and the proposed Development Plan update included in Producer's notice shall become part of the Development Plan. If Antero Water does not give such notice to Producer on or before such 60th Day, Antero Water shall be deemed to have waived its right to perform such services as Fluid Handling Services under this Agreement, except in the case of a third party services offer as provided below, and (1) Producer shall have the right to construct, own, and operate facilities to make available Fresh Water to, and/or to gather, collect, transport, process, treat, recycle, and/or dispose of Produced Water and Other Waste Water from, the Oil and Gas Interests described in such notice or (2) Producer shall have the right to solicit proposals from a third party service provider to provide such services with respect to the Oil and Gas Interests described in such notice and all Oil and Gas Interests owned by Producer within the area within two miles of any such Oil and Gas Interest for which Antero Water is not, as of the date of Producer's initial notice to Antero Water under this provision, providing Fluid Handling Services. If Producer obtains any such third party proposal, it shall, by notice to Antero Water, provide Antero Water with all the terms and conditions thereof (and, to the extent any such terms and conditions are confidential, shall use commercially reasonable efforts to obtain a waiver of any such applicable confidentiality restriction), and Antero Water shall have the right to elect, by notice to Producer on or before the 60th Day after delivery of Producer's notice containing the terms and conditions of such proposal, to provide the services described in such notice on the same terms and conditions as those offered by the third party service provider. If Antero Water does not so elect on or before such 60th Day, Producer shall have the right to contract with such third party service provider to perform such services, on terms no more favorable to such third party service provider than the terms and conditions of its proposal as provided to Antero Water, with respect to the Oil and Gas Interests described in such notice and to all Oil and Gas Interests owned by Producer within the area within two miles of any such Oil and Gas Interest for which Antero Water is not, as of the date of Producer's initial notice to Antero Water under this provision, providing Fluid Handling Services.

# ARTICLE 3 SERVICES

Section 3.1 <u>Antero Water Service Commitment</u>. Subject to and in accordance with the terms and conditions of this Agreement, during the term of this Agreement Antero Water commits to providing the following services (collectively, the "*Fluid Handling Services*") to Producer:

(a) take, or cause to be taken, at each Take Point on each Day, Fresh Water in a quantity at least equal to the lesser of (i) the maximum capacity of the Take Point Facilities as

1	3	
I	3	

they then exist at such Take Point on such Day, (ii) the maximum volume of Fresh Water that may be taken at such Take Point in accordance with Producer's rights to take Fresh Water at such Take Point, including Applicable Laws and/or any relevant Governmental Approval, and (iii) such volume of Fresh Water as shall be nominated by Producer in accordance with this Agreement;

(b) connect the Fresh Water System to Well Pads in accordance with the procedures set forth in <u>Section 4.2</u>;

(c) make available or cause to be made available, by underground or surface water lines and through the use of System Retention Facilities if applicable, and by trucking Treated Waste Water to the Well Pad if applicable, at the interconnection with the High-Rate Transfer Facilities at each Well Pad during the periods nominated by Producer in accordance with <u>Section 10.2</u> during which hydraulic fracturing operations are to be carried out at such Well Pad, Fresh Water at sufficient rates of flow so that Fresh Water may be made available at the Fresh Water Delivery Points on such Well Pad at the applicable Fresh Water Delivery Rate;

(d) at the request of Producer, as an alternative to delivery by water lines as provided in <u>Section 3.1(c)</u>, deliver Fresh Water by truck from System Retention Facilities to the inlet of the High-Rate Transfer Facilities at each Well Pad during the periods nominated by Producer in accordance with <u>Section 10.2</u> during which hydraulic fracturing operations are to be carried out at such Well Pad, Fresh Water in sufficient quantities so that Fresh Water may be made available at the Fresh Water Delivery Points on such Well Pad at the applicable Fresh Water Delivery Rate;

(e) during the periods nominated by Producer in accordance with <u>Section 10.2</u> during which hydraulic fracturing operations are to be carried out at such Well Pad, transfer Fresh Water from the Fresh Water System (to be commingled and mixed with

Treated Waste Water trucked to such Well Pad by Antero Water when applicable) through the High-Rate Transfer Facilities to the Fresh Water Delivery Point at such Well Pad at the applicable Fresh Water Delivery Rate (the "*High-Rate Transfer Services*");

(f) receive, or cause to be received, into its (or its subcontractors') trucks, or otherwise collect (i) all Attributable Produced Water from the Produced Water Receipt Points and (ii) any Other Waste Water, and deliver, or cause to be delivered, such Produced Water or Other Waste Water to the applicable disposal or treatment facility, including the Treatment Facility (the *"Transportation Services"*); and

(g) (i) from and after the in-service date of the Treatment Facility (if completed and placed into service) cause Produced Water collected from the Produced Water Receipt Points and/or Other Waste Water that in each case meets the quality standards of the Treatment Facility to be treated at the Treatment Facility and released, sold for re-use, or otherwise disposed of, up to the available capacity of the Treatment Facility from time to time and/or (ii) cause Produced Water collected from the Produced Water Receipt Points and Other Waste Water collected by Antero Water to be treated, recycled, released, sold for re-use, or otherwise disposed of, through Designated Receiving Facilities.

14

Section 3.2 <u>Priority of Fluid Handling Services</u>. All Fresh Water Services provided under this Agreement shall be Firm Services, and all other Fluid Handling Services, to the extent in the reasonable control of Antero Water, shall be performed on a first-priority basis.

# **ARTICLE 4**

# FRESH WATER SYSTEM EXPANSION AND CONNECTION OF FRESH WATER DELIVERY POINTS

### Section 4.1 Development Plan; Fresh Water Facilities Plan; Exchange and Review of Information.

The Initial Development Plan describes Producer's planned development and drilling activities relating to the (a) Service Area Properties through the date that is 18 months after the Effective Date (such plan, as updated as hereinafter provided, and including any proposed development plan that becomes part of the Development Plan pursuant to Section 2.4, the "Development Plan"). Following the Effective Date, Producer shall provide Antero Water an updated Development Plan describing the planned development and drilling activities relating to the Service Area Properties for the 18-Month period commencing on the date of such updated Development Plan on or before the last Day of each Month. Each Development Plan will include (i) information as to the Wells that Producer expects will be drilled during such period (each such Well reflected in a Development Plan, a "Planned Well"), information as to each Well Pad expected to be constructed during such period (each such Well Pad reflected in a Development Plan, a "Planned Well Pad") and the approximate locations thereof, and the earliest date on which hydraulic fracturing operations are expected to be commenced at one or more Planned Wells at each such Planned Well Pad, (ii) good faith and reasonable forecasts of the periods of time during which Fresh Water will be required at each Well Pad for the purpose of hydraulic fracturing operations for all Planned Wells on such Well Pad and the volumes of Fresh Water that will be required for such hydraulic fracturing operations at such Planned Wells during the 18-Month period following the date of such Development Plan, including for each such Planned Well the expected Fresh Water Delivery Rate, and (iii) good faith and reasonable forecasts of the volumes of Attributable Produced Water to be produced at each Well (including the Planned Wells included in such Development Plan), in each case to the extent not previously provided or, if earlier provided, as revised in Producer's good faith estimation. Producer shall make its representatives available to discuss the Development Plan from time to time with Antero Water and its representatives, in order to facilitate advance planning for expansion or improvement of the Fresh Water Facilities and to address other matters relating to the construction and installation of additions to the Fresh Water Facilities and/or the planning of the Waste Water Services. Subject to the terms of Section 4.2, Producer may provide updated or amended Development Plans to Antero Water at any time and shall provide its then-current Development Plan to Antero Water from time to time on or prior to the fifth (5<sup>th</sup>) Business Day after Antero Water's request therefor.

(b) Attached hereto as <u>Exhibit E</u> is a Fresh Water Facilities plan describing and/or depicting the Fresh Water System, including all Take Points, pipelines, Required Retention Facilities, Fresh Water Delivery Points, rights of way for surface Fresh Water lines, and all Required Pumping Stations and other major physical facilities, together with their locations, sizes and other physical specifications, operating parameters, capacities, and other relevant specifications, and together with a schedule for completing the construction and installation of the

15

planned portions thereof, in each case as currently in existence, under construction, or planned (such plan, as updated as hereinafter provided, the "*Fresh Water Facilities Plan*"). Based on the Development Plans and such other information about the expected development of the Service Area Properties as shall be provided to Antero Water by or on behalf of Producer in accordance herewith, Antero Water shall periodically update the Fresh Water Facilities Plan. Without limiting the generality of the foregoing, Antero Water shall ensure that the Fresh Water Facilities Plan reflects each Monthly Development Plan not later than 30 Days after such Development Plan is delivered. Antero Water shall make the Fresh Water Facilities Plan available for inspection by Producer and its representatives from time to time and shall make representatives of Antero Water available to discuss the Fresh Water Facilities Plan from time to time with Producer and its representatives. Antero Water shall provide Producer written updates not less frequently than Monthly on the progress of work on all facilities necessary to connect the Fresh Water System to the Planned Well Pads as set forth in the then-current Fresh Water Facilities Plan.

(c) Subject to <u>Section 4.2</u>, (i) the Parties recognize that the plans for the development of the Service Area

Properties set forth in the Development Plans, as well as all information provided by Producer to Antero Water regarding its intentions with respect to the development of the Service Area Properties, are subject to change and revision at any time at the discretion of Producer, and that such changes may impact the timing, configuration, and scope of the planned activities of Antero Water, and (ii) the exchange of such information and any changes thereto shall not give rise to any rights or liabilities as between the Parties except as expressly set forth in this Agreement, and Antero Water shall determine at its own risk the time at which it begins to work on and incur costs in connection with particular projects to expand its facilities or capacities, including the construction or installation of Water Facilities and the acquisition of rights of way, equipment, and materials necessary or desirable in connection therewith. Without limiting the generality of the foregoing, and notwithstanding anything to the contrary in this Agreement, Producer has no obligation to Antero Water under this Agreement to develop or produce any Hydrocarbons from the Service Area Properties or to pursue or complete any drilling or development on the Service Area Properties, whether or not envisioned in the Development Plan.

### Section 4.2 Expansion of Fresh Water System; Connection of Well Pads.

(a) The Fresh Water System shall be designed, developed, and constructed for the purpose of providing sufficient quantities of Fresh Water to the High-Rate Transfer Facilities at each Well Pad at a sufficient rate of flow to enable Antero Water to transfer such Fresh Water through the High-Rate Transfer Facilities to the Fresh Water Delivery Points at the applicable Fresh Water Delivery Rate as and when needed for hydraulic fracturing operations on such Well Pad, and Antero Water shall be obligated, at its sole cost and expense, subject to the provisions of this Agreement, to plan, procure, construct, install, own, and operate the Fresh Water System so as to timely extend the Fresh Water System to each Well Pad and timely deliver such quantities of Fresh Water to each Well Pad so as to permit the full scope of Fresh Water Services to be provided with respect to all the Planned Wells in accordance with this <u>Section 4.2</u>; provided, that the foregoing shall not preclude Antero Water from also designing, developing and constituting the Fresh Water System for the provision of services to third parties.

(b) Antero Water shall be obligated to extend the Fresh Water System to a particular Well Pad only if Antero Water has received from Producer a notice in the form of <u>Exhibit</u>

16

<u>F</u> hereto (or in such form as Producer and Antero Water shall otherwise agree from time to time) stating that Producer intends to drill, complete, and carry out hydraulic fracturing operations on such Wells at such Well Pad (a "Connection Notice") and setting forth the target date for the commencement of hydraulic fracturing operations (the "Target Commencement Date") at such Well Pad and the Fresh Water Delivery Rate for the Fresh Water Delivery Point at such Well Pad. Following receipt of a Connection Notice, Antero Water shall cause the necessary facilities to be constructed to connect the Fresh Water System to such Well Pads and to deliver and install at such Well Pads the necessary High-Rate Transfer Facilities, such that the Fresh Water Services can be commenced when hydraulic fracturing operations are ready to be commenced at such Well Pad. Such facilities shall be able to make Fresh Water available to such Planned Well Pad as soon as reasonably practicable following the Connection Notice and in any event on or before the later to occur of (1) the Target Commencement Date with respect to such Planned Well Pad, (2) the date that is 180 Days after the Connection Notice, (3) for any Well Pad located in the Initial Service Area but in any area outside the area described in Exhibit A as the "core area", the date specified for such area in Exhibit A, (4) for any Well Pad located outside the Initial Service Area, the date that is 18 months after the area in which such Well Pad is located became part of the Service Area, and (5) the date on which the initial Planned Well(s) at such Planned Well Pad has reached its projected depth and is ready for completion and hydraulic fracturing operations (the later of such dates, with respect to such Planned Well Pad, as it may be adjusted in accordance with this Agreement, the "Completion Deadline"). Antero Water shall provide Producer notice promptly upon Antero Water's becoming aware of any reason to believe that it may not be able to connect a Planned Well Pad to the Fresh Water System by the Target Commencement Date therefor or to otherwise complete all facilities necessary to provide the full scope of Fresh Water Services with respect to the Fresh Water Delivery Points at such Planned Well Pad by the Target Commencement Date therefor. If and to the extent Antero Water is delayed in completing and making available such facilities by a Force Majeure event or any action of Producer that is inconsistent with the cooperation requirements of Section 4.7, then the Completion Deadline for such connection shall be extended for a period of time equal to the period during which Antero Water's completion and making available of such facilities was delayed by such events or actions. If Antero Water fails to extend the Fresh Water System to a Well Pad by the Completion Date for such Well Pad, and if Antero Water fails to deliver sufficient volumes of Fresh Water to such Well Pad, by truck or otherwise, to enable Fresh Water to be delivered to the Fresh Water Delivery Points through the High-Rate Transfer Facilities at the Fresh Water Delivery Rate, by the Completion Date, then, as Producer's sole and exclusive remedies for such delay,

(i) if Antero Water notifies Producer that the required volumes of Fresh Water are available at a System Retention Facility, Producer shall have the right to pick up such volumes of Fresh Water at such System Retention Facility and truck it to the inlet to the High Rate Transfer Facilities at or in the vicinity of the Well Pad, in which case the Producer's actual cost of trucking such Fresh Water to such location shall be deducted from the Delivery Point Fee or the Cost of Service Fee, as applicable, or, if applicable, Producer shall be entitled to recover from Antero Water the excess, if any, of its third party and internal costs of trucking such Fresh Water to such location over the Delivery Point Fee or the Cost of Service Fee, as applicable, that it would have been obligated to pay Antero Water for the same volumes of Fresh Water; or

(ii) if the required volumes of Fresh Water are not available at a System Retention Facility as provided in <u>clause (i)</u> above, Producer shall have the right to obtain Fresh Water from sources other than Antero Water for hydraulic fracturing operations at such Well Pad and truck such Fresh Water to such Well Pad until such time as such Well Pad is connected to the Fresh Water System and the Fresh Water System is ready to make available Fresh Water at such Well Pad in sufficient quantity for the commencement of the Fresh Water Services with respect thereto, in which case Producer shall be entitled to recover from

Antero Water the excess, if any, of its third party and internal costs of obtaining such Fresh Water and trucking such Fresh Water to such Well Pad over the Delivery Point Fee or the Cost of Service Fee, as applicable, that it would have been obligated to pay Antero Water for the same volumes of Fresh Water; or

(iii) Producer shall have the right to complete the procurement, construction and/or installation (including through one or more subcontractors) of any rights or facilities necessary to connect the relevant Planned Well Pad to the Fresh Water System and to permit Fresh Water to be made available at such Well Pad at the applicable Fresh Water Delivery Rate, in which case Antero Water shall pay, within 30 days after presentment of an invoice therefor, to Producer an amount equal to 115% of all reasonable out of pocket costs and expenses incurred by Producer in so procuring, constructing, and/or installing such rights and facilities, and upon receipt of payment by Producer therefor, Producer shall convey all such rights and facilities owned or under the control of Producer to Antero Water (and shall use commercially reasonable efforts to obtain any applicable consents triggered by such assignment) and such rights and facilities shall thereafter be part of the Fresh Water System.

The remedies set forth in clauses (i), (ii) and (iii) above shall be applicable to Wells with Completion Deadlines that are 180 Days or more after the Effective Date. Once a Well Pad is connected to the Fresh Water System, Antero Water shall maintain such connection until such time as Producer has advised Antero Water that all hydraulic fracturing operations have been completed on all Planned Wells at such Well Pad; provided that Antero Water shall have the right to remove and re-lay temporary surface water lines from time to time as long as no delay or disruption in Producer's hydraulic fracturing operations results therefrom.

(c) If the actual commencement of hydraulic fracturing operations at a particular Well Pad is delayed through no fault of Antero Water more than 30 Days after the Target Commencement Date for such Well Pad and the Fresh Water System is connected to the Fresh Water Delivery Point at such Well Pad and available to commence providing Fresh Water to such Well Pad prior to the date such initial Planned Well is ready for hydraulic fracturing, Antero Water shall be entitled to a fee equal to interest per annum at the <u>Wall Street Journal</u> prime rate on the incremental cost and expense incurred by Antero Water to procure, construct and install the relevant rights and facilities to connect to such Well Pad and to cause such facilities to be available to commence providing Fresh Water thereto for the number of Days after the Target Commencement Date until the Day that hydraulic fracturing of the first Well at such Well Pad has commenced; provided, however, that if hydraulic fracturing of such Well has not commenced by the date that is six months after the Target Commencement Date for such Well through no fault of Antero Water or, as of an earlier date, Producer notifies Antero Water that it has elected not to fracture any Planned Wells at such Planned Well Pad, Producer shall pay, within 30 days after

18

presentment of an invoice therefor, to Antero Water an amount equal to 115% of all reasonable incremental costs and expenses incurred by Antero Water in procuring, constructing and installing such rights and facilities to connect the Fresh Water System to such Planned Well Pad and to cause such facilities to be available to commence providing Fresh Water thereto, and Antero Water shall (i) assign, transfer, and deliver to Producer all rights and facilities (including equipment, materials, work in progress, and completed construction) the costs and expenses of which have so been paid by Producer, to Producer, and (ii) use commercially reasonable efforts to obtain any applicable consents triggered by such assignment. If Producer so pays Antero Water and later completes a Well at such Well Pad which it desires to hydraulically fracture, or if such facilities are later used to connect and hydraulically fracture a Well at a different Planned Well Pad or for a third party, Antero Water shall promptly refund to Producer such amount paid by Producer, and Producer shall upon receipt of payment therefor retransfer such rights and facilities to Antero Water (and shall use commercially reasonable efforts to obtain any applicable consents triggered by such assignment).

(d) A Connection Notice shall be deemed to have been given for the Planned Wells set forth on <u>Exhibit G</u> hereto, the Target Commencement Date for which shall be as set forth on <u>Exhibit G</u>. Such Connection Notice shall be deemed to have been given for each such Planned Well 180 Days prior to such Target Commencement Date.

# Section 4.3 Installation and Operation of High-Rate Transfer Facilities.

(a) Antero Water shall be obligated, directly or through subcontractors, to engineer, procure, transport to the Well Pad or other applicable site, and erect or install on the Well Pad or on such site on or prior to the Completion Date all necessary High-Rate Transfer Facilities to enable Fresh Water to be transferred through the High-Rate Transfer Facilities to the Fresh Water Delivery Points at the Fresh Water Delivery Rate. Antero Water shall ensure that all such High-Rate Transfer Facilities remain on the Well Pad or on such site and be available to perform the High-Rate Transfer Services at all times during which Producer has notified Antero Water in accordance with Section 10.2 that hydraulic fracturing operations will be carried out on such Well Pad until such time as Producer has advised Antero Water that all hydraulic fracturing operations have been completed on all Planned Wells at such Well Pad. Antero Water shall have the right to remove and re-install or re-erect such High Rate Transfer Facilities from time to time as long as no delay or disruption in Producer's hydraulic fracturing operations results therefrom.

(b) Producer shall provide sufficient space on the Well Pad for all necessary High-Rate Transfer Facilities other than ASTs to be located on such Well Pad. Producer shall use commercially reasonable efforts to provide sufficient space on the Well Pad, or if sufficient space on such Well Pad is not available, on a site within one mile of the Well Pad, for the erection and installation of all ASTs required by Antero Water for the performance of the High-Rate Transfer Services on such Well Pad, together with rights of access to such site from a public road and easements or rights of way over which Antero Water may run hoses and temporary Fresh Water lines to the Well Pad. If Producer, through the use of commercially reasonable efforts, has been unable, by the date that is at least 120 days prior to the Target Commencement Date for such Well Pad, to obtain such a site and such rights of access rights and easements.

(c) Antero Water shall be responsible for the operation, maintenance, repair, and removal of all High-Rate Transfer Facilities, including the operation of the applicable ASTs, ensuring that such ASTs and other High-Rate Transfer Facilities are operating properly and that the Fresh Water is transferred from such ASTs at proper flow rates (such that such ASTs do not overflow), and necessary coordination with Producer's and its hydraulic fracturing contractors' personnel.

(d) If Antero Water fails to perform its obligations to timely engineer, procure, transport, erect, and install the High-Rate Transfer Facilities by the Completion Date or fails to ensure that such High-Rate Transfer Facilities remain on the Well Pad or other applicable site in each case in accordance with <u>Section 4.3(a)</u> or fails to perform the High-Rate Transfer Services in connection with the hydraulic fracturing of Wells on a Well Pad in accordance with <u>Section 3.1(e)</u>, then, as Producer's sole and exclusive remedy for such failure, Producer shall have the right to complete the procurement, transportation, erection and/or installation (including through one or more subcontractors) of replacement facilities and/or to carry out such activities itself (including through one or more subcontractors), in which case Antero Water shall pay, within 30 days after presentment of an invoice therefor, to Producer an amount equal to 115% of all reasonable out of pocket costs and expenses incurred by Producer in so engineering, procuring, transporting, erecting, and installing such facilities and carrying out such activities, and upon receipt of payment by Producer therefor, Producer shall convey all such rights (including rights under third party contracts) and facilities owned or under the control of Producer to Antero Water (and shall use commercially reasonable efforts to obtain any applicable consents triggered by such assignment).

(e) Antero Water shall have the right to deliver Treated Waste Water to the High-Rate Transfer Facilities and to commingle and mix such Treated Waste Water with Fresh Water for transfer to the Fresh Water Delivery Points as part of the High-Rate Transfer Services.

Section 4.4 <u>Take Points</u>. Antero Water shall be obligated, at Antero Water's cost, to provide Take Point Facilities with respect to the Take Points set forth on <u>Exhibit C</u>, and a connection between such Take Point Facilities and each applicable System Segment. All such Take Points shall be provided with all Take Point Facilities (including any Measurement Facilities) necessary to take volumes of Fresh Water set forth for each such Take Point on <u>Exhibit C</u> (with all expansions of capacity at such Take Point Facilities being at Producer's sole, cost, risk, and expense). Antero Water shall be responsible for obtaining all leases, easements, and other real property rights necessary for the location of Take Point Facilities; provided, however, that if Producer's rights to take Fresh Water from such Take Point (or any other surface use or other agreements of Producer) also provide Producer the right to use any lands for the purpose of installing facilities to take Fresh Water, Producer shall use commercially reasonable efforts to make such rights available to Antero Water. Producer shall have right from time to time to specify in the Development Plan or in a Connection Notice that an additional Take Point shall be added and that Fresh Water from such Take Point shall be made available to Fresh Water Delivery Points connected to a particular System Segment. If Producer so specifies, Antero Water shall, at Producer's sole cost, risk, and expense, provide Take Point Facilities for such Take Point and a connection between such Take Point Facilities and such System Segment.

Section 4.5 <u>Retention Facilities</u>. The Fresh Water Facilities Plan will describe the Retention Facilities that will be required to permit Antero Water to provide the Fresh Water

20

Services in accordance with this Agreement (each, a "*Required Retention Facility*"). Antero Water shall install, at Antero Water's cost, such Required Retention Facilities and shall own, operate and maintain such Required Retention Facilities (each such Required Retention Facility so installed by Antero Water, a "*System Retention Facility*"). For the avoidance of doubt, Antero Water shall have the right at any time to add additional Retention Facilities to the Fresh Water System as it deems necessary or appropriate to provide the Fresh Water Services and such services as it is providing to third parties.

Section 4.6Pumping Facilities.The Fresh Water Facilities Plan will describe the pumping facilities that will be requiredin order for Fresh Water to be made available at the Fresh Water Delivery Points at the rates of flow specified in Section 3.1(c) (each a<br/>"Required Pumping Station"). Antero Water shall, at Antero Water's cost, install each such Required Pumping Station and shall operate<br/>and maintain each Required Pumping Station (each such Required Pumping Station so installed by Antero Water, a "System Pumping<br/>Station"). For the avoidance of doubt, Antero Water shall have the right at any time to add additional pumping facilities to the Fresh<br/>Water System as it deems necessary or appropriate to provide the Fresh Water Services and such services as it is providing to third<br/>parties.

Section 4.7 Cooperation. Because of the interrelated nature of the actions of the Parties required to obtain the necessary Governmental Approvals from the appropriate Governmental Authorities and the necessary consents, rights of way and other authorizations from other Persons necessary to drill and complete each Planned Well and construct the required extensions of the Fresh Water System to each Planned Well Pad, the Parties agree to work together in good faith to obtain such Governmental Approvals, authorizations, consents and rights of way as expeditiously as reasonably practicable, all as provided herein. The Parties further agree to cooperate with each other and to communicate regularly regarding their efforts to obtain such Governmental Approvals, authorizations, consents and rights of way.

### ARTICLE 5

### CERTAIN PROVISIONS REGARDING WASTE WATER SERVICES

Section 5.1 <u>Access to Produced Water Receipt Points and Other Waste Water Sites</u> Producer shall be responsible for ensuring that Antero Water and its subcontractors have safe road access to all Produced Water Receipt Points and all points from which Other Waste Water is to be collected from public roadways suitable for travel by highway trucking equipment. As between Producer and

Antero Water, Producer shall be responsible for all maintenance of and damage to (and all payments in respect thereof) to all access roads from public roadways to the Produced Water Receipt Points and such other points.

# Section 5.2 <u>Dispatch Procedures</u>.

(a) Producer shall install, maintain, and operate, at Producer's cost, in its Produced Water tanks located at or in the vicinity of each Well Pad (but excluding the Wells identified in the Initial Development Plan as the "Bluestone Wells", and excluding any Off-Site Tanks) water-level sensors connected to a remote monitoring system provided by eLynx Technologies or another provider of remote monitoring services acceptable to Antero Water (the "*Monitoring Services Provider*") and cause the Monitoring Services Provider to make available

to Antero Water on an hourly or more frequent basis data regarding the level of Produced Water in each such tank and to allow Producer to view and access all such data on the Monitoring Service Provider's system, including the ability to poll for such data through the Monitoring Services Provider's system. Producer shall periodically, and no less frequently than quarterly, and at any time upon request of Antero Water if Antero Water has reason to believe there is a malfunction, inspect the tank sensors to insure that they are reading properly. Antero Water shall be responsible for the timely dispatch of trucks to all Wells at which such sensors and monitoring systems are installed and operating properly to collect Produced Water from the tanks at such Wells. In the event that Antero Water is notified or otherwise has knowledge of any outage of or malfunction in any such sensors at any such tanks or any outage of or malfunction in such monitoring system, Antero Water shall use reasonable efforts to timely dispatch trucks to collect Produced Water based on historical flow rates or on information provided by Producer but shall not otherwise be liable for any failure to timely dispatch trucks to any affected tank during any period of any such outage or malfunction.

(b) With respect to (i) the Wells identified in the Initial Development Plan as the "Bluestone Wells", (ii) all Off-Site Tanks, and (iii) all locations at or near a Well Pad where Other Waste Water is held pending collection, gathering, and/or disposal, Producer is responsible for notifying Antero Water not less than 24 hours in advance of when the Produced Water or Other Waste Water at such locations must be collected.

# Section 5.3 <u>Treatment Facility; Designated Receiving Facilities</u>.

(a) Antero Water shall treat, recycle, release, sell for re-use, or otherwise dispose of, or shall cause Antero Treatment or another Affiliate or subcontractor of Antero Water to treat, recycle, release, sell for re-use, or otherwise dispose of, all Produced Water collected at the Produced Water Receipt Points and all Other Waste Water collected by Antero Water through the Treatment Facility or through other facilities, including Antero Water's or its Affiliates' own facilities, that have been designated by Antero Water and approved by Producer (each such facility other than the Treatment Facility, a "*Designated Receiving Facility*"). If Producer approves a facility as a Designated Receiving Facility, it shall have the right, at any time upon notice to Antero Water, to withdraw such approval, and such facility shall cease to be a Designated Receiving Facility effective upon the 60<sup>th</sup> Day after such notice; provided, however, that Producer shall reimburse Antero Water for all costs reasonably incurred by Antero Water in order to utilize such facility as a Designated Receiving Facility and shall indemnify Antero Water and its Affiliates for any contractual liability incurred by Antero Water to a third party as a result of its not being able to utilize such facility as a Designated Receiving Facility.

(b) Producer shall have the right, at its sole cost and expense, to visit and observe operations at each Designated Receiving Facility operated by Antero Water, in each case during normal business hours, on reasonable notice, and subject to such reasonable safety procedures as shall be reasonably required by Antero Water. Such visits and observations shall be carried out in a manner that does not unreasonably interfere with operations at such Designated Receiving Facility. Antero Water shall use reasonable efforts to afford Producer the opportunity to visit and observe operations at (i) the Treatment Facility and (ii) each Designated Receiving Facility operated by an Antero Water subcontractor and shall at the request of Producer perform

22

such visits and observations and use reasonable efforts to include a representative of Producer among its representatives on any such visit.

Section 5.4 <u>Transportation Services Equipment</u>. Antero Water shall provide directly or through subcontractors all equipment necessary to perform the Transportation Services. The equipment shall (a) be suitable for the performance of the Transportation Services, (b) comply with the specifications for equipment used for services equivalent to the Transportation Services as required by Applicable Law, including the regulations of the USDOT, and (c) be maintained in a good, safe, and serviceable condition. Antero Water shall only use subcontractors to perform the Transportation Services that maintain a USDOT safety rating of "Satisfactory".

Section 5.5 <u>Contract Carrier Status</u>. Producer and Antero Water intend to rely on their respective rights and remedies under this Agreement and, if it would otherwise be applicable, expressly waive any and all rights and remedies under Part B, Subtitle IV of Title 49 of the United States Code that may be waived as provided in 49 USC § 14101(b)(1). Producer and Antero Water intend that the contractual arrangement documented by this Agreement be that of a contract motor carrier and for the terms and conditions of this Agreement to take precedence over any terms and conditions which might apply to a shipper and common carrier. Any use of form bills of lading, or other freight documents referring to "common carriers" and/or "tariffs", shall not alter the contract relationship created hereunder between the Parties.

# TERM

Section 6.1 <u>Term</u>. This Agreement shall become effective on the Effective Date and, unless terminated earlier by mutual agreement of the Parties, shall continue in effect until the twentieth (20th) anniversary of the Effective Date and from year to year thereafter (with the initial term of this Agreement deemed extended for each of any such additional year) until such time as this Agreement is terminated, effective upon an anniversary of the Effective Date, by notice from either Party to the other Party on or before the one hundred eightieth (180th) Day prior to such anniversary.

### ARTICLE 7 FEES AND CONSIDERATION

#### Section 7.1 Fees.

(a) Subject to the other provisions of this Agreement, including Section 7.1(j), Producer shall pay Antero Water each Month in accordance with the terms of this Agreement, for all Fluid Handling Services provided by Antero Water during such Month, an amount equal to the sum of the following:

(i) The product of (A) the aggregate volume of Fresh Water, stated in Barrels, made available by Antero Water to the Fresh Water Delivery Points, as measured at the Fresh Water Measurement Points, during such Month, excluding (1) Treated Waste Water delivered to the Fresh Water Delivery Points as contemplated in Section 4.3(e) during such Month and (2) Fresh Water delivered to the High-Rate Transfer Facilities by truck as contemplated by Section 3.1(d) during such Month, multiplied by (B) (i) in the

23

case of Fresh Water Delivery Points in West Virginia, \$3.685 and (ii) in the case of all other Fresh Water Delivery Points, \$3.635 (in each case as may be increased or decreased in accordance with <u>Section 7.1(f)</u>, the "*Delivery Point Fee*");

(ii) The sum of (A) the product of (1) the aggregate volume of Fresh Water, stated in Barrels, delivered to the High-Rate Transfer Facilities by truck as contemplated by <u>Section 3.1(d)</u> during such Month, but excluding Treated Waste Water delivered to the Fresh Water Delivery Points as contemplated in <u>Section 4.3(e)</u> during such Month, multiplied by (2) \$3.116 (as may be increased or decreased in accordance with <u>Section 7.1(f)</u>, the "*Trucked Fresh Water Fee*"), plus (B) an amount equal to all third party out-of-pocket costs actually incurred by Antero Water in trucking such Fresh Water to the applicable inlet of the applicable High-Rate Transfer Facilities, plus 3% of the amount of such out-of-pocket costs;

(iii) For Treated Waste Water delivered to the Fresh Water Delivery Points as contemplated in <u>Section 4.3(e)</u>, except to the extent that a Specified Fee is applicable to such Services in accordance with <u>Section 7.1(h)</u>, an amount equal to all third party out-of-pocket costs actually incurred by Antero Water in trucking such Treated Waste Water from the treatment plant to the applicable Well Pad, plus 3% of the amount of such out-of-pocket costs;

(iv) For the performance of the High-Rate Transfer Services, the sum of (A) an amount equal to all thirdparty out-of-pocket costs incurred by Antero Water during such Month in performing the High-Rate Transfer Services using ASTs owned by Antero Water as of the Effective Date, including all costs of engineering, procuring, transporting, installing, erecting, operating, maintaining, repairing, and removing all High-Rate Transfer Facilities used to perform High-Rate Transfer Services during such Month (but excluding the cost of procuring any ASTs in addition to those owned by Antero Water on the Effective Date), plus 3% of the amount of such out-of-pocket costs, plus (B) an amount equal to the Cost of Service Fee for all ASTs procured by Antero Water after the Effective Date;

(v) Subject to <u>Section 7.1(c)</u> and <u>Section 7.1(d)</u>, the product of (A) the aggregate volume of Waste Water, stated in Barrels, collected or received by Antero Water hereunder and processed at the Treatment Facility during such Month multiplied by (B) \$4.00 (as may be increased or decreased in accordance with <u>Section 7.1(f)</u>, the "*Treatment Facility Fee*");

(vi) For all Waste Water collected or received by Antero Water hereunder in trucks owned by Antero Water or its Affiliates during any Month, the product of (A) the aggregate volume of Waste Water so collected or received, stated in Barrels, during such Month multiplied by (B) \$5.50 (as may be increased or decreased in accordance with <u>Section 7.1(f)</u>, the *"Waste Water Trucking Fee"*);

(vii) With respect to Waste Water Services performed by subcontractors of Antero Water during such Month (to the extent that the fee referred to in Section 7.1(a)(v) is not applicable thereto), including the collection or receipt of Waste Water in Antero Water's subcontractor's trucks and the treatment, processing, recycling, or other

disposal of Waste Water at a Designated Receiving Facility not owned by Antero Water or its Affiliates and operated by a subcontractor of Antero Water, an amount equal to the sum of (A) all third party out-of-pocket costs actually incurred by Antero Water in performing such Waste Water Services ("*Reimbursable Waste Water Services Costs*"), plus (B) 3% of the amount of such costs; and

(viii) With respect to all Waste Water Services performed directly by Antero Water using facilities owned by Antero Water or its Affiliates or personnel of Antero Water or its Affiliates, other than Waste Water Services covered by the fees set forth in Section 7.1(a)(v) or Section 7.1(a)(vi), the applicable Specified Fee therefor determined as set forth in Section 7.1(h).

(b) Except as further provided in this <u>Section 7.1(b)</u> with respect to the disposal agreements and treatment agreements described in <u>Exhibit I</u> (the "*Assumed Subcontracts*"), Reimbursable Waste Water Services Costs will not include any costs incurred by Antero Water in respect of capacity reservations (however described, including minimum volume commitments, deliver-or-pay provisions, or similar provisions, and whether such reservations shall appear in disposal agreements, treatment agreements, trucking agreements, or other third party agreements) except that portion of those costs proportionately allocable to capacity actually utilized by Antero Water in the performance of Waste Water Services, it being understood that, except for its commitment under <u>Section 2.1</u>, Producer makes no commitment to provide any specific volumes of Waste Water during any period and any arrangements made by Antero Water for reserving capacity are made at Antero Water's sole risk. Notwithstanding the foregoing, Reimbursable Waste Water Services Costs will include the reservation charges under the Assumed Subcontracts for the periods set forth in <u>Exhibit I</u>, to the extent actually incurred by Antero Water, provided that Antero Water shall, to the extent within its or its Affiliates' control, deliver Waste Water collected by Antero Water hereunder to facilities under the Assumed Subcontracts in a manner so as to minimize the amount of such reservation charges actually incurred for volumes in excess of volumes actually delivered.

(c) The Treatment Facility Fee described in Section 7.1(a)(v) assumes that the Treatment Facility will have an onsite landfill owned by Antero Treatment or one of its Affiliates at which all by-products of the treatment of Waste Water at the Treatment Facility will be disposed of. If for any reason Antero Treatment is unable to dispose of such by-products at such landfill during any period (including as a result of any delay in Antero Treatment's construction of such landfill or any inability of Antero Treatment to maintain required Governmental Approvals for such landfill) and such by-products must be disposed of at a third-party landfill, then the Treatment Facility Fee applicable during such period shall be reduced by (i) during any period in which none of such by-products are disposed of at such landfill, 0.10 (as increased or decreased pursuant to Section 7.1(f), the "Landfill Credit"), or (ii) during any period in which some, but not all, such by-products are disposed of at such landfill, a pro-rata portion of the Landfill Credit, based on the quantity (in tons) of such by-products that are not disposed of at such landfills. During any such period, the Producer shall pay to Antero Water, in addition to the remainder of the Treatment Facility Fee after deduction of the Landfill Credit or applicable portion thereof, an amount equal to the sum of (A) all third party out-of-pocket costs actually incurred by Antero Water in disposing of such by-products at thirdparty landfills,

25

including transportation and landfill fees ("Reimbursable Landfill Costs"), plus (B) 3% of the amount of such costs.

The Treatment Facility Fee assumes that Antero Treatment's cost of treating Waste Water at the Treatment (d) Facility includes (i) \$0.35 per Barrel for electricity (as increased or decreased pursuant to Section 7.1(f), the "Electricity Baseline"), (ii) \$0.32 per Barrel for chemicals (as increased or decreased pursuant to Section 7.1(f), the "Chemicals Baseline"), and (iii) \$0.55 per Barrel for Gas used as fuel ("Fuel Gas") (as increased or decreased pursuant to Section 7.1(f), the "Gas Baseline"). If during any Month the costs actually incurred by Antero Treatment for electricity and/or chemicals per Barrel of waste water processed at the Treatment Facility during such Month exceed 120% of the Electricity Baseline or the Chemicals Baseline, as applicable, then Antero Water shall be entitled to a supplement to the Treatment Facility Fee for such Month equal to the product of (1) the aggregate volume of Waste Water, stated in Barrels, collected or received by Antero Water hereunder and processed at the Treatment Facility multiplied by (2) the excess of the actual per-Barrel cost for electricity and/or chemicals, as applicable, over the Electricity Baseline and/or the Chemicals Baseline, as applicable. If during any Month the costs actually incurred by Antero Treatment for Fuel Gas per Barrel of waste water processed at the Treatment Facility during such Month exceed 107.5% of the Gas Baseline, then Antero Water shall be entitled to a supplement to the Treatment Facility Fee for such Month equal to the product of (I) the aggregate volume of Waste Water, stated in Barrels, collected or received by Antero Water hereunder and processed at the Treatment Facility multiplied by (II) the excess of the actual per-Barrel cost for Fuel Gas over the Gas Baseline. The supplements to the Treatment Facility Fee provided for in this Section 7.1(d) are herein referred to as the "Treatment Facility Fee Supplement".

(e) The Waste Water Trucking Fee assumes a cost of truck or vehicle fuel of \$3.50 per gallon (the "*Truck Fuel Baseline*" and together with the Electricity Baseline, the Chemicals Baseline, and the Gas Baseline, each a "*Baseline Component*"). If during any Month Antero Water's actual cost per gallon for such fuel exceeds 107.5% of the Truck Fuel Baseline, then Antero Water shall be entitled to a supplement to the Waste Water Trucking Fee for such Month equal to the product of (i) the number of gallons of such fuel consumed in the performance of the Waste Water Services covered by the Waste Water Trucking Fee multiplied by (ii) the excess of Antero Water's actual cost per gallon over the Truck Fuel Baseline (the "*Waste Water Trucking Fee Supplement*").

(f) After each of the first five (5) Contract Years, one hundred percent (100%), and after the sixth (6<sup>th</sup>) Contract Year and each Contract Year thereafter, fifty-five percent (55%) of the Delivery Point Fee, the Trucked Fresh Water Fee, the Treatment Facility Fee, the Waste Water Trucking Fee, any Specified Fee, the Landfill Credit, and each Baseline Component shall be adjusted up or down on an annual basis in proportion to the percentage change, from the preceding year, in the All Items Consumer Price Index for All Urban Consumers (CPI-U) for the U.S. City Average, 1982-84 = 100, as published by the United States Department of Labor, Bureau of Labor Statistics ("*CPI*"). Such adjustment shall be made effective upon the first Day of each Contract Year commencing in the Contract Year beginning in 2016, and shall reflect the percentage change in the CPI as it existed for June of the preceding Contract Year from the CPI for the second immediately preceding June; *provided, however*, that the Delivery Point Fee, the Trucked Fresh Water Fee, the Treatment Facility Fee, the Waste Water Trucking Fee, any

Specified Fee, the Landfill Credit, and each Baseline Component shall never be less than the initial amounts stated in <u>Section 7.1(a)</u>, <u>Section 7.1(c)</u>, <u>Section 7.1(d)</u>, or <u>Section 7.1(e)</u>, or as applicable; nor shall such amounts be increased or decreased by more than 3% in any given Contract Year.

(g) To the extent that Antero Water is entitled under this <u>Section 7.1</u> to reimbursement for the cost of third-party services or other third-party items incurred in carrying out the Fluid Handling Services, Antero Water shall ensure that the reimbursed costs are commensurate with market costs for such services or items and shall, except in the case of emergency situations where to do so is not reasonably practicable, ensure that all such services and items are obtained through competitive bid processes.

If Antero Water desires to perform any Waste Water Services (other than the Waste Water Services covered (h) by the fees set forth in Section 7.1(a)(v) or Section 7.1(a)(vi)) or the Fresh Water Services covered by the fee set forth in Section 7.1(a) (iii) in each case directly using its own facilities and personnel, it may do so, subject to the provisions of this Section 7.1(h). Antero Water may elect to perform such Fluid Handling Services in accordance with the terms of this Agreement for the Cost of Service Fee in accordance with Section 7.1(j) (to the extent permitted under Section 7.1(j)); provided, however, that if Antero Water so elects with respect to any Waste Water disposal services, the Parties shall enter into a separate agreement covering only such disposal services and containing market standard provisions for the performance of such disposal services and providing for the Cost of Service Fee to be paid for such disposal services. Alternatively, Antero Water may, by notice to Producer, propose specific fees under this Section 7.1(h) (each, a "Specified Fee") for any such Fluid Handling Services to be performed directly by Antero Water, with such scope (including geographical scope) as shall be defined in such notice. Producer shall have the right, to be exercised by notice to Antero Water on or before the 30<sup>th</sup> day after delivery of Antero Water's proposal notice, to either (i) accept such proposal or (ii) notify Antero Water that it intends to seek fee quotes from third parties for the performance of such Fluid Handling Services. If Producer fails to timely deliver such notice, it shall be deemed to have accepted such proposal. If Producer accepts or is deemed to have accepted such proposal, then the Specified Fee proposed by Antero Water shall become effective with respect to the applicable Fluid Handling Services and Antero Water shall perform such Fluid Handling Services in accordance with this Agreement; provided, however, that if such Fluid Handling Services include disposal services, the Parties shall also enter into a separate agreement covering only such disposal services containing market standard provisions for the performance of such disposal services and providing for the Specified Fees in Antero Water's proposal to be applicable to such disposal services. If Producer timely delivers notice that it intends to seek fee quotes from third parties, it shall have a period of 30 days after the delivery of such notice to Antero Water to obtain such third party quotes. If Producer desires to accept any such third party quote and enter into a contract with such third party to perform such Fluid Handling Services for the quoted fees, it shall first deliver notice to Antero Water, including a copy of such fee quote and a description of such Fluid Handling Services covered by such quote. Antero Water shall have the right, to be exercised by notice to Producer on or before the 10th day after delivery of such notice to Antero Water, to perform any such Fluid Handling Services for the quoted fees. If Antero Water timely delivers such notice, then Antero Water shall perform such Fluid Handling Services in accordance with this Agreement, and the quoted fees shall be deemed to be the Specified Fee for the applicable Fluid Handling Services; provided, however, that if such Fluid Handling Services include Waste Water disposal, the Parties shall enter into a separate agreement covering only such disposal services containing

27

market standard provisions for the performance of such disposal services and providing for the quoted fees, which shall thereafter be deemed to be Specified Fees for purposes of this Agreement. If Antero Water fails to timely deliver such notice, Producer shall be entitled to enter into a contract with such third party covering such Fluid Handling Services, but the scope of such Fluid Handling Services shall not exceed the scope set forth in the third party quote provided to Antero Water by Producer, and the fees shall not be greater than the fee quote provided to Antero Water. If Producer does not deliver notice of any third party quotes to Antero Water on or before the end of the 30-day period referenced above, Producer shall be deemed to have accepted Antero Water's initial proposal.

(i) At the request of Antero Water, the Parties will reasonably cooperate and work together in good faith to agree on revised distance-based waste water trucking fees that would provide the same overall economics to Antero Water as the Waste Water Trucking Fee provided for in this Agreement as of the date hereof.

Notwithstanding the foregoing provisions of this Section 7.1 or any other provision to the contrary in this (j) Agreement, Antero Water shall have the right to elect to be paid for some or all Fluid Handling Services on a cost of service basis to the extent set forth in this Section 7.1(j). Antero Water shall have the right to elect to be paid on a cost of service basis for (i) any Fluid Handling Services performed with respect to Wells on Service Area Properties outside the Initial Service Area (other than the Fluid Handling Services covered by the fees set forth in Section 7.1(a)(v) and (ii) for any Waste Water Services, other than the Waste Water Services covered by the fees set forth in Section 7.1(a)(v) or Section 7.1(a)(vi), to be performed by Antero Water directly using its own facilities and personnel, whether or not within the Initial Service Area (with all other Fluid Handling Services to be performed for the fees set forth in Section 7.1(a)). With respect to such Fluid Handling Services, Antero Water may elect, by notice to Producer at least three (3) Months prior to the commencement of any Contract Year or, in the case of any water facilities acquired by Antero Water pursuant to Section 2.4(a), in the notice given by Antero Water in accordance with such Section that Antero Water will acquire such water facilities, to be paid on a cost of service basis for the Fluid Handling Services specified in such notice commencing at the beginning of such Contract Year or with the acquisition of such facilities, as applicable, and continuing for the remaining term of this Agreement, but only with respect to the facilities so acquired or discrete parts of the Water Facilities (each, a "CS Facility") that are placed into service or acquired during such Contract Year or a later Contract Year, as applicable. The Fluid Handling Services specified in such notice may be of any scope determined by Antero Water in its sole discretion and may include all eligible Fluid Handling Services or any part thereof and may include, by way of example only, making Fresh Water available at a particular Well Pad or group of Well Pads, collecting Attributable Produced Water from any Produced Water Receipt Points, providing additional System Retention Facilities or System Pumping Facilities, connecting additional Take Points, providing trucking services with respect to any Produced Water Receipt Points, providing any Waste

Water treatment or disposal services, and any other subset of the Fluid Handling Services determined by Antero Water, in each case subject to the foregoing sentence. All Fluid Handling Services provided from time to time on a cost of service basis shall be bundled together for purposes of calculating a single Monthly cost of service fee (the "*Cost of Service Fee*"), which shall be calculated with respect to each Contract Year as set forth in <u>Exhibit H</u> attached hereto.

28

# Section 7.2 <u>Minimum Daily Fresh Water Volume Commitment</u>.

(a) In addition to the fees payable by Producer to Antero Water pursuant to <u>Section 7.1(a)</u>, Producer shall pay Antero Water and, if applicable, Antero Water shall pay Producer the amounts provided for in this <u>Section 7.2</u> in respect of the Minimum Daily Fresh Water Volume Commitment.

(b) Subject to the last sentence of this <u>Section 7.2(b)</u>, with respect to each of the first three calendar quarters in each calendar year from 2016 through 2019, inclusive, Producer shall pay to Antero Water, on or before the 30th Day after receipt of Antero Water's invoice therefor (which shall be delivered not more than sixty (60) Days after the end of such calendar quarter), an amount equal to the excess, if any (such excess, the "*Quarterly MVC Shorfall Fee*"), of:

(i) the product of the Minimum Quarterly Fresh Water Volume Commitment for such calendar quarter multiplied by the Weighted Delivery Point Fee in effect on the last Day of such calendar quarter over

(ii) the product of the aggregate of the MVC Credit Volumes for such calendar quarter multiplied by the applicable Delivery Point Fee in effect on the last Day of such calendar quarter.

If there is no excess resulting in a Quarterly MVC Shortfall Fee, then no amounts shall be payable by Producer under this <u>Section 7.2(b)</u>. If there is such an excess, commencing in calendar year 2017 Producer shall be entitled to a credit against any Quarterly MVC Shortfall Fee in any calendar year in an amount not to exceed, in the aggregate, the Previous Year Credit, if any, with respect to such calendar year (but for the second and third such calendar quarters in such calendar year, only to the extent not credited in a previous calendar quarter in such calendar year), and the Quarterly MVC Shortfall Fee shall be reduced by the amount of such credit.

(c) On or before the 60<sup>th</sup> Day after the end of each calendar year from 2016 through 2019, inclusive, Antero Water shall deliver an invoice to Producer setting forth the amount resulting from the following calculation:

(i) the product of the Minimum Annual Fresh Water Volume Commitment for such calendar year multiplied by the Weighted Delivery Point Fee in effect on the last Day of such calendar year minus

(ii) the sum of (A) the product of the aggregate of the MVC Credit Volumes for such calendar year multiplied by the applicable Delivery Point Fee in effect on the last Day of such calendar year plus (B) commencing in calendar year 2017, the amount of any Previous Year Credit plus (C) the amount, if any, actually paid by Producer in respect of Quarterly MVC Shortfall Fees in such calendar year.

If the result is a positive amount, Producer shall pay such amount to Antero Water on or before the 30th Day after receipt of such invoice. If the result is a negative amount, Antero Water shall pay to Producer, concurrently with the delivery of such invoice, the lesser of (1) the absolute value

29

of such amount and (2) the sum of the amounts actually paid by Producer in respect of Quarterly MVC Shortfall Fees in such calendar year.

# ARTICLE 8 CERTAIN RIGHTS AND OBLIGATIONS OF PARTIES

Section 8.1 <u>Operational Control of Antero Water's Facilities</u>. Subject to the terms and conditions of this Agreement, Antero Water shall design, construct, own, operate, and maintain the Water Facilities at its sole cost and risk. Antero Water shall be entitled to full and complete operational control of its facilities and shall be entitled to operate and reconfigure its facilities in a manner consistent with its obligations under this Agreement.

**Section 8.2** <u>Maintenance</u>. Antero Water shall be entitled, without liability, to interrupt its performance hereunder to perform necessary or desirable inspections, maintenance, testing, alterations, modifications, expansions, connections, repairs or replacements to its facilities as Antero Water deems necessary ("*Maintenance*"), with reasonable notice provided to Producer, except in cases of emergency where such notice is impracticable or in cases where the operations of Producer will not be affected. Before the beginning of each calendar year, Antero Water shall provide Producer in writing with a projected schedule of the Maintenance to be performed during the year and the anticipated date of such Maintenance. On or before the 10<sup>th</sup> Day before the end of each Month, Antero Water shall provide Producer with its projected maintenance schedule for the following Month.

Section 8.3 <u>Right of Way and Access</u>.

Except as otherwise provided in Section 4.3(b), Section 4.4, Section 5.1 and this Section 8.3, Antero Water is (a) responsible for the acquisition of rights of way, crossing permits, licenses, use agreements, access agreements, leases, fee parcels, and other land rights necessary to construct, own, and operate the Water Facilities, and all such rights in land shall be solely for use by Antero Water and shall not be shared with Producer, except as otherwise agreed by Antero Water; provided that Producer hereby grants, without warranty of title, either express or implied, to the extent that it has the right to do so without the incurrence of material expense or liability, an easement and right of way upon all lands covered by the Service Area Properties for the purpose of installing, using, maintaining, servicing, inspecting, repairing, operating, and replacing, all or any portion of the Water Facilities to the extent necessary for the performance of this Agreement, including pipelines, meters, Retention Facilities and storage, and other equipment, and for disconnecting, and removing such Water Facilities; provided, further, that the exercise of these rights by Antero Water shall not unreasonably interfere with Producer's lease operations or with the rights of owners in fee, and will be subject to Producer's safety and other reasonable access requirements applicable to Producer's personnel. Producer shall not have a duty to maintain, and will be able to freely assign, the underlying agreements (such as leases, easements, and surface use agreements) that such grant of easement or right of way to Antero Water is based upon, and such grants of easement or right of way will terminate if Producer loses its rights to the underlying applicable property, or is unable to grant such rights to Antero Water under any contract or instrument to which Producer is a party or is otherwise bound, regardless of the reason for such loss of rights. Notwithstanding the foregoing, (i) Producer will use commercially reasonable efforts to assist Antero Water to secure replacements for such terminated grants of easement or

30

right of way, in a manner consistent with the cooperation requirements of <u>Section 4.7</u>, (ii) to the extent that Producer agrees that Antero Water's Measurement Facilities may be located on Producer's Well Pad sites, Producer shall be responsible for obtaining any necessary rights to locate such Measurement Facilities on such Well Pad sites, and (iii) Producer shall use reasonable efforts to involve Antero Water in Producer's negotiations with the owners of lands covered by the Service Area Properties so that Producer's surface use agreements and Antero Water's rights of way with respect to such lands can be concurrently negotiated and obtained. Without limiting the generality of the foregoing, Producer agrees to make space available at each Well Pad to which Fresh Water is made available pursuant to this Agreement sufficient for Antero Water to install and construct pig receiving and other facilities necessary for Antero Water to pig the Fresh Water System to each such Well Pad.

(b) Without limiting the generality of <u>Section 8.3(a)</u>, Producer hereby agrees that, to the extent that Producer can afford such rights to Antero Water without the incurrence of material expense or liability, Antero Water shall have the right to use, for the purpose of installing, using, maintaining, servicing, inspecting, repairing, operating, and replacing all or any portion of the Water Facilities to the extent necessary for the performance of this Agreement, including pipelines, meters, Retention Facilities and storage, and other equipment, and for disconnecting, and removing such Water Facilities, all Excluded Assets (as such term is defined in that certain Assignment and Bill of Sale dated as of [\_\_\_\_] between Producer and Antero Treatment, and /or that certain Amended and Restated Assignment and Bill of Sale dated of such assignments, were used or held for use for both the water business and upstream business of Producer. If Producer's rights with respect to any such Excluded Assets expire or are terminated or released, Producer will use commercially reasonable efforts to assist Antero Water to secure replacements for such Excluded Assets, in a manner consistent with the cooperation requirements of <u>Section 4.7</u>

# Section 8.4 Third Party Services; Capacity Allocations on the Fresh Water System.

(a) Subject to this <u>Section 8.4</u> and the other provisions of this Agreement, Antero Water has the right to contract with other Persons to perform services utilizing the Water Facilities on an Interruptible Service basis.

(b) If on any Day the total volumes of Fresh Water that Antero Water has agreed to take from all System Take Points on a particular System Segment, including the volumes Antero Water is obligated to take pursuant to <u>Section 3.1(a)</u>, for any reason (including Maintenance, Force Majeure, or any foreseen or unforeseen reduction in capacity) exceed the capacity of such System Segment, including any System Retention Facilities located on such System Segment, Antero Water shall reduce the volumes taken from all the System Take Points pursuant to its agreements with third parties prior to any reduction in the amounts taken at the Take Points pursuant to this Agreement.

(c) To the extent that the volumes of Fresh Water that Antero Water is obligated to make available under this Agreement at the points of interconnection to the High-Rate Transfer Facilities located at a Well Pad on a particular System Segment, including the volumes that Antero

31

Water is obligated to make available at the Fresh Water Delivery Points pursuant to <u>Section 3.1(e)</u>, for any reason (including Maintenance, Force Majeure, or any foreseen or unforeseen reduction in capacity) exceed the volume of Fresh Water available and/or the capacity of such System Segment to make available Fresh Water at such points of interconnection, then Antero Water shall interrupt or curtail volumes of Fresh Water made available to customers other than Producer (including an customers who are assignees of Producer or successors in interest to Producer with respect to any Service Area Property) prior to any reduction in the amounts made available at the Fresh Water Delivery Points.

(d) Except as otherwise provided in this <u>Section 8.4</u>, Antero Water shall be free to use any Fresh Water present in the Fresh Water System to satisfy its obligations to Producer and any third party and shall not be obligated to ensure that Fresh Water taken from any Take Point is utilized only to perform Fluid Handling Services for Producer; provided, however, that Antero Water shall

comply with any restrictions on the use of any Fresh Water taken from any Take Point and made available to any third party, and ensure that such third party also so complies, to the extent that Producer has informed Antero Water of such restrictions.

# Section 8.5 Governmental Approvals; Compliance with Applicable Law.

(a) Except as otherwise provided in this <u>Section 8.5</u>, Antero Water is responsible for obtaining all Governmental Approvals required for its performance of the Fluid Handling Services in accordance with this Agreement. Notwithstanding the foregoing, the Producer is responsible for obtaining (i) at Producer's cost, all Governmental Approvals to take Fresh Water from the Take Points (other than leases, easements, and other real property rights necessary for the location of Take Point Facilities, which are the responsibility of Antero Water) and (ii) at Antero Water's cost, any Governmental Approvals that, in accordance with Applicable Law, must be obtained by or in the name of Producer.

(b) Antero Water shall comply with all Applicable Laws and all Governmental Approvals in all material respects in its performance of the Fluid Handling Services and shall bear all costs and liabilities associated with complying with or the failure to comply with such Applicable Laws and Governmental Approvals (including Governmental Approvals obtained by or in the name of Producer as contemplated in <u>Section 8.5(a)(ii)</u>); provided, however, that if pursuant to any Applicable Law or Governmental Approval (including a Governmental Approval obtained by or in the name of Producer in accordance with <u>Section 8.5(a)(ii)</u>), a Retention Facility is required to be shut down and reclaimed or remediated during a period of 18 months after the Effective Date, Producer shall bear all cost and liability associated with shutting down and reclaiming or remediating such Retention Facility.

(c) Antero Water shall require all subcontractors to which it delegates any of the Fluid Handling Services to comply with all Applicable Laws and relevant Governmental Approvals in such subcontractor's performance of such portion of the Fluid Handling Services.

32

# ARTICLE 9 FRESH WATER DELIVERY RATES

**Section 9.1** Fresh Water Delivery Rates. Subject to the other provisions of this Agreement, Antero Water shall construct and operate the Fresh Water System in a manner so as to permit Fresh Water to be made available at the points of interconnection to the High-Rate Transfer Facilities at the rates of flow required by Section 3.1(c) and shall install and operate the High-Rate Transfer Facilities in a manner so as to make Fresh Water available at the Fresh Water Delivery Points at the applicable Fresh Water Delivery Rates.

Section 9.2 <u>Producer Facilities</u>. Producer, at its own expense, shall construct, equip, maintain, and operate all facilities necessary to receive Fresh Water at the Fresh Water Delivery Points at the applicable Fresh Water Delivery Rates.

# ARTICLE 10 FRESH WATER NOMINATION

Section 10.1 <u>Maximum Take Point Volumes</u>. Producer has informed Antero Water of the maximum volume of Fresh Water that can be taken, if any, in accordance with Producer's rights to take such Fresh Water, including any Applicable Law or Governmental Approval, at each of the currently existing Take Points. Producer shall promptly inform Antero Water of the maximum volume of Fresh Water that can be taken, if any, in accordance with Producer's rights to take such Fresh Water, including any Applicable Law or Governmental Approval, at each new Take Point established in accordance with <u>Section 4.4</u>. Producer shall notify Antero Water of any change to such maximum volumes immediately after Producer becomes aware of any such change.

**Section 10.2** Fresh Water Delivery Nominations. Producer shall regularly communicate to Antero Water the dates on which Producer plans to carry out hydraulic fracturing operations on each Well Pad and shall by notice to Producer not less than five Business Days in advance specify the dates on which Antero Water is to commence deliveries of Fresh Water at the Fresh Water Delivery Points at such Well Pad.

Section 10.3 <u>Changes in Fresh Water Delivery Rates</u>. If Producer desires that Antero Water make Fresh Water available on any Day at the Fresh Water Delivery Point on any Well Pad at flow rates greater than or less than the Fresh Water Delivery Rate specified for such Well Pad in the Connection Notice for such Well Pad, Producer may, on not less than 5 Business Days' notice to Antero Water, increase or decrease the Fresh Water Delivery Rate for such Well Pad.

# ARTICLE 11 FRESH WATER AVAILABILITY AND QUALITY; WASTE WATER QUALITY

**Section 11.1** Fresh Water Availability. The obligation of Antero Water to perform the Fresh Water Services is subject to the conditions that (a) Producer has obtained all necessary rights, including all Governmental Approvals (but excluding any leases, easements, or other real property rights necessary for the location of Take Point Facilities, which, subject to the other provisions of this Agreement, shall be the obligation of Antero Water), to take Fresh Water from the Take Points in sufficient volumes to make available Fresh Water at the Fresh Water Delivery Points in the volumes specified in Section 3.1(e), and (b) the quality of the Fresh Water available to be taken at

such Take Points is at all times in compliance with the Fresh Water Quality Standards. Antero Water shall be relieved of its obligations to provide the Fresh Water Services to the extent that it is prevented from doing so because either of such conditions is not satisfied at any time and the reason either such condition is not satisfied was not caused in whole or in part by Antero Water.

Section 11.2 <u>Take Point Fresh Water Standards</u>. Fresh Water at each Take Point shall be free from any contamination or any substances, in each case, that would result in such Fresh Water being unsuitable for use in hydraulic fracturing operations in accordance with all then-applicable general industry practices, Applicable Laws, and Governmental Approvals, or that would result in any damage to the Fresh Water Facilities; provided, however, that the presence in such Fresh Water of invasive species (including zebra mussels and quagga mussels) and other invasive or non-native pathogens or infectious agents that would be removed by treatment in ozonation facilities shall not render such Fresh Water nonconforming (the standards set forth in this <u>Section 11.2</u> being called the "*Fresh Water Quality Standards*"). Antero Water shall be responsible for the installation, operation, and maintenance of ozonation facilities at any Take Points where they are required to remove any such invasive species or other invasive or non-native pathogens or infectious agents, and Producer shall have no liability to Antero Water arising from the presence thereof in any Fresh Water at the Take Points, including any liability under the indemnities in <u>Section 11.3</u> and <u>Section 11.6</u>.

Section 11.3 <u>Non-Conforming Take Point Fresh Water</u>. If the Fresh Water quality at any Take Point does not conform to the Fresh Water Quality Standards, then Antero Water will have the right to immediately discontinue taking Fresh Water at such Take Point so long as the Fresh Water at such Take Point continues to be non-conforming. In the event that Antero Water takes receipt of non-conforming Fresh Water at any Take Point, Producer agrees to be responsible for, and to defend, indemnify, release, and hold Antero Water and its Affiliates, directors, officers, employees, agents, consultants, representatives, and invitees harmless from and against, all claims and losses of whatever kind and nature resulting from such non-conforming Fresh Water, including claims and losses resulting from any negligent acts or omissions of any indemnified party, but excluding claims and losses to the extent caused by or arising out of the gross negligence or willful misconduct of the indemnified party.

Section 11.4 Delivery Point Fresh Water Quality Standards. Antero Water shall make available Fresh Water (including, for the avoidance of doubt, Fresh Water consisting of a mixture of raw fresh water and Treated Waste Water) at each Fresh Water Delivery Point that meets the Fresh Water Quality Standards, provided that Fresh Water at the Take Points meets the Fresh Water Quality Standards, and subject to the provisions of <u>Section 11.5</u>.

Section 11.5 <u>Retention Facility Contamination</u>. Antero Water shall use reasonable efforts to ensure that Fresh Water that is held in the System Retention Facilities does not become subject to any contamination or pollution that would result in the Fresh Water held in the System Retention Facilities not meeting the Fresh Water Quality Standards. Except to the extent that Antero Water has failed to perform its obligations set forth in the immediately preceding sentence, Producer agrees to be responsible for, and to defend, indemnify, release, and hold Antero Water and its Affiliates, directors, officers, employees, agents, consultants, representatives, and invitees harmless from and against, all claims and losses of whatever kind and nature resulting from the quality of the Fresh Water in the System Retention Facilities, including any remediation obligation

34

under any Applicable Laws relating to the environment, and including claims and losses resulting from any negligent acts or omissions of any indemnified party, but excluding claims and losses to the extent caused by or arising out of the gross negligence or willful misconduct of the indemnified party.

Non-Conforming Waste Water. If the Produced Water quality at any Produced Water Receipt Point or any Section 11.6 Other Waste Water where collected by Antero Water does not conform to the Waste Water Quality Standards, then Antero Water will have the right to immediately discontinue taking Produced Water at such Produced Water Receipt Point or such Other Waste Water so long as the Produced Water at such Produced Water Receipt Point or such Other Waste Water continues to be non-conforming. In the event that Antero Water takes receipt of non-conforming Produced Water at any Produced Water Receipt Point or collects nonconforming Other Waste Water, Producer agrees to be responsible for, and to defend, indemnify, release, and hold Antero Water and its Affiliates, directors, officers, employees, agents, consultants, representatives, and invitees harmless from and against, all claims and losses of whatever kind and nature resulting from such non-conforming Produced Water or Other Waste Water, including claims and losses resulting from any negligent acts or omissions of any indemnified party, but excluding claims and losses to the extent caused by or arising out of the gross negligence or willful misconduct of the indemnified party. "Waste Water Quality Standards" means, with respect to any Waste Water, that such Waste Water is free from any contamination or any substances that would result in such Waste Water not meeting any requirements imposed by Applicable Law for transportation by truck or any quality standards of the Treatment Facility or a Designated Receiving Facility; provided, however, that such Waste Water shall not be rendered nonconforming by the presence in such Waste Water of invasive species (including zebra mussels and quagga mussels) and other invasive or non-native pathogens or infectious agents to the extent that such species, pathogens, or agents originated from Fresh Water delivered by Antero Water hereunder.

# ARTICLE 12 MEASUREMENT EQUIPMENT AND PROCEDURES

Section 12.1 <u>Measurement Equipment</u>. Antero Water shall (or shall cause Antero Treatment or another Affiliate of Antero Water to) install, own, operate, and maintain Measurement Facilities (a) to measure the volumes of Fresh Water taken at each Take Point on each Day and (b) to measure the volumes of raw Fresh Water made available at each Fresh Water Measurement Point on each Day. Antero Water shall also track and account for volumes of Fresh Water consisting of Treated Waste Water that are commingled with raw Fresh Water in the High-Rate Transfer Facilities. Producer shall have the right to install check Measurement Facilities at each Take Point and each Fresh Water Measurement Point, including the right to install check measurement equipment on Antero Water's meter tubes and orifice unions. Producer may also check Antero Water's measurements of Fresh Water using data collected by Producer or its hydraulic

fracturing contractors regarding the actual volumes of Fresh Water delivered from the High-Rate Transfer Facilities into the Fresh Water Delivery Points. Unless such check measurements show a discrepancy of greater than 2%, for all purposes of this Agreement, the volumes of Fresh Water delivered on such Day to the Fresh Water Delivery Points shall be deemed to equal the volumes of raw Fresh Water measured at the Fresh Water Measurement Point on each

35

Day. The changing and integration of the charts (if utilized for measurement purposes hereunder) and calibrating and adjusting of meters shall be performed by Antero Water.

Section 12.2 Notice of Measurement Facilities Inspection and Calibration. Each Party shall give reasonable notice to the other Party in order that the other Party may, at its option, have representatives present to observe any reading, inspecting, testing, calibrating or adjusting of Measurement Facilities or other facilities or equipment (including Producer's or its hydraulic fracturing contractor's equipment collecting data regarding Fresh Water volumes at the Fresh Water Delivery Points) used in measuring or checking the measurement of volumes of raw Fresh Water under this Agreement. The official electronic data from such Measurement Facilities or other facilities or equipment shall remain the property of the owner thereof, but copies of such records shall, upon request, be submitted, together with calculations and flow computer configurations therefrom, to the requesting Party for inspection and verification.

# Section 12.3 Measurement Accuracy Verification.

(a) Each Party shall verify the accuracy of all Measurement Facilities owned by such Party used in measuring or checking the measurement of volumes of raw Fresh Water under this Agreement no less frequently than quarterly. Neither Party shall be required to cause adjustment or calibration of such equipment more frequently than once per Month, unless a special test is requested pursuant to Section 12.4.

(b) If, during any test of such Measurement Facilities, an adjustment or calibration error is found which results in an incremental adjustment to the calculated flow rate through each meter run in excess of two percent (2%) of the adjusted flow rate (whether positive or negative and using the adjusted flow rate as the percent error equation denominator), then any previous recordings of such equipment shall be corrected to zero error for any period during which the error existed (and which is either known definitely or agreed to by the Parties) and the total flow for the period redetermined in accordance with the provisions of <u>Section 12.5</u>. If the period of error condition cannot be determined or agreed upon between the Parties, such correction shall be made over a period extending over the last one half of the time elapsed since the date of the prior test revealing the two percent (2%) error.

(c) If, during any test of such Measurement Facilities, an adjustment or calibration error is found which results in an incremental adjustment to the calculated hourly flow rate which does not exceed two percent (2%) of the adjusted flow rate, all prior recordings and electronic flow computer data shall be considered to be accurate for volume determination purpose.

**Section 12.4** <u>Special Tests</u>. In the event a Party desires a special test (a test not scheduled by a Party under the provisions of <u>Section 12.3</u>) of any Measurement Facilities used in measuring or checking the measurement of volumes of raw Fresh Water under this Agreement, seventy-two (72) hours advance notice shall be given to the other Party and both Parties shall cooperate to secure a prompt test of the accuracy of such equipment. If such Measurement Facilities tested are found to be within the two percent (2%) range of accuracy set forth in <u>Section 12.3(b)</u>, then the Party that requested the test shall pay the costs of such special test including any labor and transportation costs pertaining thereto. If such Measurement Facilities tested are found to be

36

outside the two percent (2%) range of accuracy set forth in <u>Section 12.3(b)</u>, then the Party that owns such Measurement Facilities shall pay such costs and perform the corrections according to <u>Section 12.5</u>.

Section 12.5 <u>Metered Flow Rates in Error</u>. If, for any reason, any Measurement Facilities used in measuring or checking the measurement of volumes of raw Fresh Water under this Agreement are (i) out of adjustment, (ii) out of service, or (iii) out of repair and the total calculated flow rate through each meter run is found to be in error in excess of two percent (2%) of the adjusted flow rate as described in <u>Section 12.3(b)</u>, the total volumes of Fresh Water made available shall be determined in accordance with the first of the following methods which is feasible:

(a) By using the registration of any mutually agreeable check metering facility, if installed and accurately registering (subject to testing as provided for in <u>Section 12.3</u>), or Producer's or its hydraulic fracturing contractor's data regarding Fresh Water received at the Fresh Water Delivery Points, taken together with Antero Water's data regarding the amount of such Fresh Water delivered that consisted of Treated Waste Water;

(b) Where multiple meter runs exist in series, by calculation using the registration of such meter run equipment; provided that they are measuring Fresh Water in common with the faulty metering equipment, are not controlled by separate regulators, and are accurately registering;

(c) By correcting the error by re-reading of the official data, or by straightforward application of a correcting factor to the volumes recorded for the period (if the net percentage of error is ascertainable by calibration, tests or mathematical calculation); or

(d) By estimating the volumes, based upon volumes made available during periods of similar conditions when the meter was registering accurately.

**Section 12.6** Waste Water Measurement. Volumes of Waste Water shall be determined for all relevant purposes under this Agreement based on the number of truck loads of Waste Water received or collected by Antero Water or its subcontractors and/or the number of truck loads of Waste Water delivered to the Treatment Facility or a Designated Receiving Facility and on the capacity of the trucks, assuming that each truck load consisted of the full capacity of the relevant truck.

**Section 12.7** Units of Measurement. The unit of volume for measurement of Fresh Water and Waste Water hereunder shall be one Barrel. Delivery rates for Fresh Water shall be stated in Barrels per minute.

**Section 12.8** <u>Record Retention</u>. The Party owning the Measurement Facilities shall retain and preserve all test data, flow metering data, and similar records for any calendar year for a period of at least twenty-four (24) Months following the end of such calendar year unless Applicable Law requires a longer time period or such Party has received notification of a dispute involving such records, in which case records shall be retained until the related issue is resolved.

37

# ARTICLE 13 NOTICES

Section 13.1 Notices. Unless otherwise provided herein, any notice, request, invoice, statement, or demand which either Party desires to serve upon the other regarding this Agreement shall be made in writing and shall be considered as delivered (i) when hand delivered, or (ii) when delivery is confirmed by pre-paid delivery service (such as FedEx, UPS, DHL or a similar delivery service), or (iii) if mailed by United States certified mail, postage prepaid, three (3) Business Days after mailing, or (iv) if sent by facsimile transmission, when receipt is confirmed by the equipment of the transmitting Party, or (v) when sent via email (provided that any notice sent by email shall be followed by notice sent by another form of delivery permitted by this Section 13.1 within two (2) Business Days after the date such email was sent); provided, if sent by email after normal business hours or if receipt of a facsimile transmission is confirmed after normal business hours, delivery shall be deemed to have occurred on the next Business Day. Notwithstanding the foregoing, if a Party desires to serve upon the other a notice of default under this Agreement, or if Producer desires to serve upon Antero Water a Connection Notice, the delivery of such notice shall be considered effective under this <u>Section 13.1</u> only if delivered by any method set forth in items (i) through (iv) above. Any notice shall be given to the other Party at the following address, or to such other address as either Party shall designate by notice to the other:

Producer:	ANTERO RESOURCES CORPORATION 1615 Wynkoop Street Denver, Colorado 80202 Attn: Chief Financial Officer Phone: (303) 357-7310 Fax Number: (303) 357-7315 Email:	
With copy to:	For water control, nominations & balancing: Completion Supervisor Phone: (303) 357-7310 Fax Number: (303) 357-7315 Email:	
	For accounting, financial, and legal: Controller Phone: (303) 357-7310 Fax Number: (303) 357-7315 Email:	
	38	

Antero Water:

ANTERO WATER LLC 1615 Wynkoop Street Denver, Colorado 80202

Attn: Chief Financial Officer Phone: (303) 357-7310 Fax Number: (303) 357-7315 Email:

For water control, nominations & balancing: Scheduling Coordinator Phone: (303) 357-7310 Fax Number: (303) 357-7315 Email:

For accounting, financial, and legal: Controller Phone: (303) 357-7310 Fax Number: (303) 357-7315 Email:

### ARTICLE 14 PAYMENTS

Invoices. Not later than the tenth (10th) Day following the end of each Month, Antero Water shall provide Section 14.1 Producer with a detailed statement setting forth the volumes of Fresh Water made available during such Month at the Fresh Water Delivery Points (including the volumes delivered to the inlet of the High-Rate Transfer Facilities by pipeline and the volumes delivered by truck), the volumes of Waste Water collected at the Produced Water Receipt Points or other collection points by Antero Water and processed in the Treatment Facility or any Designated Receiving Facility during such Month, and the Fresh Water Delivery Point Fee, the Trucked Fresh Water Fee, and, if applicable, the Treatment Facility Fee, any Treatment Facility Fee Supplement, the Waste Water Trucking Fee, any Waste Water Trucking Fee Supplement, Reimbursable Waste Water Services Costs, Reimbursable Landfill Costs, any Specified Fee, and the Cost of Service Fee with respect to such Month, together with all relevant data on which such fees are based, measurement summaries, run tickets, bills of lading, and third party invoices (including evidence of the actual cost of electricity, chemicals, and Fuel Gas for the Treatment Facility to the extent of any Treatment Facility Fee Supplement and the actual cost of truck and/or vehicle fuel to the extent of any Waste Water Trucking Fee Supplement), and all relevant supporting documentation. If actual data is not available on such tenth (10<sup>th</sup>) Day, Antero Water shall base such invoiced amounts on reasonable estimates, which shall be trued up in future invoices against actual data when available. To the extent that any other data is unavailable on such tenth (10th) Day, Antero Water shall be obligated to deliver such supporting documentation as soon as it becomes available. Producer shall make payment to Antero Water by the last Business Day of the Month in which such invoice is received. Such payment shall be made by wire transfer pursuant to wire transfer instructions delivered by Antero Water to Producer in writing from time to time. If any overcharge or undercharge in any form whatsoever shall at any time be found and the invoice therefor has been paid, Antero Water shall refund any amount of overcharge, and

39

Producer shall pay any amount of undercharge, on the last Business Day of the following Month, provided, however, that no retroactive adjustment will be made beyond a period of twenty-four (24) Months from the date of a statement hereunder.

Section 14.2 <u>Right to Suspend on Failure to Pay</u>. If any undisputed amount due hereunder remains unpaid for sixty (60) Days after the due date, Antero Water shall have the right to suspend or discontinue Fluid Handling Services hereunder until any such past due amount is paid.

Section 14.3 <u>Audit Rights</u>. Either Party, on not less than thirty (30) Days prior notice to the other Party, shall have the right at its expense, at reasonable times during normal business hours, but in no event more than twice in any period of twelve (12) consecutive Months, to audit the books and records of the other Party to the extent necessary to verify the accuracy of any statement, allocation, measurement, computation, charge, payment made under, or obligation or right pursuant to this Agreement. The scope of any audit shall be limited to the twenty-four (24) Month period immediately prior to the Month in which the notice requesting an audit was given. All statements, allocations, measurements, computations, charges, or payments made in any period prior to the twenty-four (24) Month period immediately prior to the Month in which the audit is requested shall be conclusively deemed true and correct and shall be final for all purposes.

Section 14.4 <u>Payment Disputes</u>. In the event of any dispute with respect to any payment hereunder, Producer shall make timely payment of all undisputed amounts, and Antero Water and Producer will use good faith efforts to resolve the disputed amounts within sixty (60) Days following the original due date. Any amounts subsequently resolved shall be due and payable within ten (10) Days of such resolution.

**Section 14.5** <u>Interest on Late Payments</u>. In the event that Producer shall fail to make timely payment of any sums, except those contested in good faith or those in a good faith dispute, when due under this Agreement, interest will accrue at an annual rate equal to ten percent (10%) from the date payment is due until the date payment is made.

**Section 14.6** <u>Credit Assurance</u>. Antero Water shall apply consistent evaluation practices to all similarly situated customers to determine Producer's financial ability to perform its payment obligations under this Agreement.

(a) If Antero Water has reasonable grounds for insecurity regarding the performance of any obligation by Producer under this Agreement (whether or not then due), Antero Water may demand Adequate Assurance of Performance from Producer (which demand shall include reasonable particulars for the demand and documentation supporting the calculation of the amount demanded), which Adequate Assurance of Performance shall be provided to Antero Water within five (5) Business Days after such demand. If Producer fails to provide such Adequate Assurance of Performance within such time, then Antero Water may suspend its performance under this Agreement until such Adequate Assurance of Performance is provided. However, any such suspension by Antero Water shall not relieve Producer of its payment obligations. The exercise by Antero Water of any right under this <u>Section 14.6</u> shall be without prejudice to any claims for damages or any other right of Antero Water under this Agreement. As used herein, "Adequate Assurance of Performance" means any of the following, as selected in Producer's discretion subject to the below:

(i) an irrevocable standby letter of credit in an amount not to exceed an amount that is equal to sixty (60) Days of Producer's payment obligations hereunder from a financial institution rated at least A- by S&P or at least A3 by Moody's in a form and substance reasonably satisfactory to Antero Water;

(ii) cash collateral in an amount not to exceed an amount that is equal to sixty (60) Days of Producer's payment obligations hereunder to be deposited in an escrow account as designated by Antero Water; Antero Water is hereby granted a security interest in and right of set-off against all such cash collateral, which is or may hereafter be delivered or otherwise transferred to such escrow account in connection with this Agreement; or

(iii) a guaranty in an amount not to exceed an amount that is equal to sixty (60) Days of Producer's payment obligations hereunder, which guaranty is reasonably acceptable to Antero Water in form and substance.

(b) The term of any security provided under this <u>Section 14.6</u> shall never exceed sixty (60) Days, after which the security shall terminate (or in the case of cash collateral, be immediately returned by Antero Water to Producer without further action by either Party). Nothing shall prohibit Antero Water, however, from requesting additional Adequate Assurance of Performance following the end of any such term, so long as the conditions triggering such a request under this <u>Section 14.6</u> exist.

# ARTICLE 15 FORCE MAJEURE

**Section 15.1** <u>Suspension of Obligations</u>. In the event a Party is rendered unable, wholly or in part, by Force Majeure to carry out its obligations under this Agreement, other than the obligation to make payments then or thereafter due hereunder, and such Party promptly gives notice and reasonably full particulars of such Force Majeure to the other Party promptly after the occurrence of the cause relied on, then the obligations of the Party giving such notice, so far as and to the extent that they are affected by such Force Majeure, shall be suspended during the continuance of any inability so caused, but for no longer period, and such cause shall so far as reasonably possible be remedied with all reasonable dispatch by the Party claiming Force Majeure.

Section 15.2 Definition of Force Majeure. The term "*Force Majeure*" as used in this Agreement shall mean any cause or causes not reasonably within the control of the Party claiming suspension and which, by the exercise of reasonable diligence, such Party is unable to prevent or overcome, including acts of God, strikes, lockouts or other industrial disturbances, acts of the public enemy, acts of terror, sabotage, wars, blockades, military action, insurrections, riots, epidemics, landslides, subsidence, lightning, earthquakes, fires, storms or storm warnings, crevasses, floods, washouts, civil disturbances, explosions, breakage or accident to wells, machinery, equipment or lines of pipe, the necessity for testing or making repairs or alterations to wells, machinery, equipment or lines of pipe, inability of any Party hereto to obtain, after the exercise of reasonable diligence, necessary materials,

4	ŀ	1

supplies, or Governmental Approvals, any action or restraint by any Governmental Authority (so long as the Party claiming suspension has not applied for or assisted in the application for, and has opposed where and to the extent reasonable, such action or restraint, and as long as such action or restraint is not the result of a failure by the claiming Party to comply with any Applicable Law), and, in the case of a Force Majeure claimed by Antero Water only, any breach of any representation or warranty of Producer or any failure by Producer to perform any obligation of Producer under either (a) that certain Amended and Restated Contribution Agreement dated November 10, 2014, by and between Producer and Antero Midstream LLC or (b) that certain Contribution, Conveyance and Assumption Agreement dated September 17, 2015, by and among Producer, Antero Midstream Partners LP, and Antero Treatment (the "*Contribution Agreement*").

Section 15.3 <u>Settlement of Strikes and Lockouts</u>. It is understood and agreed that the settlement of strikes or lockouts shall be entirely within the discretion of the Party having the difficulty, and that the above requirement that any Force Majeure shall be remedied with all reasonable dispatch shall not require the settlement of strikes or lockouts by acceding to the demands of the opposing party when such course is inadvisable in the sole discretion of the Party having the difficulty.

**Section 15.4** <u>Payments for Services Performed</u>. Notwithstanding the foregoing, it is specifically understood and agreed by the Parties that an event of Force Majeure will in no way affect or terminate Producer's obligation to make payment for Fluid Handling Services performed prior to such event of Force Majeure.

### ARTICLE 16 INDEMNIFICATION

Section 16.1 Antero Water. Subject to the terms of this Agreement, including Section 19.8,

(a) Antero Water shall release, indemnify, defend, and hold harmless Producer and its Affiliates, directors, officers, employees, agents, consultants, representatives, and invitees from and against all claims and losses to the extent arising out of or

relating to (i) the operations of Antero Water, but only to the extent that liability for such claims and losses is not otherwise allocated pursuant to the indemnification provisions of <u>Article 11</u>, <u>Article 17</u>, <u>Section 16.2(a)(iii)</u>, or <u>Section 16.2(b)</u>, and (ii) any breach of this agreement by Antero Water, including in each case claims and losses resulting from any negligent acts or omissions of any indemnified party, but excluding in each case claims and losses to the extent caused by or arising out of the gross negligence or willful misconduct of the indemnified party.

(b) Except as otherwise provided in <u>Section 16.2(a)(iii)</u>. Antero Water shall release, indemnify, defend, and hold harmless Producer and its joint interest owners and Producer's contractors and subcontractors of any tier and its and their Affiliates, directors, officers, employees, agents, consultants, representatives, and invitees (collectively, the "*Producer Group*") from and against all claims and losses for bodily injury to or death of any individual in the Antero Water Group or damage to or loss of the property of any Person in the Antero Water Group in each case arising while such individual or property is on a Well Pad or any property of the Producer adjacent to a Well Pad in connection with the performance by Antero Water of the High-Rate

42

Transfer Services or other services at such Well Pad, including in each case claims and losses resulting from any negligent acts or omissions of any indemnified party, but excluding in each case claims and losses to the extent caused by or arising out of the gross negligence or willful misconduct of the indemnified party.

Section 16.2 <u>Producer</u>. Subject to the terms of this Agreement, including <u>Section 19.8</u>,

(a) Producer shall release, indemnify, defend, and hold harmless Antero Water and its Affiliates, directors, officers, employees, agents, consultants, representatives, and invitees from and against all claims and losses to the extent arising out of or relating to (i) the operations of Producer but only to the extent that liability for such claims and losses is not otherwise allocated pursuant to the indemnification provisions of <u>Article 11</u>, <u>Article 17</u>, <u>Section 16.1(b)</u>, or (ii) any breach of this agreement by Producer or (iii) except as otherwise provided in <u>Section 11.5</u>, pollution or contamination from the emission, discharge or release of Hazardous Materials occurring (A) in the course of the performance of any Fluid Handling Services up until the point of custody and title transfer to Antero Water at the applicable Produced Water Receipt Points or (B) with respect to any Waste Water that has been treated and is in transit from the relevant treatment facility to, or is in, the High-Rate Transfer Facilities, or is otherwise being handled by Antero Water as part of the High-Rate Transfer Services, including in each case claims and losses resulting from any negligent acts or omissions of any indemnified party, but excluding in each case claims and losses to the extent caused by or arising out of the gross negligence or willful misconduct of the indemnified party.

(b) Producer shall release, indemnify, defend, and hold harmless Antero Water and its contractors and subcontractors of any tier and its and their Affiliates, directors, officers, employees, agents, consultants, representatives, and invitees (collectively, the "*Antero Water Group*") from and against all claims and losses for bodily injury to or death of any individual in the Producer Group or damage to or loss of the property of any Person in the Producer Group in each case arising while such individual or property is on a Well Pad or any property of the Producer adjacent to a Well Pad in connection with the operations of Producer at such Well Pad, including in each case claims and losses resulting from any negligent acts or omissions of any indemnified party, but excluding in each case claims and losses to the extent caused by or arising out of the gross negligence or willful misconduct of the indemnified party.

# ARTICLE 17 CUSTODY AND TITLE

Section 17.1 <u>Custody of Fresh Water</u>. As among the Parties, Producer shall be in custody, control and possession of (i) raw Fresh Water until such Fresh Water is taken into the Fresh Water Facilities at the Take Points and (ii) Fresh Water after such Fresh Water is delivered to the Fresh Water Delivery Points. As among the Parties, Antero Water shall be in custody, control and possession of all Fresh Water in the Fresh Water Facilities at all other times. Except as otherwise provided in <u>Section 16.2(a)(iii)</u>, the Party having custody and control of Fresh Water under the terms of this Agreement shall be responsible for, and shall defend, indemnify, release and hold the other Party and its Affiliates, directors, officers, employees, agents, consultants, representatives, and invitees harmless from and against, all claims and losses of whatever kind and nature for anything that may happen or arise with respect to such Fresh Water when such Fresh

43

Water is in its custody and control, including claims and losses resulting from any negligent acts or omissions of any indemnified party, but excluding all claims and losses to the extent caused by or arising out of the gross negligence or willful misconduct of the indemnified party.

Section 17.2 Custody of Waste Water. As among the Parties, Producer shall be in custody, control and possession of Waste Water until such Waste Water is received by Antero Water or its subcontractors at the Produced Water Receipt Points or at the point at which Other Waste Water is collected by Antero Water. As among the Parties, Antero Water shall be in custody, control and possession of all Waste Water from and after its receipt by Antero Water or its subcontractors at the Produced Water Receipt Points or at the point at which Other Waste Water is collected by Antero Water. As among the Parties, Antero Water shall be in custody, control and possession of all Waste Water is collected by Antero Water. Except as otherwise provided in Section 16.2(a)(iii), the Party having custody and control of Waste Water under the terms of this Agreement shall be responsible for, and shall defend, indemnify, release and hold the other Party and its Affiliates, directors, officers, employees, agents, consultants, representatives, and invitees harmless from and against, all claims and losses of whatever kind and nature for anything that may happen or arise with respect to such Waste Water when such Waste Water is in its custody and control, including claims and losses resulting from any negligent acts or omissions of any indemnified party,

but excluding claims and losses to the extent caused by or arising out of the gross negligence or willful misconduct of the indemnified party.

Section 17.3 <u>Title to Waste Water</u>. Antero Water shall take title to all Waste Water received by Antero Water or its subcontractors at the Produced Water Receipt Points or at the point where Other Waste Water is collected, as applicable. Producer shall ensure that such Waste Water is free of all liens arising by, through, or under Producer, other than liens arising by operation of law.

### ARTICLE 18 PAYMENTS FOR FRESH WATER; TAXES

**Section 18.1** Payments for Fresh Water: Taxes. To the extent that any Person is entitled to any payment in respect of Fresh Water taken from any Take Point, including any taxes, Producer shall pay or cause to be paid and agrees to hold Antero Water harmless as to the payment of all such payments or taxes. Antero Water shall not become liable for such payments or taxes, unless designated to remit those taxes on behalf of Producer by any duly constituted Governmental Authority having authority to impose such obligations on Antero Water, in which event the amount of such taxes remitted on Producer's behalf shall be reimbursed by Producer upon receipt of invoice, with corresponding documentation from Antero Water setting forth such payments. Antero Water shall pay or cause to be paid all taxes, charges and assessments of every kind and character required by statute or by order of Governmental Authorities with respect to its facilities, including the Fresh Water Facilities. Except as provided in Exhibit H attached hereto, neither Party shall be responsible nor liable for any taxes or other statutory charges levied or assessed against the facilities of the other Party, including ad valorem tax (however assessed), used for the purpose of carrying out the provisions of this Agreement or against the net worth or capital stock of such Party. Notwithstanding the foregoing, to the extent that such payments or taxes relate to Fresh Water that is made available to a third party pursuant to Section 8.4(d). Antero Water shall look only to such third party, and not to Producer, for payment or reimbursement of such payments and taxes to the extent relating to the Fresh Water made available to such third party, and shall use

44

reasonable efforts to ensure that Fresh Water not subject to such payments and taxes is made available to Producer in preference to third parties.

# ARTICLE 19 MISCELLANEOUS

**Section 19.1** <u>Rights</u>. The failure of either Party to exercise any right granted hereunder shall not impair nor be deemed a waiver of that Party's privilege of exercising that right at any subsequent time or times.

**Section 19.2** <u>Applicable Laws</u>. This Agreement is subject to all valid present and future laws, regulations, rules and orders of Governmental Authorities now or hereafter having jurisdiction over the Parties, this Agreement, or the services performed or the facilities utilized under this Agreement. To the extent that the performance of the Fluid Handling Services by Antero Water shall at any point in time become prohibited or restricted by Applicable Laws or the provisions of any Governmental Approval, Antero Water shall be relieved from its obligations to perform such Fluid Handling Services.

# Section 19.3 <u>Governing Law; Jurisdiction</u>.

(a) This Agreement shall be governed by, construed, and enforced in accordance with the laws of the State of Colorado without regard to choice of law principles.

(b) The Parties agree that the appropriate, exclusive and convenient forum for any disputes between the Parties arising out of this Agreement or the transactions contemplated hereby shall be in any state or federal court in City and County of Denver, Colorado, and each of the Parties irrevocably submits to the jurisdiction of such courts solely in respect of any proceeding arising out of or related to this Agreement. The Parties further agree that the Parties shall not bring suit with respect to any disputes arising out of this Agreement or the transactions contemplated hereby in any court or jurisdiction other than the above specified courts.

# Section 19.4 <u>Successors and Assigns</u>.

(a) This Agreement shall extend to and inure to the benefit of and be binding upon the Parties and their respective successors and permitted assigns.

(b) To the extent any Affiliate of Producer acquires any Oil and Gas Interests or any water facilities, Producer shall cause such Affiliate to comply with the obligations of Producer under Article 2 of this Agreement with respect to its Oil and Gas Interests and to enter into an agreement with Antero Water substantially the same as this Agreement.

(c) Except as set forth in <u>Section 19.4(d)</u> and <u>Section 19.4(e)</u>, neither Party shall have the right to assign its respective rights and obligations in whole or in part under this Agreement without the prior written consent of the other Party, and any assignment or attempted assignment made otherwise than in accordance with this <u>Section 19.4</u> shall be null and void *ab initio*.

using its own facilities and/or perform any or all such services through Affiliates (including Antero Treatment) or third parties, in which case references herein to the Fresh Water Facilities shall be deemed to be references to such facilities of the relevant Affiliate or third party.

(e) Notwithstanding the foregoing clause (c):

(i) Antero Water shall have the right to assign its rights and obligations under this Agreement, in whole or in part, as applicable, without the consent of Producer (A) to an Affiliate of Antero Water or (B) to any Person to which the Fresh Water Facilities or any part thereof or any facilities through which the Waste Water Services are performed has been or will be transferred who (1) hires (or retains, as applicable) operating personnel who are then operating such facilities (or has similarly experienced operating personnel itself), (2) has operated for at least two (2) years prior to such assignment systems similar to such facilities, or (3) contracts for the operation of such facilities with another Person that satisfies either of the foregoing conditions (1) or (2) in this clause (B), provided, in the case of an assignment under either of clause (A) or clause (B), that the assignee assumes in writing all of Antero Water's obligations hereunder (if applicable, to the extent of the Fresh Water Facilities or other Facilities being transferred to such Person). Notwithstanding the foregoing, Antero Water shall not be released from its obligations hereunder upon any assignment of this Agreement (in whole or in part) unless the assignee has creditworthiness as reasonably determined by Producer that is equal to or greater than the higher of Antero Water's creditworthiness as of the Effective Date and Antero Water's creditworthiness as of the date of the assignment.

(ii) Producer shall have the right to assign its rights and obligations under this Agreement, in whole or in part, as applicable, without the consent of Antero Water, to any Person to which it sells, assigns, or otherwise transfers all or any portion of the Service Area Properties and who assumes in writing all of Producer's obligations hereunder (if applicable, to the extent of the Service Area Properties being transferred to such Person). Notwithstanding the foregoing, Producer shall not be released from its obligations hereunder upon any assignment of this Agreement (in whole or in part) unless the assignee has a credit rating that is equal to or higher than the higher of Producer's credit rating as of the Effective Date and Producer's credit rating as of the date of the assignment.

(iii) Each Party shall have the right to grant a security interest in this Agreement to a lender or other debt provider (or trustee or agent on behalf of such lender) of such Party.

**Section 19.5** <u>Severability</u>. If any provision of this Agreement is determined to be void or unenforceable, in whole or in part, then (i) such provision shall be deemed inoperative to the extent it is deemed void or unenforceable, (ii) the Parties agree to enter into such amendments to this Agreement in order to give effect, to the greatest extent legally possible, to the provision that is determined to be void or unenforceable and (iii) the other provisions of this Agreement in all other respects shall remain in full force and effect and binding and enforceable to the maximum extent permitted by Applicable Law; provided, however, that in the event that a material term

46

under this Agreement is so modified, the Parties will, timely and in good faith, negotiate to revise and amend this Agreement in a manner which preserves, as closely as possible, each Party's business and economic objectives as expressed by the Agreement prior to such modification.

Section 19.6 <u>Confidentiality</u>.

(a) <u>Confidentiality</u>. Except as otherwise provided in this <u>Section 19.6</u>, each Party agrees that it shall maintain all terms and conditions of this Agreement, and all information disclosed to it by the other Party or obtained by it in the performance of this Agreement and relating to the other Party's business (including Development Plans, Fresh Water Facilities Plans, and all data relating to the production of Producer) (collectively, "*Confidential Information*") in strictest confidence, and that it shall not cause or permit disclosure of this Agreement or its existence or any provisions contained herein without the prior written consent of the other Party.

(b) <u>Permitted Disclosures</u>. Notwithstanding <u>Section 19.6(a)</u> disclosures of any Confidential Information may be made by either Party (i) to the extent necessary for such Party to enforce its rights hereunder against the other Party; (ii) to the extent to which a Party or an Affiliate of a Party is required to disclose all or part of this Agreement by a statute or by the order or rule of a Governmental Authority exercising jurisdiction over the subject matter hereof, by order, by regulations, or by other compulsory process (including deposition, subpoena, interrogatory, or request for production of documents); (iii) to the extent required by the applicable regulations of a securities or commodities exchange; (iv) to a third person in connection with a proposed sale or other transfer of a Party's interest in this Agreement, provided such third person agrees in writing to be bound by the terms of this <u>Section 19.6</u>; (v) to its own directors, officers, employees, agents and representatives; (vi) to an Affiliate of such Party; (vii) to financial advisors, attorneys, and banks, provided that such Persons are subject to a confidentiality undertaking consistent with this <u>Section 19.6(b)</u>, or (viii) in the case of Producer only, excluding any information disclosed to Producer by Antero Water pursuant to <u>Article 3</u> of this Agreement, to a royalty, overriding royalty, net profits or similar owner burdening production from the Service Area Properties, provided such royalty, overriding royalty, net profits or similar owner agrees in writing to be bound by the terms of this <u>Section 19.6</u>.

(c) <u>Notification</u>. If either Party is or becomes aware of a fact, obligation, or circumstance that has resulted or may result in a disclosure of any of the terms and conditions of this Agreement authorized by <u>Section 19.6(b)(ii)</u> or <u>(iii)</u>, it shall so notify in writing the other Party promptly and shall provide documentation or an explanation of such disclosure as soon as it is available.

(d) <u>Party Responsibility</u>. Each Party shall be deemed solely responsible and liable for the actions of its directors, officers, employees, agents, representatives and Affiliates for maintaining the confidentiality commitments of this <u>Section 19.6</u>.

(e) <u>Public Announcements</u>. The Parties agree that prior to making any public announcement or statement with respect to this Agreement or the transaction represented herein permitted under this <u>Section 19.6</u>, the Party desiring to make such public announcement or statement shall provide the other Party with a copy of the proposed announcement or statement prior to the intended release date of such announcement. The other Party shall thereafter consult

47

with the Party desiring to make the release, and the Parties shall exercise their reasonable best efforts to (i) agree upon the text of a joint public announcement or statement to be made by both such Parties or (ii) in the case of a statement to be made solely by one Party, obtain approval of the other Party to the text of a public announcement or statement. Nothing contained in this <u>Section 19.6</u> shall be construed to require either Party to obtain approval of the other Party to disclose information with respect to this Agreement or the transaction represented herein to any Governmental Authority to the extent such disclosure is required by Applicable Law or necessary to comply with disclosure requirements of the Securities and Exchange Commission, New York Stock Exchange, or any other regulated stock exchange.

(f) <u>Survival</u>. The provisions of this <u>Section 19.6</u> shall survive any expiration or termination of this Agreement; provided that other than with respect to information disclosed pursuant to <u>Article 3</u>, as to which such provisions shall survive indefinitely, such provisions shall survive only a period of one (1) year.

Section 19.7 Entire Agreement, Amendments and Waiver. The Contribution Agreement, this Agreement and the documents and instruments and other agreements specifically referred to herein or therein or delivered pursuant hereto or thereto, including the exhibits and schedules hereto and thereto, (a) constitute the entire agreement among the Parties with respect to the subject matter hereof and supersede all prior agreements and understandings, both written and oral, among the Parties with respect to the subject matter hereof and (b) are not intended to confer upon any other Person any rights or remedies hereunder except as otherwise expressly provided herein or therein. Each Party agrees that (i) no other Party (including its agents and representatives) has made any representation, warranty, covenant or agreement to or with such Party relating to this Agreement or the transactions contemplated hereby, other than those expressly set forth in the Contribution Agreement and the documents and schedules hereto and thereto, and (ii) such Party has not relied upon any representation, warranty, covenant or agreement relating to this Agreement or the transactions contemplated hereby other than those expressly representation, warranty, covenant or agreement relating to this Agreement or the transactions contemplated hereby other than those referred to in clause (i) above. No amendment, supplement, modification or waiver of this Agreement shall be binding unless executed in writing by the Parties. No waiver of any of the provisions of this Agreement shall be deemed or shall constitute a waiver of any other provision hereof (regardless of whether similar), nor shall any such waiver constitute a continuing waiver unless otherwise expressly provided.

Section 19.8 Limitation of Liability. NOTWITHSTANDING ANYTHING IN THIS AGREEMENT TO THE CONTRARY, NEITHER PARTY SHALL BE LIABLE TO THE OTHER PARTY FOR SPECIAL, INDIRECT, CONSEQUENTIAL, PUNITIVE OR EXEMPLARY DAMAGES SUFFERED BY SUCH PARTY RESULTING FROM OR ARISING OUT OF THIS AGREEMENT OR THE BREACH THEREOF OR UNDER ANY OTHER THEORY OF LIABILITY, WHETHER TORT, NEGLIGENCE, STRICT LIABILITY, BREACH OF CONTRACT, WARRANTY, INDEMNITY OR OTHERWISE, INCLUDING LOSS OF USE, INCREASED COST OF OPERATIONS, LOSS OF PROFIT OR REVENUE, OR BUSINESS INTERRUPTIONS; PROVIDED, HOWEVER, THAT THE FOREGOING LIMITATION SHALL NOT APPLY TO ANY DAMAGE CLAIM ASSERTED BY OR AWARDED TO A THIRD PARTY FOR WHICH A PARTY WOULD

48

# OTHERWISE BE LIABLE UNDER ANY INDEMNIFICATION PROVISION SET FORTH HEREIN.

Section 19.9 <u>Headings</u>. The headings of the several Articles and Sections herein are inserted for convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement.

Section 19.10 <u>Rights and Remedies</u>. Except as otherwise provided in this Agreement, each Party reserves to itself all rights, counterclaims, other remedies and defenses that such Party is or may be entitled to arising from or out of this Agreement or as otherwise provided by Applicable Law.

Section 19.11 <u>No Partnership</u>. Nothing contained in this Agreement shall be construed to create an association, trust, partnership, or joint venture or impose a trust, fiduciary or partnership duty, obligation or liability on or with regard to either Party.

# Section 19.12 Rules of Construction.

(a) The Parties agree that they have been represented by counsel during the negotiation and execution of this Agreement and therefore waive the application of any law, regulation, holding or rule of construction providing that ambiguities in an agreement or other document will be construed against the Party drafting such agreement or document.

(b) The words "this Agreement," "herein," "hereby," "hereunder" and "hereof," and words of similar import, refer to this Agreement as a whole and not to any particular subdivision unless expressly so limited. The words "this Article," "this Section" and "this clause," and words of similar import, refer only to the Article, Section or clause hereof in which such words occur. The word "or" is exclusive, and the word "including" (in its various forms) means including without limitation. (c) Pronouns in masculine, feminine or neuter genders shall be construed to state and include any other gender, and words, terms and titles (including terms defined herein) in the singular form shall be construed to include the plural and vice versa, unless the context otherwise requires.

(d) References herein to any Person shall include such Person's successors and assigns; provided, however, that nothing contained in this clause (d) is intended to authorize any assignment or transfer not otherwise permitted by this Agreement.

(e) References herein to any law shall be deemed to refer to such law as amended, reenacted, supplemented or superseded in whole or in part and in effect from time to time and also to all rules and regulations promulgated thereunder

(f) References herein to any contract mean such contract as amended, supplemented or modified (including any waiver thereto) in accordance with the terms thereof, except that with respect to any contract listed on any schedule hereto, all such amendments, supplements or modifications must also be listed on such schedule.

49

(g) Each representation, warranty, covenant and agreement contained in this Agreement will have independent significance, and the fact that any conduct or state of facts may be within the scope of two or more provisions in this Agreement, whether relating to the same or different subject matters and regardless of the relative levels of specificity, shall not be considered in construing or interpreting this Agreement.

(h) Unless otherwise expressly provided herein to the contrary, accounting terms shall have the meaning given by U.S. generally accepted accounting principles.

**Section 19.13** <u>No Third Party Beneficiaries</u>. This Agreement is for the sole benefit of the Parties and their respective successors and permitted assigns and each Person entitled to indemnity under <u>Article 11</u>, <u>Article 16</u>, or <u>Article 17</u> (but only to the extent of their entitlement to be defended, indemnified, released and held harmless thereunder, and provided that only a Party shall be entitled to enforce such entitlement on their behalf) and shall not inure to the benefit of any other Person whomsoever or whatsoever, it being the intention of the Parties that no third Person shall be deemed a third party beneficiary of this Agreement.

Section 19.14 <u>Further Assurances</u>. Each Party shall take such acts and execute and deliver such documents as may be reasonably required to effectuate the purposes of this Agreement.

**Section 19.15** <u>Counterpart Execution</u>. This Agreement may be executed in one or more counterparts, including electronic, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. In the event that any signature is delivered by facsimile transmission or by e-mail delivery of a ".pdf" format data file, such signature shall create a valid and binding obligation of the Party executing (or on whose behalf such signature is executed) with the same force and effect as if such facsimile or ".pdf" signature page were an original thereof.

Section 19.16 <u>Memorandum of Agreement</u>. Contemporaneously with the execution of this Agreement, the Parties shall execute, acknowledge, deliver and record a "short form" memorandum of this Agreement in the form of <u>Exhibit J</u> attached hereto (as modified, including by the addition of any required property descriptions, required by local law and practice to put such Memorandum of record and put third parties on notice of this Agreement), which shall be placed of record in each state and county in which the currently-existing Service Area Properties are located. Further such memoranda shall be executed and delivered by Producer as Antero Water from time to time requests to evidence the commitment of additional areas or Oil and Gas Interests under this Agreement.

50

IN WITNESS WHEREOF, the Parties have executed this Agreement on the date first set forth above.

# ANTERO RESOURCES CORPORATION

By: Name: Title:

# ANTERO WATER LLC

By: Name: Title:

Water Services Agreement Signature Page Exhibit D

Form of Secondment Agreement

# SECONDMENT AGREEMENT

# by and between

# ANTERO MIDSTREAM PARTNERS LP

# ANTERO RESOURCES MIDSTREAM MANAGEMENT LLC

# ANTERO MIDSTREAM LLC

# ANTERO WATER LLC

# ANTERO TREATMENT LLC

#### and

# ANTERO RESOURCES CORPORATION

# SEPTEMBER [ ], 2015

# TABLE OF CONTENTS

		PAGE
ARTICLE I	DEFINITIONS, CONSTRUCTION	1
1.1	Definitions	1
1.2	Construction	4
ARTICLE II	SECONDMENT	4
2.1	Seconded Employees of Antero	4
2.2	MLP's Rights	4
2.3	Antero's Rights	4
2.4	Supervision and Management of the Seconded Employees	4
2.5	Consultations	4
2.6	Additional Seconded Employees	4
2.7	Title to Items Obtained on Behalf of MLP	4
ARTICLE III	EMPLOYEES	5
3.1	Personnel	5
ARTICLE IV	REIMBURSEMENT AND BILLING PROCEDURES	6
4.1	Reimbursement	6
4.2	Billing Procedures	7
4.3	Reports	-
4.4	Audit and Examination	
ARTICLE V	STANDARD OF CARE, NEGATIVE COVENANTS	7
5.1	Standard of Care	7
5.2	Negative Covenants	7

ARTICLE VI	TAXES		8
6.1 6.2	Embedded Tax Amounts Income Taxes		8 8
ARTICLE VII	TERMINATION		8
7.1 7.2	Term Termination		8 8
ARTICLE VIII	ACCESS TO THE ASSETS		8
ARTICLE IX	INDEMNIFICATION		8
9.1 9.2 9.3 9.4 9.5	Indemnification Scope Indemnified Persons Indemnifications Damages Limitations Defense of Claims		8 8 9 9 9
ARTICLE X	FORCE MAJEURE		10
10.1	Force Majeure Event		10
		i	

ARTICLE XI	OTHER PROVISIONS	10
11.1	Assignment	10
11.2	Notices	10
11.3	Severability	11
11.4	Entire Agreement; Conflicts	11
11.5	Amendment or Modification	11
11.6	No Waiver	11
11.7	Safety Regulations	11
11.8	Relationship of Parties	11
11.9	Governing Law	11
11.10	Further Assurances	12
11.11	Counterparts	12
11.12	Rights of Third Parties	12
	<u>SCHEDULES</u>	

SCHEDULE 1	Operating Services
SCHEDULE 2	Accounting Procedures

### ii

# SECONDMENT AGREEMENT

THIS SECONDMENT AGREEMENT is made effective as of September [], 2015, by and between Antero Midstream Partners LP, a Delaware limited partnership ("*MLP*"), Antero Resources Midstream Management LLC, a Delaware limited liability company (the "*General Partner*"), Antero Midstream LLC, a Delaware limited liability company ("*Antero Midstream*"), Antero Water LLC, a Delaware limited liability company ("*Antero Water*"), Antero Treatment LLC, a Delaware limited liability company ("*Antero Treatment*") and Antero Resources Corporation, a Delaware corporation ("*Antero*"). MLP, the General Partner, Antero Midstream, Antero Water, and Antero are sometimes referred to herein separately as "*Party*" or collectively as the "*Parties*."

## RECITALS

**WHEREAS**, MLP, directly or indirectly, owns, will own or may own (i) the Gathering Facilities (as defined below) consisting of gathering pipelines, compressor stations and certain other associated midstream assets, and (ii) the Water Assets (as defined below) consisting of water delivery pipelines, water treatment and other water facilities and related assets;

WHEREAS, MLP desires that Antero provide Seconded Employees (as defined below) to perform the Operational Services (as defined below) with respect to the Gathering Facilities and the Water Assets in accordance with the following commercial agreements (i) that certain Gathering and Compression Agreement, dated as of November 10, 2014, between Antero and Antero Midstream (as amended, supplemented or restated from time to time, the "*Gathering Agreement*") and the Right of First Offer Agreement, dated as of November 10, 2014, between Antero and Antero Midstream (as amended, supplemented or restated from time to time, the "*Gathering Agreement*") and the Right of First Offer Agreement, dated as of November 10, 2014, between Antero and Antero Midstream (as amended, supplemented or restated from time to time, the "*ROFO*")
Agreement"), and (ii) that certain Water Services Agreement, dated of even date herewith, between Antero and Antero Water (as amended, supplemented or restated from time to time, the "Water Services Agreement," and together with the Gathering Agreement and the ROFO Agreement, the "Commercial Agreements");

**WHEREAS**, the Parties desire to set forth their respective rights and responsibilities with respect to Antero's secondment of employees for purposes of the operation, maintenance and management of the Gathering Facilities and the Water Assets;

NOW THEREFORE, in consideration of their mutual undertakings and agreements hereunder, the Parties undertake and agree as follows:

#### AGREEMENT

NOW, THEREFORE, the Parties hereby agree as follows:

#### ARTICLE I DEFINITIONS, CONSTRUCTION

1.1 **Definitions.** In this Agreement, capitalized terms used, but not otherwise defined, shall have the respective meanings given to such terms set forth below:

A&R Services Agreement shall have the meaning set forth in Section 11.4.

Accounting Procedures means the terms and provisions set forth in Schedule 2.

*Affiliate* means (i) with respect to Antero, any other Person that directly or indirectly through one or more intermediaries is controlled by Antero, excluding the General Partner and any other Person that directly or indirectly through one or more intermediaries is controlled by the General Partner (including MLP); and (ii) with respect to MLP, the General Partner and any other Person that directly or indirectly through one or more intermediaries is controlled by the General Partner (including MLP); and (ii) with respect to MLP, the General Partner and any other Person that directly or indirectly through one or more intermediaries is controlled by the General Partner. As used herein, the term "control" means the possession, directly or indirectly, of

1

the power to direct or cause the direction of the management and policies of a Person, whether through ownership of voting securities, by contract or otherwise.

Affiliated Group shall have the meaning set forth on Schedule 2.

Agreement means this Secondment Agreement, as the same may be amended.

Antero shall have the meaning set forth in the first paragraph.

Antero Group shall have the meaning set forth in Section 4.1.

Antero Indemnitees shall have the meaning set forth in Section 9.2.

Antero Midstream shall have the meaning set forth in the first paragraph.

Antero Water shall have the meaning set forth in the first paragraph.

*Applicable Law* means all laws, permits, rules, codes, ordinances, requirements and regulations of all federal, state or local agencies, court and other governmental bodies, including without limitation the Natural Gas Act, the Pipeline Safety Act of 1968, both as amended, and the regulations and orders of the Federal Energy Regulatory Commission and the Department of Transportation; in each case, as applicable to MLP, Antero, or the Assets.

Assets means the Water Assets and the Gathering Facilities.

Audit Committee shall have the meaning set forth in the Partnership Agreement.

Business Day means any day other than a Saturday, a Sunday, or a holiday on which national banking associations in the State of Colorado are closed.

*Capital Expenditures* means all Expenditures that are capitalized by Antero or MLP, as applicable in accordance with GAAP and the relevant Party's accounting capitalization procedures, in each case as consistently applied and as in effect from time to time.

Commercial Agreements shall have the meaning set forth in the Recitals.

Expenditure means a cost, expense or expenditure.

*Fiscal Year* means each 12 month period beginning on the first day of January of a year and ending on December 31 of the same year; *provided*, the first Fiscal Year hereunder shall begin on the date of this Agreement and shall end on December 31, 2015; *and further* 

provided, the last Fiscal Year shall end at the expiration or termination of this Agreement.

*Force Majeure* shall have the meaning set forth in Section 10.1(b).

GAAP means United States generally accepted accounting principles as in effect from time to time.

Gathering Agreement shall have the meaning set forth in the Recitals.

*Gathering Facilities* shall mean (a) the Gathering System (as defined in the Gathering Agreement), (b) any property, equipment or other assets associated with the provision of "Services" (as defined in the ROFO Agreement) under the ROFO Agreement, (c) any other assets, equipment, accessions and improvements in respect of the foregoing owned, directly or indirectly, by the Partnership Group and (d) any other assets, equipment or facilities owned by the Partnership Group other than the Water Assets.

General Partner shall have the meaning set forth in the first paragraph.

2

*Governmental Authority* means any governmental authority, agency, department, commission, bureau, board, instrumentality, court or quasi-governmental authority of any foreign nation, the United States, or any state that has or obtains jurisdiction over the matter in question, or any political subdivision thereof.

*Initial Services Agreement* means the Services Agreement dated November 10, 2014 by and among Antero, MLP and the General Partner.

*Liability* shall have the meaning set forth in Section 9.3(a).

Limited Partner shall have the meaning set forth in the Partnership Agreement.

MLP shall have the meaning set forth in the first paragraph.

MLP Indemnitees shall have the meaning set forth in Section 9.2.

*Month* means calendar month.

Operating Services have the meaning set forth in Section 2.1.

**Partnership Agreement** means the Agreement of Limited Partnership of Antero Midstream Partners LP, dated as of November 10, 2014, to which reference is hereby made for all purposes of this Agreement. No amendment or modification to the Partnership Agreement subsequent to the date hereof that relates to or otherwise affects any portion of this Agreement shall be given effect for the purposes of this Agreement unless consented to in writing by each of the Parties to this Agreement.

*Partnership Group* shall have the meaning set forth in Section 3.1(a).

*Party* or *Parties* means any of the entities named in the first paragraph to this Agreement and any respective successors or permitted assigns in accordance with the provisions of this Agreement.

*Period of Secondment* shall have the meaning set forth in Section 3.1(b).

*Permit* means all permits, licenses, franchises, consents, authorizations, certifications, exemptions, variances, and approvals, as necessary under Applicable Laws for operating the Assets.

*Person* means any natural person, corporation, limited liability company, partnership, joint venture, trust, unincorporated organization, association, joint stock company or Governmental Authority.

*Prior Contribution Agreement* means that certain Amended and Restated Contribution Agreement, dated as of November 10, 2014, between Antero and MLP, as amended, supplemented or restated from time to time.

ROFO Agreement shall have the meaning set forth in the Recitals.

Seconded Employee(s) shall have the meaning set forth in Section 3.1(b).

Service Provider shall have the meaning set forth in Section 4.2.

*Water Assets* shall have the meaning provided such term in the Water Contribution Agreement, as well as any future assets of Antero Water, Antero Treatment or any other Affiliate of MLP to the extent relating to the water business of those respective entities.

*Water Contribution Agreement* means that certain Contribution, Conveyance, and Assumption Agreement, dated as of September, 17, 2015, by and among Antero, MLP, and Antero Treatment, as amended, supplemented or restated from time to time.

Water Services Agreement shall have the meaning set forth in the Recitals.

1.2 **Construction**. In construing this Agreement, the following principles shall be followed: (a) no consideration shall be given to the captions of articles, sections or subsections; (b) no consideration shall be given to the fact or presumption that one Party had a greater or lesser hand in drafting this Agreement; (c) the word "includes" and its syntactic variants means "includes, but is not limited to" and corresponding syntactic variant expressions; (d) the plural shall be deemed to include the singular, and vice versa; (e) the words "this Agreement," "herein," "hereby," "hereunder" and "hereof," and words of similar import, refer to this Agreement as a whole and not to any particular subdivision unless expressly so limited; (f) the words "this Article," "this Section" and "this clause," and words of similar import, refer only to the Article, Section or clause hereof in which such words occur; and (g) the word "or" is exclusive, and the word "including" (in its various forms) means including without limitation.

# ARTICLE II SECONDMENT

2.1 **Seconded Employees of Antero.** Subject to the terms of this Agreement, Antero agrees to provide Seconded Employees (as defined in Section 3.1(b)) who, in their capacity as Seconded Employees of MLP, will perform the services described on Schedule 1, except as outsourced by MLP to third party service providers (the "**Operating Services**"), in order for the Partnership Group to operate the Assets in an efficient manner, and in a manner that permits the Partnership Group to comply with its obligations under the Commercial Agreements. The Seconded Employees will perform the Operating Services in accordance with the terms and conditions and subject to the limitations set forth in this Agreement.

2.2 *MLP's Rights*. The Seconded Employees shall be subject to the direction and control of MLP. Antero shall respond in a commercially reasonable manner to all instructions, notices, requests or inquiries from MLP with respect to the Seconded Employees. Decisions, acts or omissions so undertaken by the Seconded Employees or Antero with respect to the Seconded Employees pursuant to the direction and control of MLP shall not give rise to any breach of or default under this Agreement by Antero or liability to Antero provided that Antero otherwise acted in accordance with the requirements of Section 2.1.

2.3 *Antero's Rights.* MLP and its Affiliates shall have no authority to terminate a Seconded Employee's employment with Antero or otherwise to discipline a Seconded Employee, except that MLP may terminate the Seconded Employee's secondment. Antero shall, at all times, have sole authority to terminate a Seconded Employee's employment with Antero.

2.4 *Supervision and Management of the Seconded Employees*. To the extent that supervisors or managers of the Seconded Employees issue instructions to such Seconded Employees regarding the Operating Services, such supervisors and managers shall be treated for purposes of this Agreement as acting on behalf of MLP for purposes of this Agreement.

2.5 **Consultations.** Antero and MLP shall consult as frequently as reasonably necessary regarding the scope of Operating Services to be performed by the Seconded Employees and particular circumstances that may require an adjustment to Antero's obligation to provide the Seconded Employees, and shall keep each other timely informed about planned downtime, major maintenance projects, capital projects, significant operational events and other major events that are relevant to the safe and efficient operation of the Assets and the performance of the Parties' respective obligations under this Agreement.

2.6 *Additional Seconded Employees.* If, subsequent to the date hereof, additional services are required to operate the Assets that are not listed on Schedule 1, Antero shall use commercially reasonable efforts to provide Seconded Employees to operate such additional facilities and provide such additional secondment services on mutually agreeable pricing and other terms to be determined on a basis similar to the pricing and other terms set forth in this Agreement, whereupon such services shall be considered part of the Operating Services.

2.7 *Title to Items Obtained on Behalf of MLP*. To the extent that any materials, equipment, supplies, consumables, spare parts and other items are purchased or obtained by Antero or its Affiliates for or on behalf of the

4

Assets, MLP or any member of the Partnership Group (including Antero Water, Antero Treatment and Antero Midstream), title to such items shall pass immediately to and vest in MLP or such applicable Affiliate free and clear of all liens or encumbrances arising by, through and under Antero and its Affiliates but not otherwise (other than liens and security interests securing any unpaid portion of the purchase price for the same) upon passage of title from the vendor or supplier thereof. All materials, data and documents, to the extent prepared or developed by any Seconded Employee during the term of this Agreement for MLP or any of its Affiliates in connection with the Seconded Employees' performance of the Operating Services, including all manuals, data, designs, drawings, plans, specifications and reports, shall belong to MLP (or such Affiliate). All such materials, documents, and data, in whatever form, including electronic copies and databases, shall be provided promptly to MLP following any termination of this Agreement, or at such other times as MLP may reasonably direct; provided, however, that Antero shall be entitled to retain (a) copies of such materials, documents and data for document retention and compliance purposes if required by law, rules, regulations or orders of the court and (b) all electronic copies (if any) of any such materials, documents and data residing in its (and its Affiliates') automatic backup systems.

## ARTICLE III EMPLOYEES

# 3.1 Personnel.

(a) Pursuant to Section 2.1, Antero shall second, or cause to be seconded, to MLP and its subsidiaries (the "*Partnership Group*") the Seconded Employees (as defined in Section 3.1(b)) and such other Persons (including consultants and professionals, service or other organizations) as Antero deems necessary or appropriate in order to perform the Operating Services in an efficient and prudent manner. Subject to Antero's right to be reimbursed for such expenses in accordance with the Accounting Procedures, Antero shall pay all expenses incurred by it in connection with the retention of the Seconded Employees and such other Persons, including, but not limited to, compensation, salaries, wages and overhead and administrative expenses, charges to or incurred by Antero, and, if applicable, social security taxes, workers compensation insurance, retirement and insurance benefits and other such expenses. Any such Seconded Employees and other Persons retained by Antero may be union or non-union employees, and Antero shall have the sole right to negotiate the terms and provisions of any labor or other agreements with the unions to which such employees belong. Antero shall second, or cause to be seconded, all workers who will perform Operating Services.

(b) During the term of this Agreement, Antero shall, from time to time, designate certain of its employees to be seconded to MLP to perform duties for the Assets or otherwise work on behalf of the Partnership Group in accordance with and subject to the terms of this Agreement. Each such employee who Antero seconds to MLP shall, during the time that such employee is seconded to MLP under this Agreement (the "*Period of Secondment*"), be referred to individually herein as a "*Seconded Employee*" and, collectively, as the "*Seconded Employees*."

(c) Antero will give notice to each Seconded Employee at times and in the normal processes by which it gives similar notices to employees or otherwise in its discretion to the extent required in accordance with any applicable state law. The notices will include that (i) each such Seconded Employee will be a joint employee of Antero and MLP, and (ii) for any work place injury, the Seconded Employee's sole remedy against either Antero or MLP (or their respective Affiliates) will be under the workers' compensation insurance policy or qualified self-insured program of Antero. For the avoidance of doubt, the Parties acknowledge that the Seconded Employees will, during the Period of Secondment, be called upon to perform services for both MLP and Antero (and their respective Affiliates) of the same or closely-related nature. Antero retains the right to terminate the secondment of any Seconded Employee for any reason at any time or to hire or discharge the Seconded Employee for any reason at any time, upon prior written notice to Antero, but at no time will MLP have the right to terminate any Seconded Employee's employee's employment by Antero. Upon the termination of the secondment of any Seconded Employee will cease performing services for the Partnership Group.

(d) In the course and scope of performing any Seconded Employee's job functions for the Partnership Group, the Seconded Employee will report into MLP's management structure, and will be under the direct

5

management, supervision, direction and control of MLP with respect to such Seconded Employee's performance of the Operating Services and day-to-day activities.

(e) Those Seconded Employees who serve as supervisors or managers and who are called upon to oversee the work of Seconded Employees working at the Partnership Group's assets or facilities or to provide management support on behalf of MLP are designated by MLP as supervisors to act on the behalf of MLP in supervising the Seconded Employees pursuant to Section 3.1(d) above. Any Seconded Employee so designated will be acting on the behalf of MLP when supervising the work of the Seconded Employees or when they are otherwise providing management or executive support on behalf of MLP.

(f) With respect to the Partnership Group's operations in Ohio, Antero shall obtain workers' compensation coverage as defined by Ohio Revised Code Chapter 4123 on behalf of both Antero and MLP, and MLP shall be considered an employer solely for the purposes of Ohio Revised Code Chapter 4123. With respect to the Partnership Group's operations in West Virginia, Antero shall obtain workers' compensation coverage as defined by West Virginia Code Chapter 23 on behalf of both Antero and MLP, and MLP shall be considered an employer solely for the purposes of West Virginia Code Chapter 23. For the avoidance of doubt, nothing in this Agreement has any effect on the right of a Seconded Employee to prosecute a workers' compensation claim against MLP, Antero, or both.(1)

(g) MLP shall not be a participating employer in any benefit plan of Antero or any of its Affiliates. Antero shall remain solely responsible for all obligations and liabilities arising with respect to any benefit plans relating to any Seconded Employees and the Partnership Group shall not assume any benefit plan or have any obligations or liabilities arising thereunder, in each case except for costs properly chargeable to MLP hereunder.

# ARTICLE IV REIMBURSEMENT AND BILLING PROCEDURES

4.1 **Reimbursement.** Subject to and in accordance with the terms and provisions of this Article IV (but without duplication of any amounts due pursuant to the A&R Services Agreement) and such reasonable allocation and other procedures as may be agreed upon by Antero and the General Partner from time to time, MLP hereby agrees to reimburse Antero for all direct and indirect costs and expenses incurred by Antero and its Affiliates (collectively, the "Antero Group") in connection with the provision of the Operating Services to the Partnership Group, including the following:

(a) Antero shall be reimbursed by MLP for Expenditures and Capital Expenditures incurred by Antero in the performance of the Operating Services, in accordance with the Accounting Procedures; *provided*, MLP shall not be required to reimburse Antero for (i) contributions, withholding deductions or taxes measured by the wages, salaries or compensation paid to Persons employed by Antero or any of its Affiliates in connection herewith (including any Expenditures arising out of claims for non-payment of any or all of the

foregoing) or (ii) Expenditures for which Antero is required to provide indemnification to MLP or any MLP Indemnitee pursuant to Section 9.3(b); and

(b) Any payments or expenses incurred for insurance coverage, including allocable portions of premiums, and negotiated instruments (including surety bonds and performance bonds) provided by underwriters with respect to the Assets, the Partnership Group's other assets or the business of the Partnership Group; salaries and related benefits and expenses of personnel employed by the Antero Group who render Operating Services to the Partnership Group, plus general and administrative expenses to the extent associated with such personnel;

it being agreed, however, that to the extent any reimbursable costs or expenses incurred by the Antero Group consist of an allocated portion of costs and expenses incurred by the Antero Group for the benefit of both the Partnership Group and the other members of the Antero Group, such allocation shall be made on a reasonable cost reimbursement basis as determined by Antero in good faith.

(1) Code references to be confirmed.

4.2 **Billing Procedures.** MLP will reimburse Antero, or the member of the Antero Group providing the Operating Services, as applicable (each a "*Service Provider*") for billed costs no later than the later of (i) the last day of the Month following the performance Month or (ii) thirty (30) Business Days following the date of Antero's or Service Provider's billing to MLP. Billings and payments may be accomplished by inter-company accounting procedures and transfers.

6

4.3 *Reports.* Antero shall cause to be timely prepared and delivered to MLP such reports, forecasts, implementation plans, plans of action, studies and other information pertaining to the performance of the Operating Services as MLP may reasonably request from time to time. The costs incurred by Antero in preparing and delivering such reports, forecasts, plans, studies and other information shall be included in the Expenditures to be reimbursed by MLP pursuant to Section 4.1(a).

## 4.4 Audit and Examination.

(a) MLP shall have the right to review all source documentation concerning the liabilities, costs, and expenses allocated to MLP and/or the Partnership Group hereunder upon reasonable notice and during regular business hours. If any such examination establishes any inaccuracy in any billing made prior to such examination, the necessary adjustments to such billings will be made promptly without any interest charge. If any information provided to or reviewed by MLP under this Section 4.4(a) is confidential, the parties shall execute a mutually acceptable confidentiality agreement prior to such inspection or audit.

(b) Absent fraud or intentional concealment or misrepresentation by the relevant Party, a Party shall neither be required nor permitted to adjust any Expenditure incurred by such Party during a Fiscal Year unless a claim therefor is presented or adjustment is initiated within the 12 Months following such Fiscal Year, and in the absence of such timely claims or adjustments, the books and records rendered by Antero shall be conclusively established as correct. If MLP has commenced an audit within the period referenced in Section 4.4(a) but has been unable to complete the audit within such period despite its good faith efforts to do so, then MLP shall be entitled to a reasonable extension of time to complete the audit.

# ARTICLE V STANDARD OF CARE, NEGATIVE COVENANTS

5.1 *Standard of Care.* Antero shall second, or cause to be seconded, the Seconded Employees who will perform the Operating Services, and who shall carry out their responsibilities (a) in accordance with workmanlike practices common in the U.S. oil and natural gas industry, and exercise the same level of care Antero requires in the management of its own business and affairs, and (b) in compliance with all environmental laws, rules and regulations of the United States of America and the states where the Water Assets and the Gathering Facilities are located.

5.2 *Negative Covenants.* For the avoidance of doubt, no member of the Antero Group shall, without the prior written consent of MLP, do or, to the extent the same is within its reasonable control and consistent with the other terms of this Agreement, permit to occur or to continue, or permit any Seconded Employee to do or permit to occur or continue to occur, any of the following:

(a) Commit any member of the Partnership Group to, or enter into on behalf of the Partnership Group, any contract or agreement;

(b) Create or incur any lien, security interest or encumbrance upon the Assets, including without limitation any mechanics or materialmen's liens or similar encumbrances arising out of claims for work, labor or materials furnished in connection with the provision of Operating Services hereunder;

(c) Purport to sell, lease, pledge, mortgage, assign, transfer or otherwise dispose of the Assets or any of the Partnership Group now owned or hereafter acquired assets; or

(d) Commit any member of the Partnership Group to be or to become directly or contingently responsible or liable for obligations of any other Person, by assumption, guarantee, endorsement or otherwise.

#### ARTICLE VI TAXES

6.1 *Embedded Tax Amounts.* If any portion of any payment made by MLP hereunder is to reimburse Antero for any federal, state or local taxes or assessments, then Antero shall cause such taxes and assessments to be paid prior to delinquency.

6.2 *Income Taxes.* Notwithstanding anything to the contrary, Expenditures for which Antero is entitled to reimbursement pursuant to this Agreement shall not include taxes that are measured or based on Antero's income or franchise taxes or similar taxes, and all such income, franchise and similar taxes shall be the responsibility of Antero.

# ARTICLE VII TERMINATION

7.1 *Term.* Unless terminated earlier, this Agreement shall continue in effect until the twentieth (20<sup>th</sup>) anniversary of the execution of the Initial Services Agreement and from year to year thereafter (with the initial term of this Agreement deemed extended for each of any such additional year) until such time as this Agreement is terminated, effective upon an anniversary of the date hereof, by written notice from either Party to the other Party on or before the one hundred eightieth (180<sup>th</sup>) day prior to such anniversary.

# 7.2 Termination.

(a) Notwithstanding anything to the contrary in this Agreement, this Agreement may be terminated at any time (i) by mutual written agreement of the Parties and (ii) by MLP, in its sole discretion, effective upon delivery of written notice of such termination to Antero.

(b) Upon termination of this Agreement, all rights and obligations of the Parties under this Agreement shall terminate, *provided*, *however*, that such termination shall not affect or excuse the performance of any party under the provisions of Article IX which provisions shall survive the termination of this Agreement indefinitely.

# ARTICLE VIII ACCESS TO THE ASSETS

The Seconded Employees shall at all times during their performance of the Operating Services hereunder have full and free, non-exclusive access to the Assets as necessary to perform their obligations under this Agreement, and all such Persons shall comply with all safety and other procedures from time to time imposed by MLP in connection with any access to or work performed on or about the Assets.

# ARTICLE IX INDEMNIFICATION

9.1 **Indemnification Scope.** IT IS IN THE BEST INTERESTS OF THE PARTIES THAT CERTAIN RISKS RELATING TO THE MATTERS GOVERNED BY THIS AGREEMENT SHOULD BE IDENTIFIED AND ALLOCATED AS BETWEEN THEM. IT IS THEREFORE THE INTENT AND PURPOSE OF THIS AGREEMENT TO PROVIDE FOR THE INDEMNITIES SET FORTH HEREIN TO THE MAXIMUM EXTENT ALLOWED BY LAW. ALL PROVISIONS OF THIS ARTICLE SHALL BE DEEMED CONSPICUOUS WHETHER OR NOT CAPITALIZED OR OTHERWISE EMPHASIZED.

9.2 **Indemnified Persons.** Wherever "MLP" or "Antero" appears as an indemnitee in this Article, the term shall include that entity and its Affiliates, and the respective agents, officers, directors, employees, representatives and contractors and subcontractors of any tier of the foregoing entities involved in actions or duties to act on behalf of the indemnified Party. These groups will be the "**MLP Indemnitees**" or the "**Antero Indemnitees**" as applicable, *provided*, *however*, that for the avoidance of doubt, the MLP Indemnitees shall not include Antero and its Affiliates,

8

and the Antero Indemnitees shall not include the General Partner or any member of the Partnership Group. "Third parties" shall not include any MLP Indemnitees or Antero Indemnitees.

# 9.3 Indemnifications.

(a) MLP SHALL RELEASE, DEFEND, INDEMNIFY, AND HOLD HARMLESS THE ANTERO INDEMNITEES FROM AND AGAINST ANY AND ALL CLAIMS, CAUSES OF ACTION, DEMANDS, LIABILITIES, LOSSES, DAMAGES, FINES, PENALTIES, JUDGMENTS, EXPENSES AND COSTS, INCLUDING REASONABLE ATTORNEYS' FEES AND COSTS OF INVESTIGATION AND DEFENSE (EACH, A "*LIABILITY*") (INCLUDING, WITHOUT LIMITATION, ANY LIABILITY FOR (i) DAMAGE, LOSS OR DESTRUCTION OF THE ASSETS, (ii) BODILY INJURY, ILLNESS OR DEATH OF ANY PERSON, EXCEPT TO THE EXTENT SUCH PERSON IS A SECONDED EMPLOYEE, AND (iii) LOSS OF OR DAMAGE TO EQUIPMENT OR PROPERTY OF ANY PERSON) ARISING FROM OR RELATING TO THE SECONDED EMPLOYEES' PERFORMANCE OF THIS AGREEMENT, EXCEPT TO THE EXTENT SUCH LIABILITY IS CAUSED BY THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF THE ANTERO INDEMNITEES. (b) ANTERO SHALL RELEASE, DEFEND, INDEMNIFY, AND HOLD HARMLESS THE MLP INDEMNITEES FROM AND AGAINST ANY AND ALL LIABILITIES (INCLUDING, WITHOUT LIMITATION, ANY LIABILITY FOR (i) DAMAGE, LOSS OR DESTRUCTION OF THE ASSETS, (ii) BODILY INJURY, ILLNESS OR DEATH OF ANY PERSON, EXCEPT TO THE EXTENT SUCH PERSON IS A SECONDED EMPLOYEE, AND (iii) LOSS OF OR DAMAGE TO EQUIPMENT OR PROPERTY OF ANY PERSON) ARISING FROM OR RELATING TO THE SECONDED EMPLOYEES' PERFORMANCE UNDER THIS AGREEMENT TO THE EXTENT SUCH LIABILITY IS CAUSED BY THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF THE ANTERO INDEMNITEES.

9.4 **Damages Limitations.** Any and all damages recovered by either Party pursuant to this Article IX or pursuant to any other provision of or actions or omissions under this Agreement shall be limited to actual damages. CONSEQUENTIAL DAMAGES (INCLUDING WITHOUT LIMITATION BUSINESS INTERRUPTIONS AND LOST PROFITS) AND EXEMPLARY AND PUNITIVE DAMAGES SHALL NOT BE RECOVERABLE UNDER ANY CIRCUMSTANCES EXCEPT TO THE EXTENT THOSE DAMAGES ARE INCLUDED IN THIRD PARTY CLAIMS FOR WHICH A PARTY HAS AGREED HEREIN TO INDEMNIFY THE OTHER PARTY. EACH PARTY ACKNOWLEDGES IT IS AWARE THAT IT HAS POTENTIALLY VARIABLE LEGAL RIGHTS UNDER COMMON LAW AND BY STATUTE TO RECOVER CONSEQUENTIAL, EXEMPLARY, AND PUNITIVE DAMAGES UNDER CERTAIN CIRCUMSTANCES, AND, EXCEPT AS PROVIDED IN THE PRECEDING SENTENCE WITH RESPECT TO THIRD PARTY CLAIMS, EACH PARTY NEVERTHELESS WAIVES, RELEASES, RELINQUISHES, AND SURRENDERS RIGHTS TO CONSEQUENTIAL PUNITIVE AND EXEMPLARY DAMAGES TO THE FULLEST EXTENT PERMITTED BY LAW WITH FULL KNOWLEDGE AND AWARENESS OF THE CONSEQUENCES OF THE WAIVER REGARDLESS OF THE NEGLIGENCE OR FAULT OF EITHER PARTY.

9.5 **Defense of Claims.** The indemnifying Party shall defend, at its sole expense, any claim, demand, loss, liability, damage, or other cause of action within the scope of the indemnifying Party's indemnification obligations under this Agreement, *provided* that the indemnified Party notifies the indemnifying Party promptly in writing of any claim, loss, liability, damage, or cause of action against the indemnified Party and gives the indemnifying Party information, and assistance at the reasonable expense of the indemnifying Party in defense of the matter. The indemnified Party may be represented by its own counsel (at the indemnified Party's sole expense) and may participate in any proceeding relating to a claim, loss, liability, damage, or cause of action in which the indemnified Party or both Parties are defendants, *provided, however*, the indemnifying Party shall, at all times, control the defense and any appeal or settlement of any matter for which it has indemnification obligations under this Agreement so long as any such settlement includes an unconditional release of the indemnified Party from all liability arising out of such claim, demand, loss, liability, damage, or other cause of action and does not include a statement as to or an admission of fault, culpability or a failure to act by or on behalf of the indemnified Party. Should the Parties both be named as defendants in any third-party claim or cause of action arising out of or relating to the Assets or Operating Services, the Parties will cooperate with each other in the joint defense of

9

their common interests to the extent permitted by law, and will enter into an agreement for joint defense of the action if the Parties mutually agree that the execution of the same would be beneficial.

#### ARTICLE X FORCE MAJEURE

# 10.1 Force Majeure Event.

(a) In the event a Party is rendered unable, wholly or in part, by Force Majeure to carry out its obligations under this Agreement, other than the obligation to make payments then or thereafter due hereunder, and such Party promptly gives notice and reasonably full particulars of such Force Majeure to the other Party promptly after the occurrence of the cause relied on, then the obligations of the Party giving such notice, so far as and to the extent that they are affected by Force Majeure, shall be suspended during the continuance of any inability so caused, but for no longer period, and such cause shall so far as reasonably possible be remedied with all reasonably dispatch by the Party claiming Force Majeure.

(b) *"Force Majeure"* as used in this Agreement shall mean any cause or causes not reasonably within the control of the Party claiming suspension and which, by the exercise of reasonable diligence, such Party is unable to prevent or overcome, including acts of God, strikes, lockouts or other industrial disturbances, acts of the public enemy, acts of terror, sabotage, wars, blockades, military action, insurrections, riots, epidemics, landslides, subsidence, lightning, earthquakes, fires, storms or storm warnings, crevasses, floods, washouts, civil disturbances, explosions, breakage or accidents to wells, machinery, equipment or lines of pipe; the necessity for testing or making repairs or alterations to wells, machinery, equipment or lines of pipe, freezing of wells, equipment or lines of pipe; inability of any Party hereto to obtain, after the exercise of reasonable diligence, necessary materials, supplies or governmental approvals, and action or restraint by any Governmental Authority (so long as the Party claiming suspension has not applied for or assisted in the application for, and has opposed where and to the extent reasonable, such action or restraint, and as long as such action or restraint is not the result of a failure by the claiming Party to comply with any Applicable Law.

(c) The settlement of any strikes or lockouts will be entirely within the discretion of Antero, and settlement of strikes, lockouts or other labor disturbances when that course is considered inadvisable is not required.

# ARTICLE XI OTHER PROVISIONS

11.1 *Assignment.* This Agreement shall be binding upon and inure to the benefit of the Parties named herein. No Party may assign or otherwise transfer either this Agreement or any of its rights, interests or obligations hereunder without the prior written approval of the other Parties, which approval shall not be unreasonably withheld, conditioned or delayed.

11.2 **Notices.** All notices or requests or consents provided for by, or permitted to be given pursuant to, this Agreement must be in writing and must be given by depositing the same in the United States mail, addressed to the Person to be notified, postpaid, and registered or certified with return receipt requested or by delivering such notice in person or by facsimile or e-mail to such Party. Notice given by personal delivery or mail shall be effective upon actual receipt. Notice given by facsimile or e-mail shall be effective upon actual receipt if received during the recipient's normal business hours or at the beginning of the recipient's next Business Day after receipt if not received during the recipient's normal business hours. All notices to be sent to a Party pursuant to this Agreement shall be sent to or made at the address set forth below such Party's signature to this Agreement or at such other address as such Party may stipulate to the other Parties in the manner provided in this Section 11.2.

Partnership Group:

Antero Midstream Partners LP 1615 Wynkoop Street

10

Denver, Colorado 80202 Attn: Chief Financial Officer Fax: (303) 357-7315

Antero:

Antero Resources Corporation 1615 Wynkoop Street Denver, Colorado 80202 Attn: Chief Financial Officer Fax: (303) 357-7315

11.3 *Severability.* If any provision of this Agreement shall be finally determined to be unenforceable, illegal or unlawful, such provision shall, so long as the economic and legal substance of the transactions contemplated hereby is not affected in any materially adverse manner as to any Party, be deemed severed from this Agreement and the remainder of this Agreement shall remain in full force and effect.

11.4 **Entire Agreement; Conflicts.** This Agreement, the Amended and Restated Services Agreement dated of even date herewith among MLP, General Partner and Antero (the "**A&R Services Agreement**"), the Water Contribution Agreement, the Prior Contribution Agreement, the Commercial Agreements, any exhibits or schedules to the foregoing and any other transaction documents executed in connection herewith or therewith constitute the entire agreement of the Parties relating to the matters contained herein and therein, superseding all prior contracts or agreements, whether oral or written, relating to the matters contained herein and therein. In the event of a conflict between the terms of this Agreement and the terms of the A&R Services Agreement with respect to the coverage of any individual and/or services provided, the terms of this Agreement shall control.

11.5 *Amendment or Modification.* This Agreement may be amended or modified from time to time only by the written agreement of all the Parties hereto. Each such instrument shall be reduced to writing and shall be designated on its face an "Amendment" or an "Addendum" to this Agreement.

11.6 *No Waiver.* Failure of either MLP or Antero to require performance of any provision of this Agreement shall not affect either Party's right to full performance thereof at any time thereafter, and the waiver by either MLP or Antero of a breach of any provision hereof shall not constitute a waiver of any similar breach in the future or of any other breach or nullify the effectiveness of such provision.

11.7 *Safety Regulations.* All employees of each Party when on the property of the other Party will conform to the rules, regulations and procedures concerning safety of such other Party. From time to time, each Party shall furnish the other Party with complete, accurate and current copies of all such rules, regulations and procedures.

11.8 *Relationship of Parties.* This Agreement does not create a partnership, joint venture, or relationship of trust or agency between the Parties.

11.9 **Governing Law.** Each of the Parties hereby irrevocably consents and agrees that any dispute arising out of or relating to this Agreement or any related document shall exclusively be brought in the courts of the State of Colorado, in Denver County or the federal courts located in the District of Colorado. The Parties agree that, after such a dispute is before a court as specified in this Section 11.9 and during the pendency of such dispute before such court, all actions with respect to such dispute, including any counterclaim, cross-claim or interpleader, shall be subject to the exclusive jurisdiction of such court. The Parties also agree that after such a dispute is before a court as specified in this Section 11.9, and during the pendency of such dispute before such court, each of the Parties hereby waives, and agrees not to assert, as a defense in any legal dispute, that it is not subject thereto or that such dispute may not be brought or is not maintainable in such court or that its property is exempt or immune from execution, that the dispute is brought in an inconvenient forum or that the venue of the dispute is improper. Each Party agrees that a final judgment in any dispute described in this Section 11.9 after the expiration of any period permitted for appeal and subject to any stay during appeal shall be conclusive and may be

11

11.10 Further Assurances. In connection with this Agreement and all transactions contemplated by this Agreement, each signatory party hereto agrees to execute and deliver such additional documents and instruments and to perform such additional acts as may be necessary or appropriate to effectuate, carry out and perform all of the terms, provisions and conditions of this Agreement and all such transactions.

11.11 Counterparts. This Agreement may be executed in one or more counterparts, including electronic, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. In the event that any signature is delivered by facsimile transmission or by e-mail delivery of a ".pdf" format data file, such signature shall create a valid and binding obligation of the Party executing (or on whose behalf such signature is executed) with the same force and effect as if such facsimile or ".pdf" signature page were an original thereof.

Rights of Third Parties. The provisions of this Agreement are enforceable solely by the Parties to this Agreement, and 11.12 no third party (including any Limited Partner of MLP) shall have the right, separate and apart from the Parties to this Agreement, to enforce any provision of this Agreement or to compel any Party to this Agreement to comply with the terms of this Agreement.

# [Signatures on following page]

12

IN WITNESS WHEREOF, the Parties have executed this Agreement as of the date first set forth above.

# ANTERO MIDSTREAM PARTNERS LP

Antero Resources Midstream Management LLC, its general By: partner

By:

Name: Title:

#### ANTERO RESOURCES MIDSTREAM MANAGEMENT LLC

By:

Name: Title:

# ANTERO MIDSTREAM LLC

By:	
Name: Title:	
ANTER	O WATER LLC
By:	
Name:	
Title:	
ANTED	O TDE ATMENT LLC

#### ANTERO TREATMENT LLC

By:

Name:

# Title: ANTERO RESOURCES CORPORATION

By:

Name: Title:

Signature Page to Services and Secondment Agreement

# SCHEDULE 1

The services shall include the personnel necessary for the provision of comprehensive Operating Services. Antero shall second, or cause to be seconded, Seconded Employees to perform the following Operating Services in connection with the operation and maintenance of the Assets, all in accordance with, and subject to, the requirements of this Agreement:

# **OPERATING SERVICES: WATER ASSETS**

The Seconded Employees shall perform the following Operational Services with respect to the Water Assets:

- 1. provide, or procure (in the name of MLP or its applicable Affiliate) and manage, those services (including operation, maintenance, engineering and construction services) necessary for the operation of the Water Assets and to maintain the Water Assets in sound operating condition and good repair;
- 2. perform routine maintenance, preventative maintenance and capitalized repairs;
- 3. perform corrosion and inspection services;
- 4. perform or cause to be performed waste water fluid handling services;
- 5. submit the applications for and in the name of MLP or its applicable Affiliate, pursue the issuance of, and maintain in force, the environmental and all other permits necessary for the operation of the Water Assets;
- 6. prepare, sign and file, or cause to be prepared, signed and filed, in each case in the name of MLP or its applicable Affiliate, all filings required to be filed by an operator of comparable water transportation and disposal assets with any Governmental Authority with respect to the Water Assets or the operation thereof; and
- 7. perform or cause to be performed any other services with respect to the Water Assets and associated business reasonably requested by MLP.

# **OPERATING SERVICES: GATHERING FACILITIES**

The Seconded Employees shall perform the following Operational Services with respect to the Gathering Facilities:

- conduct, or cause to be conducted, all operations with respect to the Gathering Facilities, and shall procure and furnish, or cause to be procured or furnished in the name of MLP or its applicable Affiliate, all materials, equipment, services, supplies, and labor necessary for the operation and maintenance of the Gathering Facilities, engineering support for these activities, and related warehousing and security, including the following:
  - i. Maintain and operate flow and pressure control, monitoring, and over-pressure protection;
  - ii. Maintain, repair, recondition, overhaul, and replace equipment, as needed, to keep the Gathering Facilities in good working order;
  - iii. Operate the Gathering Facilities in a manner consistent with the standard of conduct set forth in the applicable Commercial Agreements; and
  - iv. Conduct all other routine day-to-day operations of the Gathering Facilities.
- 2. provide, manage and conduct, or cause to be provided, managed and conducted, the business operations associated with the Gathering Facilities, including without limitation, the following:
  - i. Transportation and logistics, including commercial operations;

- ii. Contract administration;
- iii. Gas control;
- iv. Gas measurement;
- v. GIS mapping;
- vi. Database mapping, reporting and maintenance;
- vii. Rights of way;
- viii. Materials management;
- ix. Engineering support (including facility design and optimization); and
- x. perform or cause to be performed any other services with respect to the Gathering Facilities and associated business reasonably requested by MLP.
- 3. coordinate and direct, or cause to be coordinated and directed, the activities of Persons (including contractors, subcontractors, consultants, professionals, service and other organizations) required to perform the duties and responsibilities necessarily for the provision of the Operating Services. Such persons may include employees of Antero or its affiliates or employees of one or more third persons; provided, however, that any contracts or agreements with respect to third party services shall be entered into in the name of MLP or its applicable Affiliates unless otherwise agreed by MLP in writing.

# <u>SCHEDULE 2</u> ACCOUNTING PROCEDURES

1.0 **Statements and Billings.** Antero shall bill MLP in accordance with Section 4.2 of this Agreement. If requested by MLP, Antero will promptly provide reasonably sufficient support for the Expenditures anticipated to be incurred for the following Month. Bills will be summarized by appropriate classifications indicative of the nature thereof and will be accompanied by such detail and supporting documentation as MLP may reasonably request.

2.0 *Records.* The Parties shall maintain accurate books and records covering all performance of the Services.

3.0 **Purchase of Materials.** All material, equipment and supplies used or consumed on behalf of the Assets will be owned by MLP or the relevant member of the Partnership Group, as applicable, and purchased or furnished for its account. So far as is reasonably practical and consistent with efficient, safe and economical operation as determined by Antero, only such material shall be obtained for the Assets as may be required for immediate or near-term use, and the accumulation of surplus stock shall be avoided. To the extent reasonably possible, the Seconded Employees shall take advantage of discounts available by early payments and pass such benefits (or an allocable portion thereof) on to MLP.

# 4.0 Accounting Procedures.

(a) Antero is part of an affiliated group of companies (the "*Affiliated Group*") that as of the date of this Agreement is engaged in the exploration and production of natural gas. Accounting, purchasing, and risk management (among other functions and services) as of the date of this Agreement are managed or provided by Antero or one of its Affiliates to the Affiliated Group. The costs and expenses incurred by Antero or such Affiliate in managing or providing such functions and services are accrued on the books and records of Antero in accordance with GAAP and are allocated (where applicable) among the members of the Affiliated Group in accordance with GAAP (the "*Accounting Procedures*").

(b) The costs and expenses incurred by Antero Water, Antero Treatment or Antero Midstream in managing or providing functions and services as at the date of this Agreement are to be accrued on the books and records of such entity in accordance with the Accounting Procedures.

(c) Antero shall apply the Accounting Procedures in determining the Expenditures. The Accounting Procedures shall provide a mechanism for validating an Expenditure and all allocations of an Expenditure. If MLP believes that the determination or allocation of any Expenditure is inconsistent with the Accounting Procedures, then MLP shall notify Antero in writing of the specific manner in which MLP regards such determination or allocation to be deficient or objectionable. Antero shall either correct or change such determination or allocation in accordance with the notice, or, if Antero disagrees with MLP's notice, shall reasonably cooperate with MLP in addressing such changes. If MLP's auditors and the Audit Committee of the General Partner make reasonable suggestions on accounting relating to this Agreement, Antero will reasonably cooperate with MLP and any other applicable members of the Partnership Group in addressing such changes; *provided*, the implementation of such suggestions shall be subject to the mutual agreement of MLP and Antero.

(d) There shall be no duplication of charges for the same Expenditure. Likewise, no duplication of an Expenditure that has been charged to MLP under any other agreement between the Parties may occur.

## <u>Exhibit E</u>

#### Form of Amended and Restated Services Agreement

# AMENDED AND RESTATED SERVICES AGREEMENT

This AMENDED AND RESTATED SERVICES AGREEMENT (this "Agreement") dated as of [ ], 2015, is entered into by and among Antero Midstream Partners LP, a Delaware limited partnership (the "Partnership"), Antero Resources Midstream Management LLC, a Delaware limited liability company (the "General Partner"), and Antero Resources Corporation, a Delaware corporation ("Antero"). The Partnership, the General Partner and Antero may be referred to herein individually as "Party" or collectively as "Parties."

# RECITALS

**WHEREAS**, the Parties entered into a Services Agreement dated November 10, 2014 (the "**Initial Services Agreement**"), and the Parties intend to amend and restate such Initial Services Agreement in its entirety as set forth herein;

WHEREAS, the Partnership desires that Antero perform the Services (as defined below); and

WHEREAS, the Parties desire to set forth their respective rights and responsibilities with respect to the provision of the Services.

**NOW THEREFORE**, in consideration of their mutual undertakings and agreements hereunder, the Parties agree that the abovedescribed Initial Services Agreement shall hereby be amended and restated in its entirety as follows:

# ARTICLE 1 PERFORMANCE OF SERVICES

1.1 <u>Agreement to Provide Services</u>. Antero hereby agrees to provide, or cause to be provided to, the Partnership and its subsidiaries (collectively, the "**Partnership Group**") with certain centralized corporate, general and administrative services, such as accounting, audit, billing, business development, corporate record keeping, treasury services, cash management and banking, real property/land, legal, engineering, planning, budgeting, geology/geophysics, investor relations, risk management, information technology, insurance administration and claims processing, regulatory compliance and government relations, tax, payroll, human resources and environmental, health and safety, including without limitation permit filing, support for permit filing and maintenance (collectively, the "**Services**"). Antero shall provide, or cause to be provided to, the Partnership Group with such Services in a manner consistent in nature and quality to the services of such type that Antero performs in the management of its own business and affairs.

#### ARTICLE 2 RELATIONSHIP OF ANTERO AND THE PARTNERSHIP

2.1 <u>Independent Contractor</u>. Antero is an independent contractor and shall perform the Services hereunder as an independent contractor under the direction and control of the General Partner. Nothing hereunder shall be construed as creating any other relationship between Antero, on the one hand, and the General Partner and the Partnership, on the other hand, including but not limited to a partnership, agency or fiduciary relationship, joint venture, limited liability company,

association, or any other enterprise. Except as provided in that certain Secondment Agreement dated of even date herewith among the Partnership, the General Partner, Antero Midstream LLC, Antero Water LLC, Antero Treatment LLC and Antero (the "Secondment Agreement"), neither Party nor its employees shall be deemed to be an employee of the other Party and the Partnership's interest is only in the performance of the Services by Antero in accordance with this Agreement.

2.2 <u>The Partnership's Right to Observe</u>. The Partnership shall at all times have the right to observe and consult with Antero in connection with Antero's performance of its obligations under this Agreement. The Partnership shall comply with all reasonable requirements of Antero prior to such observation or witnessing, including but not limited to safety requirements.

# 2.3 <u>Standard of Conduct of Antero</u>.

(a) <u>General Standard</u>. Antero shall (1) perform the Services and carry out its responsibilities hereunder, and shall require all contractors, subcontractors and materialmen furnishing labor, material or services for the performance of the Services to carry out their responsibilities in accordance with workmanlike practices common in the Partnership's industry, and (2) exercise the same level of care Antero exercises in the management of its own business and affairs.

(b) <u>Compliance with Procedures and Laws</u>. Antero shall perform the Services under this Agreement in compliance with all laws, permits, rules, codes, ordinances, requirements and regulations of all federal, state or local agencies, court and

other governmental bodies, including without limitation the Natural Gas Act, the Pipeline Safety Act of 1968, both as amended, and the regulations and orders of the Federal Energy Regulatory Commission and the U.S. Department of Transportation, which are applicable to (1) Antero's business and (2) the performance of the Services or any other obligation of Antero hereunder.

# ARTICLE 3 REIMBURSEMENT AND BILLING PROCEDURES

3.1 <u>Reimbursement by the Partnership</u>. Subject to and in accordance with the terms and provisions of this Article 3 and such reasonable allocation and other procedures as may be agreed upon by Antero and the General Partner from time to time, the Partnership hereby agrees to reimburse Antero for all direct and indirect costs and expenses incurred by Antero and its Affiliates (collectively, the "Antero Group") in connection with the provision of the Services to the Partnership Group, including the following:

(a) any payments or expenses incurred for insurance coverage, including allocable portions of premiums, and negotiated instruments (including surety bonds and performance bonds) provided by underwriters with respect to the assets or the business of the Partnership Group;

(b) salaries and related benefits and expenses of personnel employed by the Antero Group who render Services to the Partnership Group, plus general and administrative expenses to the extent associated with such personnel;

2

(c) any taxes or other direct operating expenses paid by the Antero Group for the benefit of the Partnership Group (including any state income, franchise or similar tax paid by the Antero Group resulting from the inclusion of the Partnership Group in a combined or consolidated state income, franchise or similar tax report with Antero as required by applicable law as opposed to the flow through of income attributable to the Antero Group's ownership interest in the Partnership Group); *provided, however*, that the amount of any such reimbursement shall be limited to the tax that the Partnership Group would have paid had it not been included in a combined or consolidated group with Antero; and

(d) all expenses and expenditures incurred by the Antero Group as a result of the Partnership being a publicly traded entity, including costs associated with annual and quarterly reports, tax return and Schedule K-1 preparation and distribution, independent auditor fees, partnership governance and compliance, registrar and transfer agent fees, legal fees and independent director compensation;

it being agreed, however, that to the extent any reimbursable costs or expenses incurred by the Antero Group consist of an allocated portion of costs and expenses incurred by the Antero Group for the benefit of both the Partnership Group and the other members of the Antero Group, such allocation shall be made on a reasonable cost reimbursement basis as determined by Antero in good faith. For purposes of this Agreement, "Affiliate" means (A) with respect to Antero, any other Person that directly or indirectly through one or more intermediaries is controlled by Antero, excluding the General Partner and any other Person that directly or indirectly through one or more intermediaries is controlled by the General Partner (including the Partnership); and (B) with respect to the Partnership, the General Partner and any other Person that directly or indirectly through one or more intermediaries is controlled by the General Partner (including the Partnership); and (B) with respect to the Partnership, the General Partner and any other Person that directly or indirectly through one or more intermediaries is controlled by the General Partner. As used herein, the term "control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through ownership of voting securities, by contract or otherwise, and the term "**Person**" means any natural person, corporation, limited liability company, partnership, joint venture, trust, unincorporated organization, association, joint stock company or governmental authority.

3.2 <u>Billing Procedures</u>. The Partnership will reimburse Antero, or the members of the Antero Group providing the Services, as applicable (each a "Service Provider"), for billed costs no later than the later of (a) the last day of the month following the performance month or (b) thirty (30) business days following the date of the Service Provider's billing to the Partnership. Billings and payments may be accomplished by inter-company accounting procedures and transfers. The Partnership shall have the right to review all source documentation concerning the liabilities, costs, and expenses allocated to the Partnership and/or Partnership Group hereunder upon reasonable notice and during regular business hours.

# ARTICLE 4 TERM AND TERMINATION

4.1 <u>Term</u>. Unless terminated earlier, this Agreement shall continue in effect until the twentieth (20<sup>th</sup>) anniversary of the execution of the Initial Services Agreement and from year to year thereafter (with the initial term of this Agreement deemed extended for each of any such additional year) until such time as this Agreement is terminated, effective upon an anniversary of

3

the execution of the Initial Services Agreement, by written notice from either Party to the other Party on or before the one hundred eightieth (180<sup>th</sup>) day prior to such anniversary.

4.2 <u>Termination</u>.

(a) <u>Methods of Termination</u>. Notwithstanding anything to the contrary in this Agreement, this Agreement may be terminated at any time (1) by mutual written agreement of the Parties and (2) by the Partnership, in its sole discretion, effective upon

delivery of written notice of such termination to Antero.

(b) <u>Effect of Termination</u>. Upon termination of this Agreement, all rights and obligations of the Parties under this Agreement shall terminate, *provided*, *however*, that such termination shall not affect or excuse the performance of any party under the provisions of Article 5 which provisions shall survive the termination of this Agreement indefinitely.

# ARTICLE 5 INDEMNITY

5.1 Indemnification Scope. IT IS IN THE BEST INTERESTS OF THE PARTIES THAT CERTAIN RISKS RELATING TO THE MATTERS GOVERNED BY THIS AGREEMENT SHOULD BE IDENTIFIED AND ALLOCATED AS BETWEEN THEM. IT IS THEREFORE THE INTENT AND PURPOSE OF THIS AGREEMENT TO PROVIDE FOR THE INDEMNITIES SET FORTH HEREIN TO THE MAXIMUM EXTENT ALLOWED BY LAW. ALL PROVISIONS OF THIS ARTICLE SHALL BE DEEMED CONSPICUOUS WHETHER OR NOT CAPITALIZED OR OTHERWISE EMPHASIZED.

5.2 Indemnified Persons. Wherever "the Partnership" or "Antero" appears as an indemnitee in this Article, the term shall include that entity and its Affiliates, and the respective agents, officers, directors, employees, representatives and contractors and subcontractors of any tier of the foregoing entities involved in actions or duties to act on behalf of the indemnified Party. These groups will be the "**Partnership Indemnitees**" or the "**Antero Indemnitees**" as applicable, provided, however, that for the avoidance of doubt, the Partnership Indemnitees shall not include Antero and its Affiliates, and the Antero Indemnitees shall not include any member of the Partnership Group or the General Partner. "Third parties" shall not include any Partnership Indemnitees.

# 5.3 <u>Indemnifications</u>.

(a) EXCEPT AS OTHERWISE PROVIDED IN THE SECONDMENT AGREEMENT, THE PARTNERSHIP SHALL RELEASE, DEFEND, INDEMNIFY, AND HOLD HARMLESS THE ANTERO INDEMNITEES FROM AND AGAINST ANY AND ALL CLAIMS, CAUSES OF ACTION, DEMANDS, LIABILITIES, LOSSES, DAMAGES, FINES, PENALTIES, JUDGMENTS, EXPENSES AND COSTS, INCLUDING REASONABLE ATTORNEYS' FEES AND COSTS OF INVESTIGATION AND DEFENSE (EACH, A "**LIABILITY**") (INCLUDING, WITHOUT LIMITATION, ANY LIABILITY FOR (1) DAMAGE, LOSS OR DESTRUCTION OF THE ASSETS OR THE BUSINESS OF THE PARTNERSHIP GROUP, (2) BODILY INJURY, ILLNESS OR DEATH OF ANY PERSON, AND (3) LOSS OF OR DAMAGE TO EQUIPMENT OR PROPERTY OF ANY PERSON)

4

ARISING FROM OR RELATING TO THE GENERAL PARTNER'S, PARTNERSHIP'S OR ANTERO'S PERFORMANCE OF THIS AGREEMENT, EXCEPT TO THE EXTENT SUCH LIABILITY IS CAUSED BY THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF THE ANTERO INDEMNITEES.

(b) EXCEPT AS OTHERWISE PROVIDED IN THE SECONDMENT AGREEMENT, ANTERO SHALL RELEASE, DEFEND, INDEMNIFY, AND HOLD HARMLESS THE PARTNERSHIP INDEMNITEES FROM AND AGAINST ANY AND ALL LIABILITIES (INCLUDING, WITHOUT LIMITATION, ANY LIABILITY FOR (1) DAMAGE, LOSS OR DESTRUCTION OF THE ASSETS OR THE BUSINESS OF THE PARTNERSHIP GROUP, (2) BODILY INJURY, ILLNESS OR DEATH OF ANY PERSON AND (3) LOSS OF OR DAMAGE TO EQUIPMENT OR PROPERTY OF ANY PERSON) ARISING FROM OR RELATING TO ANTERO'S PERFORMANCE UNDER THIS AGREEMENT TO THE EXTENT SUCH LIABILITY IS CAUSED BY THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF THE ANTERO INDEMNITEES.

5.4 Damages Limitations. Any and all damages recovered by either Party pursuant to this Article 5 or pursuant to any other provision of or actions or omissions under this Agreement shall be limited to actual damages. CONSEQUENTIAL DAMAGES (INCLUDING WITHOUT LIMITATION BUSINESS INTERRUPTIONS AND LOST PROFITS) AND EXEMPLARY AND PUNITIVE DAMAGES SHALL NOT BE RECOVERABLE UNDER ANY CIRCUMSTANCES EXCEPT TO THE EXTENT THOSE DAMAGES ARE INCLUDED IN THIRD PARTY CLAIMS FOR WHICH A PARTY HAS AGREED HEREIN TO INDEMNIFY THE OTHER PARTY. EACH PARTY ACKNOWLEDGES IT IS AWARE THAT IT HAS POTENTIALLY VARIABLE LEGAL RIGHTS UNDER COMMON LAW AND BY STATUTE TO RECOVER CONSEQUENTIAL, EXEMPLARY, AND PUNITIVE DAMAGES UNDER CERTAIN CIRCUMSTANCES, AND, EXCEPT AS PROVIDED IN THE PRECEDING SENTENCE WITH RESPECT TO THIRD PARTY CLAIMS, EACH PARTY NEVERTHELESS WAIVES, RELEASES, RELINQUISHES, AND SURRENDERS RIGHTS TO CONSEQUENTIAL PUNITIVE AND EXEMPLARY DAMAGES TO THE FULLEST EXTENT PERMITTED BY LAW WITH FULL KNOWLEDGE AND AWARENESS OF THE CONSEQUENCES OF THE WAIVER REGARDLESS OF THE NEGLIGENCE OR FAULT OF EITHER PARTY.

5.5 Defense of Claims. The indemnifying Party shall defend, at its sole expense, any claim, demand, loss, liability, damage, or other cause of action within the scope of the indemnifying Party's indemnification obligations under this Agreement, *provided* that the indemnified Party notifies the indemnifying Party promptly in writing of any claim, loss, liability, damage, or cause of action against the indemnified Party and gives the indemnifying Party information and assistance at the reasonable expense of the indemnifying Party in defense of the matter. The indemnified Party may be represented by its own counsel (at the indemnified Party's sole expense) and may participate in any proceeding relating to a claim, loss, liability, damage, or cause of action in which the indemnified Party or both Parties are defendants, *provided*, *however*, the indemnifying Party shall, at all times, control the defense and any appeal or settlement of any matter for which it has indemnification obligations under this Agreement so long as any such settlement includes an unconditional release of the indemnified Party from all liability arising out of such claim, demand, loss, liability, damage, or other cause of action and does not require any

remediation or other action other than the payment of money which the indemnifying party will be responsible for hereunder and does not include a statement as to or an admission of fault, culpability or a failure to act by or on behalf of the indemnified Party. Should the Parties both be named as defendants in any third-party claim or cause of action arising out of or relating to the Services, the Parties will cooperate with each other in the joint defense of their common interests to the extent permitted by law, and will enter into an agreement for joint defense of the action if the Parties mutually agree that the execution of the same would be beneficial.

# ARTICLE 6 <u>NOTICES</u>

Either Party may give notices to the other Party by first class mail postage prepaid, by overnight delivery service, or by facsimile with receipt confirmed at the following addresses or other addresses furnished by a Party by written notice. Any telephone numbers below are solely for information and are not for Agreement notices.

#### If to the General Partner or the Partnership to:

Antero Midstream Partners LP 1615 Wynkoop Street Denver, Colorado 80202 Attn: Chief Financial Officer Fax: (303) 357-7315

#### If to Antero to:

Antero Resources Corporation 1615 Wynkoop Street Denver, Colorado 80202 Attn: Chief Financial Officer Fax: (303) 357-7315

# ARTICLE 7 GENERAL

7.1 <u>Succession and Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the Parties named herein. No Party may assign or otherwise transfer either this Agreement or any of its rights, interests or obligations hereunder without the prior written approval of the other Parties, which approval shall not be unreasonably withheld, conditioned or delayed.

7.2 <u>Governing Law</u>. This Agreement will be governed by and construed and enforced in accordance with the laws of the State of Colorado, excluding any choice of Law rules which may direct the application of the laws of another jurisdiction.

7.3 <u>Consent to Jurisdiction, Etc.</u>; Waiver of Jury Trial. Each of the Parties hereby irrevocably consents and agrees that any dispute arising out of or relating to this Agreement or any related document shall exclusively be brought in the courts of the State of Colorado, in Denver County or the federal courts located in the District of Colorado. The Parties agree that, after such

a dispute is before a court as specified in this Section 7.3 and during the pendency of such dispute before such court, all actions with respect to such dispute, including any counterclaim, cross-claim or interpleader, shall be subject to the exclusive jurisdiction of such court. The Parties also agree that after such a dispute is before a court as specified in this Section 7.3, and during the pendency of such dispute before such court, each of the Parties hereby waives, and agrees not to assert, as a defense in any legal dispute, that it is not subject thereto or that such dispute may not be brought or is not maintainable in such court or that its property is exempt or immune from execution, that the dispute is brought in an inconvenient forum or that the venue of the dispute is improper. Each Party agrees that a final judgment in any dispute described in this Section 7.3 after the expiration of any period permitted for appeal and subject to any stay during appeal shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by laws. THE PARTIES HEREBY WAIVE IRREVOCABLY ANY AND ALL RIGHTS TO DEMAND A TRIAL BY JURY IN CONNECTION WITH THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREBY OR ANY DOCUMENT CONTEMPLATED HEREIN OR OTHERWISE RELATED HERETO.

7.4 <u>Non-waiver of Future Default</u>. No waiver of any Party of any one or more defaults by the other in performance of any of the provisions of this Agreement shall operate or be construed as a waiver of any other existing or future default or defaults, whether of a like or different character.

7.5 <u>Audit and Maintenance of Records; Reporting</u>. Notwithstanding the payment by the Partnership of any charges, the Partnership shall have the right to review and contest the charges in accordance with this Section 7.5. For a period of two years from the end of any calendar year, the Partnership shall have the right, upon reasonable notice and at reasonable times, to inspect and audit all the records, books, reports, data and processes related to the Services performed by Antero to ensure Antero's compliance with the terms of this Agreement. If any information provided to or reviewed by the Partnership or its representatives pursuant to this Section 7.5 is

confidential, the parties shall execute a mutually acceptable confidentiality agreement prior to such inspection or audit.

7.6 Entire Agreement; Amendments and Schedules. This Agreement shall be amended or waived only by an instrument in writing executed by both Parties. This Agreement, the Secondment Agreement, the Water Contribution Agreement (as defined in the Secondment Agreement), the Prior Contribution Agreement (as defined in the Secondment Agreement), the Prior Contribution Agreement (as defined in the Secondment Agreement), the Prior Contribution Agreement (as defined in the Secondment Agreement), the Commercial Agreements (as defined in the Secondment Agreement), any exhibits or schedules to the foregoing and any other transaction documents executed in connection herewith or therewith constitute the entire agreement of the Parties relating to the matters contained herein and therein, superseding all prior contracts or agreements, whether oral or written, relating to the matters contained herein and therein. In the event of a conflict between the terms of this Agreement and the terms of the Secondment Agreement with respect to the coverage of any individual and/or services provided, the Secondment Agreement shall control.

#### 7.7 Force Majeure.

(a) If either Party is rendered unable, wholly or in part, by force majeure to carry out its obligations under this Agreement, other than to make payments due, the obligations

7

of that Party, so far as they are affected by force majeure, will be suspended during the continuance of any inability so caused, but for no longer period. The Party whose performance is affected by force majeure will provide notice to the other Party, which notice may initially be oral, followed by a written notification, and will use commercially reasonable efforts to resolve the event of force majeure to the extent reasonably possible.

(b) **"Force majeure"** as used in this Agreement shall mean any cause or causes not reasonably within the control of the Party claiming suspension and which, by the exercise of reasonable diligence, such Party is unable to prevent or overcome, including acts of God, strikes, lockouts or other industrial disturbances, acts of the public enemy, acts of terror, sabotage, wars, blockades, military action, insurrections, riots, epidemics, landslides, subsidence, lightning, earthquakes, fires, storms or storm warnings, crevasses, floods, washouts, civil disturbances, explosions, breakage or accidents to wells, machinery, equipment or lines of pipe; freezing of wells, equipment or lines of pipe; inability of any Party hereto to obtain, after the exercise of reasonable diligence, necessary materials, supplies or governmental approvals, and action or restraint by any Governmental Authority (so long as the Party claiming suspension has not applied for or assisted in the application for, and has opposed where and to the extent reasonable, such action or restraint, and as long as such action or restraint is not the result of a failure by the claiming Party to comply with any Applicable Law. The settlement of strikes or lockouts will be entirely within the discretion of the Party having the difficulty, and settlement of strikes, lockouts, or other labor disturbances when that course is considered inadvisable is not required.

7.8 <u>Counterpart Execution</u>. This Agreement may be executed in one or more counterparts, including electronic, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. In the event that any signature is delivered by facsimile transmission or by e-mail delivery of a ".pdf" format data file, such signature shall create a valid and binding obligation of the Party executing (or on whose behalf such signature is executed) with the same force and effect as if such facsimile or ".pdf" signature page were an original thereof.

7.9 <u>Third Parties</u>. The provisions of this Agreement are enforceable solely by the Parties to this Agreement, and no third party (including any Limited Partner of the Partnership) shall have the right, separate and apart from the Parties to this Agreement, to enforce any provision of this Agreement or to compel any Party to this Agreement to comply with the terms of this Agreement.

7.10 <u>Severability</u>. If any provision of this Agreement shall be finally determined to be unenforceable, illegal or unlawful, such provision shall, so long as the economic and legal substance of the transactions contemplated hereby is not affected in any materially adverse manner as to any Party, be deemed severed from this Agreement and the remainder of this Agreement shall remain in full force and effect.

7.11 <u>Further Assurances</u>. In connection with this Agreement and all transactions contemplated by this Agreement, each signatory party hereto agrees to execute and deliver such additional documents and instruments and to perform such additional acts as may be necessary or

8

appropriate to effectuate, carry out and perform all of the terms, provisions and conditions of this Agreement and all such transactions.

9

The Parties have caused this Agreement to be signed by their duly authorized representatives effective as of the date first written above.

#### ANTERO RESOURCES CORPORATION

By:

Name:Alvyn A. SchoppTitle:Chief Administrative Officer, Regional<br/>Vice President and Treasurer

# ANTERO MIDSTREAM PARTNERS LP

By: Antero Resources Midstream Management LLC, its general partner

By: Name:

Name:Alvyn A. SchoppTitle:Chief Administrative Officer, Regional<br/>Vice President and Treasurer

# ANTERO RESOURCES MIDSTREAM MANAGEMENT LLC

By:

Name: Alvyn A. Schopp Title: Chief Administrative Officer, Regional Vice President and Treasurer

Signature Page – Management Services Agreement

1

16

16

17

17

# COMMON UNIT PURCHASE AGREEMENT

# by and among

# ANTERO MIDSTREAM PARTNERS LP

and

#### THE PURCHASERS NAMED ON SCHEDULE A HERETO

# **TABLE OF CONTENTS**

#### ARTICLE I

# DEFINITIONS

Section 1.1 Definitions

# **ARTICLE II**

# AGREEMENT TO SELL AND PURCHASE

Sale and Purchase	6
Closing	6
Mutual Conditions	6
Each Purchaser's Conditions	7
The Partnership's Conditions	7
Partnership Deliveries	8
Purchaser Deliveries	9
Independent Nature of Purchasers' Obligations and Rights	9
	Closing Mutual Conditions Each Purchaser's Conditions The Partnership's Conditions Partnership Deliveries Purchaser Deliveries

# ARTICLE III

# REPRESENTATIONS AND WARRANTIES OF THE PARTNERSHIP

Section 3.1	Formation and Qualification of the Partnership Entities	10
Section 3.2	Purchased Units; Capitalization	10
Section 3.3	No Conflict	12
Section 3.4	No Default	12
Section 3.5	Authority	12
Section 3.6	No Consents	13
Section 3.7	Authorization, Execution and Delivery of the Common Unit Purchase Agreement	13
Section 3.8	Authorization, Execution, Delivery and Enforceability of Certain Agreements	13
Section 3.9	Contribution Agreement	13
Section 3.10	Authorization of Contribution Equity Consideration	14
Section 3.11	Valid Issuance; No Options or Preemptive Rights of Common Units	14
Section 3.12	No Registration Rights	14
Section 3.13	Periodic Reports	15
Section 3.14	Financial Statements	15
Section 3.15	Independent Registered Public Accounting Firm	15
Section 3.16	Litigation	15
Section 3.17	No Material Adverse Changes	15
Section 3.18	Title to Properties	16
Section 3.19	Rights of Way	16
Section 3.20	License and Permits	16
	i	

Section 3.22 Insurance

Section 3.23 No Labor Dispute; No Notice of Labor Law Violations

Section 3.24 Environmental Compliance

Section 3.25	Tax Returns	17
Section 3.26	No Employment Law Violations	17
Section 3.27	No Unlawful Payments	18
Section 3.28	Compliance with Money Laundering Laws	18
Section 3.29	OFAC	18
Section 3.30	Certain Fees	18
Section 3.31	No Side Agreements	19
Section 3.32	No Registration	19
Section 3.33	No Integration	19
Section 3.34	MLP Status	19
Section 3.35	Qualifying Income of Contributed Assets	19
Section 3.36	Investment Company	19
Section 3.37	Disclosure Controls	19
Section 3.38	Accounting Controls	20
Section 3.39	Placement Agent Reliance	20
Section 3.40	Legal Sufficiency of the Contribution Agreement	20
Section 3.41	Absence of Price Manipulation	20

# ARTICLE IV

# REPRESENTATIONS AND WARRANTIES OF THE PURCHASERS

Section 4.1	Existence	21
Section 4.2	Authorization, Enforceability	21
Section 4.3	No Breach	21
Section 4.4	Certain Fees	21
Section 4.5	No Side Agreements	22
Section 4.6	Investment	22
Section 4.7	Nature of Purchaser	22
Section 4.8	Restricted Securities	22
Section 4.9	Legend	22
Section 4.10	Company Information	23
Section 4.11	Short Selling	23

# ARTICLE V

# COVENANTS

Section 5.2	Taking of Necessary Action Other Actions Contribution Transaction Expenses	23 24 24 24
-------------	---	----------------------

# ii

# ARTICLE VI

# INDEMNIFICATION

Section 6.1	Indemnification by the Partnership	24
Section 6.2	Indemnification by Purchasers	25
Section 6.3	Indemnification Procedure	25

# ARTICLE VII

# MISCELLANEOUS

Section 7.1	Interpretation	26
Section 7.2	Survival of Provisions	27
Section 7.3	No Waiver; Modifications in Writing	27
Section 7.4	Binding Effect; Assignment	27
Section 7.5	Confidentiality	28
Section 7.6	Communications	28
Section 7.7	Removal of Legend	28
Section 7.8	Entire Agreement	29
Section 7.9	Governing Law	29
Section 7.10	Execution in Counterparts	29
Section 7.11	Termination	30
Section 7.12	Recapitalization, Exchanges, Etc. Affecting the Common Units	30

Schedule A — List of Purchasers and Commitment Amounts

Exhibit A — Form of Registration Rights Agreement Exhibit B — Form of Opinion of Vinson & Elkins L.L.P.

iii

#### COMMON UNIT PURCHASE AGREEMENT

This COMMON UNIT PURCHASE AGREEMENT, dated as of September 17, 2015 (this "<u>Agreement</u>"), is by and among ANTERO MIDSTREAM PARTNERS LP, a Delaware limited partnership (the "<u>Partnership</u>"), and each of the purchasers listed on <u>Schedule A</u> hereof (each a "<u>Purchaser</u>" and collectively, the "<u>Purchasers</u>").

WHEREAS, the Partnership, Antero Treatment LLC, a Delaware limited liability company and wholly-owned subsidiary of the Partnership ("<u>Antero Treatment</u>") and Antero Resources Corporation, a Delaware corporation ("<u>Antero</u>"), intend to enter into to the Contribution Agreement (as defined below), pursuant to which Antero will, on the terms and subject to the conditions set forth in the Contribution Agreement, contribute (i) the Treatment Assets (as defined below) to Antero Treatment and (ii) 100% of the outstanding limited liability company interests in Antero Water (as defined below) to the Partnership, in exchange for aggregate consideration of \$1.050 billion, consisting of cash, assumed debt and Common Units (the "<u>Unit Consideration</u>"); and

WHEREAS, to fund a portion of the purchase price for the Contribution (as defined below), the Partnership desires to sell to the Purchasers, and the Purchasers desire to purchase from the Partnership, certain Common Units (as defined below), in accordance with the provisions of this Agreement; and

WHEREAS, the Partnership and the Purchasers will enter into a registration rights agreement (the "<u>Registration Rights</u> <u>Agreement</u>"), substantially in the form attached hereto as <u>Exhibit A</u>, pursuant to which the Partnership will provide the Purchasers with certain registration rights with respect to the Common Units acquired pursuant hereto.

NOW THEREFORE, in consideration of the mutual covenants and agreements set forth herein and for good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Partnership and each of the Purchasers, severally and not jointly, hereby agree as follows:

# ARTICLE I

#### DEFINITIONS

Section 1.1 <u>Definitions</u>. As used in this Agreement, and unless the context requires a different meaning, the following terms have the meanings indicated:

"<u>Affiliate</u>" means, with respect to any Person, any other Person that directly or indirectly through one or more intermediaries controls, is controlled by or is under common control with, the Person in question. As used herein, the term "control" means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through ownership of voting securities, by contract or otherwise.

"<u>Aggregate Purchase Price</u>" means the product of (i) the Common Unit Price multiplied by (ii) the aggregate number of Purchased Units purchased by the Purchasers.

1

"Agreement" has the meaning set forth in the introductory paragraph.

"Antero" has the meaning set forth in the recitals.

"Antero Treatment" has the meaning set forth in the recitals.

"Antero Treatment LLC Agreement" has the meaning specified in Section 3.2(h).

"Antero Water" means Antero Water LLC, a Delaware limited liability company.

"Antero Water LLC Agreement" has the meaning specified in Section 3.2(i).

"Business Day" means a day other than (i) a Saturday or Sunday or (ii) any day on which banks located in New York, New York, U.S.A. are authorized or obligated to close.

"<u>Closing</u>" has the meaning specified in <u>Section 2.2</u>.

"Closing Date" has the meaning specified in Section 2.2.

"Code" has the meaning specified in Section 3.26.

"Commission" means the United States Securities and Exchange Commission.

"Common Unit Price" has the meaning specified in Section 2.1(b).

"Common Units" means common units representing limited partnership interests in the Partnership.

"Consent" has the meaning specified in Section 3.6.

"<u>Contribution</u>" means, collectively, the contribution, conveyance, transfer and assignment pursuant to the Contribution Agreement by Antero of (i) 100% of the equity interests in Antero Water to the Partnership and (ii) the Treatment Assets to Antero Treatment.

"<u>Contribution Agreement</u>" means the Contribution, Conveyance and Assumption Agreement, dated as of September 17, 2015, by and among the Partnership, Antero Treatment and Antero.

"Delaware LLC Act" means the Delaware Limited Liability Company Act.

"Delaware LP Act" means the Delaware Revised Uniform Limited Partnership Act.

"Enforceability Exceptions" has the meaning specified in Section 3.7.

"Environmental Laws" has the meaning specified in Section 3.24.

"ERISA" has the meaning specified in Section 3.26.

2

"Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time, and the rules and regulations of the Commission promulgated thereunder.

"Existing Registration Rights Agreement" means the Registration Rights Agreement, dated November 10, 2014, between the Partnership and Antero.

"Fundamental Representations" has the meaning specified in Section 7.2.

"General Partner" means Antero Resources Midstream Management LLC, a Delaware limited liability company.

"General Partner Interest" has the meaning specified in Section 3.2(d).

"Governmental Authority" means, with respect to a particular Person, any country, state, county, city and political subdivision in which such Person or such Person's property is located or that exercises valid jurisdiction over any such Person or such Person's property, and any court, agency, department, commission, board, bureau or instrumentality of any of them and any monetary authority that exercises valid jurisdiction over any such Person or such Person's property. Unless otherwise specified, all references to Governmental Authority herein with respect to the Partnership mean a Governmental Authority having jurisdiction over the Partnership, its Subsidiaries or any of their respective properties or assets.

"Incentive Distribution Rights" means all of the incentive distribution rights representing limited partner interests in the Partnership.

"Investment Company Act" has the meaning specified in Section 3.36.

"Law" means any federal, state, local or foreign order, writ, injunction, judgment, settlement, award, decree, statute, law, rule or regulation.

"Lien" means any interest in property securing an obligation owed to, or a claim by, a Person other than the owner of the property, whether such interest is based on the common law, statute or contract, and whether such obligation or claim is fixed or contingent, and including the lien or security interest arising from a mortgage, encumbrance, pledge, security agreement, conditional sale or trust receipt or a lease, consignment or bailment for security purposes. For the purpose of this Agreement, a Person shall be deemed to be the owner of any property that it has acquired or holds subject to a conditional sale agreement, or leases under a financing lease or other arrangement pursuant to which title to the property has been retained by or vested in some other Person in a transaction intended to create a financing.

"Material Adverse Effect" has the meaning specified in Section 3.1.

"Money Laundering Laws" has the meaning specified in Section 3.28.

"<u>NYSE</u>" means The New York Stock Exchange, Inc.

"Operating Company" means Antero Midstream LLC, a Delaware limited liability company.

"Operating Company LLC Agreement" means the Limited Liability Company Agreement of the Operating Company dated as of January 16, 2014.

"<u>Operative Documents</u>" means, collectively, this Agreement and the Registration Rights Agreement and any amendments, supplements, continuations or modifications thereto.

"Organizational Documents" has the meaning specified in Section 3.11.

"Partnership" has the meaning set forth in the introductory paragraph.

"Partnership Agreement" means the Agreement of Limited Partnership of the Partnership dated as of November 10, 2014.

"<u>Partnership Entities</u>" and each a "<u>Partnership Entity</u>" means the Partnership, the General Partner, the Operating Company and Antero Treatment and, after giving effect to the Contribution, Antero Water.

"Partnership Related Parties" has the meaning specified in Section 6.2.

"Permits" has the meaning specified in Section 3.20.

"<u>Person</u>" means an individual or a corporation, limited liability company, partnership, joint venture, trust, unincorporated organization, association, government agency or political subdivision thereof or other form of entity.

"Placement Agent" means Barclays Capital Inc.

"<u>Placement Agent Engagement Letter</u>" means that certain Placement Agent Engagement Letter, dated as of September 14, 2015, between the Partnership and the Placement Agent.

"<u>Purchase Price</u>" means, with respect to a particular Purchaser, the dollar amount set forth opposite such Purchaser's name in the column titled "Purchase Price" set forth on <u>Schedule A</u> hereto, as adjusted in accordance with <u>Section 7.12</u>, if applicable; *provided* that in no event shall the Purchase Price applicable to such Purchaser be increased without the prior written consent of such Purchaser.

"<u>Purchased Units</u>" means, with respect to a particular Purchaser, the number of Common Units (rounded, if necessary, to the nearest whole number) equal to the quotient of (i) the Purchase Price applicable to such Purchaser divided by (ii) the Common Unit Price.

"Purchaser" and "Purchasers" have the meanings set forth in the introductory paragraph.

"Purchaser Related Parties" has the meaning specified in Section 6.1.

"Registration Rights Agreement" has the meaning set forth in the recitals hereto.

4

"Registration Statement" has the meaning set forth in the Registration Rights Agreement.

"<u>Representatives</u>" of any Person means the Affiliates, officers, directors, managers, employees, agents, counsel, accountants, investment bankers and other representatives of such Person.

"<u>Revolving Credit Facility</u>" means the Credit Agreement, dated as of February 28, 2014, among the Operating Company, J.P. Morgan Chase Bank, N.A., as administrative agent, and the guarantors and lenders party thereto, as amended.

"rights-of-way" has the meaning specified in Section 3.19.

"Sanctions" has the meaning specified in Section 3.29.

"SEC Reports" means reports and statements filed by the Partnership under the Exchange Act and statements filed by the Partnership under the Securities Act (in the form that became effective), including all amendments, exhibits and schedules thereto.

"Securities Act" means the Securities Act of 1933, as amended from time to time, and the rules and regulations of the Commission promulgated thereunder.

"Short Sales" means, without limitation, all "short sales" as defined in Rule 200 promulgated under Regulation SHO under the Exchange Act, whether or not against the box, and forward sale contracts, options, puts, calls, short sales, "put equivalent positions" (as

defined in Rule 16a-1(h) under the Exchange Act) and similar arrangements, and sales and other transactions through non-U.S. broker dealers or foreign regulated brokers.

"Sponsor Units" has the meaning specified in Section 3.2(f).

"Subordinated Units" has the meaning set forth in the Partnership Agreement.

"<u>Subsidiary</u>" means, with respect to any Person, (a) a corporation of which more than 50% of the voting power of shares entitled (without regard to the occurrence of any contingency) to vote in the election of directors or other governing body of such corporation is owned, directly or indirectly, at the date of determination, by such Person, by one or more Subsidiaries of such Person or a combination thereof, (b) a partnership (whether general or limited) in which such Person or a Subsidiary of such Person is, at the date of determination, a general or limited partner of such partnership, but only if more than 50% of the partnership interests of such partnership (considering all of the partnership interests of the partnership as a single class) is owned, directly or indirectly, at the date of determination, by such Person, by one or more Subsidiaries of such Person, or a combination thereof, or (c) any other Person (other than a corporation or a partnership) in which such Person, one or more Subsidiaries of such Person, or a combination thereof, directly or indirectly or indirectly, at the date of determination, has (i) at least a majority ownership interest or (ii) the power to elect or direct the election of a majority of the directors or other governing body of such Person.

"Treatment Assets" has the meaning given such term in the Contribution Agreement.

5

"Unit Consideration" has the meaning set forth in the recitals hereto.

"Walled Off Person" has the meaning specified in Section 4.12.

"<u>Water Assets</u>" has the meaning given to such term in the Contribution Agreement.

#### ARTICLE II

#### AGREEMENT TO SELL AND PURCHASE

#### Section 2.1 <u>Sale and Purchase</u>.

(a) Subject to the terms and conditions hereof, the Partnership hereby agrees to issue and sell to each Purchaser and each Purchaser hereby agrees, severally and not jointly, to purchase from the Partnership, its respective Purchased Units, and each Purchaser agrees, severally and not jointly, to pay the Partnership the Common Unit Price for each Purchased Unit as set forth in paragraph (b) below.

(b) The amount per Common Unit each Purchaser will pay to the Partnership to purchase the Purchased Units (the "<u>Common Unit Price</u>") hereunder shall be \$18.84. If the Closing Date is after the record date for the distribution to the Partnership's holders of Common Units with respect to the quarter ending September 30, 2015, the Common Unit Price shall be reduced by an amount equal to such per unit distribution and <u>Schedule A</u> hereto shall be updated accordingly.

Section 2.2 <u>Closing</u>. Subject to the terms and conditions hereof, the consummation of the purchase and sale of the Purchased Units hereunder (the "<u>Closing</u>") shall take place at the offices of Vinson & Elkins L.L.P., 1001 Fannin Street, Suite 2500 Houston, Texas 77002, or such other location as mutually agreed by the parties, and upon the first Business Day following the satisfaction or waiver of the conditions set forth in <u>Sections 2.3, 2.4</u> and <u>2.5</u> (other than those conditions that are by their terms to be satisfied at the Closing) (the date of such closing, the "<u>Closing Date</u>"). The parties agree that the Closing may occur via delivery of facsimiles or photocopies of the Operative Documents and the closing deliverables contemplated hereby and thereby. Unless otherwise provided herein, all proceedings to be taken and all documents to be executed and delivered by all parties at the Closing will be deemed to have been taken and executed simultaneously, and no proceedings will be deemed to have been taken or documents executed or delivered until all have been taken, executed or delivered.

Section 2.3 <u>Mutual Conditions</u>. The respective obligations of each party to consummate the purchase and issuance and sale of the Purchased Units shall be subject to the satisfaction on or prior to the Closing Date of each of the following conditions (any or all of which may be waived by a party on behalf of itself in writing, in whole or in part, to the extent permitted by applicable Law):

(a) No Law shall have been enacted or promulgated, and no action shall have been taken, by any Governmental Authority of competent jurisdiction that temporarily, preliminarily or permanently restrains, precludes, enjoins or otherwise prohibits the consummation of the transactions contemplated hereby or makes the transactions contemplated hereby illegal;

(b) There shall not be pending any suit, action or proceeding by any Governmental Authority seeking to restrain, preclude, enjoin or prohibit the transactions contemplated by this Agreement; and

(c) The closing of the Contribution shall have occurred, or shall occur concurrently with the Closing, in which case all conditions set forth in Article VII of the Contribution Agreement shall have been satisfied in all material respects or the fulfillment of any such conditions shall have been waived, except for those conditions which, by their nature, will be satisfied concurrently with the Closing.

Section 2.4 <u>Each Purchaser's Conditions</u>. The obligation of each Purchaser to consummate the purchase of its Purchased Units shall be subject to the satisfaction on or prior to the Closing Date of each of the following conditions (any or all of which may be waived by a particular Purchaser on behalf of itself in writing with respect to its Purchased Units, in whole or in part, to the extent permitted by applicable Law):

(a) The Partnership shall have performed and complied with the covenants and agreements contained in this Agreement that are required to be performed and complied with by the Partnership on or prior to the Closing Date;

(b) (i) The (x) Fundamental Representations and (y) representations and warranties of the Partnership contained in this Agreement that are qualified by materiality or a Material Adverse Effect shall be true and correct when made and as of the Closing Date and (ii) all other representations and warranties of the Partnership shall be true and correct in all material respects when made and as of the Closing Date, in each case as though made at and as of the Closing Date (except that representations and warranties made as of a specific date shall be required to be true and correct as of such date only);

(c) The NYSE shall have authorized, upon official notice of issuance, the listing of the Purchased Units;

(d) No notice of delisting from the NYSE shall have been received by the Partnership with respect to the Common Units;

(e) The Common Units shall not have been suspended by the Commission or the NYSE from trading on the NYSE nor shall suspension by the Commission or the NYSE have been threatened in writing by the Commission or the NYSE;

(f) No Material Adverse Effect shall have occurred and be continuing;

(g) The Partnership shall have received an Aggregate Purchase Price of not less than \$150 million; and

(h) The Partnership shall have delivered, or caused to be delivered, to the Purchasers at the Closing, the Partnership's closing deliveries described in <u>Section 2.6</u>.

Section 2.5 <u>The Partnership's Conditions</u>. The obligation of the Partnership to consummate the issuance and sale of the Purchased Units to a Purchaser shall be

7

subject to the satisfaction on or prior to the Closing Date of each of the following conditions with respect to such Purchaser (any or all of which may be waived by the Partnership in writing, in whole or in part, to the extent permitted by applicable Law):

(a) (i) The representations and warranties of such Purchaser contained in this Agreement that are qualified by materiality shall be true and correct when made and as of the Closing Date and (ii) all other representations and warranties of such Purchaser shall be true and correct in all material respects as of the Closing Date (except that representations of such Purchaser made as of a specific date shall be required to be true and correct as of such date only); and

(b) Such Purchaser shall have delivered, or caused to be delivered, to the Partnership at the Closing, such Purchaser's closing deliveries described in <u>Section 2.7</u>.

Section 2.6 <u>Partnership Deliveries</u>. At the Closing, subject to the terms and conditions hereof, the Partnership will deliver, or cause to be delivered, to each Purchaser:

(a) evidence of the Purchased Units credited to book-entry accounts maintained by the transfer agent of the Partnership, bearing the legend or restrictive notation set forth in <u>Section 4.9</u>, free and clear of all Liens, other than transfer restrictions under the Partnership Agreement and applicable federal and state securities laws;

(b) the Registration Rights Agreement in the form attached to this Agreement as <u>Exhibit A</u>, which shall have been duly executed by the Partnership;

(c) A certificate of the Secretary of State of the State of Delaware, dated a recent date, to the effect that each of the General Partner, the Partnership, the Operating Company, Antero Treatment and Antero Water is in good standing;

(d) An opinion addressed to the Purchasers from Vinson & Elkins L.L.P., legal counsel to the Partnership, dated as of the Closing, in the form and substance attached hereto as Exhibit B;

(e) A certificate, dated the Closing Date and signed by each of the Chief Financial Officer and the Chief Administrative Officer of the General Partner, on behalf of the Partnership, in their capacities as such, stating that:

(i) The Partnership has performed and complied with the covenants and agreements contained in this Agreement that are required to be performed and complied with by the Partnership on or prior to the Closing Date; and

(ii) The representations and warranties of the Partnership contained in this Agreement that are qualified by materiality or Material Adverse Effect are true and correct as of the Closing Date and all other representations and warranties of the Partnership are, individually and in the aggregate, true and correct in all material respects as of the Closing Date (except that representations and warranties made as of a specific date shall be required to be true and correct as of such date only);

8

(f) A certificate of the Secretary or Assistant Secretary of the General Partner, on behalf of the Partnership, certifying as to (1) the Amended and Restated Certificate of Limited Partnership of the Partnership and the Partnership Agreement, (2) board resolutions authorizing the execution and delivery of the Operative Documents and the consummation of the transactions contemplated thereby, including the issuance of the Purchased Units, and (3) the incumbency of the officers authorized to execute the Operative Documents, setting forth the name and title and bearing the signatures of such officers; and

(g) A receipt, dated the Closing Date, executed by the Partnership to the effect that the Partnership has received the Aggregate Purchase Price with respect to the Purchased Units issued and sold to the Purchasers.

Section 2.7 <u>Purchaser Deliveries</u>. At the Closing, subject to the terms and conditions hereof, each Purchaser will deliver, or cause to be delivered, to the Partnership:

(a) Payment to the Partnership of the Purchase Price applicable to such Purchaser by wire transfer of immediately available funds to an account designated by the Partnership in writing at least two Business Days prior to the Closing Date; *provided* that such delivery shall be required only after delivery of the Purchased Units as set forth in <u>Section 2.6(a)</u>; and

(b) The Registration Rights Agreement in the form attached to this Agreement as <u>Exhibit A</u>, which shall have been duly executed by such Purchaser.

Section 2.8 Independent Nature of Purchasers' Obligations and Rights. The obligations of each Purchaser under any Operative Document are several and not joint with the obligations of any other Purchaser, and no Purchaser shall be responsible in any way for the performance of the obligations of any other Purchaser under any Operative Document. The failure or waiver of performance under any Operative Document by any Purchaser does not excuse performance by any other Purchaser or by the Partnership with respect to the other Purchasers. It is expressly understood and agreed that each provision contained in the Operative Documents is between the Partnership and a Purchaser, solely, and not between the Partnership and the Purchasers collectively and not between and among the Purchasers. Nothing contained herein or in any other Operative Document, and no action taken by any Purchaser pursuant thereto, shall be deemed to constitute the Purchasers as a partnership, an association, a joint venture or any other kind of group or entity, or create a presumption that the Purchasers are in any way acting in concert or as a group for purposes of Section 13(d) of the Exchange Act or otherwise with respect to such obligations or the transactions contemplated by the Operative Documents. Each Purchaser shall be entitled to independently protect and enforce its rights, including without limitation, the rights arising out of this Agreement or out of the other Operative Documents, and it shall not be necessary for any other Purchaser to be joined as an additional party in any proceeding for such purpose.

# ARTICLE III

# REPRESENTATIONS AND WARRANTIES OF THE PARTNERSHIP

The Partnership represents and warrants to each Purchaser as follows:

9

Section 3.1 Formation and Qualification of the Partnership Entities. Each of the Partnership Entities has been duly organized and is validly existing and in good standing under the laws of its jurisdiction of organization, is duly qualified to do business and is in good standing in each jurisdiction in which its ownership or lease of property or the conduct of its business requires such qualification, and has all power and authority necessary to own or hold its properties and to conduct the business in which it is engaged, except where the failure to be so qualified, in good standing or have such power or authority would not, individually or in the aggregate, (A) have a material adverse effect on the business, properties, management, financial position or results of operations of the Partnership Entities taken as a whole; (B) materially impair the ability of any of the Partnership Entities to consummate the Contribution or to perform their respective obligations under this Agreement or the other Operative Documents (each of clause (A) and (B), a "<u>Material Adverse Effect</u>"); or (C) subject the limited partners of the Partnership to any material liability or disability. Each of the Partnership Entities has all power and authority necessary to own or hold its properties and to conduct the business in which it is engaged. The Partnership does not own or control, directly or indirectly, any corporation, association or other entity other than the Operating Company and Antero Treatment and, after giving effect to the Contribution, Antero Water.

Section 3.2 Purchased Units; Capitalization.

(a) On the Closing Date, the Purchased Units shall have those rights, preferences, privileges and restrictions governing the Common Units as set forth in the Partnership Agreement.

(b) *General Partner*. The General Partner has, and at the Closing Date will have, full limited liability company power and authority to serve as general partner of the Partnership. The General Partner is the sole general partner of the Partnership.

(c) *Common Units Held.* As of the date hereof, the issued and outstanding partnership interests of the Partnership consist of (i) 29,940,957 Common Units and 75,940,957 Subordinated Units and the Incentive Distribution Rights, which are the only limited partner interests of the Partnership issued and outstanding (other than limited partner interests issued under the Partnership's Long-Term Incentive Plan), and (ii) the General Partner Interest; all of such Common Units have been duly authorized and validly issued pursuant to the Partnership Agreement and are fully paid (to the extent required under the Partnership Agreement) and nonassessable (except as such nonassessability may be affected by Sections 17-303, 17-607 and 17-804 of the Delaware LP Act).

(d) Ownership of the General Partner Interest in the Partnership. The General Partner is, and on the Closing Date will be, the sole general partner of the Partnership, with a noneconomic general partner interest in the Partnership (the "General Partner Interest"); such General Partner Interest has been duly authorized and validly issued in accordance with the Partnership Agreement; and the General Partner owns such General Partner Interest free and clear of all Liens (except for (i) restrictions on transferability contained in the Partnership Agreement and (ii) Liens created or arising under the Delaware LP Act).

10

(e) Ownership of the Incentive Distribution Rights. The General Partner owns, and on the Closing Date will own, all of the Incentive Distribution Rights; the Incentive Distribution Rights and the limited partner interests represented thereby have been duly authorized and validly issued in accordance with the Partnership Agreement and are fully paid (to the extent required under the Partnership Agreement) and nonassessable (except as such nonassessability may be affected by Sections 17-303, 17-607 and 17-804 of the Delaware LP Act); and the General Partner owns such Incentive Distribution Rights free and clear of all Liens (except for (i) restrictions on transferability contained in the Partnership Agreement and (ii) Liens created or arising under the Delaware LP Act).

(f) Ownership of the Sponsor Units. On the Closing Date, and after giving effect to the Contribution and the other transactions contemplated by the Contribution Agreement, Antero will own 40,929,378 Common Units and 75,940,957 Subordinated Units (collectively, the "Sponsor Units"); the Sponsor Units and the limited partner interests represented thereby have been duly authorized and validly issued in accordance with the Partnership Agreement and are fully paid (to the extent required under the Partnership Agreement) and nonassessable (except as such nonassessability may be affected by Sections 17-303, 17-607 and 17-804 of the Delaware LP Act); and Antero owns such Sponsor Units free and clear of all Liens (except for (i) restrictions on transferability contained in the Partnership Agreement, (ii) Liens created or arising under the Delaware LP Act and (iii) Liens created or arising under Antero's revolving credit facility).

(g) Ownership of Operating Company. The Partnership owns, and on the Closing Date will own, 100% of the limited liability company interests in the Operating Company; such limited liability company interests have been duly authorized and validly issued in accordance with the Operating Company LLC Agreement and are fully paid (to the extent required under the Operating Company LLC Agreement) and nonassessable (except as such nonassessability may be affected by Sections 18-607 and 18-804 of the Delaware LLC Act); and such limited liability company interests are owned free and clear of all Liens (except for (i) restrictions on transferability contained in the Operating Company LLC Agreement, (ii) Liens created or arising under the Delaware LLC Act and (iii) Liens created or arising under the Revolving Credit Facility).

(h) Ownership of Antero Treatment. The Partnership owns, and on the Closing Date will own, 100% of the limited liability company interests in Antero Treatment; such limited liability company interests have been duly authorized and validly issued in accordance with the limited liability company agreement of Antero Treatment (the "Antero Treatment LLC Agreement") and are fully paid (to the extent required under the Antero Treatment LLC Agreement) and nonassessable (except as such nonassessability may be affected by Sections 18-607 and 18-804 of the Delaware LLC Act); and such limited liability company interests are owned free and clear of all Liens (except for (i) restrictions on transferability contained in the Antero Treatment LLC Agreement, (ii) Liens created or arising under the Delaware LLC Act and (iii) Liens created or arising under the Revolving Credit Facility).

(i) *Ownership of Antero Water*. On the Closing Date, after giving effect to the Contribution, the Partnership will own 100% of the limited liability company interests in Antero Water; such limited liability company interests will have been duly authorized and validly issued in accordance with the limited liability company agreement of Antero Water (the "<u>Antero Water</u>

11

<u>LLC Agreement</u>") and will be fully paid (to the extent required under the Antero Water LLC Agreement) and nonassessable (except to the extent such nonassessability may be affected by Sections 18-607 and 18-804 of the Delaware LLC Act); and such limited liability company interests will be owned free and clear of all Liens (except for (i) restrictions on transferability contained in the Antero Water LLC Agreement, (ii) Liens created or arising under the Delaware LLC Act and (iii) Liens created or arising under the Revolving Credit Facility).

(j) *No Other Subsidiaries.* On the Closing Date, and after giving effect to the Contribution, the General Partner will not own, directly or indirectly, any equity or long-term debt securities of any corporation, partnership, limited liability company, joint venture, association or other entity, other than the Partnership, the Operating Company, Antero Treatment and Antero Water. On the Closing Date, after giving effect to the Contribution, the Partnership will not own, directly or indirectly, any equity or long-term debt securities of any corporation, partnership, limited liability company, joint venture, association or other entity, other than the Operating Company, Antero Treatment and Antero Water.

Section 3.3 <u>No Conflicts</u>. The execution, delivery and performance by the Partnership Entities of this

Agreement and each of the other Operative Documents to which they are a party, the issuance and sale of the Purchased Units, the consummation of the Contribution and any other transactions contemplated by this Agreement and the other Operative Documents and the application of the proceeds from the sale of the Purchased Units will not (i) conflict with or result in a breach or violation of any of the terms or provisions of, or constitute a default under, or result in the creation or imposition of any lien, charge or encumbrance upon any property or assets of any of the Partnership Entities pursuant to, any indenture, mortgage, deed of trust, loan agreement, license, lease or other agreement or instrument to which any of the Partnership Entities is a party or by which any of the Partnership Entities is bound or to which any of the property, right or assets of any of the Partnership Entities; or (iii) result in any violation of the provisions of the Organizational Documents of any of the Partnership Entities; or (iii) result in any violation of any law or statute or any judgment, order, decree, rule or regulation of any court or arbitrator or governmental or regulatory authority, except, in the case of clauses (i) and (iii) above, for any such conflict, breach, violation or default that would not, individually or in the aggregate, have a Material Adverse Effect.

Section 3.4 <u>No Defaults</u>. None of the Partnership Entities is (i) in violation of its Organizational Documents; (ii) in default, and no event has occurred that, with notice or lapse of time or both, would constitute such a default, in the due performance or observance of any term, covenant or condition contained in any indenture, mortgage, deed of trust, loan agreement or other agreement or instrument to which any of the Partnership Entities is a party or by which any of the Partnership Entities is bound or to which any of the property or assets of any of the Partnership Entities is subject; or (iii) in violation of any law or statute or any judgment, order, rule or regulation of any court or arbitrator or governmental or regulatory authority; except, in the case of clauses (ii) and (iii) above, for any such default or violation that would not, individually or in the aggregate, have a Material Adverse Effect.

Section 3.5 <u>Authority</u>. The Partnership has full right, power and authority to execute and deliver this Agreement and to perform its obligations hereunder. Each of the Partnership Entities has full right, power and authority to execute and deliver each of the other

12

Operative Documents to which such Partnership Entity is a party and to perform its obligations thereunder. The Partnership has all requisite limited partnership power and authority to issue, sell and deliver the Purchased Units, in accordance with and upon the terms and conditions set forth in this Agreement and the Partnership Agreement. On the Closing Date, all limited partnership or limited liability company action, as the case may be, required to be taken by the General Partner or the Partnership for the authorization, issuance, sale and delivery of the Purchased Units, the execution and delivery of the Operative Documents and the consummation of the transactions contemplated hereby, shall have been validly taken.

Section 3.6 <u>No Consents</u>. No consent, approval, authorization or order of, or filing, registration or qualification ("<u>Consent</u>") of or with any court or arbitrator or governmental or regulatory authority is required for (i) the execution, delivery and performance by any of the Partnership Entities of any of the Operative Documents; (ii) the issuance and sale of the Purchased Units; (iii) the consummation of the Contribution or any other transactions contemplated by this Agreement, the Contribution Agreement or the other Operative Documents; or (iv) the application of the proceeds from the sale of the Purchased Units, except (A) such as have been, or prior to the Closing Date will be, obtained or made, and (B) for the registration of the Purchased Units under the Securities Act and Consents as may be required under the Exchange Act, applicable state securities laws, and the rules of the Financial Industry Regulatory Authority, Inc. in connection with the purchase and sale of the Purchased Units by the Purchasers, and (C) for such consents that, if not obtained, have not or would not, individually or in the aggregate, have a Material Adverse Effect.

Section 3.7 <u>Authorization, Execution and Delivery of the Common Unit Purchase Agreement</u>. This Agreement has been duly authorized and validly executed and delivered by or on behalf of the Partnership and constitutes a valid and legally binding agreement of the Partnership, enforceable against the Partnership in accordance with its term; *provided*, that the enforceability thereof may be limited by (A) applicable bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally or by equitable principles (whether considered in a proceeding at law or in equity) relating to enforceability and (B) public policy, applicable law relating to fiduciary duties and indemnification and an implied covenant of good faith and fair dealing (collectively, the "<u>Enforceability Exceptions</u>").

Section 3.8 <u>Authorization, Execution, Delivery and Enforceability of Certain Agreements</u>. On the Closing Date, each of the Operative Documents will have been duly authorized, executed and delivered by the Partnership Entities party thereto and will be a valid and legally binding agreement of such Partnership Entity, enforceable against such Partnership Entity in accordance with its terms, subject to the Enforceability Exceptions.

# Section 3.9 <u>Contribution Agreement.</u>

(a) On the Closing Date, the Contribution Agreement will have been duly authorized, executed and delivered by (i) the Partnership and Antero Treatment and (ii) to the knowledge of the Partnership, Antero. Assuming the due authorization of the parties thereto other than the Partnership and Antero Treatment, the Contribution Agreement will constitute a valid and legally

binding agreement of the Partnership and Antero Treatment, enforceable against the Partnership and Antero Treatment in accordance with its terms, subject to the Enforceability Exceptions.

available to the Purchasers, a copy of the Contribution Agreement (other than exhibits and schedules, except to the extent they will be filed with the Commission within four business days of the date hereof) that is complete in all material respects.

Section 3.10 <u>Authorization of Contribution Equity Consideration</u>. The Unit Consideration to be issued by the Partnership pursuant to the Contribution Agreement, and the limited partner interests represented thereby, have been duly authorized and, when issued and delivered in accordance with the terms of the Partnership Agreement and the Contribution Agreement as consideration therefor as provided therein, will be fully paid (to the extent required under the Partnership Agreement) and non-assessable (except as such non-assessability may be affected by Section 17-303, 17-607 or 17-804 of the Delaware LP Act).

Section 3.11 <u>Valid Issuance; No Options or Preemptive Rights of Common Units</u>. The Purchased Units to be issued and sold by the Partnership and the limited partner interests represented thereby have been duly authorized in accordance with the Partnership Agreement and, when issued and delivered to the Purchasers against payment therefor in accordance with the terms hereof, will be validly issued, fully paid (to the extent required under the Partnership Agreement) and non-assessable (except as such nonassessability may be affected by matters described in Sections 17-303, 17-607 and 17-804 of the Delaware LP Act). Except as provided in the Operative Documents and the Partnership Agreement, there are no options, warrants, preemptive rights, rights of first refusal or other rights to subscribe for or to purchase, or any restriction upon the voting or transfer of, any equity securities of any of the Partnership Entities pursuant to any of their certificate of limited partnership, formation or incorporation, agreement of limited partnership, limited liability company agreement, the Registration Rights Agreement and the Existing Registration Rights Agreement, neither the filing of the Registration Statement pursuant to the Registration Rights Agreement nor the offering or sale of the Common Units or other securities of the Partnership.

Section 3.12 <u>No Registration Rights</u>. Except as contemplated by this Agreement and the Registration Rights Agreement or pursuant to the Partnership Agreement or the Existing Registration Rights Agreement, there are no contracts, agreements or understandings between any of the Partnership and any Person granting such Person the right to require the Partnership to file a registration statement under the Securities Act with respect to any securities of the Partnership owned or to be owned by such person or to require the Partnership to include such securities in the Registration Statement or in any securities registered or to be registered pursuant to any registration statement filed by or required to be filed by the Partnership under the Securities Act.

14

Section 3.13 <u>Periodic Reports</u>. The SEC Reports have been filed with the Commission on a timely basis. The SEC Reports, including, without limitation, any audited or unaudited financial statements and any notes thereto or schedules included therein, at the time filed (or in the case of registration statements, solely on the dates of effectiveness) (except to the extent corrected by a subsequent SEC Report) (a) did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading and (b) complied in all material respects with the applicable requirements of the Exchange Act and the Securities Act, as the case may be.

Section 3.14 <u>Financial Statements</u>. The historical financial statements of the Partnership (including the related notes and supporting schedules) included in the SEC Reports comply as to form in all material respects with the requirements of Regulation S-X under the Securities Act and present fairly in all material respects the financial condition, results of operations and cash flows of the entities purported to be shown thereby at the dates and for the periods indicated and have been prepared in conformity with accounting principles generally accepted in the United States applied on a consistent basis throughout the periods involved.

Section 3.15 <u>Independent Registered Public Accounting Firm</u>. KPMG LLP, which has certified certain financial statements of the Partnership and its subsidiaries is an independent public accounting firm with respect to the Partnership and its subsidiaries within the applicable rules and regulations adopted by the Commission and the Public Company Accounting Oversight Board (United States) and as required by the Securities Act.

Section 3.16 <u>Litigation</u>. There are no legal, governmental or regulatory investigations, actions, suits or proceedings pending to which any of the Partnership Entities is or may be a party or to which any property, right or asset of the Partnership Entities is or may be the subject that, individually or in the aggregate, if determined adversely to the Partnership Entities, could reasonably be expected to have a Material Adverse Effect; and to the knowledge of the Partnership Entities, no such investigations, actions, suits or proceedings are threatened or contemplated by any governmental or regulatory authority or by others.

Section 3.17 <u>No Material Adverse Changes</u>. Since the date of the most recent audited financial statements included in the SEC Reports, (i) there has not been any change in the equity or long-term debt of the Partnership Entities, or any dividend or distribution of any kind declared, set aside for payment, paid or made by the Partnership Entities on any class of equity interests, or any material adverse change, or any development that would reasonably be expected to result in a material adverse change, in or affecting the business, properties, management, financial position or results of operations of the Partnership Entities taken as a whole; (ii) none of the Partnership Entities has entered into any transaction or agreement that is material to the Partnership Entities taken as a whole or incurred any liability or obligation, direct or contingent, that is material to the Partnership Entities taken as a whole; (iii) none of the Partnership Entities has sustained any material loss or interference with its business or operation from fire, explosion, flood or other calamity, or from any labor disturbance or dispute or any action, order or decree of any court or arbitrator or governmental or regulatory authority; and (iv) none of the Partnership Entities has issued or granted any securities; except in each case as

otherwise disclosed in the SEC Reports and except as contemplated by the Operative Documents.

Section 3.18 <u>Title to Properties</u>. Except with respect to rights-of-way (as defined below), which are the subject of <u>Section 3.19</u>, each of the Partnership Entities has good and marketable title to, or valid rights to lease or otherwise use, all items of real property and personal property that are material to the respective businesses of the Partnership Entities, in each case free and clear of all Liens except those that (i) do not materially interfere with the use made and proposed to be made of such property by the Partnership Entities or (ii) could not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect.

Section 3.19 <u>Rights of Way</u>. Each of the Partnership Entities, directly or indirectly, has such consents, easements, rights-of-way, permits or licenses from each person (collectively, "<u>rights-of-way</u>") as are necessary to conduct its business in the manner described in the SEC Reports, if any, except for (i) qualifications, reservations and encumbrances with respect thereto that would not have a Material Adverse Effect and (ii) such rights-of-way that, if not obtained, would not have, individually or in the aggregate, a Material Adverse Effect; each of the Partnership Entities has fulfilled and performed, in all material respects, its obligations with respect to such rights-of-way and no event has occurred that allows, or after notice or lapse of time would allow, revocation or termination thereof or would result in any impairment of the rights of the holder of any such rights-of-way, except for such revocations, terminations and impairments that, individually or in the aggregate, would not have a Material Adverse Effect; and none of such rights-of-way contains any restriction that would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

Section 3.20 <u>License and Permits</u>. Except with respect to permits related to Environmental Law (as defined below), which are the subject of <u>Section 3.24</u>, each of the Partnership Entities possesses all licenses, certificates, permits and other authorizations issued by, and have made all declarations and filings with, the appropriate federal, state, local or foreign governmental or regulatory authorities ("<u>Permits</u>") that are necessary for the ownership or lease of their respective properties or the conduct of their respective businesses as described in the SEC Reports, except where the failure to possess or make the same would not, individually or in the aggregate, have a Material Adverse Effect; and none of the Partnership Entities has received notice of any revocation or modification of any such Permits or has any reason to believe that any such Permits will not be renewed in the ordinary course.

Section 3.21 <u>Intellectual Property</u>. Each of the Partnership Entities own or possess adequate rights to use all material patents, patent applications, trademarks, service marks, trade names, trademark registrations, service mark registrations, copyrights, licenses and know-how (including trade secrets and other unpatented and/or unpatentable proprietary or confidential information, systems or procedures) necessary for the conduct of their respective businesses, except as could not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect.

Section 3.22 <u>Insurance</u>. Each of the Partnership Entities has insurance covering their respective properties, operations, personnel and businesses, which insurance is in reasonable amounts and insures against such losses and risks as are reasonably adequate to

16

protect the Partnership Entities and their respective businesses; and none of the Partnership Entities has (i) received notice from any insurer or agent of such insurer that capital improvements or other expenditures are required or necessary to be made in order to continue such insurance or (ii) any reason to believe that it will not be able to renew its existing insurance coverage as and when such coverage expires or to obtain similar coverage at reasonable cost from similar insurers as may be necessary to continue its business.

Section 3.23 <u>No Labor Dispute; No Notice of Labor Law Violations</u>. No labor disturbance by, or dispute with, the employees of the Partnership Entities exists or, to the knowledge of each of the Partnership Entities, is contemplated or threatened, and the Partnership is not aware of any existing or imminent labor disturbance by, or dispute with, the employees of the Partnership Entities, except as could not reasonably be expected to have a Material Adverse Effect.

Section 3.24 Environmental Compliance. (i) The Partnership Entities (x) are and, during the relevant time periods specified in all applicable statutes of limitations, have been in compliance with all applicable federal, state, local and foreign laws, rules, regulations, requirements, decisions and orders relating to the protection of human health or safety (to the extent such human health or safety protection is related to exposure to hazardous or toxic substances or wastes, pollutants or contaminants), the environment, natural resources, hazardous or toxic substances or wastes, pollutants or contaminants (collectively, "Environmental Laws"), (y) have received and are in compliance with all permits, licenses, certificates or other authorizations or approvals required of them under applicable Environmental Laws to conduct their respective businesses and (z) have not received any written notice of any actual or potential liability under or relating to any Environmental Laws, including for the investigation or remediation of any disposal or release of hazardous or toxic substances or wastes, pollutants or contaminants, and have no knowledge of any event or condition that would reasonably be expected to result in any such notice; (ii) there are no costs or liabilities associated with Environmental Laws of or relating to the Partnership Entities, except in the case of each of (i) and (ii) above, for any such failure to comply, or failure to receive required permits, licenses or approvals, or cost or liability, as would not, individually or in the aggregate, have a Material Adverse Effect; and (iii) there are no proceedings that are pending or, to the knowledge of the Partnership Entities, threatened against the Partnership Entities under any Environmental Laws in which a governmental authority is also a party, other than such proceedings regarding which it is reasonably believed no monetary sanctions of \$100,000 or more will be imposed.

Section 3.25 <u>Tax Returns</u>. Each of the Partnership Entities has paid all federal, state, local and foreign taxes and filed all tax returns required to be paid or filed through the date hereof, and except as would not, individually or in the aggregate, have a Material Adverse Effect, there is no tax deficiency that has been, or could reasonably be expected to be, asserted against the Partnership Entities or any of their respective properties or assets.

Section 3.26

No Employment Law Violations. Each employee benefit plan, within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), that is maintained, administered or contributed to by the Partnership or any of its affiliates for employees or former employees of the Partnership and its affiliates has been maintained in compliance in all material respects with its terms and the

17

requirements of any applicable statutes, orders, rules and regulations, including, but not limited to, ERISA and the Internal Revenue Code of 1986, as amended (the "Code"); no prohibited transaction, within the meaning of Section 406 of ERISA or Section 4975 of the Code, has occurred with respect to any such plan excluding transactions effected pursuant to a statutory or administrative exemption, and transactions which, individually or in the aggregate, would not have a Material Adverse Effect, and no such plan is subject to the funding rules of Section 412 of the Code or Section 302 of ERISA; and neither the Partnership nor any of its subsidiaries has any reasonable expectation of incurring any liabilities under Title IV of ERISA.

Section 3.27 No Unlawful Payments. None of the Partnership Entities nor, to the knowledge of the Partnership Entities, any director, officer, agent, employee or other person associated with or acting on behalf of the Partnership Entities, has (i) used its funds for any unlawful contribution, gift, entertainment or other unlawful expense relating to political activity; (ii) made any direct or indirect unlawful payment to any foreign or domestic government official or employee from corporate funds; (iii) violated or is in violation of any provision of the U.S. Foreign Corrupt Practices Act of 1977; (iv) violated or is in violation of any provision of the Bribery Act 2010 of the United Kingdom; or (v) made any bribe, rebate, payoff, influence payment, kickback or other unlawful payment.

Compliance with Money Laundering Laws. The operations of the Partnership Entities are and Section 3.28 have been conducted at all times in compliance with applicable financial recordkeeping and reporting requirements of the Currency and Foreign Transactions Reporting Act of 1970, as amended, the money laundering statutes of all jurisdictions, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any governmental agency (collectively, the "Money Laundering Laws") and no action, suit or proceeding by or before any court or governmental agency, authority or body or any arbitrator involving the Partnership Entities with respect to the Money Laundering Laws is pending or, to the knowledge of the Partnership Entities, threatened.

OFAC. None of the Partnership Entities nor, to the actual knowledge of the Partnership Entities, Section 3.29 any director, officer, agent, employee or affiliate of the Partnership Entities is currently the subject or the target of any sanctions ("Sanctions") administered or enforced by the U.S. Office of Foreign Assets Control of the U.S. Department of the Treasury ("OFAC"); and the Partnership will not directly or indirectly use the proceeds of the offering of the Common Units hereunder, or lend, contribute or otherwise make available such proceeds to any subsidiary, joint venture partner or other person or entity (i) to fund any activities of or business with any person, or in any country or territory, that, at the time of such funding, is the subject of Sanctions or (ii) in any other manner that will result in a violation by any person (including any person participating in the transaction, whether as underwriter, advisor, investor or otherwise) of Sanctions.

Section 3.30 Certain Fees. Other than as described in the Placement Agent Engagement Letter, none of the Partnership Entities is a party to any contract, agreement or understanding with any person (other than this Agreement) that would give rise to a valid claim against any of them or the Purchasers for a brokerage commission, finders' fee or like payment in connection with the offering and sale of the Purchased Units. The Partnership agrees that it

18

will indemnify and hold harmless each Purchaser from and against any and all claims, demands or liabilities for broker's, finder's, placement or other similar fees or commissions incurred by the Partnership in connection with the purchase of the Purchased Units or the consummation of the transactions contemplated by this Agreement.

No Side Agreements. There are no agreements by, among or between the Partnership or any of Section 3.31 its Affiliates, on the one hand, and any Purchaser or any of their Affiliates, on the other hand, with respect to the transactions contemplated hereby other than the Operative Documents nor promises or inducements for future transactions between or among any of such parties.

Section 3.32 No Registration. Assuming the accuracy of the representations and warranties of the Purchaser contained in Section 4.6 and Section 4.7, the issuance and sale of the Purchased Units pursuant to this Agreement is exempt from registration requirements of the Securities Act, and neither the Partnership nor, to the knowledge of the Partnership, any authorized Representative acting on its behalf has taken or will take any action hereafter that would cause the loss of such exemption.

Section 3.33 No Integration. The Partnership has not, directly or through any agent, issued, sold, offered for sale, solicited offers to buy or otherwise negotiated in respect of, any security (as defined in the Securities Act), that is or will be integrated with the issuance and sale of the Purchased Units contemplated by this Agreement pursuant to the Securities Act, the rules and regulations thereunder or the interpretations thereof by the Commission.

Section 3.34 MLP Status. The Partnership is, and has been since November 10, 2014, properly treated as a partnership for United States federal income tax purposes, and 90% or more of the combined gross income of the Partnership for the most recent four complete calendar quarters ending before the Closing Date for which the necessary financial information is available are from sources treated as "qualifying income" within the meaning of Section 7704(d) of the Internal Revenue Code of 1986, as amended.

Section 3.35 <u>Qualifying Income of Contributed Assets</u>. The Partnership expects that at least 90% of the combined gross income of the Partnership, Antero Water, Antero Treatment and the operations conducted with the Water Assets in 2015 after the closing date of the transactions contemplated by the Contribution Agreement will be "qualifying income" within the meaning of Section 7704(d) of the Internal Revenue Code of 1986, as amended.

Section 3.36 <u>Investment Company</u>. None of the Partnership Entities is and, as of the Closing Date after giving effect to the offer and sale of the Purchased Units and the application of the proceeds therefrom, none of them will be, (i) an "investment company" or a company "controlled" by an "investment company" within the meaning of the Investment Company Act of 1940, as amended, and the rules and regulations of the Commission thereunder (collectively, the "<u>Investment Company Act</u>") or (ii) a "business development company" (as defined in Section 2(a)(48) of the Investment Company Act).

Section 3.37 <u>Disclosure Controls</u>. The Partnership Entities maintain an effective system of disclosure controls and procedures (as defined in Rule 13a-15(e) of the

19

Exchange Act) that is designed to ensure that information required to be disclosed by the Partnership Entities in reports that the Partnership files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the Commission's rules and forms, including controls and procedures designed to ensure that such information is accumulated and communicated to the Partnership's management, including the principal executive officer(s) and principal financial officer(s) of the General Partner, as appropriate to allow timely decisions regarding required disclosure to be made. As of the date of the Partnership's most recent audited financial statements included in an SEC Report, the Partnership's disclosure controls and procedures were effective in all material respects to perform the functions for which they were established.

Section 3.38 <u>Accounting Controls</u>. The Partnership Entities maintain systems of "internal control over financial reporting" (as such term is defined in Rule 15d-15(f) of the Exchange Act) that complies with the requirements of the Exchange Act and that has been designed by, or under the supervision of, the General Partner's principal executive officer(s) and principal financial officer(s), to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles in the United States, including, but not limited to, internal accounting controls sufficient to provide reasonable assurance that (i) transactions are executed in accordance with management's general or specific authorization; (ii) transactions are recorded as necessary to permit preparation of the Partnership's consolidated financial statements in conformity with U.S. generally accepted accounting principles and to maintain asset accountability; (iii) access to assets is permitted only in accordance with management's general or specific authorization; (iv) the recorded accountability for assets is compared with existing assets at reasonable intervals and appropriate action is taken with respect to any differences; and (v) interactive data in eXtensible Business Reporting Language is prepared in accordance with the Commission's rules and guidelines applicable thereto. As of the date of the most recent balance sheet of the Partnership and its consolidated subsidiaries reviewed or audited by KPMG LLP, there were no material weaknesses or significant deficiencies in the internal controls of the Partnership Entities.

Section 3.39 <u>Placement Agent Reliance</u>. The Partnership acknowledges that the Placement Agent may rely upon the representations and warranties made by the Partnership to each Purchaser in this Agreement.

Section 3.40 <u>Legal Sufficiency of the Contribution Agreement</u>. The Contribution Agreement is legally sufficient to transfer or convey to the Partnership (i) all of the transferor's right, title and interest in and to the Water Assets and (ii) all of the ownership interests, assets and rights in and to the Water Assets purported to be transferred thereby, subject to the conditions, reservations, encumbrances and limitations contained in the Contribution Agreement. The Partnership, upon consummation of the transactions contemplated by the Contribution Agreement, will directly or indirectly succeed in all material respects to the Water Assets.

Section 3.41 <u>Absence of Price Manipulation</u>. Neither the Partnership nor, to the knowledge of the Partnership, any of its Affiliates or its or their respective directors or officers, has taken, or will take, directly or indirectly, any action designed to, or that might

20

reasonably be expected to, cause or result in stabilization or manipulation of the price of the Common Units to facilitate the sale or resale of the Purchased Units in violation of Regulation M under the Exchange Act.

# ARTICLE IV

## **REPRESENTATIONS AND WARRANTIES OF THE PURCHASERS**

Each Purchaser, severally and not jointly, hereby represents and warrants to the Partnership that:

Section 4.1 <u>Existence</u>. Such Purchaser is duly organized and validly existing and in good standing under the Laws of its jurisdiction of organization, with all requisite power and authority to own, lease, use and operate its properties and to conduct its business as currently conducted, except where the failure to have such power or authority would not prevent the consummation of the transactions contemplated by this Agreement and the Registration Rights Agreement.

Section 4.2 <u>Authorization, Enforceability</u>. Such Purchaser has all necessary corporate, limited liability

company or partnership power and authority to execute, deliver and perform its obligations under this Agreement and the Registration Rights Agreement and to consummate the transactions contemplated thereby, and the execution, delivery and performance by such Purchaser of this Agreement and the Registration Rights Agreement has been duly authorized by all necessary action on the part of such Purchaser; and this Agreement and the Registration Rights Agreement constitute the legal, valid and binding obligations of such Purchaser, enforceable in accordance with their terms, subject to the Enforceability Exceptions.

Section 4.3 <u>No Breach</u>. The execution, delivery and performance of this Agreement and the Registration Rights Agreement by such Purchaser and the consummation by such Purchaser of the transactions contemplated hereby and thereby will not (a) conflict with or result in a breach or violation of any of the terms or provisions of, or constitute a default under, any material agreement to which such Purchaser is a party or by which such Purchaser is bound or to which any of the property or assets of such Purchaser is subject, (b) conflict with or result in any violation of the provisions of the Organizational Documents of such Purchaser, or (c) violate any statute, order, rule or regulation of any court or governmental agency or body having jurisdiction over such Purchaser or the property or assets of such Purchaser, except in the cases of clauses (a) and (c), for such conflicts, breaches, violations or defaults as would not prevent the consummation of the transactions contemplated by this Agreement and the Registration Rights Agreement.

Section 4.4 <u>Certain Fees</u>. No fees or commissions are or will be payable by such Purchaser to brokers, finders, or investment bankers with respect to the purchase of any of the Purchased Units or the consummation of the transaction contemplated by this Agreement. Such Purchaser agrees that it will indemnify and hold harmless the Partnership from and against any and all claims, demands, or liabilities for broker's, finder's, placement, or other similar fees or commissions incurred by such Purchaser in connection with the purchase of the Purchased Units or the consummation of the transactions contemplated by this Agreement.

21

Section 4.5 <u>No Side Agreements</u>. There are no other agreements by, among or between such Purchaser and any of its Affiliates, on the one hand, and the Partnership or any of its Affiliates, on the other hand, with respect to the transactions contemplated hereby other than the Operative Documents nor promises or inducements for future transactions between or among any of such parties.

Section 4.6 Investment. The Purchased Units are being acquired for such Purchaser's own account, the account of its Affiliates, or the accounts of clients for whom such Purchaser exercises discretionary investment authority (all of whom such Purchaser hereby represents and warrants are "accredited investors" within the meaning of Rule 501(a) of Regulation D promulgated by the Commission pursuant to the Securities Act), not as a nominee or agent, and with no present intention of distributing the Purchased Units or any part thereof, and such Purchaser has no present intention of selling or granting any participation in or otherwise distributing the same in any transaction in violation of the securities laws of the United States or any state, without prejudice, however, to such Purchaser's right at all times to sell or otherwise dispose of all or any part of the Purchased Units under a registration statement under the Securities Act and applicable state securities laws or under an exemption from such registration available thereunder (including, without limitation, if available, Rule 144 promulgated thereunder). If such Purchaser should in the future decide to dispose of any of the Purchased Units, the Purchaser understands and agrees (a) that it may do so only in compliance with the Securities Act and applicable state securities law, as then in effect, including a sale contemplated by any registration statement pursuant to which such securities are being offered, or pursuant to an exemption from the Securities Act, and (b) that stop-transfer instructions to that effect will be in effect with respect to such securities.

Section 4.7 <u>Nature of Purchaser</u>. Such Purchaser represents and warrants to, and covenants and agrees with, the Partnership that, (a) it is an "accredited investor" within the meaning of Rule 501 of Regulation D promulgated by the Commission pursuant to the Securities Act and (b) by reason of its business and financial experience it has such knowledge, sophistication and experience in making similar investments and in business and financial matters generally so as to be capable of evaluating the merits and risks of the prospective investment in the Purchased Units, is able to bear the economic risk of such investment and, at the present time, would be able to afford a complete loss of such investment.

Section 4.8 <u>Restricted Securities</u>. Such Purchaser understands that the Purchased Units are characterized as "restricted securities" under the federal securities Laws inasmuch as they are being acquired from the Partnership in a transaction not involving a public offering and that under such Laws and applicable regulations such securities may not be resold absent registration under the Securities Act or an exemption therefrom. In this connection, such Purchaser represents that it is knowledgeable with respect to Rule 144 of the Commission promulgated under the Securities Act.

Section 4.9 <u>Legend</u>. Such Purchaser understands that the book entry evidencing the Purchased Units will bear the legend required by the Partnership Agreement as well as a legend substantively consistent with the following legend: "These securities have not been registered under the Securities Act of 1933, as amended (the "<u>Securities Act</u>"). These securities may not be sold or offered for sale except pursuant to an effective registration

statement under the Securities Act or pursuant to an exemption from registration thereunder, in each case in accordance with all applicable securities laws of the states or other jurisdictions, and in the case of a transaction exempt from registration, such securities may only be transferred if the transfer agent for such securities has received documentation satisfactory to it that such transaction does not require registration under the Securities Act."

made available to such Purchaser (through EDGAR, the Company's website or otherwise) all SEC Reports, as well as all press releases or investor presentations issued by the Company through the date of this Agreement that are included in a filing by the Company on Form 8-K or clearly posted on the Company's website.

Section 4.11 <u>Short Selling</u>. Such Purchaser represents that it has not entered into any Short Sales of the Common Units owned by it since the time it first began discussions with the Partnership or the Placement Agent about the transactions contemplated by this Agreement; provided, however, subject to such Purchaser's compliance with its obligations under the U.S. federal securities laws and its internal policies, the above shall not apply, in the case of a Purchaser that is a large multi-unit investment or commercial banking organization, to activities in the normal course of trading units of such Purchaser; provided, further, that subject to such Purchaser's compliance with its obligations under the U.S. federal securities laws and its internal policies: (a) such Purchaser, for purposes hereof, shall not be deemed to include any employees, subsidiaries or Affiliates that are effectively walled off by appropriate "Chinese Wall" information barriers approved by such Purchaser's legal or compliance department (and thus have not been privy to any information concerning this transaction) (a "Walled Off Person") and (b) the foregoing representations in this paragraph shall not apply to any transaction by or on behalf of such Purchaser that was effected by a Walled Off Person in the ordinary course of trading without the advice or participation of such Purchaser or receipt of confidential or other information regarding this transaction provided by such Purchaser to such entity.

#### ARTICLE V

#### **COVENANTS**

Section 5.1 <u>Taking of Necessary Action</u>. Each of the parties hereto shall use its commercially reasonable efforts promptly to take or cause to be taken all action and promptly to do or cause to be done all things necessary, proper or advisable under applicable Law and regulations to consummate and make effective the transactions contemplated by this Agreement. Without limiting the foregoing, the Partnership and each Purchaser shall use its commercially reasonable efforts to make all filings and obtain all Consents of Governmental Authorities that may be necessary or, in the reasonable opinion of the other parties, as the case may be, advisable for the consummation of the transactions contemplated by the Operative Documents. The Partnership shall promptly and accurately respond, and shall use its commercially reasonable efforts to cause its transfer agent to respond, to reasonable requests for information (which is otherwise not publicly available) made by a Purchaser or its auditors relating to the actual holdings of such Purchaser or its accounts; *provided*, that the Partnership shall not be obligated to provide any such information that could reasonably result in a violation of applicable Law or conflict with the Partnership's insider trading policy or a confidentiality

23

obligation of the Partnership. The Partnership shall use its commercially reasonable efforts to cause its transfer agent to reasonably cooperate with each Purchaser to ensure that the Purchased Units are validly and effectively issued to such Purchaser and that such Purchaser's ownership of the Purchased Units following the Closing is accurately reflected on the appropriate books and records of the Partnership's transfer agent.

Section 5.2 <u>Other Actions</u>. The Partnership shall file prior to the Closing a supplemental listing application with the NYSE to list the Purchased Units.

Section 5.3 <u>Contribution Transaction</u>. The Partnership will distribute to Antero an amount equal to the net proceeds of this offering as additional cash consideration and will reduce the Unit Consideration by a number equal to the number of Common Units issued in this offering. If the transactions contemplated by the Contribution Agreement are not closed on the same Business Day as the Closing, the Partnership shall return to each Purchaser its Purchase Price within two Business Days of receipt thereof and each Purchaser shall promptly return its Purchased Units to the Partnership.

Section 5.4 <u>Expenses</u>. The Partnership shall pay up to \$75,000 of legal fees of Andrews Kurth LLP, counsel to the Purchasers, incurred in connection with the negotiation, execution, delivery and performance of this Agreement and Registration Rights Agreement and the transactions contemplated hereby and thereby, provided that any request for such payment is accompanied by a satisfactory written invoice for such expenses. If any action at law or equity is necessary to enforce or interpret the terms of any Operative Document, the prevailing party shall be entitled to reasonable attorney's fees, costs and necessary disbursements in addition to any other relief to which such Party may be entitled. Any legal fees of Andrews Kurth LLP in excess of \$75,000 shall be paid pro rata by all the Purchasers in proportion to the number of Purchased Units purchased by each.

#### ARTICLE VI

# **INDEMNIFICATION**

Section 6.1 <u>Indemnification by the Partnership</u>. The Partnership agrees to indemnify each Purchaser and its Representatives (collectively, "<u>Purchaser Related Parties</u>") from, and hold each of them harmless against, any and all actions, suits, proceedings (including any investigations, litigation or inquiries), demands, and causes of action, and, in connection therewith, and promptly upon demand, pay or reimburse each of them for all costs, losses, liabilities, damages, or expenses of any kind or nature whatsoever, including, without limitation, the reasonable fees and disbursements of counsel and all other reasonable expenses incurred in connection with investigating, defending or preparing to defend any such matter that may be incurred by them or asserted against or involve any of them as a result of, arising out of, or in any way related to the breach of any of the representations, warranties or covenants of the Partnership contained herein, provided that such claim for indemnification relating to a breach of the representations or warranties is made prior to the expiration of the survival period for such representations or warranties; and provided further, that no Purchaser Related Party shall be entitled to recover special, consequential (including lost profits) or punitive damages.

Notwithstanding anything to the contrary, consequential damages shall not be deemed to include diminution in value of the Purchased Units, which is specifically included in damages covered by Purchaser Related Parties' indemnification above.

Section 6.2 <u>Indemnification by Purchasers</u>. Each Purchaser agrees, severally and not jointly, to indemnify the Partnership, the General Partner and their respective Representatives (collectively, "<u>Partnership Related Parties</u>") from, and hold each of them harmless against, any and all actions, suits, proceedings (including any investigations, litigation or inquiries), demands, and causes of action, and, in connection therewith, and promptly upon demand, pay or reimburse each of them for all costs, losses, liabilities, damages, or expenses of any kind or nature whatsoever, including, without limitation, the reasonable fees and disbursements of counsel and all other reasonable expenses incurred in connection with investigating, defending or preparing to defend any such matter that may be incurred by them or asserted against or involve any of them as a result of, arising out of, or in any way related to the breach of any of the representations, warranties or covenants of such Purchaser contained herein, *provided* that such claim for indemnification relating to a breach of the representations and warranties is made prior to the expiration of such representations and warranties; and *provided further*, that no Partnership Related Party shall be entitled to recover special, consequential (including lost profits or diminution in value) or punitive damages.

Indemnification Procedure. Promptly after receipt by an indemnified party under this Article VI Section 6.3 of notice of any claim or the commencement of any action, the indemnified party shall, if a claim in respect thereof is to be made against the indemnifying party under this Article VI, notify the indemnifying party in writing of the claim or the commencement of that action; provided, however, that the failure to notify the indemnifying party shall not relieve it from any liability which it may have under Sections 6.1 or 6.2 of this Article VI except to the extent it has been materially prejudiced (through the forfeiture of substantive rights and defenses) by such failure and, provided, further, that the failure to notify the indemnifying party shall not relieve it from any liability which it may have to an indemnified party otherwise than under this Article VI. If any such claim or action shall be brought against an indemnified party, and it shall notify the indemnifying party thereof, the indemnifying party shall be entitled to participate therein and, to the extent that it wishes, jointly with any other similarly notified indemnifying party, to assume the defense thereof with counsel reasonably satisfactory to the indemnified party. After notice from the indemnifying party to the indemnified party of its election to assume the defense of such claim or action, the indemnifying party shall not be liable to the indemnified party under this Article VI for any legal or other expenses subsequently incurred by the indemnified party in connection with the defense thereof other than reasonable costs of investigation; provided, however, that the indemnified party shall have the right to employ counsel to represent jointly the indemnified party and those other indemnified parties and their respective directors, officers, employees and controlling persons who may be subject to liability arising out of any claim in respect of which indemnity may be sought under this Article VI if (i) the indemnified party and the indemnifying party shall have so mutually agreed; (ii) the indemnifying party has failed within a reasonable time to retain counsel reasonably satisfactory to the indemnified party; (iii) the indemnified party and its directors, officers, employees and controlling persons shall have reasonably concluded that there may be legal

25

defenses available to them that are different from or in addition to those available to the indemnifying party; or (iv) the named parties in any such proceeding (including any impleaded parties) include both the indemnified parties or their respective directors, officers, employees or controlling persons, on the one hand, and the indemnifying party, on the other hand, and representation of both sets of parties by the same counsel would be inappropriate due to actual or potential differing interests between them, and in any such event the fees and expenses of such separate counsel shall be paid by the indemnifying party. No indemnifying party shall (x) without the prior written consent of the indemnified parties (which consent shall not be unreasonably withheld), settle or compromise or consent to the entry of any judgment with respect to any pending or threatened claim, action, suit or proceeding in respect of which indemnification or contribution may be sought hereunder (whether or not the indemnified parties are actual or potential parties to such claim or action) unless such settlement, compromise or consent includes an unconditional release of each indemnified party from all liability arising out of such claim, action, suit or proceeding and does not include a statement as to, or an admission of fault, culpability or a failure to act by or on behalf of any indemnified party, or (y) be liable for any settlement of any such action effected without its written consent (which consent shall not be unreasonably withheld), but if settled with the consent of the indemnifying party or if there be a final judgment for the plaintiff in any such action, the indemnifying party agrees to indemnify and hold harmless any indemnified party from and against any loss or liability by reason of such settlement or judgment. Notwithstanding the foregoing sentence, if at any time an indemnified party shall have requested an indemnifying party to reimburse the indemnified party for fees and expenses of counsel to the extent required by Sections 6.1 and 6.2 hereof, the indemnifying party agrees that it shall be liable for any settlement of any proceeding effected without its written consent if (i) such settlement is entered into more than 30 days after receipt by such indemnifying party of the aforesaid request and (ii) such indemnifying party shall not have reimbursed the indemnified party in accordance with such request or disputed in good faith the indemnified party's entitlement to such reimbursement prior to the date of such settlement.

# ARTICLE VII

#### **MISCELLANEOUS**

Section 7.1 <u>Interpretation</u>. Article, Section, Schedule, and Exhibit references are to this Agreement, unless otherwise specified. All references to instruments, documents, contracts, and agreements are references to such instruments, documents, contracts, and agreements are references to such instruments, documents, contracts, and agreements are references to such instruments, documents, specified. The word "including" shall mean "including but not limited to." Whenever any party has an obligation under the Operative Documents, the expense of complying with that obligation shall be an expense of such party unless otherwise specified. Whenever any

determination, consent, or approval is to be made or given by any Purchaser, such action shall be in such Purchaser's sole discretion unless otherwise specified in this Agreement. If any provision in the Operative Documents is held to be illegal, invalid, not binding, or unenforceable, such provision shall be fully severable and the Operative Documents shall be construed and enforced as if such illegal, invalid, not binding, or unenforceable provision had

never comprised a part of the Operative Documents, and the remaining provisions shall remain in full force and effect. The Operative Documents have been reviewed and negotiated by sophisticated parties with access to legal counsel and shall not be construed against the drafter.

Section 7.2 <u>Survival of Provisions</u>. The representations and warranties set forth in <u>Sections 3.1, 3.2, 3.5, 3.7, 3.8, 3.9, 3.30, 3.32, 3.39</u> and 3.40 (collectively, the "<u>Fundamental Representations</u>") shall survive indefinitely, <u>Sections 3.10, 3.11, 3.12, 3.19, 3.20, 3.21, 3.22, 3.23, 3.24, 3.25, 3.26, 3.27, 3.28, 3.29, 4.4, 4.5, 4.7, 4.8 and 4.9 hereunder shall survive the execution and delivery of this Agreement for two years, and the other representations and warranties set forth herein shall survive for a period of twelve (12) months following the Closing Date regardless of any investigation made by or on behalf of the Partnership or any Purchaser. The covenants made in this Agreement shall survive the Closing of the transactions described herein and remain operative and in full force and effect regardless of acceptance of any of the Partnership and the Purchasers pursuant to this Agreement and the provisions of <u>Article VI</u> shall remain operative and in full force and effect unless such obligations are expressly terminated in a writing by the parties, regardless of any purported general termination of this Agreement.</u>

# Section 7.3 <u>No Waiver; Modifications in Writing</u>.

(a) <u>Delay</u>. No failure or delay on the part of any party in exercising any right, power, or remedy hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such right, power, or remedy preclude any other or further exercise thereof or the exercise of any other right, power, or remedy. The remedies provided for herein are cumulative and are not exclusive of any remedies that may be available to a party at law or in equity or otherwise.

(b) <u>Specific Waiver</u>. Except as otherwise provided herein, no amendment, waiver, consent, modification, or termination of any provision of this Agreement or any other Operative Document shall be effective unless signed by each of the parties hereto or thereto affected by such amendment, waiver, consent, modification, or termination. Any amendment, supplement or modification of or to any provision of this Agreement, any waiver of any provision of this Agreement, and any consent to any departure by the Partnership from the terms of any provision of this Agreement shall be effective only in the specific instance and for the specific purpose for which made or given. Except where notice is specifically required by this Agreement, no notice to or demand on the Partnership in any case shall entitle the Partnership to any other or further notice or demand in similar or other circumstances.

Section 7.4 Binding Effect; Assignment.

(a) <u>Binding Effect</u>. This Agreement shall be binding upon the Partnership, the Purchasers, and their respective successors and permitted assigns. Except as expressly provided in this Agreement, this Agreement shall not be construed so as to confer any right or benefit upon any Person other than the parties to this Agreement and their respective successors and permitted assigns.

27

(b) <u>Assignment of Rights</u>. All or any portion of the rights and obligations of any Purchaser under this Agreement may be transferred by such Purchaser to any Affiliate of such Purchaser without the consent of the Partnership. No portion of the rights and obligations of any Purchaser under this Agreement may be transferred by such Purchaser to a non-Affiliate without the written consent of the Partnership (which consent shall not be unreasonably withheld by the Partnership).

Section 7.5 <u>Confidentiality</u>. Notwithstanding anything herein to the contrary, to the extent that any Purchaser has executed or is otherwise bound by a confidentiality agreement in favor of the Partnership, such Purchaser shall continue to be bound by such confidentiality agreement in accordance with the terms thereof.

Section 7.6 <u>Communications</u>. All notices and demands provided for hereunder shall be in writing and shall be given by registered or certified mail, return receipt requested, telecopy, air courier guaranteeing overnight delivery or personal delivery to the following addresses:

(a) If to any Purchaser, to the respective address listed on <u>Schedule A</u> to the Registration Rights Agreement; and

(b) If to the Partnership:

Antero Midstream Partners LP 1615 Wynkoop Street Denver, Colorado 80202 Attention: Glen C. Warren, Jr. Facsimile: (303) 357-7315 Vinson & Elkins L.L.P. 1001 Fannin Street Suite 2500 Houston, Texas 77002 Attention: W. Matthew Strock Facsimile: (713) 615-5650

or to such other address as the Partnership or such Purchaser may designate in writing. All notices and communications shall be deemed to have been duly given: at the time delivered by hand, if personally delivered; at the time of transmittal, if sent via electronic mail; upon actual receipt if sent by certified mail, return receipt requested, or regular mail, if mailed; when receipt acknowledged, if sent via facsimile; and upon actual receipt when delivered to an air courier guaranteeing overnight delivery.

28

#### Section 7.7 <u>Removal of Legend</u>.

The Partnership, at its sole cost, shall remove the legend described in Section 4.9 (or instruct its transfer agent to so (a)remove such legend) from the certificates evidencing Purchased Units issued and sold to each Purchaser pursuant to this Agreement if (i) such Purchased Units are sold pursuant to an effective registration statement under the Securities Act, (ii) such Purchased Units are sold or transferred pursuant to Rule 144 (if the transferor is not an Affiliate of the Partnership), or (iii) such Purchased Units are eligible for sale under Rule 144, without the requirement for the Partnership to be in compliance with the current public information required under Rule 144(c)(1) (or Rule 144(i)(2), if applicable) as to such securities and without volume or manner-of-sale restrictions. Each Purchaser agrees to provide the Partnership, its counsel and/or the transfer agent with evidence reasonably requested by it in order to cause the removal of the legend described in Section 4.9, including, as may be appropriate, any information the Partnership deems necessary to determine that the legend is no longer required under the Securities Act or applicable state laws, including a certification that the holder is not an Affiliate of the Partnership (and a covenant to inform the Partnership if it should thereafter become an Affiliate and to consent to exchange its certificates for certificates bearing an appropriate restrictive legend) and regarding the length of time the Purchased Units have been held. Any fees (with respect to the transfer agent, Partnership counsel or otherwise) associated with the issuance of any legal opinion required by the Partnership's transfer agent or the removal of such legend shall be borne by the Partnership. If a legend is no longer required pursuant to the foregoing, the Partnership will use commercially reasonable efforts to, no later than three (3) Business Days following the delivery by a Purchaser to the Partnership or the transfer agent (with notice to the Partnership) of a legended certificate or instrument representing Purchased Units (endorsed or with stock powers attached, signatures guaranteed, and otherwise in form necessary to affect the reissuance and/or transfer) and any representation letter or certification as may be requested by the Partnership, deliver or cause to be delivered to such Purchaser a certificate or instrument (as the case may be) representing such Purchased Units that is free from all restrictive legends.

Section 7.8 <u>Entire Agreement</u>. This Agreement, the other Operative Documents and the other agreements and documents referred to herein are intended by the parties as a final expression of their agreement and intended to be a complete and exclusive statement of the agreement and understanding of the parties hereto in respect of the subject matter contained herein and therein. There are no restrictions, promises, representations, warranties or undertakings, other than those set forth or referred to herein or the other Operative Documents with respect to the rights granted by the Partnership or any of its Affiliates or any Purchaser or any of its Affiliates set forth herein or therein. This Agreement, the other Operative Documents and the other agreements and documents referred to herein or therein or therein agreements and understandings between the parties with respect to such subject matter.

# Section 7.9 <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of New York without regard to conflict of laws principles (other than Section 5-1401 of the General Obligations Law).

Section 7.10 <u>Execution in Counterparts</u>. This Agreement may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which counterparts, when so executed and delivered, shall be deemed to be an original and all of which counterparts, taken together, shall constitute but one and the same Agreement. In the event

29

that any signature is delivered by facsimile transmission or by e-mail delivery of a ".pdf" format data file, such signature shall create a valid and binding obligation of the party executing (or on whose behalf such signature is executed) with the same force and effect as if such facsimile or ".pdf" signature page were an original thereof.

Section 7.11 <u>Termination</u>.

(a) Notwithstanding anything herein to the contrary, this Agreement may be terminated at any time at or prior to the Closing by any Purchaser (with respect to such Purchaser only), upon a breach in any material respect by the Partnership of any covenant or agreement set forth in this Agreement.

(b) Notwithstanding anything herein to the contrary, this Agreement shall automatically terminate at any time at or prior to the Closing
(i) if a statute, rule, order, decree or regulation shall have been enacted or promulgated, or if any action shall have been taken by any Governmental Authority of competent jurisdiction that permanently restrains, permanently precludes, permanently enjoins or otherwise permanently prohibits the consummation of the transactions contemplated by this Agreement or makes the transactions contemplated by this Agreement illegal;

- (ii) upon the termination of the Contribution Agreement; or
- (iii) if the Closing shall not have occurred by September 25, 2015.

(c) In the event of the termination of this Agreement as provided in this <u>Section 7.11</u>, this Agreement shall forthwith become null and void. In the event of such termination, there shall be no liability on the part of any party hereto, except as set forth in <u>Section 5.3</u> and <u>Article VI</u> of this Agreement.

Section 7.12 <u>Recapitalization, Exchanges, Etc. Affecting the Common Units</u>. The provisions of this Agreement shall apply to the full extent set forth herein with respect to any and all equity interests of the Partnership or any successor or assign of the Partnership (whether by merger, consolidation, sale of assets or otherwise) which may be issued in respect of, in exchange for or in substitution of, the Common Units, and shall be appropriately adjusted for combinations, unit splits, recapitalizations and the like occurring after the date of this Agreement and prior to the Closing.

### [Signature pages follow.]

30

IN WITNESS WHEREOF, the parties hereto execute this Agreement, effective as of the date first above written.

### ANTERO MIDSTREAM PARTNERS LP

By: ANTERO RESOURCES MIDSTREAM MANAGEMENT LLC (its General Partner)

By: /s/ Alvyn A. Schopp

Name: Alvyn A. Schopp Title: Chief Administrative Officer, Regional Vice President and Treasurer

Signature Page to Common Unit Purchase Agreement

#### AT MLP FUND, LLC

By: /s/ Chris Linder Name: Chris Linder Title: Senior Vice President

#### **CENTER COAST MLP & INFRASTRUCTURE FUND**

By: /s/ Billy Branch Name: Billy Branch Title: CFO

#### CLEARBRIDGE ENERGY MLP FUND INC.

- By: ClearBridge Investments, LLC, as discretionary manager
- By: /s/ Terrence J. Murphy Name: Terrance J. Murphy Title: Chief Executive Officer

# CLEARBRIDGE ENERGY MLP TOTAL RETURN FUND INC.

By: ClearBridge Investments, LLC, as discretionary manager

By: /s/ Terrance J. Murphy

Name: Terrance J. Murphy Title: Chief Executive Officer

# CLEARBRIDGE AMERICAN ENERGY MLP FUND INC.

- By: ClearBridge Investments, LLC, as discretionary manager
- By: /s/ Terrance J. Murphy Name: Terrance J. Murphy Title: Chief Executive Officer

# COHEN & STEERS INFRASTRUCTURE FUND, INC.

By: /s/ Robert Becker Name: Robert Becker Title: Vice President

## COHEN & STEERS MLP INCOME AND ENERGY OPPORTUNITY FUND, INC.

By: /s/ Robert Becker

Name: Robert Becker Title: Vice President

# COHEN & STEERS GLOBAL INFRASTRUCTURE FUND, INC.

By: /s/ Robert Becker

Name: Robert Becker Title: Vice President

# COHEN & STEERS MLP & ENERGY OPPORTUNITY FUND, INC.

By: /s/ Robert Becker Name: Robert Becker Title: Vice President

# TEACHER'S RETIREMENT SYSTEM OF OKLAHOMA

- By: Cushing Asset Management, LP, its investment adviser
- By: Swank Capital, LLC, its General Partner

By: /s/ Robert Becker

Name:Jerry V. SwankTitle:Managing Member

#### CUSHING/SALI MLP ALPHA TOTAL RETURN INSURANCE FUND SERIES OF SALI MULTI-SERIES FUND, LP

- By: Cushing Asset Management, LP, its investment adviser
- By: Swank Capital, LLC, its General Partner
- By: /s/ Jerry V. Swank
  - Name:Jerry V. SwankTitle:Managing Member

# GOLDMAN SACHS MLP ENERGY INFRASTRUCTURE FUND

- By: Goldman Sachs Asset Management, L.P., its Investment Adviser
- By: <u>/s/ Ken Topping</u> Name: Ken Topping Title: Managing Director

## **SELECT 40 FUND**

By: <u>/s/ Brian D. Watson</u> Name: Brian D. Watson Title: Portfolio Manager

### SALIENT MLP FUND, L.P.

- By: Salient Capital Advisors, LLC, its Investment Manager
- By: /s/ Gregory A. Reid Name: Gregory A. Reid Title: Managing Director

# SALIENT MLP & ENERGY INFRASTRUCTURE FUND II

- By: Salient Capital Advisors, LLC, its Investment Manager
- By: /s/ Gregory A. Reid Name: Gregory A. Reid Title: Managing Director

## SALIENT MLP TOTAL RETURN FUND, L.P.

- By: Salient Capital Advisors, LLC, its Investment Manager
- By: /s/ Gregory A. Reid Name: Gregory A. Reid

Title: Managing Director

# COMMONWEALTH OF PENNSYLVANIA PUBLIC SCHOOL RETIREMENT SYSTEM

By: Salient Capital Advisors, LLC, its Investment Manager

By: <u>/s/ Gregory A. Reid</u> Name: Gregory A. Reid Title: Managing Director

# TINICUM L.P.

- By: Tinicum Lantern III, LLC, its general partner
- By: <u>/s/ Gabriel Yuen</u> Name: Gabriel Yuen Title: Member

## TINICUM EMPLOYEES L.P.

By: Tinicum Lantern III, LLC, its general partner

By: <u>/s/ Gabriel Yuen</u> Name: Gabriel Yuen Title: Member

#### TINICUM PARALLEL L.P.

- By: Tinicum Lantern III, LLC, its general partner
- By: /s/ Gabriel Yuen Name: Gabriel Yuen Title: Member

## TTE CP1 HP MORNING, L.P.

- By: Tinicum Lantern III, LLC, its general partner
- By: /s/ Gabriel Yuen Name: Gabriel Yuen Title: Member

## TORTOISE ENERGY INFRASTRUCTURE CORPORATION

By: /s/ James Mick Name: James Mick Title: Managing Director

# TORTOISE POWER AND ENERGY INFRASTRUCTURE FUND, INC.

By: /s/ James Mick Name: James Mick

Title: Managing Director

## TORTOISE MLP FUND, INC.

By: /s/ James Mick

Title: Managing Director

# TORTOISE PIPELINE & ENERGY FUND, INC.

By: /s/ James Mick Name: James Mick Title: Managing Director

# TORTOISE ENERGY INDEPENDENCE FUND, INC.

By: /s/ James Mick Name: James Mick Title: Managing Director

## **TORTOISE MLP & PIPELINE FUND**

By: /s/ James Mick Name: James Mick Title: Managing Director

# **TORTOISE VIP MLP & PIPELINE PORTFOLIO**

By: /s/ James Mick Name: James Mick Title: Managing Director

## ZP ENERGY FUND, L.P.

- By: ZP Energy GP, LLC, its general partner
- By: /s/ Stuart J. Zimmer Name: Stuart J. Zimmer Title: Sole Member

## ZP MASTER UTILITY FUND, LTD.

- By: Zimmer Partners, LP, its Investment Manager
- By: /s/ Stuart J. Zimmer Name: Stuart J. Zimmer Title: CEO

# P ZIMMER LTD.

- By: Zimmer Partners, LP, its Investment Manager
- By: /s/ Stuart J. Zimmer Name: Stuart J. Zimmer Title: CEO

Purchaser	Purchased Units	Со	mmitment Amount
AT MLP FUND, LLC	796,000	\$	14,996,640.00
Center Coast MLP & Infrastructure Fund	398,000	\$	7,498,320.00
ClearBridge Energy MLP Fund Inc.			
ClearBridge Energy MLP Total Return Fund Inc.			
ClearBridge American Energy MLP Fund Inc.			
Subtotal	928,500	\$	17,492,940.00
Cohen & Steers Infrastructure Fund, Inc.	576,300	\$	10,857,492.00
Cohen & Steers MLP Income and Energy Opportunity Fund, Inc.	143,700	\$	2,707,308.00
Cohen & Steers Global Infrastructure Fund, Inc.	58,100	\$	1,094,604.00
Cohen & Steers MLP & Energy Opportunity Fund, Inc.	17,900	\$	337,236.00
Teacher's Retirement System of Oklahoma	325,000	\$	6,123,000.00
Cushing/Sali MLP Alpha Total Return Insurance Fund Series of Sali Multi-Series Fund, LP	45,000	\$	847,800.00
Goldman Sachs MLP Energy Infrastructure Fund	1,592,000	\$	29,993,280.00
Oppenheimer SteelPath Inc Select 40 Fund	1,250,000	\$	23,550,000.00
Salient MLP Fund, L.P.	67,625	\$	1,274,055.00
Salient MLP & Energy Infrastructure Fund II	279,792	\$	5,271,281.28
Salient MLP Total Return Fund, L.P.	169,321	\$	3,190,007.64
Commonwealth of Pennsylvania Public School Employees' Retirement System	199,262	\$	3,754,096.08
Tinicum L.P.	1,259,587	\$	23,730,619.08
TTE CP1 HP Morning, L.P.	159,525	\$	3,005,451.00
Tinicum Employees L.P.	29,190	\$	549,939.60
Tinicum Parallel L.P.	11,198	\$	210,970.32
Tortoise Energy Infrastructure Corporation	831,336	\$	15,662,370.24
Tortoise MLP Fund Inc.	428,930	\$	8,081,041.20
Tortoise Power and Energy Infrastructure Fund, Inc.	28,210	\$	531,476.40
Tortoise Energy Independence Fund, Inc.	37,454	\$	705,633.36
Tortoise MLP & Pipeline Fund	223,951	\$	4,219,236.84
Tortoise Pipeline & Energy Fund, Inc.	41,789	\$	787,304.76
Tortoise VIP MLP & Pipeline Portfolio	330	\$	6,217.20
ZP Energy Fund, L.P.	550	Ψ	0,217.20
ZP Master Utility Fund, LTD			
P Zimmer, LTD			
Subtotal	3,000,000	\$	56,520,000.00
Total	12,898,000	\$	242,998,320.00
	12,070,000	Ψ	<u></u> ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,

Schedule A to Common Unit Purchase Agreement

# Exhibit A

# FORM OF

# **REGISTRATION RIGHTS AGREEMENT**

# BY AND AMONG

# ANTERO MIDSTREAM PARTNERS LP

# AND

# THE PURCHASERS NAMED ON SCHEDULE A HERETO

Exhibit A to Common Unit Purchase Agreement

# TABLE OF CONTENTS

1

# ARTICLE I DEFINITIONS

	ction 1.01 ction 1.02	Definitions Registrable Securities	1 3
ART	ICLE II REGIS	TRATION RIGHTS	3
Se	ction 2.01	Registration	3
Se	ction 2.02	Piggyback Rights	5

Section 2.03	Delay Rights	7
Section 2.04	Underwritten Offerings	8
Section 2.05	Sale Procedures	8
Section 2.06	Cooperation by Holders	12
Section 2.07	Restrictions on Public Sale by Holders of Registrable Securities	12
Section 2.08	Expenses	12
Section 2.09	Indemnification	13
Section 2.10	Rule 144 Reporting	15
Section 2.11	Transfer or Assignment of Registration Rights	15
Section 2.12	Limitation on Subsequent Registration Rights	16
ARTICLE III M	IISCELLANEOUS	16
Section 3.01	Communications	16
Section 3.02	Successor and Assigns	17
Section 3.03	Assignment of Rights	17
Section 3.04	Recapitalization, Exchanges, Etc. Affecting the Units	17
Section 3.05	Aggregation of Registrable Securities	17
Section 3.06	Specific Performance	17
Section 3.07	Counterparts	17
Section 3.08	Headings	17
Section 3.09	Governing Law	17
Section 3.10	Severability of Provisions	18
Section 3.11	Entire Agreement	18
Section 3.12	Amendment	18
Section 3.13	No Presumption	18
Section 3.14	Obligations Limited to Parties to Agreement	18
Section 3.15	Independent Nature of Purchaser's Obligations	18
Section 3.16	Interpretation	19

Schedule A — Purchaser List; Notice and Contact Information; Opt-Out Election

#### **REGISTRATION RIGHTS AGREEMENT**

This REGISTRATION RIGHTS AGREEMENT (this "<u>Agreement</u>") is made and entered into as of September [\_], 2015, by and among Antero Midstream Partners LP, a Delaware limited partnership (the "<u>Partnership</u>"), and each of the Persons set forth on <u>Schedule</u> <u>A</u> to this Agreement (each, a "<u>Purchaser</u>" and collectively, the "<u>Purchasers</u>").

WHEREAS, this Agreement is made and entered into in connection with the Closing of the issuance and sale of the Purchased Units pursuant to the Common Unit Purchase Agreement, dated as of September 17, 2015, by and among the Partnership and the Purchasers (the "<u>Common Unit Purchase Agreement</u>"); and

WHEREAS, the Partnership has agreed to provide the registration and other rights set forth in this Agreement for the benefit of the Purchasers pursuant to the Common Unit Purchase Agreement.

NOW THEREFORE, in consideration of the mutual covenants and agreements set forth herein and for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged by each party hereto, the parties hereby agree as follows:

## ARTICLE I DEFINITIONS

Section 1.01 <u>Definitions</u>. Capitalized terms used herein without definition shall have the meanings given to them in the Common Unit Purchase Agreement. The terms set forth below are used herein as so defined:

"<u>Affiliate</u>" means, with respect to any Person, any other Person that directly or indirectly through one or more intermediaries controls, is controlled by or is under common control with, the Person in question. As used herein, the term "control" means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through ownership of voting securities, by contract or otherwise.

"<u>Aggregate Purchase Price</u>" means the product of (i) the Common Unit Price multiplied by (ii) the aggregate number of Purchased Units purchased by the Purchasers.

"Agreement" has the meaning specified therefor in the introductory paragraph of this Agreement.

"Commission" means the U.S. Securities and Exchange Commission.

"Common Unit Price" has the meaning given to such term in the Common Unit Purchase Agreement.

"Common Unit Purchase Agreement" has the meaning specified therefor in the recitals of this Agreement.

"Effectiveness Period" has the meaning specified therefor in Section 2.01(a) of this Agreement.

"General Partner" means Antero Resources Midstream Management LLC, a Delaware limited liability company.

"Holder" means the record holder of any Registrable Securities.

"Included Registrable Securities" has the meaning specified therefor in Section 2.02(a) of this Agreement.

"Liquidated Damages" has the meaning specified therefor in Section 2.01(b) of this Agreement.

"<u>Liquidated Damages Multiplier</u>" means, with respect to a particular Purchaser, (i) the product of the Common Unit Price multiplied by (ii) the number of Purchased Units purchased by such Purchaser that may not be disposed of without restriction and without the need for current public information pursuant to any section of Rule 144 (or any similar provision then in effect) under the Securities Act.

"Losses" has the meaning specified therefor in Section 2.09(a) of this Agreement.

"<u>Managing Underwritter</u>" means, with respect to any Underwritten Offering, the book-running lead manager or managers of such Underwritten Offering.

"Opt-Out Notice" has the meaning specified therefor in Section 2.02(a) of this Agreement.

"Parity Securities" has the meaning specified therefor in Section 2.02(b) of this Agreement.

"Partnership" has the meaning specified therefor in the introductory paragraph of this Agreement.

"<u>Person</u>" means an individual or a corporation, limited liability company, partnership, firm, joint venture, trust, unincorporated organization, association, government agency or political subdivision thereof or other entity.

"Purchased Units" has the meaning given to such term in the Common Unit Purchase Agreement.

"Purchaser" and "Purchasers" have the meanings specified therefor in the introductory paragraph of this Agreement.

"<u>Registrable Securities</u>" means (i) the Common Units comprising the Purchased Units and (ii) any Common Units issued as Liquidated Damages pursuant to <u>Section 2.01(b)</u> of this Agreement, in each case, as subject to exchange, substitution or adjustment pursuant to <u>Section</u>

2

3.04 of this Agreement, all of which Registrable Securities are subject to the rights provided herein until such rights terminate pursuant to the provisions hereof.

"Registration Expenses" has the meaning specified therefor in Section 2.08(b) of this Agreement.

"Registration Statement" has the meaning specified therefor in Section 2.01(a) of this Agreement.

"Selling Expenses" has the meaning specified therefor in Section 2.08(b) of this Agreement.

"Selling Holder" means a Holder who is selling Registrable Securities pursuant to a registration statement.

"Selling Holder Indemnified Persons" has the meaning specified therefor in Section 2.09(a) of this Agreement.

"<u>Underwritten Offering</u>" means an offering (including an offering pursuant to a Registration Statement) in which Common Units are sold to an underwriter on a firm commitment basis for reoffering to the public or an offering that is a "bought deal" with one or more investment banks.

Section 1.02 <u>Registrable Securities</u>. Any Registrable Security will cease to be a Registrable Security (a) when a registration statement covering such Registrable Security becomes or has been declared effective by the Commission and such Registrable Security has been disposed of pursuant to such effective registration statement; (b) when such Registrable Security has been disposed of pursuant to any section of Rule 144 (or any similar provision then in effect) under the Securities Act; (c) when such Registrable Security is held by the Partnership or one of its subsidiaries or Affiliates; (d) when such Registrable Security has been sold or disposed of in a private transaction in which the transferor's rights under this Agreement are not assigned to the transferee of such securities pursuant to <u>Section 2.11</u> hereof or (e) when such Registrable Security becomes eligible for resale without restriction and without the need for current public information pursuant to any section of Rule 144 (or any similar provision then in effect) under the Securities Act.

## ARTICLE II REGISTRATION RIGHTS

Section 2.01 Registration.

(a) <u>Effectiveness Deadline</u>. Following the date hereof, but no later than 30 days following the Closing Date, the Partnership shall prepare and file a registration statement under the Securities Act to permit the public resale of Registrable Securities then outstanding from time to time as permitted by Rule 415 (or any similar provision then in effect) under the Securities Act with respect to all of the Registrable Securities (the "<u>Registration Statement</u>"). The Registration Statement filed pursuant to this <u>Section 2.01(a)</u> shall be on such appropriate registration form or forms of the Commission as shall be selected by the Partnership so long as it

3

permits the continuous offering of the Registrable Securities pursuant to Rule 415 (or any similar provision then in effect) under the Securities Act at then-prevailing market prices. The Partnership shall use its commercially reasonable efforts to cause the Registration Statement to become effective on or as soon as practicable after the filing thereof. Any Registration Statement shall provide for the resale pursuant to any method or combination of methods legally available to, and requested by, the Holders of any and all Registrable Securities covered by such Registration Statement. The Partnership shall use its commercially reasonable efforts to cause the Registration Statement filed pursuant to this <u>Section 2.01(a)</u> to be effective, supplemented and amended to the extent necessary to ensure that it is available for the resale of all Registrable Securities by the Holders until all Registrable Securities covered by such Registration Statement have ceased to be Registrable Securities (the "<u>Effectiveness Period</u>"). The Registration Statement when effective (including the documents incorporated therein by reference) will comply as to form in all material respects with all applicable requirements of the Securities Act and the Exchange Act and will not contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading (in the case of any prospectus contained in such Registration Statement or documents incorporated therein by reference, in the light of the circumstances under which a statement is made). As soon as practicable following the date that the Registration Statement becomes effective, but in any event within two (2) Business Days of such date, the Partnership shall provide the Holders with written notice of the effectiveness of the Registration Statement.

Failure to Go Effective. If the Registration Statement required by Section 2.01(a) is not declared effective within 90 (b)days after the Closing Date, then each Purchaser shall be entitled to a payment (with respect to the Purchased Units of each such Purchaser), as liquidated damages and not as a penalty, of 0.25% of the Liquidated Damages Multiplier per 30-day period, that shall accrue daily, for the first 30 days following the 90th day, increasing by an additional 0.25% of the Liquidated Damages Multiplier per 30day period, that shall accrue daily, for each subsequent 30 days, up to a maximum of 1.00% of the Liquidated Damages Multiplier per 30day period (the "Liquidated Damages"). The Liquidated Damages payable pursuant to the immediately preceding sentence shall be payable within ten (10) Business Days after the end of each such 30-day period. Notwithstanding anything to the contrary contained herein, in no event shall the aggregate of all Liquidated Damages payable by the Partnership hereunder exceed 5.00% of the Aggregate Purchase Price. Any Liquidated Damages shall be paid to each Purchaser in immediately available funds; provided, however, if the Partnership certifies that it is unable to pay Liquidated Damages in cash because such payment would result in a breach of or constitute a default under a credit facility or other debt instrument, then the Partnership shall pay such Liquidated Damages using as much cash as is permitted without causing a breach of or default under such credit facility or other debt instrument and shall pay the balance of any such Liquidated Damages in kind in the form of the issuance of additional Common Units. Upon any issuance of Common Units as Liquidated Damages, the Partnership shall promptly (i) prepare and file an amendment to the Registration Statement prior to its effectiveness adding such Common Units to such Registration Statement as additional Registrable Securities and (ii) prepare and file a supplemental listing application with the NYSE (or such other national securities exchange on which the Common Units are then-listed and traded) to list such additional Common Units. The determination of the number of Common Units to be issued as Liquidated Damages shall be equal to quotient of (i) the dollar amount of the balance of such Liquidated Damages due to each such Purchaser divided by (ii) the volumeweighted average

4

trading price of the Common Units on the NYSE, or any other national securities exchange on which the Common Units are then-traded, for the ten (10) trading days ending on the first trading day immediately preceding the date on which the Liquidated Damages payment is due, less a discount to such average closing price of 2.00%. The payment of Liquidated Damages to a Purchaser shall cease at the earlier of (i) the Registration Statement becoming effective or (ii) when such Purchaser no longer holds Registrable Securities, assuming that each Holder is not an Affiliate of the Partnership, and any payment of Liquidated Damages shall be prorated for any period of less than 30 days in which the payment of Liquidated Damages ceases. If the Partnership is unable to cause a Registration Statement to go effective within 90 days after the Closing Date as a result of an acquisition, merger, reorganization, disposition or other similar transaction, then the Partnership may request a waiver of the Liquidated Damages, and each Purchaser may individually grant or withhold its consent to such request in its discretion.

(c) <u>Termination of Purchaser's Rights</u>. A Purchaser's rights (and any transferee's rights pursuant to <u>Section 2.11</u> of this Agreement) under this <u>Section 2.01</u> shall terminate upon the termination of the Effectiveness Period.

Section 2.02 Piggyback Rights.

(a) <u>Participation</u>. If the Partnership proposes to file (i) a shelf registration statement other than the Registration Statement contemplated by <u>Section 2.01(a)</u>, (ii) a prospectus supplement to an effective shelf registration statement, other than the Registration Statement contemplated by <u>Section 2.01(a)</u> of this Agreement and Holders may be included without the filing of a post-effective

amendment thereto, or (iii) a registration statement, other than a shelf registration statement, in each case, for the sale of Common Units in an Underwritten Offering for its own account and/or another Person, then as soon as practicable following the engagement of counsel by the Partnership to prepare the documents to be used in connection with an Underwritten Offering, the Partnership shall give notice (including, but not limited to, notification by electronic mail) of such proposed Underwritten Offering to each Holder (together with its Affiliates) holding at least \$15 million of the then-outstanding Registrable Securities (based on the Common Unit Price) and such notice shall offer such Holders the opportunity to include in such Underwritten Offering such number of Registrable Securities (the "Included Registrable Securities") as each such Holder may request in writing; provided, however, that if the Partnership has been advised by the Managing Underwriter that the inclusion of Registrable Securities for sale for the benefit of the Holders will have an adverse effect on the price, timing or distribution of the Common Units in the Underwritten Offering, then (A) if no Registrable Securities can be included in the Underwritten Offering in the opinion of the Managing Underwriter, the Partnership shall not be required to offer such opportunity to the Holders or (B) if any Registrable Securities can be included in the Underwritten Offering in the opinion of the Managing Underwriter, then the amount of Registrable Securities to be offered for the accounts of Holders shall be determined based on the provisions of Section 2.02(b). Any notice required to be provided in this Section 2.02(a) to Holders shall be provided on a Business Day pursuant to Section 3.01 hereof and receipt of such notice shall be confirmed by the Holder. Each such Holder shall then have two (2) Business Days (or one (1) Business Day in connection with any overnight or bought Underwritten Offering) after notice has been delivered to request in writing the inclusion of Registrable Securities in the Underwritten Offering. If no written request for inclusion from a Holder is received within the specified time, each such Holder shall have no

5

further right to participate in such Underwritten Offering. If, at any time after giving written notice of its intention to undertake an Underwritten Offering and prior to the closing of such Underwritten Offering, the Partnership shall determine for any reason not to undertake or to delay such Underwritten Offering, the Partnership may, at its election, give written notice of such determination to the Selling Holders and, (x) in the case of a determination not to undertake such Underwritten Offering, shall be relieved of its obligation to sell any Included Registrable Securities in connection with such terminated Underwritten Offering, and (y) in the case of a determination to delay such Underwritten Offering, shall be permitted to delay offering any Included Registrable Securities for the same period as the delay in the Underwritten Offering. Any Selling Holder shall have the right to withdraw such Selling Holder's request for inclusion of such Selling Holder's Registrable Securities in such Underwritten Offering by giving written notice (an "<u>Opt-Out Notice</u>") to the Partnership requesting that such Holder not receive notice from the Partnership of any proposed Underwritten Offering; *provided, however*, that such Holder may later revoke any such Opt-Out Notice in writing. Following receipt of an Opt-Out Notice from a Holder (unless subsequently revoked), the Partnership shall not be required to deliver any notice to such Holder pursuant to this <u>Section 2.02(a)</u> and such Holder shall no longer be entitled to participate in Underwritten Offerings by the Partnership pursuant to this <u>Section 2.02(a)</u>. The Holders indicated on <u>Schedule A</u> hereto as having opted out shall each be deemed to have delivered an Opt-Out Notice and the hereof.

(b) <u>Priority</u>. If the Managing Underwriter or Underwriters of any proposed Underwritten Offering advises the Partnership that the total amount of Registrable Securities that the Selling Holders and any other Persons intend to include in such offering exceeds the number that can be sold in such offering without being likely to have an adverse effect on the price, timing or distribution of the Common Units offered or the market for the Common Units, then the Common Units to be included in such Underwritten Offering shall include the number of Registrable Securities that such Managing Underwriter or Underwriters advises the Partnership can be sold without having such adverse effect, with such number to be allocated (i) first, to the Partnership and (ii) second, pro rata among the Selling Holders who have requested participation in such Underwritten Offering and any other holder of securities of the Partnership having rights of registration that are neither expressly senior nor subordinated to the Registrable Securities (the "<u>Parity Securities</u>"). The pro rata allocations for each Selling Holder who has requested participation in such Underwritten Offering multiplied by (b) the fraction derived by dividing (x) the number of Registrable Securities owned on the Closing Date by such Selling Holder by (y) the aggregate number of Registrable Securities owned on the Closing Date by all Selling Holders plus the aggregate number of Parity Securities owned on the Closing Date by all holders of Parity Securities owned on the Closing Date by all holders plus the aggregate number of Parity Securities owned on the Closing Date by all Selling Holders plus the aggregate number of Parity Securities owned on the Closing Date by all holders of Parity Securities owned on the Closing Date by all holders plus the aggregate number of Parity Securities owned on the Closing Date by all holders of Parity Securities owned on the Closing Date by all holders plus the aggregate number of fering.

(c) <u>Termination of Piggyback Registration Rights</u>. Each Holder's rights under this <u>Section 2.02</u> shall terminate upon such Holder (together with its Affiliates) ceasing to hold at least \$15 million of Registrable Securities (based on the Common Unit Price). Each Holder shall notify the Partnership in writing when such Holder holds less than \$15 million of Registrable Securities (based on the Common Unit Price).

6

#### Section 2.03 Delay Rights.

Notwithstanding anything to the contrary contained herein, the Partnership may, upon written notice to any Selling Holder whose Registrable Securities are included in the Registration Statement or other registration statement contemplated by this Agreement, suspend such Selling Holder's use of any prospectus which is a part of the Registration Statement or other registration statement contemplated by this Agreement (in which event the Selling Holder shall discontinue sales of the Registrable Securities pursuant to the Registration Statement or such other registration statement but may settle any previously made sales of Registrable Securities) if (i) the Partnership is pursuing an acquisition, merger, reorganization, disposition, financing or other similar transaction and the Partnership determines in good faith that the Partnership's ability to pursue or consummate such a transaction would be materially adversely affected by any required disclosure of such transaction in the Registration Statement or such other registration statement or (ii) the Partnership has experienced some other material non-public event, the disclosure of which at such time, in the good faith judgment of the Partnership, would materially adversely affect the Partnership; *provided, however*, in no event shall the Selling Holders be suspended from selling Registrable Securities pursuant to the Registration Statement or such other registration statement for a period that exceeds an aggregate of 60 days in any 180-day period or 105 days in any 365-day period, in each case, exclusive of days covered by any lock-up agreement executed by a Selling Holder in connection with any Underwritten Offering. Upon disclosure of such information or the termination of the condition described above, the Partnership shall provide prompt notice to the Selling Holders whose Registrable Securities are included in the Registration Statement or other registration statement contemplated by this Agreement, and shall promptly terminate any suspension of sales it has put into effect and shall take such other reasonable actions to permit registered sales of Registrable Securities as contemplated in this Agreement.

If (i) the Selling Holders shall be prohibited from selling their Registrable Securities under the Registration Statement or other registration statement contemplated by this Agreement as a result of a suspension pursuant to the immediately preceding paragraph in excess of the periods permitted therein or (ii) the Registration Statement or other registration statement contemplated by this Agreement is filed and declared effective but, during the Effectiveness Period, shall thereafter cease to be effective or fail to be usable for its intended purpose without being succeeded within 30 Business Days by a post-effective amendment thereto, a supplement to the prospectus or a report filed with the Commission pursuant to Section 13(a), 13(c), 14 or 15(d) of the Exchange Act, then, until the suspension is lifted or a post-effective amendment, supplement or report is filed with the Commission, but not including any day on which a suspension is lifted or such amendment, supplement or report is filed and declared effective, if applicable, the Partnership shall pay the Selling Holders an amount equal to the Liquidated Damages, following the earlier of (x) the date on which the suspension period exceeded the permitted period and (y) the thirty-first (31st) Business Day after the Registration Statement or other registration statement contemplated by this Agreement ceased to be effective or failed to be useable for its intended purposes, as liquidated damages and not as a penalty (for purposes of calculating Liquidated Damages, the date in (x) or (y) above shall be deemed the "90th day," as used in the definition of Liquidated Damages). For purposes of this paragraph, a suspension shall be deemed lifted on the date that notice that the suspension has been terminated is delivered to the Selling Holders. Liquidated Damages purposes purposes of this paragraph, a suspension shall be deemed lifted on the date that notice that the suspension has been

7

upon the Purchased Units held of record by such Holder becoming eligible for resale without restriction and without the need for current public information under any section of Rule 144 (or any similar provision then in effect) under the Securities Act, assuming that such Holder is not an Affiliate of the Partnership, and any payment of Liquidated Damages shall be prorated for any period of less than 30 days in which the payment of Liquidated Damages ceases.

Section 2.04 <u>Underwritten Offerings</u>.

General Procedures. In connection with any Underwritten Offering under this Agreement, the Partnership shall be (a) entitled to select the Managing Underwriter or Underwriters. In connection with an Underwritten Offering contemplated by this Agreement in which a Selling Holder participates, each Selling Holder and the Partnership shall be obligated to enter into an underwriting agreement that contains such representations, covenants, indemnities and other rights and obligations as are customary in underwriting agreements for firm commitment offerings of securities. No Selling Holder may participate in such Underwritten Offering unless such Selling Holder agrees to sell its Registrable Securities on the basis provided in such underwriting agreement and completes and executes all questionnaires, powers of attorney, indemnities and other documents reasonably required under the terms of such underwriting agreement. Each Selling Holder may, at its option, require that any or all of the representations and warranties by, and the other agreements on the part of, the Partnership to and for the benefit of such underwriters also be made to and for such Selling Holder's benefit and that any or all of the conditions precedent to the obligations of such underwriters under such underwriting agreement also be conditions precedent to its obligations. No Selling Holder shall be required to make any representations or warranties to or agreements with the Partnership or the underwriters other than representations, warranties or agreements regarding such Selling Holder, its authority to enter into such underwriting agreement and to sell, and its ownership of, the securities being registered on its behalf, its intended method of distribution and any other representation required by Law. If any Selling Holder disapproves of the terms of an underwriting, such Selling Holder may elect to withdraw therefrom by notice to the Partnership and the Managing Underwriter; provided, however, that such withdrawal must be made up to and including the time of pricing of such Underwritten Offering. No such withdrawal or abandonment shall affect the Partnership's obligation to pay Registration Expenses. The Partnership's management may but shall not be required to participate in a roadshow or similar marketing effort in connection with any Underwritten Offering.

(b) <u>No Demand Rights</u>. Notwithstanding any other provision of this Agreement, no Holder shall be entitled to any "demand" rights or similar rights that would require the Partnership to effect an Underwritten Offering solely on behalf of the Holders.

Section 2.05 <u>Sale Procedures</u>. In connection with its obligations under this <u>Article II</u>, the Partnership will, as expeditiously as possible:

(a) prepare and file with the Commission such amendments and supplements to the Registration Statement and the prospectus and any prospectus supplement used in connection therewith as may be necessary to keep the Registration Statement effective for the Effectiveness Period and as may be necessary to comply with the provisions of the Securities Act with respect to the disposition of all Registrable Securities covered by the Registration Statement;

(b) if a prospectus or prospectus supplement will be used in connection with the marketing of an Underwritten Offering from the Registration Statement and the Managing Underwriter at any time shall notify the Partnership in writing that, in the sole judgment of such Managing Underwriter, inclusion of detailed information to be used in such prospectus or prospectus supplement is of

material importance to the success of the Underwritten Offering of such Registrable Securities, the Partnership shall use its commercially reasonable efforts to include such information in such prospectus or prospectus supplement;

(c) furnish to each Selling Holder (i) as far in advance as reasonably practicable before filing the Registration Statement or any other registration statement contemplated by this Agreement or any supplement or amendment thereto, upon request, copies of reasonably complete drafts of all such documents proposed to be filed (including exhibits and each document incorporated by reference therein to the extent then required by the rules and regulations of the Commission), and provide each such Selling Holder the opportunity to object to any information pertaining to such Selling Holder and its plan of distribution that is contained therein and make the corrections reasonably requested by such Selling Holder with respect to such information prior to filing the Registration Statement or such other registration statement or supplement or amendment thereto, and (ii) such number of copies of the Registration Statement or such other such Selling Holder may reasonably request in order to facilitate the public sale or other disposition of the Registrable Securities covered by such Registration Statement or other registration statement;

(d) if applicable, use its commercially reasonable efforts to register or qualify the Registrable Securities covered by the Registration Statement or any other registration statement contemplated by this Agreement under the securities or blue sky laws of such jurisdictions as the Selling Holders or, in the case of an Underwritten Offering, the Managing Underwriter, shall reasonably request; *provided, however*, that the Partnership will not be required to qualify generally to transact business in any jurisdiction where it is not then required to so qualify or to take any action that would subject it to general service of process in any such jurisdiction where it is not then so subject;

(e) promptly notify each Selling Holder, at any time when a prospectus relating thereto is required to be delivered by any of them under the Securities Act, of (i) the filing of the Registration Statement or any other registration statement contemplated by this Agreement or any prospectus or prospectus supplement to be used in connection therewith, or any amendment or supplement thereto, and, with respect to such Registration Statement or any such other registration statement or any post-effective amendment thereto, when the same has become effective; and (ii) the receipt of any written comments from the Commission with respect to any filing referred to in clause (i) and any written request by the Commission for amendments or supplements to the Registration Statement or any such other registration statement or any prospectus or prospectus supplement thereto;

(f) promptly notify each Selling Holder of (i) the happening of any event as a result of which the prospectus or prospectus supplement contained in the Registration Statement or any other registration statement contemplated by this Agreement, as then in effect, includes an untrue statement of a material fact or omits to state any material fact required to be stated therein or

9

necessary to make the statements therein not misleading (in the case of any prospectus or prospectus supplement contained therein, in the light of the circumstances under which such statement is made); (ii) the issuance or express threat of issuance by the Commission of any stop order suspending the effectiveness of the Registration Statement or any other registration statement contemplated by this Agreement, or the initiation of any proceedings for that purpose; or (iii) the receipt by the Partnership of any notification with respect to the suspension of the qualification of any Registrable Securities for sale under the applicable securities or blue sky laws of any jurisdiction. Following the provision of such notice, the Partnership agrees, subject to Section 2.03 of this Agreement, to as promptly as practicable amend or supplement the prospectus or prospectus supplement or take other appropriate action so that the prospectus or prospectus supplement does not include an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading in the light of the circumstances then existing and to take such other commercially reasonable action as is necessary to remove a stop order, suspension, threat thereof or proceedings related thereto;

(g) upon request and subject to appropriate confidentiality obligations, furnish to each Selling Holder copies of any and all transmittal letters or other correspondence with the Commission or any other governmental agency or self-regulatory body or other body having jurisdiction (including any domestic or foreign securities exchange) relating to such offering of Registrable Securities;

(h) in the case of an Underwritten Offering, furnish upon request, (i) an opinion of counsel for the Partnership dated the date of the closing under the underwriting agreement and (ii) a "comfort" letter, dated the pricing date of such Underwritten Offering and a letter of like kind dated the date of the closing under the underwriting agreement, in each case, signed by the independent public accountants who have certified the Partnership's financial statements included or incorporated by reference into the applicable registration statement, and each of the opinion and the "comfort" letter shall be in customary form and covering substantially the same matters with respect to such registration statement (and the prospectus and any prospectus supplement included therein) as have been customarily covered in opinions of issuer's counsel and in accountants' letters delivered to the underwriters in Underwritten Offerings of securities by the Partnership and such other matters as such underwriters and Selling Holders may reasonably request;

(i) otherwise use its commercially reasonable efforts to comply with all applicable rules and regulations of the Commission, and make available to its security holders, as soon as reasonably practicable, an earnings statement, which earnings statement shall satisfy the provisions of Section 11(a) of the Securities Act and Rule 158 promulgated thereunder;

(j) make available to the appropriate representatives of the Managing Underwriter and Selling Holders access to such information and Partnership personnel as is reasonable and customary to enable such parties to establish a due diligence defense under the Securities Act; *provided*, that the Partnership need not disclose any non-public information to any such representative unless and until such representative has entered into a confidentiality agreement with the Partnership;

(k) cause all such Registrable Securities registered pursuant to this Agreement to be listed on each securities exchange or nationally recognized quotation system on which similar securities issued by the Partnership are then listed;

(1) use its commercially reasonable efforts to cause the Registrable Securities to be registered with or approved by such other governmental agencies or authorities as may be necessary by virtue of the business and operations of the Partnership to enable the Selling Holders to consummate the disposition of such Registrable Securities;

(m) provide a transfer agent and registrar for all Registrable Securities covered by such registration statement not later than the effective date of such registration statement;

(n) enter into customary agreements and take such other actions as are reasonably requested by the Selling Holders or the underwriters, if any, in order to expedite or facilitate the disposition of such Registrable Securities; and

(o) if requested by a Selling Holder, (i) incorporate in a prospectus or prospectus supplement or post-effective amendment to the Registration Statement or any other registration statement contemplated by this Agreement such information as such Selling Holder reasonably requests to be included therein relating to the sale and distribution of Registrable Securities, including information with respect to the number of Registrable Securities being offered or sold, the purchase price being paid therefor and any other terms of the offering of the Registrable Securities to be sold in such offering and (ii) make all required filings of such prospectus or prospectus supplement or post-effective amendment after being notified of the matters to be incorporated in such prospectus or prospectus supplement or posteffective amendment.

The Partnership shall not name a Holder as an underwriter as defined in Section 2(a)(11) of the Securities Act in any registration statement without such Holder's consent. If the staff of the Commission requires the Partnership to name any Holder as an underwriter as defined in Section 2(a)(11) of the Securities Act, or the Partnership deems it advisable, on the advice of counsel, to so name any Holder, and such Holder does not consent thereto, then such Holder's Registrable Securities shall not be included on the Registration Statement (or any other registration statement contemplated by this Agreement), such Holder shall no longer be entitled to receive Liquidated Damages under this Agreement with respect thereto, the Partnership shall have no further obligations hereunder with respect to Registrable Securities held by such Holder and such Holder shall be deemed to have terminated this Agreement with respect to such Holder.

Each Selling Holder, upon receipt of notice from the Partnership of the happening of any event of the kind described in <u>subsection (f)</u> of this <u>Section 2.05</u>, shall forthwith discontinue offers and sales of the Registrable Securities by means of a prospectus or prospectus supplement until such Selling Holder's receipt of the copies of the supplemented or amended prospectus or prospectus supplement contemplated by <u>subsection (f)</u> of this <u>Section 2.05</u> or until it is advised in writing by the Partnership that the use of the prospectus or prospectus supplement may be resumed and has received copies of any additional or supplemental filings incorporated by reference in the prospectus or prospectus supplement, and, if so directed by the Partnership, such Selling Holder will, or will request the Managing Underwriter(s), if any, to deliver to the Partnership (at the Partnership's expense) all copies in their possession or control, other than

11

permanent file copies then in such Selling Holder's possession, of the prospectus or prospectus supplement covering such Registrable Securities current at the time of receipt of such notice.

Section 2.06 <u>Cooperation by Holders</u>. The Partnership shall have no obligation to include in the Registration Statement, or in an Underwritten Offering pursuant to <u>Section 2.02(a)</u>. Registrable Securities of a Holder who has failed to timely furnish such information that the Partnership determines, after consultation with its counsel, is reasonably required in order for the registration statement or prospectus supplement, as applicable, to comply with the Securities Act.

Section 2.07 <u>Restrictions on Public Sale by Holders of Registrable Securities</u>. Each Holder of Registrable Securities agrees, if requested by the underwriters of an Underwritten Offering, to enter into a customary letter agreement with such underwriters providing such Holder will not effect any public sale or distribution of Registrable Securities during the 60 calendar day period beginning on the date of a prospectus or prospectus supplement filed with the Commission with respect to the pricing of any Underwritten Offering, *provided* that (i) the duration of the foregoing restrictions shall be no longer than the duration of the shortest restriction generally imposed by the underwriters on the Partnership or the officers, directors or any other Affiliate of the Partnership on whom a restriction is imposed and (ii) the restrictions set forth in this <u>Section 2.07</u> shall not apply to any Registrable Securities that are included in such Underwritten Offering by such Holder. In addition, this <u>Section 2.07</u> shall not apply to any Holder that is not entitled to participate in such Underwritten Offering, whether because such Holder delivered an Opt-Out Notice prior to receiving notice of the Underwritten Offering or because such Holder holds less than \$15 million of the then-outstanding Registrable Securities.

## Section 2.08 Expenses.

(a) <u>Expenses</u>. The Partnership will pay all reasonable Registration Expenses as determined in good faith, including, in the case of an Underwritten Offering, whether or not any sale is made pursuant to such Underwritten Offering. Each Selling Holder shall pay its pro rata share of all Selling Expenses in connection with any sale of its Registrable Securities hereunder. In addition, except as otherwise provided in <u>Section 2.09</u> hereof, the Partnership shall not be responsible for professional fees incurred by Holders in connection with the exercise of such Holders' rights hereunder.

(b) <u>Certain Definitions</u>. "<u>Registration Expenses</u>" means all expenses incident to the Partnership's performance under or compliance with this Agreement to effect the registration of Registrable Securities on the Registration Statement pursuant to <u>Section 2.01(a)</u> or an Underwritten Offering covered under this Agreement, and the disposition of such Registrable Securities, including, without limitation, all registration, filing, securities exchange listing and NYSE fees, all registration, filing, qualification and other fees and expenses of complying with securities or blue sky laws, fees of the Financial Industry Regulatory Authority, fees of transfer agents and registrars, all word processing, duplicating and printing expenses, any transfer taxes and the fees and disbursements of counsel and independent public accountants for the Partnership, including the expenses of any special audits or "comfort" letters required by or incident to such performance and compliance. "<u>Selling Expenses</u>" means all underwriting fees,

discounts and selling commissions or similar fees or arrangements allocable to the sale of the Registrable Securities.

## Section 2.09 <u>Indemnification</u>.

By the Partnership. In the event of a registration of any Registrable Securities under the Securities Act pursuant to this (a) Agreement, the Partnership will indemnify and hold harmless each Selling Holder thereunder, its directors, officers, employees and agents and each Person, if any, who controls such Selling Holder within the meaning of the Securities Act and the Exchange Act, and its directors, officers, employees or agents (collectively, the "Selling Holder Indemnified Persons"), against any losses, claims, damages, expenses or liabilities (including reasonable attorneys' fees and expenses) (collectively, "Losses"), joint or several, to which such Selling Holder Indemnified Person may become subject under the Securities Act, the Exchange Act or otherwise, insofar as such Losses (or actions or proceedings, whether commenced or threatened, in respect thereof) arise out of or are based upon any untrue statement or alleged untrue statement of any material fact (in the case of any prospectus or prospectus supplement, in the light of the circumstances under which such statement is made) contained in the Registration Statement or any other registration statement contemplated by this Agreement, any preliminary prospectus, preliminary prospectus supplement, free writing prospectus or final prospectus or prospectus supplement contained therein, or any amendment or supplement thereof, or arise out of or are based upon the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein (in the case of a prospectus or prospectus supplement, in the light of the circumstances under which they were made) not misleading, and will reimburse each such Selling Holder Indemnified Person for any legal or other expenses reasonably incurred by them in connection with investigating or defending any such Loss or actions or proceedings; provided, however, that the Partnership will not be liable in any such case if and to the extent that any such Loss arises out of or is based upon an untrue statement or alleged untrue statement or omission or alleged omission so made in conformity with information furnished by such Selling Holder Indemnified Person in writing specifically for use in the Registration Statement or such other registration statement contemplated by this Agreement, any preliminary prospectus, preliminary prospectus supplement, free writing prospectus, or final prospectus or prospectus supplement contained therein, or any amendment or supplement thereof. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of such Selling Holder Indemnified Person, and shall survive the transfer of such securities by such Selling Holder.

(b) <u>By Each Selling Holder</u>. Each Selling Holder agrees severally and not jointly to indemnify and hold harmless the Partnership, the General Partner, its directors, officers, employees and agents and each Person, if any, who controls the Partnership within the meaning of the Securities Act or of the Exchange Act, and its directors, officers, employees and agents, to the same extent as the foregoing indemnity from the Partnership to the Selling Holder, but only with respect to information regarding such Selling Holder furnished in writing by or on behalf of such Selling Holder expressly for inclusion in the Registration Statement or any other registration statement contemplated by this Agreement, any preliminary prospectus, preliminary prospectus supplement, free writing prospectus or final prospectus or prospectus supplement contained therein, or any amendment or supplement thereof; *provided, however*, that the liability of each Selling Holder shall not be greater in amount than the dollar amount of the proceeds (net

13

of any Selling Expenses) received by such Selling Holder from the sale of the Registrable Securities giving rise to such indemnification.

Notice. Promptly after receipt by an indemnified party hereunder of notice of the commencement of any action, such indemnified party shall, if a claim in respect thereof is to be made against the indemnifying party hereunder, notify the indemnifying party in writing thereof, but the omission so to notify the indemnifying party shall not relieve it from any liability that it may have to any indemnified party other than under this Section 2.09. In any action brought against any indemnified party, it shall notify the indemnifying party of the commencement thereof. The indemnifying party shall be entitled to participate in and, to the extent it shall wish, to assume and undertake the defense thereof with counsel reasonably satisfactory to such indemnified party and, after notice from the indemnifying party to such indemnified party of its election so to assume and undertake the defense thereof, the indemnifying party shall not be liable to such indemnified party under this Section 2.09 for any legal expenses subsequently incurred by such indemnified party in connection with the defense thereof other than reasonable costs of investigation and of liaison with counsel so selected; provided, however, that, (i) if the indemnifying party has failed to assume the defense or employ counsel reasonably acceptable to the indemnified party or (ii) if the defendants in any such action include both the indemnified party and the indemnifying party and counsel to the indemnified party shall have concluded that there may be reasonable defenses available to the indemnified party that are different from or additional to those available to the indemnifying party, or if the interests of the indemnified party reasonably may be deemed to conflict with the interests of the indemnifying party, then the indemnified party shall have the right to select a separate counsel and to assume such legal defense and otherwise to participate in the defense of such action, with the reasonable expenses and fees of such separate counsel and other reasonable expenses related to such participation to be reimbursed by the indemnifying party as incurred. Notwithstanding any other provision of this Agreement, no indemnifying party shall settle any action brought against any indemnified party with respect to which such indemnified party is entitled to indemnification hereunder without the consent of the indemnified party, unless the settlement thereof

imposes no liability or obligation on, and includes a complete and unconditional release from all liability of, the indemnified party.

(d) <u>Contribution</u>. If the indemnification provided for in this <u>Section 2.09</u> is held by a court or government agency of competent jurisdiction to be unavailable to any indemnified party or is insufficient to hold them harmless in respect of any Losses, then each such indemnifying party, in lieu of indemnifying such indemnified party, shall contribute to the amount paid or payable by such indemnified party as a result of such Loss in such proportion as is appropriate to reflect the relative fault of the indemnifying party on the one hand and of such indemnified party on the other in connection with the statements or omissions that resulted in such Losses, as well as any other relevant equitable considerations; *provided, however*, that in no event shall such Selling Holder be required to contribute an aggregate amount in excess of the dollar amount of proceeds (net of Selling Expenses) received by such Selling Holder from the sale of Registrable Securities giving rise to such indemnification. The relative fault of the indemnifying party on the one hand and the indemnified party on the other shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact has been made by, or relates to, information supplied by such party, and the parties' relative intent, knowledge, access to information and opportunity to

14

correct or prevent such statement or omission. The parties hereto agree that it would not be just and equitable if contributions pursuant to this paragraph were to be determined by pro rata allocation or by any other method of allocation that does not take account of the equitable considerations referred to herein. The amount paid by an indemnified party as a result of the Losses referred to in the first sentence of this paragraph shall be deemed to include any legal and other expenses reasonably incurred by such indemnified party in connection with investigating or defending any Loss that is the subject of this paragraph. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who is not guilty of such fraudulent misrepresentation.

(e) <u>Other Indemnification</u>. The provisions of this <u>Section 2.09</u> shall be in addition to any other rights to indemnification or contribution that an indemnified party may have pursuant to law, equity, contract or otherwise.

Section 2.10 <u>Rule 144 Reporting</u>. With a view to making available the benefits of certain rules and regulations of the Commission that may permit the sale of the Registrable Securities to the public without registration, the Partnership agrees to:

(a) use its commercially reasonable efforts to make and keep public information regarding the Partnership available, as those terms are understood and defined in Rule 144 under the Securities Act, at all times from and after the date hereof;

(b) use its commercially reasonable efforts to file with the Commission in a timely manner all reports and other documents required of the Partnership under the Securities Act and the Exchange Act at all times from and after the date hereof; and

(c) so long as a Holder owns any Registrable Securities, furnish (i) to the extent accurate, forthwith upon request, a written statement of the Partnership that it has complied with the reporting requirements of Rule 144 under the Securities Act, and (ii) unless otherwise available via EDGAR, to such Holder forthwith upon request a copy of the most recent annual or quarterly report of the Partnership, and such other reports and documents so filed as such Holder may reasonably request in availing itself of any rule or regulation of the Commission allowing such Holder to sell any such securities without registration.

Solely for purposes of this <u>Section 2.10</u>, the term "Registrable Securities" shall be read without regard to the limitation set forth in <u>Section 1.02(e)</u>.

Section 2.11 <u>Transfer or Assignment of Registration Rights</u>. The rights to cause the Partnership to register Registrable Securities granted to the Purchasers by the Partnership under this <u>Article II</u> may be transferred or assigned by any Purchaser to one or more transferees or assignees of Registrable Securities; *provided, however*, that (a) unless the transferee or assignee is an Affiliate of, and after such transfer or assignment continues to be an Affiliate of, such Purchaser, the amount of Registrable Securities transferred or assigned to such transferee or assignee shall represent at least \$15 million of Registrable Securities (based on the Common Unit Price), (b) the Partnership is given written notice prior to any said transfer or assignment, stating the name and address of each such transferee or assignee and identifying the securities with respect to which such registration rights are being transferred or assigned, (c) each such

15

transferee or assignee assumes in writing responsibility for its portion of the obligations of such Purchaser under this Agreement and (d) the transferor or assignor is not relieved of any obligations or liabilities hereunder arising out of events occurring prior to such transfer.

Section 2.12 <u>Limitation on Subsequent Registration Rights</u>. From and after the date hereof, the Partnership shall not, without the prior written consent of the Holders of a majority of the Registrable Securities, enter into any agreement with any current or future holder of any securities of the Partnership that would allow such current or future holder to require the Partnership to include securities in any registration statement filed by the Partnership on a basis other than *pari passu* with, or expressly subordinate to the rights of, the Holders of Registrable Securities hereunder.

# ARTICLE III MISCELLANEOUS

Section 3.01 <u>Communications</u>. All notices and other communications provided for or permitted hereunder shall be made in writing by facsimile, electronic mail, courier service or personal delivery:

- (a) if to a Purchaser, to the respective address listed on <u>Schedule A</u> hereof;
- (b) if to a transferee of a Purchaser, to such Holder at the address provided pursuant to <u>Section 2.11</u> above; and
- (c) if to the Partnership:

Antero Midstream Partners LP 1615 Wynkoop Street Denver, Colorado 80202 Attention: Glen C. Warren, Jr. Facsimile: (303) 357-7315)

with a copy to:

Vinson & Elkins L.L.P. 1001 Fannin Street Suite 2500 Houston, Texas 77002 Attention: W. Matthew Strock Facsimile: (713) 615-5650

All such notices and communications shall be deemed to have been received at the time delivered by hand, if personally delivered; when receipt acknowledged, if sent via facsimile or sent via Internet electronic mail; and when actually received, if sent by courier service or any other means.

16

Section 3.02 <u>Successor and Assigns</u>. This Agreement shall inure to the benefit of and be binding upon the successors and permitted assigns of each of the parties, including subsequent Holders of Registrable Securities to the extent permitted herein.

Section 3.03 <u>Assignment of Rights</u>. All or any portion of the rights and obligations of any Purchaser under this Agreement may be transferred or assigned by such Purchaser only in accordance with <u>Section 2.11</u> hereof.

Section 3.04 <u>Recapitalization, Exchanges, Etc. Affecting the Units</u>. The provisions of this Agreement shall apply to the full extent set forth herein with respect to any and all units of the Partnership or any successor or assign of the Partnership (whether by merger, consolidation, sale of assets or otherwise) that may be issued in respect of, in exchange for or in substitution of, the Registrable Securities, and shall be appropriately adjusted for combinations, unit splits, recapitalizations, pro rata distributions of units and the like occurring after the date of this Agreement.

Section 3.05 <u>Aggregation of Registrable Securities</u>. All Registrable Securities held or acquired by Persons who are Affiliates of one another shall be aggregated together for the purpose of determining the availability of any rights and applicability of any obligations under this Agreement.

Section 3.06 <u>Specific Performance</u>. Damages in the event of breach of this Agreement by a party hereto may be difficult, if not impossible, to ascertain, and it is therefore agreed that each such Person, in addition to and without limiting any other remedy or right it may have, will have the right to an injunction or other equitable relief in any court of competent jurisdiction, enjoining any such breach, and enforcing specifically the terms and provisions hereof, and each of the parties hereto hereby waives any and all defenses it may have on the ground of lack of jurisdiction or competence of the court to grant such an injunction or other equitable relief. The existence of this right will not preclude any such Person from pursuing any other rights and remedies at law or in equity that such Person may have.

Section 3.07 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which counterparts, when so executed and delivered, shall be deemed to be an original and all of which counterparts, taken together, shall constitute but one and the same Agreement. In the event that any signature is delivered by facsimile transmission or by e-mail delivery of a ".pdf" format data file, such signature shall create a valid and binding obligation of the party executing (or on whose behalf such signature is executed) with the same force and effect as if such facsimile or ".pdf" signature page were an original thereof.

Section 3.08 <u>Headings</u>. The headings in this Agreement are for convenience of reference only and shall not limit or otherwise affect the meaning hereof.

Section 3.09 <u>Governing Law.</u> THIS AGREEMENT WILL BE CONSTRUED IN ACCORDANCE WITH, AND GOVERNED BY, THE LAWS OF THE STATE OF NEW YORK.

Section 3.10 <u>Severability of Provisions</u>. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof or affecting or impairing the validity or enforceability of such provision in any other jurisdiction.

Section 3.11 <u>Entire Agreement</u>. This Agreement, the Common Unit Purchase Agreement and the other agreements and documents referred to herein are intended by the parties as a final expression of their agreement and intended to be a complete and exclusive statement of the agreement and understanding of the parties hereto in respect of the subject matter contained herein. There are no restrictions, promises, warranties, representations or undertakings, other than those set forth or referred to herein with respect to the rights granted by the Partnership set forth herein. This Agreement and the Common Unit Purchase Agreement supersede all prior agreements and understandings between the parties with respect to such subject matter.

Section 3.12 <u>Amendment</u>. This Agreement may be amended only by means of a written amendment signed by the Partnership and the Holders of a majority of the then outstanding Registrable Securities; *provided, however,* that no such amendment shall materially and adversely affect the rights of any Holder hereunder without the consent of such Holder.

Section 3.13 <u>No Presumption</u>. If any claim is made by a party relating to any conflict, omission or ambiguity in this Agreement, no presumption or burden of proof or persuasion shall be implied by virtue of the fact that this Agreement was prepared by or at the request of a particular party or its counsel.

Section 3.14 <u>Obligations Limited to Parties to Agreement</u>. Each of the parties hereto covenants, agrees and acknowledges that no Person other than the Purchasers (and their permitted transferees and assignees) and the Partnership shall have any obligation hereunder and that, notwithstanding that one or more of the Purchasers may be a corporation, partnership or limited liability company, no recourse under this Agreement or under any documents or instruments delivered in connection herewith or therewith shall be had against any former, current or future director, officer, employee, agent, general or limited partner, manager, member, stockholder or Affiliate of any of the foregoing, whether by the enforcement of any assessment or by any legal or equitable proceeding, or by virtue of any applicable Law, it being expressly agreed and acknowledged that no personal liability whatsoever shall attach to, be imposed on or otherwise be incurred by any former, current or future director, officer, employee, agent, general or limited partner, director, officer, employee, agent, general or limited partner, manager, member, stockholder or Affiliate of any of the Purchasers or any former, current or future director, officer, employee, agent, general or limited partner, manager, member, stockholder or Affiliate of any of the Purchasers or any former, current or future director, officer, employee, agent, general or limited partner, manager, member, stockholder or Affiliate of any of the Purchasers or any former, current or future director, officer, employee, agent, general or limited partner, manager, member, stockholder or Affiliate of any of the Purchasers or any former, current or future director, officer, employee, agent, general or limited partner, manager, member, stockholder or Affiliate of any of the Purchasers or any former, current or future director, officer, employee, agent, general or limited partner, manager, member, stockholder or Affiliate of any of the Purchasers or any former, current or future director, offic

Section 3.15 <u>Independent Nature of Purchaser's Obligations</u>. The obligations of each Purchaser (and their permitted transferees and assignees) under this Agreement are several and

18

not joint with the obligations of any other Purchaser, and no Purchaser shall be responsible in any way for the performance of the obligations of any other Purchaser under this Agreement. Nothing contained herein, and no action taken by any Purchaser pursuant thereto, shall be deemed to constitute the Purchasers as a partnership, an association, a joint venture or any other kind of group or entity, or create a presumption that the Purchasers are in any way acting in concert or as a group with respect to such obligations or the transactions contemplated by this Agreement. Each Purchaser shall be entitled to independently protect and enforce its rights, including without limitation, the rights arising out of this Agreement, and it shall not be necessary for any other Purchaser to be joined as an additional party in any proceeding for such purpose.

Section 3.16 <u>Interpretation</u>. Article and Section references to this Agreement, unless otherwise specified. All references to instruments, documents, contracts and agreements are references to such instruments, documents, contracts and agreements as the same may be amended, supplemented and otherwise modified from time to time, unless otherwise specified. The word "including" shall mean "including but not limited to." Whenever any determination, consent or approval is to be made or given by a Holder under this Agreement, such action shall be in such Holder's sole discretion unless otherwise specified.

# [Signature pages to follow]

19

IN WITNESS WHEREOF, the parties hereto execute this Agreement, effective as of the date first above written.

## ANTERO MIDSTREAM PARTNERS LP

By: ANTERO RESOURCES MIDSTREAM MANAGEMENT LLC (its General Partner)

# [PURCHASER]

By:

Name: Title:

Signature Page to Registration Rights Agreement

## Schedule A

### Purchaser Name; Notice and Contact Information; Opt-Out Election

 Notice and Contact
 Opt-Out Election per

 Notice and Contact
 Section 2.02(a)

 Information
 Tax I.D. Number
 [Please indicate "Yes 

 Purchaser Name
 [Please provide address,
 [Please provide for
 Opt Out" or "No-Not

 [Please list each fund]
 phone and email]
 each fund]
 Opting Out"]

Schedule A to Registration Rights Agreement

#### Exhibit B — Form of Opinion of Vinson & Elkins L.L.P.

Based on the foregoing, and subject to the qualifications and limitations set forth herein, we are of the opinion that:

- i. Each of the Partnership Entities has been duly formed and is validly existing as a limited partnership or limited liability company, as applicable, in good standing under the laws of the State of Delaware, with all limited partnership or limited liability company, as the case may be, power and authority necessary to (A) conduct its business as described in the SEC Reports, and (B) enter into and perform its obligations under this Agreement, the Contribution Agreement and each of the other Operative Documents to which it is a party;
- ii. Each of the Partnership Entities is duly qualified to do business as a foreign limited liability company or limited partnership, as applicable, in good standing in all jurisdictions listed on <u>Annex A</u> hereto;
- iii. The General Partner is the sole general partner of the Partnership and owns a noneconomic general partner interest in the Partnership; the General Partner Interest has been duly authorized and validly issued in accordance with the Partnership Agreement, is fully paid (to the extent required under the Partnership Agreement) and conforms in all material respects to the description thereof contained in the SEC Reports; and the General Partner Interest is owned free and clear of any Liens (1) in respect of which a financing statement under the Uniform Commercial Code of the State of Delaware naming the General Partner as debtor is on file in the office of the Delaware Secretary of State or (2) otherwise known to us, without independent investigation, other than (A) restrictions on transferability contained in the Partnership Agreement and (B) Liens created by or arising under the Delaware LP Act;
- iv. As of the date hereof, immediately after the offer, issuance and sale of the Purchased Units to the Purchasers in accordance with this Agreement and after giving effect to the Contribution and the other transactions contemplated by the Contribution Agreement, the issued and outstanding limited partner interests in the Partnership (other than limited partner interests issued under the Partnership's Long-Term Incentive Plan) consist of (1) the 116,870,335 Sponsor Units (consisting of 40,929,378 Common Units and 75,940,957 Subordinated Units) held by Antero, (2) 58,922,054 Common Units held by investors other than Antero (including the 12,898,000 Purchased Units issued and sold pursuant to the Purchasers pursuant to this Agreement), and (3) the Incentive Distribution Rights issued to the General Partner; and such limited partner interests on preemptive rights with respect to the Common Units under federal law, the Delaware LP Act or any other instrument to which the Partnership is a party; and, except as set forth in the SEC Reports, no options, warrants or preemptive or other rights to purchase, agreements or other obligations to issue, or

rights to convert any obligations into or exchange any securities of or ownership interests in the Partnership are outstanding;

- v. The General Partner owns all of the Incentive Distribution Rights, and Antero owns all of the Sponsor Units; the Incentive Distribution Rights and the Sponsor Units, and the limited partner interests represented thereby, have been duly authorized and validly issued in accordance with the Partnership Agreement and are fully paid (to the extent required under the Partnership Agreement) and nonassessable (except as such nonassessability may be affected by Sections 17-303, 17-607 and 17-804 of the Delaware LP Act); and the Incentive Distribution Rights and the Sponsor Units are owned free and clear of any Liens (1) in respect of which a financing statement under the Uniform Commercial Code of the State of Delaware naming the General Partner or Antero, as applicable, as debtor is on file in the office of the Delaware Secretary of State or (2) otherwise known to us, without independent investigation, other than (A) restrictions on transferability contained in the Partnership Agreement, (B) Liens created by or arising under the Delaware LP Act and (C) pledges of equity interests in connection with Antero's revolving credit facility;
- vi. The Purchased Units to be issued and sold by the Partnership, and the limited partner interests represented thereby, have been duly authorized for issuance and sale to each Purchaser in accordance with this Agreement and the Partnership Agreement and, when issued and delivered to the Purchasers against payment therefor in accordance with the terms of this Agreement, will be validly issued, fully paid (to the extent required under the Partnership Agreement) and nonassessable (except as such nonassessability may be affected by Sections 17-303, 17-607 and 17-804 of the Delaware LP Act);
- vii. The Partnership is the sole member of the Operating Company and owns 100% of the limited liability company interests in the Operating Company; such limited liability company interests have been duly authorized and validly issued in accordance with the Operating Company LLC Agreement and are fully paid (to the extent required under the Operating Company LLC Agreement) and nonassessable (except as such nonassessability may be affected by Sections 18-607 and 18-804 of the Delaware LLC Act); and such limited liability company interests are owned free and clear of any Liens (1) in respect of which a financing statement under the Uniform Commercial Code of the State of Delaware naming the Partnership as debtor is on file in the office of the Delaware Secretary of State or (2) otherwise known to us, without independent investigation, other than (A) restrictions on transferability contained in the Operating Company LLC Agreement, (B) Liens created by or arising under the Delaware LLC Act and (C) pledges of equity interests in connection with the Revolving Credit Facility;
- viii. The Partnership is the sole member of Antero Treatment and owns 100% of the limited liability company interests in Antero Treatment; such limited liability company interests have been duly authorized and validly issued in accordance with the Antero Treatment LLC Agreement and are fully paid (to the extent

required under the Antero Treatment LLC Agreement) and nonassessable (except as such nonassessability may be affected by Sections 18-607 and 18-804 of the Delaware LLC Act); and such limited liability company interests are owned free and clear of any Liens (1) in respect of which a financing statement under the Uniform Commercial Code of the State of Delaware naming the Partnership as debtor is on file in the office of the Delaware Secretary of State or (2) otherwise known to us, without independent investigation, other than (A) restrictions on transferability contained in the Antero Treatment LLC Agreement, (B) Liens created by or arising under the Delaware LLC Act and (C) pledges of equity interests in connection with the Revolving Credit Facility;

- ix. After giving effect to the Contribution, the Partnership will be the sole member of Antero Water and will own 100% of the limited liability company interests in Antero Water; such limited liability company interests have been duly authorized and validly issued in accordance with the Antero Water LLC Agreement and are fully paid (to the extent required under the Antero Water LLC Agreement) and nonassessable (except as such nonassessability may be affected by Sections 18-607 and 18-804 of the Delaware LLC Act); and, after giving effect to the Contribution, such limited liability company interests will be owned by the Partnership free and clear of any Liens (1) in respect of which a financing statement under the Uniform Commercial Code of the State of Delaware naming the Partnership as debtor is on file in the office of the Delaware Secretary of State or (2) otherwise known to us, without independent investigation, other than (A) restrictions on transferability contained in the Antero Water LLC Agreement, (B) Liens created by or arising under the Delaware LLC Act and (C) pledges of equity interests in connection with the Revolving Credit Facility;
- x. Each of the Contribution Agreement, this Agreement and the Registration Rights Agreement has been duly authorized, executed and delivered by the Partnership Entities party thereto; the Contribution Agreement and, assuming the due authorization, execution and delivery by the Purchasers, the Registration Rights Agreement constitute valid and legally binding agreements of the Partnership Entities party thereto, enforceable against the Partnership Entities party thereto in accordance with their terms, subject to the Enforceability Exceptions;
- xi. The Contribution Agreement is in a form legally sufficient as between the parties thereto to transfer or convey to the Partnership all of Antero's right, title and interest in and to the Water Assets and the limited liability company interests in Antero Water, subject to the conditions, reservations, encumbrances and limitations described therein;

xii. The offering, issuance and sale of the Purchased Units, the execution, delivery and performance of the Contribution Agreement, this Agreement and the other Operative Documents by the Partnership Entities party thereto, the consummation of the Contribution or any other transactions contemplated by this Agreement or the other Operative Documents and the application of the proceeds from the sale of the Purchased Units, in each case, do not and will not result in a breach or

violation of any of the terms and provisions of, or constitute a default under, or result in the imposition of any Lien upon any property or assets of the Partnership Entities pursuant to (1) the Organizational Documents of any of the Partnership Entities, (2) the Delaware LLC Act, the Delaware LP Act or applicable U.S. federal law or any order, judgment, decree or injunction known to us of any U.S. federal or Delaware court or governmental agency or body having jurisdiction over the Partnership Entities or any of their properties in a proceeding in which any of them or their respective properties is a party or (3) any agreement or instrument filed as an exhibit to the SEC Reports; except in the case of clauses (2) and (3) for such breaches, violations, defaults and Liens as would not, individually or in the aggregate, have a Material Adverse Effect, it being understood with respect to clause (2) above, we express no opinion as to the application of any state securities or Blue Sky laws or federal or state antifraud laws, rules or regulations;

- xiii. No consent, approval, authorization or order of, or filing with, any person (including any governmental agency or body or any court) is required to be obtained or made by any of the Partnership Entities in connection with the Contribution or the execution, delivery and performance by the Partnership Entities of the Contribution Agreement, this Agreement and the other Operative Documents, except (A) for the approvals required by the Commission in connection with the Partnership's obligations under the Registration Rights Agreement, (B) such as have been obtained or made, or (C) such as may be required under federal or state securities laws or by the FINRA rules, in each case except where the failure to obtain such consent, approval, authorization or order of, or filing with, would not reasonably be expected to materially impair the ability of the Partnership Parties to consummate the Contribution or the transactions contemplated by the Contribution Agreement, this Agreement or the other Operative Documents;
- xiv. Assuming the accuracy of the representations and warranties of each of the Purchasers and the Partnership contained in this Agreement, the sale and issuance of the Purchased Units by the Partnership to the Purchasers solely in the manner contemplated by this Agreement are exempt from the registration requirements of the Securities Act; provided, that, we will express no opinion as to any subsequent sale or resale; and
- xv. None of the Partnership Entities is now and, after giving effect to the issuance and sale of the Purchased Units by the Partnership and the application of the proceeds therefrom, none of the Partnership Entities will be required to register as an "investment company" within the meaning of the Investment Company Act.



## Antero Midstream Announces Acquisition of Water Business, Private Placement of Common Units & Increased Guidance

**Denver, Colorado, September 18, 2015—Antero Midstream Partners LP** (NYSE: AM) ("Antero Midstream" or the "Partnership") announced today that it has entered into a definitive agreement to acquire Antero Resources Corporation's ("Antero Resources") integrated water business for an aggregate \$1.05 billion in a combination of cash, assumed debt and Partnership common units. In addition, Antero Resources is entitled to receive two potential \$125 million earn out payments at year-end 2019 and 2020 if certain fresh water volumetric delivery targets are met. The acquisition includes Antero Resources' fresh water delivery business, together with an exclusive right to provide fresh water for Antero Resources' well completion operations in West Virginia and Ohio, the to-be-constructed advanced wastewater treatment complex, and the exclusive right to provide fluid handling and disposal services for Antero Resources. The acquisition is expected to close on September 23, 2015, subject to the satisfaction of customary closing conditions, including the consummation of the financing described below.

#### Announcement Highlights:

- Antero Midstream is acquiring Antero Resources' integrated water business for an aggregate \$1.05 billion in cash, assumed debt and Partnership common units plus a total of \$250 million of potential earn out payments
- Purchase price to be funded by common units issued to Antero Resources, cash from Antero Midstream's balance sheet and expanded credit facility and cash proceeds from common units issued in a private placement
- Acquisition is expected to be accretive to distributable cash flow per unit
- Increases 2015 EBITDA guidance to \$170 to \$180 million and distributable cash flow guidance to \$150 to \$160 million
- Expected coverage ratio in excess of the high end of previously stated 1.1x to 1.2x range with reaffirmed distribution growth target of 28% to 30% per year through 2017

### **Transaction Financing**

As consideration for the acquisition, the Partnership has agreed to pay Antero Resources a cash distribution equal to \$552 million, less any assumed debt, and issue 23,886,421 common units representing limited partner interests in the Partnership to Antero Resources. Upon completion of the Partnership's concurrent private placement of common units described below, the net proceeds will be paid to Antero Resources and the number of common units issuable to Antero Resources will be reduced by an equivalent number to that issued in the private placement. The Partnership's credit facility capacity will be increased by \$500 million to a total of \$1.5 billion in conjunction with the completion of this transaction. Pro-forma for the transaction, Antero Midstream would have had \$439 million in debt and approximately \$1.1 billion in availability under its revolving credit facility as of June 30, 2015.

On September 17, 2015, the Partnership priced the private placement of 12,898,000 common units for gross proceeds of approximately \$243 million. The closing of the private placement is subject to the closing of the acquisition and certain other customary conditions, and the private placement is expected to close concurrently with the Partnership's acquisition of Antero Resources' water business. Upon completion of the private placement, the total cash and unit consideration paid by the Partnership to Antero Resources will be \$794 million and 10,988,421 common units, respectively, plus the potential earn out payments described below. The securities offered in the private placement have not been registered under the Securities Act of 1933, as amended (the "Securities Act"), or any state securities laws, and may not be offered or sold in the United States absent registration or an applicable exemption from the registration requirements of the Securities Act and applicable state securities laws. This press release shall not constitute an offer to sell or a solicitation of an offer to buy the securities described herein.

Barclays is acting as sole placement agent for the offering.

Paul Rady, Chairman and CEO commented, "This transaction is a win-win for both entities. Antero Midstream takes another step towards becoming a full value chain midstream services provider in the Appalachian Basin, and Antero Resources reduces debt by \$794 million. The fixed fee terms and minimum volume commitments on fresh water delivery volumes significantly reduce the potential volatility of water business cash flows, making it an excellent business for the Partnership. The transaction is accretive to Antero Midstream as the coverage ratio is expected to exceed the high end of the previously disclosed guidance of 1.1 times to 1.2 times while continuing to target peer-leading distribution growth of 28% to 30% through 2017. In light of current market conditions we believe preserving flexibility through higher distributable cash flow coverage is prudent given our already peer-leading distribution

growth and robust backlog of existing organic growth opportunities. We will continue to monitor the markets and revisit this policy if warranted."

# **Transaction Structure and Minimum Volume Commitments**

In conjunction with the acquisition of the fresh water delivery business, Antero Midstream will enter into a 20-year water services agreement covering Antero Resources' 534,000 net acres in West Virginia and Ohio, with a right of first offer on all future areas of operation. Under the agreement, Antero Midstream will receive a fixed fee of \$3.69 per barrel in West Virginia and \$3.64 per barrel in

Ohio for fresh water deliveries by pipeline directly to the well site, subject to annual CPI adjustments.

In connection with the acquisition, Antero Resources will commit to pay a fee on a minimum volume of fresh water deliveries in calendar years 2016 through 2019. Minimum volume commitments are 90,000 barrels per day in 2016, 100,000 barrels per day in 2017 and 120,000 barrels per day in each of 2018 and 2019. Assuming average barrels of water used to complete a Marcellus and Utica well of 250,000 barrels and 275,000 barrels, respectively, and based on an average lateral length of 9,000 feet for 2015, the minimum volume commitment equates to 125 to 135 completions, which is less than the total completions currently targeted by Antero Resources for 2016.

Additionally, the transaction includes a total of \$250 million of potential earn out payments to be paid to Antero Resources at the end of 2019 and 2020, contingent on meeting specific average volume thresholds. Antero Midstream will pay Antero Resources \$125 million if Antero Midstream's delivered fresh water volumes average 161,000 barrels per day or more between January 1, 2017 and December 31, 2019 and an additional \$125 million if Antero Midstream's delivered fresh water volumes average 161,000 barrels per day or more between January 1, 2017 and December 31, 2019 and an additional \$125 million if Antero Midstream's delivered fresh water volumes average 200,000 barrels per day or more during the period between January 1, 2018 and December 31, 2020.

## Water Business Overview

## Fresh Water Delivery Business

The acquisition will include 100% of Antero Resources' fresh water delivery business, which consists of two independent systems that deliver water from the Ohio River and other regional water sources for well completion operations in both the Marcellus and Utica Shales. The Marcellus and Utica systems consist of a combined 150 miles of permanent buried pipelines, 80 miles of surface water pipelines, 35 fresh water impoundments with storage capacity of approximately 5 million barrels and 15,000 horsepower of water pump capacity. As of June 30, 2015, over 300 wells have utilized Antero Resources' extensive network of fresh water delivery pipelines, eliminating an estimated 400,000 truck trips, materially reducing Antero Resources' overall environmental impact.

Glen Warren, President and CFO, commented, "The acquisition of the water business is expected to be accretive to 2016 distributable cash flow per unit and adds additional scale and scope to Antero Midstream's core footprint in Appalachia. The initial purchase price of \$1.05 billion implies an approximate 8.5 times to 9.0 times multiple of the forecasted 2016 water business EBITDA. This transaction multiple excludes the potential value of the two earn out payments. The minimum volume payments and Antero Resources' significant investment to date in the existing permanent water delivery infrastructure largely de-risks the fresh water delivery business, which is expected to have significant free cash flow in 2016."

## Advanced Wastewater Treatment Complex

Under the terms of the transaction, Antero Midstream will acquire the right to construct the advanced wastewater treatment complex previously announced by Antero Resources. The state-of-the-art advanced wastewater treatment facility is designed to treat 60,000 barrels per day of flowback and produced water and will be centrally located in Antero Resources' footprint in the southwestern core of the Marcellus Shale with the ability to serve Antero Resources' development in both the Marcellus and Utica Shale plays. The advanced wastewater treatment complex is expected to allow Antero Resources and other third-party customers to treat and reuse flowback and produced water rather than permanently dispose of wastewater in water injection wells. Further, water treated at the advanced wastewater treatment facility is expected to displace the water currently sent to third party water recycling or "polishing" facilities which is subsequently used in completion operations (estimated to be approximately 10% of total volumes currently used in completions activities).

Antero Midstream's share of the capital investment for the advanced wastewater treatment complex is estimated to be \$265 million, and also includes site preparation and construction, byproducts processing equipment and five miles of water pipeline that will connect the treatment facility to its existing fresh water distribution system. At full capacity, expected to be achieved within 36 months of the inservice date based on current drilling plans, the facility is expected to have the capability to generate \$55 to \$65 million in annual EBITDA, excluding the potential sale of commercial byproducts.

$\mathbf{r}$
2
_

For additional details regarding the previously announced advanced wastewater treatment facility, please refer to Antero Resources' announcement on August 19, 2015, which can be found on the Antero Resources' website at www.anteroresources.com.

Under the terms of the water services agreement, Antero Resources has agreed to exclusively use Antero Midstream in Ohio and West Virginia, and has granted a right of first offer on all other areas for flowback and produced water services. Antero Midstream will charge Antero Resources a fixed fee of \$4.00 per barrel for wastewater treatment at the advanced wastewater treatment complex under a 20-year agreement, subject to annual CPI-based adjustments.

# Financial Guidance Update

Driven by the drop down acquisition, the Partnership is raising 2015 EBITDA guidance to a range of \$170 to \$180 million from the previous range of \$150 to \$160 million. Distributable cash flow is expected to be \$150 to \$160 million in 2015 versus prior guidance of \$135 to \$145 million. Additionally, the Partnership is maintaining capital expenditure guidance of \$425 to \$450 million as capital reductions on gathering projects offsets anticipated capital investments in the water business for the remainder of 2015.

Commenting on the capital budget, Mr. Warren added, "The reduction in 2015 natural gas gathering capital is a direct result of Antero Resources' shift in its drilling program focus during the year to well pad locations directly adjacent to in-service midstream infrastructure. This results in the deferral of certain gathering projects into 2016 and highlights the flexibility and just-in-time capital investment strategy of Antero Midstream. The current 2016 capital budget contemplates a return to gathering capital expenditure levels more in line with the

original 2015 budget as Antero Resources continues development in new areas dedicated to Antero Midstream. The water infrastructure capital is primarily related to the initial construction costs of the recently announced advanced wastewater treatment facility. We are forecasting approximately \$75 million of capital associated with the facility in 2015 with the remaining capital being ratably invested until completion in the fourth quarter of 2017."

	Year End December 31, 2015			
Guidance Comparison	Prior Guidance	Updated Guidance	Increase/ (decrease)	
Adjusted EBITDA (\$MM)	\$150 - \$160	170 - 180	\$20	
Distributable Cash Flow (\$MM)	\$135 - \$145	150 - 160	\$15	
Year-Over-Year Distribution Growth(1)	28%-30%	28% - 30%	_	
Low Pressure Gathering	165 - 170	\$90 - \$95	\$(75)	
High Pressure Gathering	85 - 90	70 - 75	\$(15)	
Compression	160 - 165	165 - 170	\$5	
Condensate Gathering	5 - 10	5 - 5	\$(5)	
Water Infrastructure		80 - 90	\$80-90	
Maintenance Capital (\$MM)	10 - 15	15 - 15	\$5	
Total Antero Midstream Capital Budget (\$MMs)	\$425 - \$450	425 - 450	_	

(1) Year-over-year distribution growth reflects the expected distribution in the fourth quarter of 2015 vs. the minimum quarterly distribution ("MQD") of \$0.17/unit (not full year 2015 distributions vs. the annualized MQD)

The terms of the transaction were approved by the board of directors of Antero Midstream's general partner, after the conflicts committee of the board of directors of Antero Midstream's general partner unanimously recommended approval, and the board of directors of Antero Resources, after the special committee of the board of directors of Antero Resources unanimously recommended approval. The conflicts committee and special committee are each comprised of independent directors. On financial matters, the conflicts committee of the Partnership was advised by Evercore Group LLC and the special committee of the Partnership, Vinson & Elkins LLP advised the special committee of the Partnership, Vinson & Elkins LLP advised the special committee of Antero Resources and Latham & Watkins LLP advised Barclays Capital Inc.

# Presentation

An updated presentation has been posted to the Partnership's website which can be found at www.anteromidstream.com on the homepage. Information on the Partnership's website does not constitute a portion of this press release.

3

## **Non-GAAP Financial Measures**

As used in this news release, adjusted EBITDA means net income plus interest expense, depreciation and amortization expense, income tax expense (if applicable), and non-cash equity-based compensation expense. As used in this news release, distributable cash flow means adjusted EBITDA less cash interest expense and maintenance capital expenditures. Distributable cash flow should not be viewed as indicative of the actual amount of cash that the Partnership has available for distributions from operating surplus or that the Partnership plans to distribute. Adjusted EBITDA and distributable cash flow are non-GAAP supplemental financial measures that management and external users of the Partnership's consolidated financial statements, such as industry analysts, investors, lenders and rating agencies, use to assess:

- the Partnership's operating performance as compared to other publicly traded partnerships in the midstream energy industry without regard to historical cost basis or, in the case of adjusted EBITDA, financing methods;
- the ability of the Partnership's assets to generate sufficient cash flow to make distributions to the Partnership's unitholders;
- the Partnership's ability to incur and service debt and fund capital expenditures; and
- the viability of acquisitions and other capital expenditure projects and the returns on investment of various investment opportunities.

The Partnership believes that adjusted EBITDA and distributable cash flow provide useful information to investors in assessing the Partnership's financial condition and results of operations. Adjusted EBITDA and distributable cash flow should not be considered as alternatives to net income, operating income, net cash provided by operating activities or any other measure of financial performance or liquidity presented in accordance with GAAP. Adjusted EBITDA and distributable cash flow have important limitations as analytical tools because they exclude some items that affect net income and net cash provided by operating activities. Additionally, because adjusted EBITDA and distributable cash flow may be defined differently by other partnerships in its industry, the Partnership's definition of adjusted EBITDA and distributable cash flow may not be comparable to similarly titled measures of other partnerships, thereby diminishing their utility.

The partnership does not provide financial guidance for projected net income or changes in working capital, and, therefore, is unable to provide a reconciliation of its adjusted EBITDA and distributable cash flow projections to net income, operating income, or net cash flow provided by operating activities, the most comparable financial measures calculated in accordance with GAAP.

Antero Midstream Partners LP is a limited partnership that owns, operates and develops midstream gathering and compression, as well as integrated water assets that service Antero Resources' production located in the Appalachian Basin in West Virginia, Ohio and Pennsylvania.

This release includes "forward-looking statements" within the meaning of federal securities laws. Such forward-looking statements are subject to a number of risks and uncertainties, many of which are beyond the Partnership's control. All statements, other than historical facts included in this release, are forward-looking statements. All forward-looking statements speak only as of the date of this release. Although the Partnership believes that the plans, intentions and expectations reflected in or suggested by the forward-looking statements are reasonable, there is no assurance that these plans, intentions or expectations will be achieved. Therefore, actual outcomes and results could materially differ from what is expressed, implied or forecasted in such statements.

The Partnership cautions you that these forward-looking statements are subject to all of the risks and uncertainties, most of which are difficult to predict and many of which are beyond our control, incident to the gathering and compression business. These risks include, but are not limited to, Antero Resources' expected future growth, Antero Resources' ability to meet its drilling and development plan, commodity price volatility, inflation, environmental risks, drilling and completion and other operating risks, regulatory changes, the uncertainty inherent in projecting future rates of production, cash flow and access to capital, the timing of development expenditures, and the other risks described under "Risk Factors" in Antero Midstream's Annual Report on Form 10-K for the year ended December 31, 2014.

Antero Midstream's ability to make future distributions is substantially dependent upon the development and drilling plan of Antero Resources, which itself is substantially dependent upon the review and approval by the board of directors of Antero Resources of its capital budget on an annual basis. In connection with the review and approval of the annual capital budget, the board of directors will take into consideration many factors, including expected commodity prices and the existing contractual obligations and capital resources and liquidity of Antero Resources at the time.

For more information, contact Michael Kennedy – VP Finance, at (303) 357-6782 or mkennedy@anteroresources.com.

4